**Resilience, Adolescents and Outdoor Education:** 

Is Resilience Context Specific?

By

**Casie-Anne Chalman** 

A thesis submitted in the total fulfilment of the requirements for the degree

of Doctor of Philosophy

College of Education Victoria University Melbourne, Australia March 2019

### DECLARATION

I, Casie-Anne Chalman, declare that the PhD thesis entitled Resilience, Adolescents and Outdoor Education: Is resilience context specific?, is no more than 100,000 words in length, including quotes and exclusive of tables, figures, appendices, bibliography, references, and footnotes. This thesis contains no material that has been submitted previously, in whole or in part, for the award of any other academic degree or diploma. Except where otherwise indicated, this thesis is my own work.

Signature:

### ABSTRACT

This exploratory study investigated the impact of participation in a threeweek journey style outdoor education program upon levels of resilience attributes and coping skills of adolescents. Globalisation and modernisation has increased the social burdens of the 21<sup>st</sup> century and amplifies pressures to conform to unrealistic expectations in society, resulting in negative impacts on young people's mental health and well-being. These life stressors, along with the excessive amount of time that young people spend using technology, is impacting their development and causing young people to experience increased amounts of psychological distress.

In order to manage these stressors, young people often require the development of adaptive coping skills and resilience attributes. Schools can assist their students by supporting the development of resilience attributes and coping skills which are crucial for the future success of young people, to thrive, cope with adversity, and live at an optimal level of human functioning.

Outdoor education programs are regularly delivered by schools to students worldwide as an effective method to facilitate the development and enrichment of personal and social attributes. However, this field is undermined by ad hoc theory and limited research that supports the ability to enhance levels of resilience and coping skills through outdoor education programs with adolescents in mainstream school settings.

The purpose of this longitudinal study was to investigate the effects of an extended journey style outdoor education program on levels of resilience

attributes and coping skills with adolescents. Specifically, the study used a mixed method approach to investigate if the resilience attributes and coping skills were context specific to the setting in which they were developed. The research is framed around particular theories including experiential learning, transactional theory, optimal arousal theory of play, the adventure experience paradigm theory, developmental theory and behaviouristic theories.

This research examined two groups of Year 10 boys who were aged between 14 and 17 years (N = 111). The program group (n = 69), who participated in an extended journey-style outdoor education program, and the control group (n = 42), who did not participate in any outdoor education programs, completed the same survey measures. To identify which particular attributes of resilience and coping that were strengthened and developed through participation in the threeweek program, both groups answered the Resilience Scale and the Brief COPE scale questionnaires within a similar timeframe.

Post-positivist theory was used to analyse the quantitative data, and constructivist theory was used to analyse the qualitative data. The quantitative results revealed that the program group reported higher scores of resilience and resilience attributes compared with the control group following participation in the outdoor education program. Repeated measures t-tests showed significant increases in Overall Resilience and the Existential Aloneness, Perseverance and Purposeful Life subscales. A mixed-design analysis of variance model (ANOVA) revealed a main effect for group and Self-Reliance, F(1, 109) = 7.31, p = .008, and an interaction effect for both time and group was also found for Overall Resilience, F(1, 109) = 3.86, p = .043 and Existential Aloneness, F(1, 109) 9.40, p = .003. Compared to the control group, the program group showed reductions in coping skills that have the potential to undermine well-being in youth, such as Substance Use and Behavioural Disengagement, while increasing in adaptive coping skills such as Active Coping, and Planning after the program.

The qualitative phase of the research addressed the question of whether the resilience attributes and coping skills developed during the program were transferred and drawn upon by the participants in their lives back at school sixmonths after the program. Qualitative data was collected through two means; observation data, and small group semi-structured interviews. The researcher conducted field observations of one group's experience during the three-week program. The observation data provided an insight into the goals and activities of the program and helped to inform the design of the semi-structured interview guides. Small group semi-structured interviews were conducted with the program. Follow-up semi-structured interviews were then conducted with the same 18 participants six-months after the program.

Semi-structured interviews conducted immediately after completion of the program highlighted that participants increased their capacity to demonstrate Overall Resilience, as well as Self-Reliance and Independence, Mental Strength, Determination, and various Developmental Tasks. The students also reported developing positive relationships with their peers, leaders, and the natural environment. An awareness and increased levels of appreciation was shown towards their relationships with family members and technology. A range of coping skills were also developed and applied by the boys during the program, including Putting Things into Perspective; Removing Oneself from the Stressor; Ability to Accept Social Support; Addressing the Issue; Chunking; Coming to Terms with Difficulties; Cognitive Reframing and Applying Positive Thinking; Distraction and Avoidance. Thematic analysis of researcher observations and both sets of semi-structured interview responses revealed three main themes that impacted the development and transference of resilience attributes and coping skills during and after the program. These themes included the program design, the group leader and the learner. The findings showed that most participants struggled to make links between the different contexts of learning. Some of the attributes and skills developed did transfer; however, a common finding was that most participants felt their learnings had dissipated in the six months after the program.

This research supports the notion that outdoor education programs are an effective method to develop resilience attributes and coping skills in young people, however, for the transfer of learning to occur in other contexts, it is recommended that practitioners re-assess their intervention's program design and implement more strategies to improve the transfer of learning. Overall, the findings of the thesis are discussed in terms of the development and refinement of program design of extended outdoor education programs which aim to foster the transference of resilience and positive coping skills into other contexts. Future research directions and implications of the results in relation to professional practices associated with the development of resilience and adaptive coping skills through outdoor education programs are also presented.

Keywords: Outdoor education; Adolescents; Resilience; Stress; Coping.

### ACKNOWLEDGEMENTS

Firstly, I would like to acknowledge the outstanding bunch of young men, the schools and the outdoor education company that graciously agreed to participate in my research. Without you, this research would not have been possible.

Secondly, I would like to recognise my past educators and work colleagues. For me, it all started back in high school with Peter Ridgeway, my outdoor education teacher, sparking my love for nature and outdoor pursuits. He instilled confidence in me, supported me, challenged me and encouraged me to follow my dreams at a critical time of adolescent development. To David Marsden, the concept of embarking in a PhD all started with you. You planted the seed in my mind and fed my curiosity for post-graduate studies. I sincerely thank all my teachers past and present for the knowledge, skills, guidance and support.

Thirdly, to my family and dearest friends. I love you dearly. Your support and unwavering belief in my ability means the absolute world to me. I would especially like to thank my Mother and Father. Your continual support and unconditional love throughout my lifetime is the most precious gift.

Lastly, but especially not least, I would love to acknowledge my amazing supervisors; Dr Peter Burridge and Dr Anthony Watt. I will be eternally grateful for your high expectations, commitment, time, valuable constructive feedback, patience, support and wisdom.

### **DEDICATION**

To all the individuals and families that have been affected by mental illness: It is a difficult road. I feel your pain and suffering.

To the youth of today: You are the reason for conducting this research. There is hope and a better way. It is my vision to help young people develop the skills and knowledge to assist them in navigating the stressors of modern daily life; and to support them to thrive and live happy, fulfilling and purposeful lives.

To my family: You are the most import thing in the world to me. On reflection, I wish that someone specifically taught us effective coping skills to manage life stressors when we were adolescents. Prevention is always better than cure.

Undertaking this research study has been an invaluable learning experience. I have gained an understanding of the complex nature of research involving adolescents, the concept of resilience development and outdoor education. Research involving humans and personal growth has many challenges and has proven to be problematic as these areas of research in their own right are complex, dynamic and multidimensional.

This study has helped me to become constructively critical of not only my own professional practice, values, program design and guidelines but also in examining the professional organisations I have been a practitioner for. The research process has encouraged me to use different theories and practices in the in my own life. The growth of my conceptual understandings, critical thinking and analytic skills during this process has formed the basis for my recommendations for future practice.

Ironically, throughout this study based on positive adaption, I have been confronted with many personal adversities and struggled with my own mental health issues. It has been a process where I have grown spiritually, emotionally, mentally and physically. Mental illness can be crippling, and it affects every aspect of your life. Unlike a physical ailment, I feel that adverse mental health is still not considered to be a legitimate illness in today's western society. I think that it is time to speak up, to show power and strength through vulnerability, and to remove the stigma attached to mental health, to pave the way for others. Finally, I will leave you with the quote that inspired me and kept me going on this journey:

First, it is a challenge. Secondly, you have to learn to prepare meticulously, for your life may depend on the thoroughness and extent of your planning. You have to get off your tail and spur yourself to get going. You have to leave your comfortable slot and go out where things are rough. You have to push into the background the worry of the less likely hazards and make some bold judgements about the more probable ones.

You learn not to be frightened by fear. You discover what a fine piece of machinery the human body is and that it can take a tremendous amount of stress before it breaks down. You learn to make decisions, and gradually you find you make fewer and fewer mistakes. Your confidence grows, and you discover human resources which are ready to be called upon in time of future crises. You learn

ix

something about human frailties and develop sympathy for those weaker or less competent than yourself; you learn to make a team out of a group of individuals.

Adventurous experiences out-of-doors are perceived to kindle the enthusiasm of the young, to develop their concern for others, for their community and for the environment. Such experiences provide the means of self-discovery, self-expression and enjoyment which are at once both stimulating and fulfilling.

It thus emerges that, for young people and adults alike, outdoor adventure is perceived as a vehicle for building values and ideals, for developing creativity and enterprise, for enhancing a sense of citizenship, and for widening physical and spiritual horizons.

- Lord Hunt of Llanfair Waterdin

# TABLE OF CONTENTS

DECLARATION	
ABSTRACT	111
ACKNOWLEDGEMENTS	VII
DEDICATION	VIII
TABLE OF CONTENTS	XI
LIST OF TABLES	XXI
LIST OF FIGURES	XXIV
LIST OF APPENDICES	XXVII
CHAPTER 1: INTRODUCTION TO THE PROBLEM	1
Problem Statement	
Purpose	6
Thesis Structure and Chapter Organisation	6
Significance of the Study	7
CHAPTER 2: LITERATURE REVIEW	
Definition of Terms	
Adolescence	
Stress and Challenge	
Psychological Resilience	
Coping	
Outdoor Education Program	

Understanding Adolescence	13
Understanding Stress and Coping	24
The Concept of Stress	24
The Concept of Coping	25
Theoretical Underpinnings of Stress and Coping.	26
Types of coping	29
Understanding the Concept of Resilience	32
Risk and Protective Factors	34
Developmental assets as protective factors	35
Resilience as a Trait or System	38
Individual level: Resilience as a trait.	38
Systems level: Resilience as a dynamic process	41
Resilience and Social-Ecology	41
Defining Resilience	48
Related Constructs of Resilience	50
Executive Functioning	51
Developing Floribility	51
rsychological riexidinty	
Self-efficacy	52
Self-efficacy	52 53
Self-efficacy Internal Locus of Control Measures of Resilience and Coping	52 53 54
Fychological Flexibility      Self-efficacy      Internal Locus of Control      Measures of Resilience and Coping      Approaches for the Development of Resilience and Coping	52 53 54 58
Psychological Flexibility      Self-efficacy      Internal Locus of Control      Measures of Resilience and Coping      Approaches for the Development of Resilience and Coping      Cognitive Behavioural Approaches	52 53 54 58 60

Positive Psychology	63
Understanding Outdoor Education	66
Adventure and Wilderness Therapy	68
Outdoor Education Program Design	70
Concepts of Outdoor Education	71
Constructivist theory	72
Experiential learning	72
Transfer of learning	77
Key Elements of Outdoor Education Programs	81
The role of the leader	81
The role of the natural environment	85
The role of challenge	88
The role of the social group	92
Outdoor Education as a Resilience and Coping Intervention	94
Chapter Summary	. 101
CHAPTER 3: METHODS	106
Research Question 1	. 107
Research Question 2	. 107
Research Question 3	. 108
Research Question 4	. 108
Research Paradigm	. 110
Post-Positivist Paradigm	. 111
Constructivist Paradigm	. 113

Sample Selection 115
Selection of Schools
Outdoor Education Program
Overview
Group Leaders and Teaching Staff 126
Program Outline
Phase 1: Frontloading phase 127
Phase 2: Immersion phase
Phase 3: Integration phase 129
Ethical Issues and Considerations
Data Collection Process
Phase I: Survey Data Collection and Procedures
Instruments
The Resilience Scale
The Brief COPE 138
Survey Data Collection and Procedure
Survey Data Analysis
Reliability Analysis
Phase II: Program Observation Data146
Program Observation Analysis148
Trustworthiness of Phase II Data
Phase III: Interview Data Collection and Procedures
Semi-Structured Interviews

Semi-structured interview guide 1 151
Semi-structured interview guide 2 152
Interview Data Collection and Procedure 154
Interview Data Analysis 155
Trustworthiness of Phase III Data 157
Chapter Summary 160
CHAPTER 4: QUANTITATIVE RESULTS- PHASE I162
Impact of an Outdoor Education Program on Resilience and Coping skills 162
Aims
Phase I: Quantitative Results
Resilience Scale Descriptive and Inferential Statistics
Resilience Scale Frequencies 167
Resilience Scale Internal Consistencies 168
Brief COPE Descriptive and Inferential Statistics
Brief COPE Frequencies
Brief COPE Exploratory Factor Analysis 175
Initial analysis of the Exploratory Factor Analysis 175
Coping dimension categories 178
Final analysis of the Exploratory Factor Analysis
Brief COPE Internal Consistencies
Correlations
Quantitative Discussion
Chapter Summary 193

CHAPTER 5: QUALITATIVE RESULTS- PHASE II & PHASE III	195
Phase II and Phase III: Qualitative Data Results	196
Theme 1: Challenge	200
Physical Challenges	203
Emotional Challenges	205
Social Challenges	212
Theme Summary	225
Theme 2: Response to Challenge	226
Problem-Focused Coping Strategies	227
Putting things into perspective	227
Removing oneself from the stressor.	231
Addressing the issue.	232
Ability to accept social support	236
Chunking	241
Coming to terms with difficulties.	243
Cognitive reframing and applying positive thinking.	245
Emotion-Focused Coping Strategies	247
Distraction	247
Avoidance	251
Theme Summary	254
Theme 3: Personal Development	255
Impacts of Program Design.	256
Sense of achievement	257

Sense of adventure.	. 261
Awareness and appreciation of one's circumstances, privileges and	
relationships	. 264
Appreciation of family.	. 266
Perceptions of technology.	. 268
Developing and Demonstrating Overall Resilience	. 270
Internal Developmental Assets	. 275
Developmental tasks.	. 275
Self-reliance and independence.	. 277
Mental strength.	. 282
Determination.	. 284
Unconscious Learnings	. 286
Follow-Up Interview Insights and Understandings	. 287
Level of difficulty of the program challenges	. 288
Link between challenge and fun.	. 290
Theme Summary	. 292
Theme 4: Relationships	. 293
External Developmental Assets	. 293
Relationship with the natural environment.	. 294
Relationships with leaders.	. 296
Leaders role in relationship development.	. 297
Relationships with peers.	. 304
Theme Summary	. 311

Theme 5: Transfer of Learning
Transference of Coping Skills
Transference of Internal Assets
Transfer of self-reliance and independence
Transfer of tolerance
Transference of External Assets
Transference of positive relationships
Transference of positive relationships with the environment
Appreciation of relationships with family
Participant's Struggle to Make Links Between Contexts
Theme Summary
Factors Impacting the Development of Resilience and Coping Skills and the
Transfer of Learning
Transactional Processes of Stress Appraisal
Environment filter
Level of challenge
Individual perception filter
Application or non-application of skills and assets filter
Application of coping skills
Application of internal and external assets
Unable to apply skills and assets
Possible outcomes filter
Exploring the interaction of elements that lead to adaption 362

Transfer of Learning into Different Contexts	373
Chapter Summary	377
CHAPTER 6: DISCUSSION AND IMPLICATIONS FOR RESEARCH AND	
PRACTICE	.381
Final Discussion	381
Importance of Understanding the Concept of Resilience	382
Importance of Development-by-Challenge	390
Importance of Making Connections Between Contexts to Support the Trans	sfer
of Learning	395
Importance of Outdoor Education Programs in Schools	402
Addressing the Research Questions	404
Research Question 1:	405
Research Question 2:	408
Research Question 3:	412
Research Question 4:	413
Is Resilience Context Specific?	416
Recommendations for Practice	419
Program Design to Foster the Development and Transference of Resilience	and
Coping Skills	420
Program design planning	426
Phase 1: Frontloading Phase.	428
Phase 2: Immersion Phase	430
Phase 3: Integration Phase	432

Strengths of the Study	435
Limitations of the Study	437
Implications and Recommendations for Future Research	439
Chapter Summary	442
CHAPTER 7: SUMMARY AND CONCLUSION	448
Summary	449
Conclusions	452
Concluding Comments	455
REFERENCES	457
APPENDICES	496

### LIST OF TABLES

Table 2.1 Examples of characteristics of resilient individuals 40
Table 2.2 Examples of risk and protective factors at microsystem systems levels 45
Table 2.3 Examples of bio-psycho-social-cultural systems and processes that can
implicate fostering resilience in young people
Table 2.4 Values in Action (VIA) framework of character strengths 65
Table 2.5 Summary of research on resilience and adolescents in outdoor
education
Table 3.1 Sample selection criteria and results for both sample groups
Table 3.2 Socio-educational rating factors for both private boys' schools
Table 3.3 Number of participants who completed the questionnaire data
Table 3.4 Data collection timeline with descriptions of each phase
Table 3.5 Emotion-focused and problem-focused examples of coping within the 14
subscales of the Brief COPE
Table 4.1 Resilience Scale means, standard deviations and t-tests 165
Table 4.2 ANOVA results for the Resilience Scale 167
Table 4.3 Resilience Scale percentage scores for program and control groups 168
Table 4.4 Brief COPE means, standard deviations and repeated measures t-tests
Table 4.5 ANOVA results for the Brief COPE scale 172

Table 4.6 Brief COPE percentage scores for program and control groups 174
Table 4.7 Factor loadings for the Exploratory Factor Analysis with oblimin
rotation for the Brief COPE177
Table 4.8 Emotion-focused and problem-focused coping dimensions of the Brief
<i>COPE</i>
Table 4.9 New factor structure, labels and coping dimensions of the Brief COPE
Table 4.10 Correlations between the Resilience Scale subscales and EFA factors
of the Brief COPE 183
Table 5.1 Group 9 participant pseudonyms 197
Table 5.2 Participant pseudonyms: Interview and activity group breakdowns 198
Table 5.3 Results and overview of the five major qualitative themes    199
Table 5.4 Types of challenges, definitions and participant comments    202
Table 5.5 Common problem-focused and emotion-focused coping skills applied
Table 5.6 Description of the three approaches to 'put things into perspective'. 228
Table 5.7 Various approaches and participant comments of 'putting things into
perspective'
Table 5.8 Examples of Group comments about the effects of nature on their
perspectives about technology

Table 5.9 Participants definitions of resilience 272
Table 5.10 Examples of participant's reporting the application of problem-
focused coping strategies
Table 6.1 Quantitative and qualitative results comparisons of resilience attributes
Table 6.2 Quantitative and qualitative results comparisons of coping skills 408
Table 6.3 Psychosocial attributes that were transferred post-Immersion Phase 413
Table 6.4 Resilience Development Program Design 424

## LIST OF FIGURES

<i>Figure 1.1.</i> Reflecting on life in the wilderness
<i>Figure 2.1</i> . Helping one another through a river crossing
<i>Figure 2.2.</i> The mental health dual axis model
<i>Figure 2.3.</i> Lazarus's transactional model of stress and coping
Figure 2.4. The ecology of human development systems model
<i>Figure 2.5.</i> The four-stage experiential learning cycle74
<i>Figure 2.6.</i> Yerkes-Dodson law of optimal arousal
Figure 3.1. Watching the sunrise 106
<i>Figure 3.2.</i> Research process flowchart
Figure 3.3. The journey begins: Group 9, Day 1 hiking leg 128
<i>Figure 4.1.</i> Self-guided raft on the Mitta Mitta River
Figure 5.1. The last day of the journey 195
Figure 5.2. Group 9 on the mountain biking leg 204
Figure 5.3. Students resting after hiking to the top of Mount Misery 205
<i>Figure 5.4.</i> Wild Brumbies in the high plains
Figure 5.5. Group 9 undertaking teamwork and initiative activities designed to
build trust
Figure 5.6. Day 1: Arriving at the first track junction

Figure 5.7. Seeking navigation assistance from the group leader
Figure 5.8. A normal campsite setup 218
Figure 5.9. An example of students not talking with one another 221
Figure 5.10. An example of a duty group preparing food 222
Figure 5.11. Day 7: Group 9 struggling to step up their group tarp in the wind. 224
<i>Figure 5.12.</i> The jump rock challenge, Mitta Mitta River
Figure 5.13. Day 17: Group 9's taco wrap shelter
Figure 5.14. Day 10: A student resting after the last uphill leg of the mountain
biking section
Figure 5.15. Students joking around at camp after a long day of hiking 251
Figure 5.16. Group 9 sitting around the campfire
Figure 5.17. Sunrise on the top of Mt Kosciuszko
Figure 5.18. Group 9 after the jump rock challenge
Figure 5.19. Self-identified developmental assets enhanced as a result of
participation in 'City to Summit'
Figure 5.20. A student practising setting up a shelter
Figure 5.21. Day 3: Leader's and participant's playing games
Figure 5.22. Group 9 complaining to one another at the top of Mt Misery 300
<i>Figure 5.23.</i> The 'positive thinking model' from the leader of Group 9 301
<i>Figure 5.24.</i> A duty group working together

<i>Figure 5.25.</i> Peers having fun during hiking
Figure 5.26. Transactional process of stress appraisal in response to program
challenges
<i>Figure 5.27.</i> Goldilocks approach to program design
Figure 5.28. Inset 1: Individual perception filter
Figure 5.29. Inset 2: Application or non-application of skills assets filter 351
Figure 5.30. Inset 3: Possible outcomes filter and the breakthrough diagram 370
Figure 6.1. Celebrating summiting Mt Kosciusko
<i>Figure 6.2.</i> Resilience capacity flourishing flower
Figure 7.1. Last day sunrise hike to Mt Kosciusko
Note. Photographs in Figure 3.3 and Figure 5.17 were taken by the Group Leader of Group 9. All other photographs
throughout this thesis were taken by the researcher in 2012 during the Immersion Phase of the program.

# LIST OF APPENDICES

Appendix A: Human Research Ethics Approval
Appendix B: Invitation Letter to Program Group Principal
Appendix C: Invitation Letter to Control Group Principal
Appendix D: Information to Program Group (School)
Appendix E: Information to Control Group (School)
Appendix F: Program Group Consent Form (School)
Appendix G: Control Group Consent Form (School)
Appendix H: Information to Program Group (Parents and Participants)
Appendix I: Information to Control Group (Parents and Participants)
Appendix J: Program Group Consent Form for Questionnaires
Appendix K: Program Group Consent Form for Small Group Interviews
Appendix L: Control Group Consent Form for Questionnaires
Appendix M: Program Outline
Appendix N: Phase I- Resilience Scale
Appendix O: Phase I- Brief COPE
Appendix P: Phase III- Interview Schedule
Appendix Q: Phase III- Semi Structured Interview Guide 1
Appendix R: Phase III- Semi Structured Interview Guide 2
Appendix S: Cycle 1- All Coded Themes

Appendix T: Cycle 2- First-Order and Second-Order Themes Breakdown	.551
Appendix U: Cycle 3- Major Themes Breakdown	.557
Appendix V: Group 9 Consent for Use of Photos	.561
Appendix W: Progressive Levels of Difficulties of Adventure Activities	.564

### **CHAPTER 1: INTRODUCTION TO THE PROBLEM**

Change and growth take place when a person has risked themselves and dares to become involved in experimenting with their own life

– Herbert Otto



Figure 1.1. Reflecting on life in the wilderness.

Globalisation and modernisation has increased the social burdens of the 21-century and amplifies pressures to conform to unrealistic expectations in society. Life stressors, excessive amounts of time using technology and the increasing problem of 'information and sensory overload' in this 'digital era' is contributing to young people experiencing increased amounts of psychological distress (Australian Bureau of Statistics, 2012; Barrett, Cooper, & Guajardo, 2014; Hanewald, 2011; Herrman et al., 2011; Sawyer et al., 2012; Statista, 2018b). This modern engagement in information processing and additional stressors is resulting in negative impacts on young people's mental health and

well-being; with statistics revealing that suicide is the leading cause of death in young people in Australia aged 15 to 24 (Australian Institute of Health and Welfare, 2016; Mission Australia, 2016), and one-fifth of Australian adolescent's aged 4 to 17 years report experiencing very high levels of psychological distress (Lawrence et al., 2015). These statistics may be influenced by the 'destigmatisation' of mental health over the past 20 years and the rising number of people recognising mental health issues and seeking help (Rice-Oxley, 2019).

Currently, the existing population of adolescents is the largest recorded in human history (Patton et al., 2016). In 2017 there were over 1.8 billion young people aged between 10-24 years old, making up more than a quarter of the world's population (United Nations Department of Economic and Social Affairs: Population Division, 2017). With this number expected to rise to 2 billion by the year 2032, the Lancet Commission (2016) on adolescent health and well-being proposed that we must take action and invest in the development of adolescents (Patton et al., 2016; Sheehan et al., 2017). The importance of working with youth in a proactive capacity, through health promotion is prevalent worldwide (World Health Organization, 2018). For instance, Ki-moon (2016, p. 2357) posits that "young people are the world's greatest untapped resource" and investing in the health and well-being of adolescents will create immense benefits, not only in mental and physical health but it will also generate high economic returns, especially in low-income countries (Patton et al., 2016; Sawyer et al., 2012; Sheehan et al., 2017).

2

Whilst there are many different variables that can affect an individual's ability to thrive, cope or adapt to stressful events, building psychological resilience and developing coping skills in young people can be the contributing factors in minimising the risk and severity of adverse mental health conditions developed during adolescence and early adulthood (Durlak & Wells, 1997; Friedli, 2009; Wagnild, 2010). As the World Health Organization (2012a) advocates, there is no health without mental health. Consequently, developing psychological resilience is a fundamental element of mental health and psychological well-being, and is strongly associated with a young person's sense of confidence, happiness, success, productivity and quality of life (Friedli, 2009; Wagnild, 2010).

In addition to young people feeling distressed, it has been argued that young people in developed countries have been 'wrapped in cotton wool' and their inability to apply coping skills in response to stress may be attributed to overprotective parenting. Research into 'helicopter parenting' types indicates that parents who 'overprotect' their children and try to 'safeguard' them from stressors, challenges or feeling pain, may block young people from developing essential life skills and reduce their emotional coping capacity and self-reliance (Grolnick, Kurowski, Dunlap, & Hevey, 2000; Hunt Jr, 2008; Levine, 2006; Marano, 2008; Schiffrin et al., 2013).

In Western culture, we primarily live in a reactive society. Our physical and mental health care system is grounded in a disease-based model. In the past, psychological theories and therapy interventions have used a disease-based model, which is a problem-focused approach that attempts to address the concerns, diseases, illnesses, maladaptation, or incompetence of an individual, after the issue already exists (Benard, 1991). However, for students to meet the new challenges of the 21<sup>st</sup> century, different methods than the disease-based approach to mental health are required. There is a vital need for the implementation of proactive, prevention and education programs in school curriculum (Friedli, 2009; World Health Organization, 2009).

An approach to enhancing an individual's psychological resilience is to support them to increase their repertoire of coping strategies. Several studies have indicated that outdoor education interventions are a positive tool to enhance resilience (Hattie, Marsh, Neill, & Richards, 1997; Hayhurst, Hunter, Kafka, & Boyes, 2015; Neill & Dias, 2001) and positive coping skills (Booth, 2015; Yoshino, 2008). The nature of outdoor education programming allows participants to experience challenges through exposure to real and perceived risks that may not be available in classroom settings. However, there is little evidence to indicate that participation in mainstream, school delivered, outdoor education programs enhances resilience and coping skills in young people. What is also unclear is which particular attributes of resilience and coping are strengthened through participation outdoor education programs (Neill, 2001; Neill & Richards, 1998). Furthermore, if these interventions affect an individual's resilience and coping capacities, there is little evidence to indicate whether young people apply these skills in other contexts of their lives.

4

It is evident that schools play a critical role in supporting young people with emotional and behavioural problems, as young people spend their majority of time in schools and 95.9% of young people in Australia aged 4 to 17 attend the mainstream school education system (Lawrence et al., 2015). Outdoor education programs are increasingly becoming part of school curriculum across the world and have the capacity to be used as a positive intervention for mental health promotion. The potential exposure for schools to proactively foster the development of resilience attributes in young people is huge, as it is expected that there will be over 700,000 students enrolled in Grade 4 to Year 12 in Victorian schools (Australian Bureau of Statistics, 2008b) and that almost half of these students will participate in outdoor programs by the year 2021. Considering this participation rate, school-based outdoor education programs have the capacity to be used as a platform to promote positive mental health, through fostering the development of psychological resilience and coping strategies in young people.

### **Problem Statement**

Limited research has examined the role that mainstream outdoor education interventions play in helping young people to build resilience and coping skills. Furthermore, there is a lack of understanding as to whether the skills developed during these interventions benefit the participants after the program and support the transference of these life skills into other contexts of their lives. A critical question that remains is; 'is resilience context specific?'

#### **Purpose**

The purpose of this study is to investigate if resilience is context specific to the field of outdoor education. The research specifically aims to examine whether students who participated in an extended journey style outdoor education program increased their levels of resilience and coping skills, and to see if they were able to transfer these skills into other contexts of their lives. In addition, how resilience and coping skills can be best developed and transferred to other contexts of learning will be explored. The results of this study aim to provide insights into best practices for outdoor education programs to assist adolescents in becoming more resilient young adults, who have a greater capacity to cope and thrive in difficult circumstances.

### **Thesis Structure and Chapter Organisation**

Images of one outdoor education group's (Group 9) experience are scattered throughout this thesis to provide context of the wilderness environment, the adventurous activities, program challenges and the group during the outdoor education program. This chapter provides an overview and background to the study, and briefly considers the literature. It then describes the problem and purpose of the study and explains the contribution to knowledge expected from this thesis.

Next, Chapter 2 provides a comprehensive review of the literature on adolescence and adolescent development, stress and coping, and unpacks the concept of resilience. This is followed by a review of outdoor education and its purpose in being used as an intervention for building resilience attributes and coping skills in young people.

Chapter 3 outlines the research methodology for all three phases of this study. It provides a detailed explanation of the aims, research questions, rationale, and theoretical framework. The scope and rigour of the mixed methods approach chosen in this study provides a solid research platform for understanding the role that outdoor education interventions play in building resilience attributes in young people. The next two Chapters present and discuss the results of this study. Chapter 4 provides the quantitative results of Phase I and discusses these results accordingly. This is followed by Chapter 5 that presents and discusses the qualitative results of Phase II and Phase III.

In the final discussion, Chapter 6 commences with a discussion of the key conclusions of the research in relation to the results and provides insights into how resilience and coping can be best supported and developed through participation in outdoor education programs. Responses to the research questions are summarised and that section is followed by a presentation of applied recommendations for practice. The chapter then outlines the strengths and limitations of the research, and implications for future research. Chapter 7 provides closing summaries and conclusions of the research.

### Significance of the Study

Considering the statistics and current status of young people's mental health in Australia, taking a proactive approach to help young people develop 7

psychological resilience and coping skills is critical for assisting youth to gain the skills to help manage their responses to stressors and maintain positive mental health and well-being. Taking a proactive approach to help young people develop psychological resilience and coping skills is important if we are going to assist young people to gain the skills to manage their responses to stressors of modern life and maintain positive mental health and well-being. The current focus of responding to young people once they are in need of support is no longer sufficient (Seligman, 2011).

School-based outdoor education programs have the potential to develop psychological resilience and coping skills (Neill, 2008). Theoretically, these programs can serve as a platform or 'learning playground' whereby participants can practice and experiment with the application of different resilience attributes and coping strategies when immersed in stressful circumstances (Priest & Gass, 2018a; Walsh & Golins, 1976; Yoshino, 2008).

However, there is a lack of research and evidence indicating how effectively resilience can be best developed, enhanced, and transferred into the everyday lives of young people through outdoor education interventions (Ahern, Ark, & Byers, 2008; Neill & Dias, 2001; Neill & Richards, 1998). In addition, the role that coping strategies play in developing resilience through outdoor education programs has been overlooked. Very little research has been conducted on the impact of coping skills in outdoor education programs (Booth, 2015; Yoshino, 2008).
Furthermore, there is a limited number of recent studies that examine the participant perceptions as to whether or not the outcomes and benefits gained in outdoor education programs are maintained (e.g., Buckner et al., 2005; Hayhurst et al., 2015) and are transferred into their daily lives (e.g., Holman & McAvoy, 2005; Sibthorp, 2003). Researchers have emphasised that transferability of resilience attributes and coping skills from outdoor education programs into participants everyday lives warrants further investigation (e.g., Booth, 2015; Neill & Dias, 2001, p.6). Consequently, there is a clear gap and need for ongoing research which can inform program design and practical applications to support the transfer of learning skills developed through outdoor education interventions into other contexts of young people's lives.

The significance of this study can be seen through four major needs: (1) the need to know what psychological resilience attributes and coping skills improve as a result of participating in an outdoor education program; (2) the need to understand the connection between the theories and previous empirical findings of outdoor education, resilience and coping; (3) the need to investigate if the skills developed during outdoor education interventions are transferable into other contexts; and (4) the need to explore the role of outdoor education programming in contributing to fostering the development of psychological resilience and coping skills in young people.

In summary, this thesis offers a substantial theoretical review and empirical analysis of the effects of an extended journey style outdoor education program on levels of resilience and coping skills in adolescent boys. The study

explores the underlying theories, evidence-based research and recommended practices that promote a greater understanding of what and how resilience attributes and coping skills are transferred and used in other contexts. Knowledge gained in this study will enhance the current literature and add to the findings of empirical studies in outdoor education. The findings will provide practical recommendations for practitioners and assist in creating and implementing effective outdoor education program designs that foster the development of resilience and coping skills. In addition, this will open the field of study to further investigation and exploration into the outcomes of this research.

### **CHAPTER 2: LITERATURE REVIEW**

In the middle of difficulty lies opportunity – Albert Einstein



Figure 2.1. Helping one another through a river crossing.

This study aimed to investigate whether students who participated in an extended journey style outdoor education program increased their levels of resilience and coping skills, and to see if they were able to transfer these skills into other contexts of their lives. This chapter provides a synopsis of the relevant theory and literature surrounding adolescence, resilience, coping skills and outdoor education in relation to this study.

The chapter begins by providing definitions of the key concepts. This leads into the second section that presents a brief overview of literature pertaining to adolescent development. The third part explores the concept and importance of resilience and coping for adolescents, and the fourth section examines the role of outdoor education as a tool for resilience development. The literature review finishes by bringing together these four core concepts to provide an understanding of how outdoor education has the potential to support the development of resilience and coping skills in young people.

## **Definition of Terms**

#### Adolescence

Adolescence is defined as the period in which a young person transitions from childhood to adulthood (Furlong, 2009). In this study, adolescents are referred to as young people.

### **Stress and Challenge**

The terms 'stress', 'stressors', 'challenge' and 'challenging experiences' are used interchangeably in this study. This was deemed appropriate as 'challenge' is a universal construct used in the context of outdoor education, as well as 'challenge' being regarded as part of the stress appraisals process (Lazarus & Folkman, 1984). In addition, the terms 'program challenges' relate specifically to the challenges that are included as part of the program design and curriculum of the outdoor education program. Whereas, the term 'difficult' or 'difficulty' relate specifically to the perceived level of 'stress' or 'challenge'.

### **Psychological Resilience**

"Resilience describes a psychological quality that allows a person to cope with, and respond effectively to, life stressors" (Neill & Dias, 2001, p. 5).

### Coping

Conceptually, coping is defined as the constant changing cognitive and behavioural efforts that an individual takes to manage external and/or internal demands (Frydenberg, 2010; Lazarus & Folkman, 1984).

### **Outdoor Education Program**

An outdoor education program is an 'educational design' in which participants are part of a facilitated small group that is exposed to 'perceived or real' risk, challenge and/or adventure in a natural environment. The intentions of programs generally focus on the growth of personal and interpersonal skills. In the context of this research, an outdoor education program refers to an extended journey style outdoor education program which is; (a) the duration of two to four weeks; (b) located in a semi-wilderness setting; (c) consists of a small groups (less than 16 participants and 3 staff); (d) daily and intense interactions within a small group setting requiring problem-solving and decision making; (e) gradual exposure to a range of mentally and/or physically challenging adventure activities, such as hiking steep mountains or whitewater rafting down Class III rapids; and (g) the participants are engaged in a 'journey' where they transition through locations and campsites to reach their final destination (Hattie et al., 1997).

### **Understanding Adolescence**

In the context of western culture and society, adolescence refers to the period of human development that spans between childhood and adulthood (Furlong, 2009). This stage of development occurs in the second decade of a young person's life which coincides with the stages of puberty and continues as they mature and reach their early twenties (Schunk & Meece, 2006; UNICEF, 2011). They form an immensely complex group of people as they experience the youth to young adult transition. As they move through the biologically driven process of puberty; physical, emotional, social, cognitive, and behavioural changes occur in complex ways (Sawyer et al., 2012). Young people experience a range of changes during adolescence such as brain development and connections of new neural pathways (Johnson, S, Blum, & Giedd, 2009; UNICEF, 2011). Parts of the brain that start to mature during this period include areas that are critical for higher-order processes and executive functioning, such as memory, decision making, planning and impulse control (Anderson, Anderson, Northam, Jacobs, & Catroppa, 2001; Johnson, S, Blum, & Giedd, 2009).

Broader changes during puberty include growth spurts, sexual maturation and hormonal changes. These processes are different for each person, as they are not only experiencing the physical changes associated with adolescence, but also the cultural, emotional and social factors that will influence ideas of how they view themselves, how they view their place in the world and their interpretation of how others perceived them (Cole, Vindurampulle, & Vindurampulle, 2006).

No longer children and not yet adults, adolescents start to move away from the security, comfort, and influence of their parents or primary caregivers and move towards taking ownership of their own choices, decisions and abilities (Kroger, 2004). As they strive to become more independent and explore their

place in society, they are progressively confronted with an array of new life experiences.

In early research of adolescent development, the founder of adolescent psychology, Granville Stanley Hall (1904) defined adolescence as a period of 'storm and stress' (Lerner & Steinberg, 2009, p. 1). This deficit view of adolescent development concluded that young people, in general, lacked the ability and the necessary skills to function as effectively as adults in an adult society and that they are unable to make logical, rational decisions under pressure due to incomplete brain development (Arnett, 1999; Cole et al., 2006). Arnett (1999) suggests that while emotional distress is more likely in adolescence than any other period of the lifespan, not everyone experiences this emotional turmoil.

This deficit standpoint may have been influenced by the early research which demonstrated that the majority of early studies on adolescent development suggest that the public agreed with this perception, that adolescence is a time of 'storm and stress' (Arnett, 1999; Lerner & Steinberg, 2009). For example, Buchanan et al. (1990, p. 372) reported that the majority of parents and teachers in their study believed adolescence to be a difficult time of life, agreeing with statements like "early adolescence is a difficult time of life for children and their parents/teachers". However, it is now evident that this is not a period of 'storm and stress' but one of physical and social development. Even though adolescence is a period of change and has the potential for some youth to experience turmoil, it is also a period of opportunity for positive growth (Coulson, 2017; UNICEF, 2011).

Even when faced with the challenges of maturing through adolescence, many young people transition through this developmental period with few issues, becoming competent, adapted and well-functioning adults of society, who operate at an optimal level of human functioning. In other words, they have the capacity to manage stress and live out their daily lives without feeling high levels of anxiety (Jaffe, 1998). Resilient Youth Australia collected data from over 90,000 Australian youth enrolled in Year 3 to Year 12 from over 400 different communities around the country (Resilient Youth Australia as cited in Coulson, 2017). The data revealed that 43% of girls and 40% of boys have good to high levels of resilience and emotional well-being (Coulson, 2017). These numbers provide evidence that on average approximately 40% of youth in Australia are thriving, managing well with challenges and operating at optimal levels. This also illustrates that adolescents do have the capacity to be capable and powerful agents of personal change (Sawyer et al., 2012). On the contrary, the statistics also indicated that more than half of our youth need support to develop the skills required to cope with life's challenges.

Keyes' (2006) research that aimed to determine whether young people between the ages of 12 and 18 were flourishing revealed a correlation between the age and the mental health of the participants (r = -.07; p < .02). This demonstrated that the mental health of the participants declined the older they were. For instance, people aged between 12–14 demonstrated higher scores in flourishing, while young people aged 15–18 demonstrated moderate mental health. The level of optimal human functioning varies for each individual depending on their

culture, age, experience, skills, situation and education. However, the greater the young person's capacity to understand change and adapt to the adversities they encounter, the more likely they are to experience positive effects in almost all areas of their life, now and in the future (Constantine & Derald Wing, 2006).

As Figure 2.2 indicates, mental health and well-being is not the same as mental illness. The four quadrants in the dual axis model demonstrate that individuals can have no mental illness but still have poor mental health and wellbeing (bottom-right quadrant) and likewise that an individual can have a mental illness but still have positive mental health and well-being (top-left quadrant) (Tudor, 1996). If an individual is flourishing, this means that they are living within an optimal range of human functioning, that includes growth, productivity, purpose, psychological flexibility and adaptability (Fredrickson & Losada, 2005). The model portrays that everyone can improve their mental well-being whether they have a mental health issue or not. It views mental health in positive terms rather than by the absence of mental illness (Fredrickson & Losada, 2005; Keyes, 2006).



Note. Adapted from Tudor (1996).

Figure 2.2. The mental health dual axis model.

Along with individual developmental changes, new life experiences and responsibilities, comes the accompanying life hurdles that young people will need to overcome. For example, as young people commence adolescence, they will experience the challenging transition from education in a primary school setting, into a secondary school setting. This transition occurs during an age where many young people's well-being and resilience declines significantly. Australian research is consistent with other global data that indicates that a young person's well-being peaks at around 8 years of age when they are in primary school and decreases consistently through until the age of 16-17 years when they are in secondary school (Australian Bureau of Statistics, 2016; Bruckauf, 2017; Commissioner for Children and Young People Western Australia, 2015). This finding by 'Resilient Youth Australia', indicates the need to promote and continually develop a young person's well-being and resilience through their secondary school years (Resilient Youth Australia as cited in Coulson, 2017).

Adolescence has always been a period of personal change and development, where youth are confronted with many different types of challenges at any one point. However, growing up in the 21<sup>st</sup> century adds another layer of complexity (Sawyer et al., 2012). Modern technology has resulted in the young people of today growing up in the digital and information age. The digital era has influenced the way youth learn, process information, interact, and develop relationships. Young people now use social media, accessed on the world wide web, as a virtual platform to communicate, access and share information 24 hours a day. This new medium of connecting has created a new generation of people whose identities are defined by their online connections and what they produce online (Dye, 2007).

'Facebook' is an example of an online platform that is one of the leading online social networking sites and has over 2.2 billion users worldwide. Of these 2.2 billion users are 44.15 million young people aged between 13-24 years old (Statista, 2018a, 2018b). This virtual world can directly impact youth development and increase the likelihood of external inputs and stressors, which

can be driven by social pressures and lead to psychological turmoil. Some examples include access to inappropriate adult information, around the clock exposure to social interaction and the increased potential for experiencing online bullying or peer pressure (Barrett et al., 2014). The new challenges that come along with the digital age may require different coping strategies to manage or overcome these stressors, compared to coping strategies that may have been useful ten years ago, when transitioning through adolescence (Sawyer et al., 2012). In addition to the pressures of the digital era, other social changes have occurred, providing a range of challenges for adolescents to cope with. For example, the rise of such issues as divorce or unemployment of parents, alcohol and substance abuse, internet or other addictions, and other random traumatic events (Australian Bureau of Statistics, 2012; Hanewald, 2011; Herrman et al., 2011; Humphrey, 2002).

Given all the pressures and challenges that young people face, it is understandable that adolescence is a period which is associated with an increased risk of mental health concerns. For many people, mental health issues come to light during adolescence (UNICEF, 2011). An increased number of studies worldwide are finding that more young people are suffering from mental health issues (Australian Bureau of Statistics, 2007, 2016; Bruckauf, 2017; Commissioner for Children and Young People Western Australia, 2015; UNICEF, 2011). During 2012-2016, one in four young Australian's aged between 15 and 19 met the criteria for having a serious mental illness (Bullot, Cave, Fildes, Hall, & Plummer, 2017; Mission Australia, 2016). This is equivalent to more than 560,000 youth with mental disorders, and of these young people, 278,000 reported having anxiety disorders, 112,000 had major depressive disorders, and 83,600 had conduct disorders (Lawrence et al., 2015). The problem continues to increase with the proportion of those indicating mental health as a national concern rose from 14.9% in 2015 to 33.7% in 2017 (Bullot et al., 2017). If these conditions are left untreated or unsupported, mental health disorders can lead down the path to more critical problems including suicide.

Worldwide, road traffic injuries were the leading cause of death in adolescents (World Health Organization, 2018). However, according to the Australian Institute of Health and Welfare (2016) suicide is the leading cause of death in youth aged between 15 and 24 years. These statistics highlight the importance of acknowledging the needs of young people and supporting their capacity to learn coping skills in order to manage and adapt to adversity and hardship when it occurs.

In addition, adolescents are at greater risk of experiencing mental health disorders compared to adults (Australian Bureau of Statistics, 2016; Australian Institute of Health and Welfare, 2016; Bullot et al., 2017). Mental health problems in young people often stem from having low self-esteem as well as the increased social pressures of the 21<sup>st</sup> century to conform to unrealistic expectations (UNICEF, 2011). There are many contributing factors that affect mental health. However, the main factors affecting young people include violence, abuse, neglect, and bullying (UNICEF, 2011). These sorts of hardships can often lead to feelings of vulnerability and instability, resulting in an increased likelihood of

poor decision-making, unhealthy risk-taking behaviour and a susceptibility to further problems (Ahern et al., 2008; Hanewald, 2011). Factors such as peer pressure and trying to 'show off' in front of others or being under the influence of intoxicants (e.g., drugs and alcohol) are examples of factors that can lead to poor decision making. Decision making is crucial during adolescence, with the potential for poor decisions and momentary actions resulting in serious unwanted consequences that may impact the rest of their lives (e.g., teen pregnancy, underage drinking, permanent mental or physical injury, breaking the law, dropping out of school, juvenile delinquency).

As young people mature into later adolescence, they develop greater independence and undertake additional responsibilities. The acquisition of more control over decision making promotes a greater sense of freedom. For example, in society, this is illustrated by having the opportunity to obtain a Learner's Driver's Licence or gain paid employment. However, even though young people's expectations and responsibilities in society may increase, young people may not have had the necessary life experiences or opportunities to develop the coping skills required to manage the challenges of adult life. General life hurdles that young people may endure range from seemingly mundane challenges of adult life, such as saving money, budgeting and paying for acquired phone bills, to more complex hurdles, such as maintaining employment or dealing with relationship conflict. An inability for young people to cope or manage challenges of adult life may be attributed to the fact that they may have;

• experienced 'over-controlling' parents or 'helicopter parenting';

- not been taught effective coping skills required to overcome challenges;
- not had the opportunities to practice and apply coping skills in various contexts;
- not had the guidance and opportunities for self-reflection and personal growth; or
- not been exposed to challenges with that level of difficulty before (Coulson, 2017; Marano, 2008; Schiffrin et al., 2013).

Some would argue that 'wrapping kids in cotton wool' by not exposing them to challenges or allowing for healthy decision making around risk, could have an adverse effect on their capacity to positively adapt to challenging situations (Booth, 2015; Coulson, 2017; Schiffrin et al., 2013).

To avoid poor decision making and increase a young person's capacity to cope with challenges and adversity, it is important to assist adolescents to develop positive human characteristics, such as a positive sense of self-agency, selfesteem, self-efficacy and resilience. For instance, a young person with a positive personal agency would have an increased capacity to originate and direct actions for a given purpose (Zimmerman & Cleary, 2006, p. 45). Youth that hold a positive sense of agency are more resilient and better able to negotiate activities that are determined as high-risk activities that they may be exposed to, such as, alcohol, illicit drug use, smoking and early sexual activity (Bandura, 2006).

Developing a positive sense of agency and healthy decision-making skills coincide with the adolescent stage of cognitive development where individuals start to develop formal operational thinking (Piaget, 1964, 1972). This occurs later in adolescence and enables them to think abstractly, allowing for self-reflection and to be able to predict possible outcomes of different types of behaviour in various contexts (Jaffe, 1998; Piaget, 1964, 1972; UNICEF, 2011). Increasing a young person's personal agency gives them an increased control of their decisions and actions, which allows them to become agents for change and in control of their own future (Bandura, 1977, 1986).

#### **Understanding Stress and Coping**

#### The Concept of Stress

Previously, psychological stress has been defined in two ways; one approach defines stress an environmental stimulus (stressor), and the other defines stress as "the nonspecific response of the body to any demand made upon it" (Selye, 1974, p. 137). However, Lazarus (2006) later provided an alternative to the 'stimulus' and 'response' definitions of psychological stress by combining the two to form a relational approach of viewing stress that acknowledges the direct relationship between the person (response) and the environment (stressor).

Selye (1975) categorises two types of stress; eustress and distress. In Selye's (1976) theory, he indicates that 'stress' is the initial response to the stressor; however, the eustress and distress are the effects of the stressor. Eustress, also known as positive stress, is regarded as a healthy form of stress which can be used to motivate participants to apply coping mechanisms and test their problemsolving capabilities (Priest & Gass, 2018b; Selye, 1974, 1976). Whereas distress, also known as negative stress, can cause adverse responses, such as anxiety, overwhelm or the inability to adapt (Selye, 1975). According to the transactional perspective of stress and coping, stress is the relationship between the individual and the environment in which one perceives the adaptive demands of the stress as being either difficult but achievable or exceeding their perceived ability to apply coping mechanisms to meet the demands of the stressor (Neill, 2008).

### The Concept of Coping

Coping is a multidimensional phenomenon, which includes an individual demonstrating a variety of strategies or behavioural responses as a reaction to challenge or adversity (Duhachek, 2005). Coping and resilience have been used interchangeably in the literature (Markstrom, Marshall, & Tryon, 2000), with some researchers advocating that coping, stress, competence, and resilience should be viewed as the one construct (Haggerty, Sherrod, Garmezy, & Rutter, 1997; Masten & Obradović, 2006). However, while resilience and coping are similar and interrelated, they should be identified as a separate construct.

Coping is a dynamic process and refers to the constantly changing cognitive and behavioural efforts of an individual to manage external or internal demands, which are dependent on their levels of perceived stress (Frydenberg, 2010; Lazarus & Folkman, 1984). Coping occurs in response to a perceived challenging situation or anticipation of future challenging events. Similarly, adaption refers to the actions that enhance one's ability to cope with external stressors (Brooks, 2003). The coping response occurs before the individual's resilience capacity is demonstrated. Therefore, common sense seems to dictate that for resilience to be demonstrated, stressor/s must be present as an individual's adaptive capacity only becomes apparent in the face of hardship or adversity. This statement is in line with Toland and Carrigan (2011) who states that "individuals are not considered to have displayed resilience if there has never been a significant threat to their development" (p. 97).

A study by Markstrom et al. (2000) demonstrates how resilience and coping are interrelated. For example, the findings of the study showed that coping mechanisms such as wishful thinking and avoidance were negatively linked to resilience. Conversely, those who applied direct, problem-solving mechanisms increased their likelihood of responding positively to difficult situations.

#### **Theoretical Underpinnings of Stress and Coping.**

Exploring theoretical constructs of coping first began with Lazarus in the 1960's. His theories remain relevant in coping literature (Carver, 1997; Hulbert-Williams, Morrison, Wilkinson, & Neal, 2013; Lazarus, 1966; Matthews & Zeidner, 2003; Swanson, 2000; Wong, Reker, & Peacock, 2006). Lazarus (1984) developed the transactional model of stress and coping which is an appraisal-based model that describes the subjective processes of an individual's cognitive appraisals and coping responses (Lazarus & Folkman, 1987). This model presents coping as an interactive process which is associated with three dimensions'; (a) primary appraisal, whereby a person perceives a threat; (b) secondary appraisal, which is the consideration of the appropriate action; and (c) coping, which is when the person carries out that action (Carver, Scheier, & Weintraub, 1989; Lazarus, 1984; Lazarus & Folkman, 1987; Wong et al., 2006) (see Figure 2.3). This model details coping to be a dynamic system which changes from moment to

moment. The transactions within this model depend on the effect that the environment or external stressors have on the individual. The impact of the environment is then mediated by an individual's repeated appraisal (assessment) of the challenge and their coping responses to the stressor. These responses to stress can be physiological, emotional, cognitive, behavioural and/or physical.

As Figure 2.3 demonstrates, there are three types of primary appraisal: (1) positive, (2) dangerous, and (3) irrelevant. If an individual's interaction with the environment and stressor is perceived to be positive, benign-positive appraisals occur. These appraisals are characterised by pleasurable emotions such as joy, love, happiness, exhilaration, or peacefulness (Lazarus & Folkman, 1984). Dangerous or 'stressful' appraisals include harm/loss, threat and challenge. Harm and loss can be physical, social, financial or psychological and implies the 'damage' to an individual has already occurred (Lazarus & Folkman, 1984). For example, damage to self-esteem or social embarrassment. Whereas, a 'threat' is when the 'harm or loss' is anticipated but has not yet occurred. Threats tend to initiate negative emotions such as anxiety, fear and anger. On the contrary, challenge appraisals are seen to focus on the gain or potential growth of an experience. Challenge may be expressed as pleasurable emotions such as excitement, motivation and exhilaration. On the other hand, if the environment has no implication on an individual's well-being, their appraisal may deem the stressor as irrelevant. Irrelevance is where there is no value, need, commitment or anything to be lost or gain in their transaction with the stressor (Lazarus & Folkman, 1984).



Note. Adapted from Lazarus & Folkman (1984, 1987)

Figure 2.3. Lazarus's transactional model of stress and coping.

### Types of coping.

Lazarus's (1984) model as summarised in Figure 2.3 demonstrates the need for young people to be able to draw upon and utilise sufficient internal and/or external resources or assets (e.g., personal motivation, coping skills, support from family) in order to minimise stress when they are confronted with challenges or adversity. The model clearly represents two types of coping; problem-focused and emotion-focused coping. These coping skills are required for a person to change the situation or perspective, to be able to overcome the stressor/s. Lazarus and Folkman (1984) suggests that coping responses should not be labelled adaptive or maladaptive, or seen as inherently good or bad. Instead, coping skills should be viewed as either emotion-focused or problem-focused coping skills. In addition to these two broad types of coping, researchers have added unproductive as another type of coping (Carver et al., 1989; Fanshawe & Burnett, 1991). Unproductive coping refers to how appropriate the application of the coping skills is for the specific stressor (Carver et al., 1989).

If a challenging situation is perceived as unchangeable, such as experiencing grief from a death in the family, emotion-focused coping strategies are more likely to be appropriate (Frydenberg, 2008). Whereas, if an individual perceives they can alter the situation, they are more likely to apply problemfocused coping strategies to confront the problem, such as acquiring more information or looking for alternative solutions (Breinbauer & Maddaleno, 2005).

Emotion-focused coping is typically based around an emotional response to a challenge or stressor, such as fear, anxiety, embarrassment, excitement or frustration. This type of coping naturally involves the person trying to lower the effect of the negative emotional response due to the challenge or stress. Breinbauer and Maddaleno (2005) suggests that emotion-focused coping strategies, such as choosing to ignore a problem, are only effective in the short term, while problem-focused coping strategies are more effective for long-term solutions. Some emotion-focused coping strategies are seen to be proactive, such as venting or seeking social support, whilst other strategies such as avoidance or denial are seen as reactive (Carver et al., 1993).

On the contrary, problem-focused coping typically tackles the causes of challenge or stress in practical ways and aims to deal with the challenge or reduce the stress. Problem-focused coping normally comprises practical strategies such as problem-solving or time management. These types of coping strategies are considered as engaging coping strategies, as the actions involve a person confronting the problem and adopting active coping skills, such as seeking support from counsellor or family and friends (Breinbauer & Maddaleno, 2005). For example, a study by Lee et al. (2017) examined coping responses in high school students from Year 7 to Year 12 in South Korea (N = 1446). Researchers used three different scales; the Risk Factors Scale (Kim, 2004) to measure risk factors of individuals; the Korean version (Kim, 2001) of the Coping Inventory for Stress Situations (Endler & Parker, 1990) to measure typical patterns of coping strategies for stressful situations; and Kim (2004) combined a self-report measure from other scales to measure school adaptability levels. While the study

showed strength by using a large sample size of adolescents, using two self-report measures that were newly derived for Korean populations constituted a limitation.

Nevertheless, based on their results, participants were classified into four groups. If a participant showed high adaption and high risk, they were classified as to be in the resilient group (Lee et al., 2017). If the group showed high adaption but low risk, they were classified as competent. Whereas, if the participant scored high risk levels, but low adaption, they were classified to be in the struggling group. Finally, if the participant demonstrated low risk and low adaption, they were classified as the vulnerable group. Interestingly, only 214 (14.8%) of the participants were classified to be in the resilient group. Whereas, 747 students (51.7%) were classified into either the struggling group (n = 358, 24.8%) or the vulnerable group (n = 389, 26.9%). Results from a descriptive discriminant analysis revealed that both problem-focused and emotion-focused coping skills were used simultaneously in the resilient group. Whereas, the competent group used predominantly only problem-focused coping skills and the struggling group mainly used emotion-focused coping strategies. One of the most interesting results is that the vulnerable group used neither emotion-focused or problemfocused coping strategies. Due to the fact that both the competent and resilient groups used both problem-focused coping strategies, this finding reinforces that problem-focused copings skills are very useful in overcoming challenge or adversity.

Figure 2.3 also shows the need for self-reflection and continual assessment to gain awareness and learning in order to apply the skills in future challenging

situations. Once the emotion-focused or problem-focused coping strategies have been implemented, the individual is required to evaluate the strategies as to whether they are successful in mitigating the challenge. If the strategy is unsuccessful, a reappraisal is required, and the flow of the model starts again. An example may be a young person feeling stress relating to the submission of an assignment for their science subject at school. Their initial appraisal may result in an understanding that there is not any direct danger or harm to self. However, they may find they are exposed to 'loss', such as the loss of grade's due to the quality of their assignment. This may lead to a reduction of their personal pride, self-esteem or they may fail the subject. To avoid stress and overcome the challenge, a secondary appraisal of the situation may involve drawing on their required resources. This may include gaining support from a classmate or science tutor or applying effective coping mechanisms, such as problem-solving skills or time management to complete the assignment to a satisfactory standard. It is important to note that appraisals can be either conscious or unconscious reactions to the stressor and the reactions can be manipulated by context, time and personal factors (Hulbert-Williams et al., 2013; Schwarzer & Knoll, 2003).

### **Understanding the Concept of Resilience**

The concept of resilience is widely investigated and applied across various disciplines including engineering, ecology, education, economics, psychology, psychiatry, sociology, and biological disciplines (Herrman et al., 2011, p. 259; Windle, 2011). The diversity of research areas and theories of this

multidimensional construct has caused confusion and difficulty in researchers agreeing on a clear definition.

Resilience researchers in the domain of psychology have provided significant variations in definitions over the past four decades, which has created further confusion and misunderstandings of the key concepts (Masten, 2001, 2012; Richardson, 2002; Werner & Smith, 1982). However, the commonality between the definitions of resilience in all fields of study is that resilience is the capacity to cope with stress (Neill & Dias, 2001).

The American Psychological Association (2011) defines resilience as the "process of adapting well in the face of adversity, trauma, tragedy, threats or significant sources of stress- such as family and relationship problems, serious health problems or workplace and financial stressors. It means 'bouncing back' from difficult experiences'" (p. 1). This understanding of resilience and the idea of 'bouncing back' can etymologically be traced back to the Latin words *resilio, resilire, resilui*; meaning *to leap back, spring back, rebound, recoil or retreat* (Lewis, 1890). This foundational understanding has informed the commonly used, simplistic description of resilience as one's ability to 'bounce back' or 'rebound' from stress or adversity (Bounce Back, 2017; Netuveli, Wiggins, Montgomery, Hildon, & Blane, 2008; B. W. Smith, Tooley, Christopher, & Kay, 2010; Spangler, Koesten, Fox, & Radel, 2012; Tugade & Fredrickson, 2004; West, 2012). However, while this simple explanation of resilience addresses the critical aspect of positive adaption to stress, the definition fails to encompass the

interaction between the individual and their environment throughout the process of adaption.

### **Risk and Protective Factors**

Within the domain of building resilience, it is essential to understand the factors that may hinder (risk factors) or strengthen (protective factors) an individual's capacity for adaption. Risk and protective factors are the individual, family and environmental characteristics that may contribute negatively or positively to an individual's ability for optimal human functioning. These variables include personal, behavioural, biomedical, genetic, environmental and demographic factors (Australian Institute of Health and Welfare, 2011; Banatao, 2011; Herrman et al., 2011). Starting from birth, an individual's risk and protective factors are moderated by the environment in which they are immersed.

Rarely occurring in isolation, risk factors are associated with an increased probability of negative outcomes, whereas protective factors or promotive factors are associated with the likelihood of positive outcomes (Barrett et al., 2014; Masten, 2001; O'Dougherty Wright & Masten, 2015; World Health Organization, 2012b). As risk factors accumulate (also referred to as cumulative risk factors) in people's lives their ability to demonstrate resilience becomes less likely and negative outcomes increase (Evans, Li, & Whipple, 2013; Obradovic, Shaffer, & Masten, 2012). Protective factors (originally referred to as positive risk factors), help to reduce the probability of an individual being affected negatively by risks or stressors (Benard, 1991, 1995; Ungar, 2004). The more protective factors an individual has, the more likely they are to be able to cope with adversity and

return to a baseline level of optimal human functioning (Benard, 1991, 1995; Ungar, 2004).

People can be supported to build their resilience through exposure to various challenging experiences occurring throughout a person's lifetime (Shonkoff, 2015). Studies, such as Bartels and Hudziak (2007) and Caspi, Taylor, Moffitt, and Plomin (2000) have shown that biological factors can predict positive developmental outcomes. In addition, both positive and negative experiences help to shape an individual's capacity to positively adapt to challenges (Fredrickson & Losada, 2005; Shonkoff, 2015; Shonkoff et al., 2009). For example, an individual may have a negative experience (e.g., being unsuccessful in making the cut for the school soccer team or not passing their school maths exam) which then can set off a trigger to self-reflect. This may lead to self-motivation by the person, to practice the skills required to overcome the challenge and be more likely to be successful in the future.

#### Developmental assets as protective factors.

Developmental assets are protective factors that focus on the growth of core skills that are required for a young person's future success (Havighurst, 1948; O'Dougherty Wright, Masten, & Narayan, 2013; Scales & Leffert, 2004; Schoeppe, Haggard, & Havighurst, 1953; The Search Institute, 2016). Examples include fostering empowerment from challenging experiences, the development of positive identity, and actively developing supportive, positive relationships. Developmental theory details that youth who acquire developmental assets and tasks have a higher probability of coping with challenge and are more likely to positively adapt and overcome adverse situations (Benson, 1997, 2007; Benson, Leffert, Scales, & Blyth, 2012; Benson, Scales, Leffert, & Roehlkepartain, 1999).

Developmental assets are categorised into either internal or external assets. Internal assets are the personal skills, self-perceptions, and values that contribute to young people taking responsibility of their own lives, making good decisions and being more independent (e.g., achievement motivation, personal responsibility, planning, self-efficacy and decision making). Internal assets include the personal strengths, characteristics or traits of an individual, such as problem-solving, autonomy, social competence and a sense of purpose (Benard, 1991, 1995, 2004). External assets are the supports, opportunities, and relationships that young people need across all areas of their lives (Benard, 1991, 1995, 2004). External assets, otherwise known as environmental assets, refer to the meaningful relationships between an individual and their surroundings such as school, family, community and peers (Benard, 1991, 1995, 2004).

Developmental assets, also considered as psychosocial attributes, are different to developments tasks. Developmental assets relate to the positive relationships, competencies, opportunities, values and self-perceptions that youth need to acquire to succeed in life (Scales & Leffert, 2004). Whereas, developmental tasks are age-related standards of behaviour which span across a variety of areas of an individual's life, including emotional, cognitive, moral, behavioural, social, spiritual and physical areas of achievement or functioning (e.g., self-sufficiency, learning to get along with peers and acquiring a set of values to guide their behaviour) (McCormick, Kuo, & Masten, 2011). These tasks generally serve as the criteria to judge how well a person is managing their life (O'Dougherty Wright et al., 2013; Piaget, 1972; Scales & Leffert, 2004).

Research has shown that we have the capacity to develop internal and external assets of resilience through direct personal experiences (Passarelli, Hall, & Anderson, 2010; Scales & Leffert, 2004; Sesma Jr, Roehlkepartain, Benson, & Van Dulmen, 2003; Vera & Shin, 2006). For example, Jain, Buka, Subramanian, and Molnar (2011) conducted a community-based multilevel longitudinal study over seven years with young people aged 11-16 years (N = 1,166) to determine if protective factors (developmental assets) would build emotional resilience among an ethnically diverse sample of at-risk youth who were exposed to violence. Their study found that supportive relationships were strong predictors of emotional resilience. Participants of the study, who spent hours on structured activities that provided meaningful experiences in their early adolescent years, reported a significant influence on building their emotional resilience. However, interestingly this was only the case for the group who was not exposed to violence. Groups who were either a victim or witness to violence were not affected by participation in structured activities. The young people in this study who had the highest risk factors also demonstrated the lowest number of average assets, highlighting that these young people would benefit from developing external assets such as positive relationships with peers, teachers and their families to help them cope with adversity. These findings demonstrate the importance of youth having access to programs and experiences that aim to develop resilience and developmental assets.

### **Resilience as a Trait or System**

Theories of resilience detail the concept of resilience from two approaches; at an individual level and at a systems level. Firstly, at an individual level, resilience has been viewed as the psychological capacity that enables someone to maintain or regain mental health while overcoming adversity (Goldstein & Brooks, 2005; Herrman et al., 2011; Kaplan, 2005; Neill & Dias, 2001; Wagnild, 2009). This is also referred to as psychological resilience, mental resilience or mental toughness (Booth & Neill, 2017; Lin, Mutz, Clough, & Papageorgiou, 2017; Neill & Dias, 2001). Psychological resilience is categorised differently from other types of resilience (e.g., physical resilience, environmental resilience, social resilience etc.) as it specifically refers to an individual's ability to maintain, regain or improve one's mental health (Neill & Dias, 2001).

The second approach examines resilience at a systems level, where resilience is viewed as the dynamic interaction of a person's social and physical environments that contribute to the action taken to overcome adversity and maintain or regain mental health and well-being (Masten, 2015). Analysing resilience from these two approaches raises the issue of whether to define resilience as a trait, outcome or dynamic process.

# Individual level: Resilience as a trait.

Originally, resilience inquiry identified personal characteristics or traits of individuals who survived and thrived through high-risk situations, stressors or adversity. Virtually all early studies of resilience focused on the individual capacity of the person. Interactions between biological and environmentalsystemic factors (e.g., family, school, institutions, or community) that assist in optimal human functioning were acknowledged, but in these earlier studies the focus remained on the qualities of the individual, not the surrounding social and physical environment (Kaplan, 1999; Ungar, Dumond, & McDonald, 2005; Wagnild & Young, 1993). Earlier research defined resilience as a set of traits (inherited features) or characteristics (personal distinctive qualities) that acted as protective factors to support the individual through adverse situations (Masten & O'Connor, 1989; Rutter, 1985, 1987; Werner & Smith, 1982, 1992). This initial approach formed a crucial foundation in determining the resilient characteristics and traits deemed to be beneficial to assist people to not only survive but thrive in adverse situations.

Internal assets (personal characteristics) such as having a purposeful life (Edward, Welch, & Chater, 2009), self- efficacy (Bandura, 1977), self-esteem (Dumont & Provost, 1999), perseverance (Wagnild, 2009), self-reliance (Samuels & Pryce, 2008) and equanimity (Stagman-Tyrer, 2014) are some examples of resilience attributes that have been associated with improving positive adaption in the face of adversity. Miller (2016) provides a clear example of 42 traits and characteristics that have been identified in resilient people (see Table 2.1).

## Table 2.1

### Examples of characteristics of resilient individuals

Resilience characteristics		
Able to problem solve	Courage	Persistence
Adaptability	Empathy	Presence
Altruism	Energy	Purpose
Attunement	Faith	Regulate behaviours
Autonomy	Flexibility	Regulate emotions
Awareness	Норе	Regulate thoughts
Calm	Intention	Resonance
Clarity	Intimacy	Resources
Commitment	Joy	Responsibility
Communication	Love	Self-acceptance
Compassion	Meaning	Self-confidence
Competence	Open-mindedness	Stability
Connection	Patience	Trust
Consistency	Perseverance	Value

Note. Adapted from Miller (2016)

These traits and characteristics can be extended further to incorporate different strengths. For example, problem-solving encompasses planning and resourcefulness; autonomy involves self-efficacy, self-awareness, and mindfulness (Bandura, 1977); social competence, comprises of social skills, communication skills, empathy and caring, and the ability to prompt positive responses from others (responsiveness) (Benard, 2004; Masten, 2001); and a sense of purpose entails goal direction, achievement motivation, optimism, and hope (Benard, 1991, 1995, 2004).

#### Systems level: Resilience as a dynamic process.

Conceptualising resilience as an individual's traits or characteristics ignores the wider contextual factors of the social and physical environment. Resilience is not a personality characteristic (Toland & Carrigan, 2011), resilience is now regarded as a dynamic process of interactions between internal predispositions, protective factors, experiences and external assets (Bandura, 1994; Burdette & Whitaker, 2005; Ewert & Yoshino, 2008; C. Hammond, 2004; Luthar, Cicchetti, & Becker, 2000; Masten, 2001, 2009; Miller, 2015, 2016; Rutter, 2008; Ungar, 2008, 2011; Windle, 2011). A primary issue of past resilience theory has been the inability to develop a definition of resilience which encapsulates the dual focus of both the individual and an individual's socialecology (Ungar, 2008).

#### **Resilience and Social-Ecology**

The development of an individual is the direct result of the relationships and interactions between themselves and their environments within their ecological systems (Bronfenbrenner, 1981; Ungar, 2012). Human development, including the development of resilience, is consequently the outcome of the complex interactions between a growing individual and their environment (Bronfenbrenner, 1981).

Bronfenbrenner's (1977) ecological model details the interactions between an individual's biological, psychological and social-cultural systems (see Figure 2.4). The model demonstrates that an individual's development is dependent on their relationships with their surrounding microsystems which include, family, school, and community systems. Mesosystem interactions are the interactions between microsystems. Within the micro- and mesosystems, a young person's protective factors are generally identified across three levels of functioning:

- 1. Individual (e.g., psychological, neurobiological, behavioural).
- 2. Social (e.g., peer support, family cohesion, parental support).
- 3. Community/societal (e.g., support systems generated through social and political capital, and institutional and economic factors, such as schools) (Garmezy, 1985; Werner, 1995, 2000; Windle, 2011).

Developmental resources within mesosystems that are designed to support personal growth, such as counselling, school camps and academic support programs, have a positive correlation with people's ability to cope with challenges (Ungar, 2011). Consistent with other researchers, Bronfenbrenner (1977, 1981) also suggests that positive growth occurs when conditions are optimal (Hanson & Kim, 2007; Olsson, Bond, Burns, Vella-Brodrick, & Sawyer, 2003). An example of optimal conditions for positive growth would be when an individual is feeling loved and supported not only in their home environment, but they are also feeling supported and encouraged while they are in their school and community environments. Positive adaption is not the result of individual characteristics nor social environments alone, but rather involving the complex interplay between the two.



Note. Adapted from (Garbarino, 1982) and (Bronfenbrenner, 1977, 1981)

Figure 2.4. The ecology of human development systems model.

It is the external assets (e.g., positive relationships with peers, family and other significant adults) which make up a person's social-ecology that is more likely to impact developmental pathways of resilience, rather than the impact of individual internal assets alone (e.g., positive values, positive identity etc.) (Ungar, 2011) (also see Table 2.1). In other words, 'nurture trumps nature'. Table 2.2 provides examples of risk and protective factors which occur for a person across their microsystems and mesosystems. The interactions between these systems are demonstrated in Figure 2.4. It should be noted that changes which occur in one system have the potential to then flow over and impact other systems (Masten & Cicchetti, 2010). This can have either positive or negative outcomes depending on the system affected. For example, if an individual has risk factors in their family environment, such as family conflict, it is more likely the individual will be affected negatively in another system, such as their school life and academic achievement. Protective factors operate in a similar manner with positive development occurring in other systems (see Table 2.2) (Ungar, 2012).
# Table 2.2

Examples of risk and protective factors at microsystem systems levels

Microsystem	Risk factors	Promotive and protective factors
Individual	Difficult temperament	Easy temperament
	Poor social skills	Adequate nutrition
	Low self-esteem	Problem-solving skills
	Physical and intellectual disabilities	School achievement
	Impulsivity	Above average intelligence
	Low intelligence	Internal locus of control
	Alienation	Social skills and competence
	Chronic illness	Optimism
	Insecure attachment	Positive self-related cognitions
		Values and moral beliefs
Family	Homelessness	Supportive, caring parents
	Divorce or family break up	Responsibility within family
	Family conflict	Strong family norms and morality
	Violence in the home	Supportive relationship with adult other
	Physical, sexual or emotional abuse	than parents
School	Bullying	Sense of belonging/ connectedness
	Peer rejection	Positive peer relationships
	Inadequate behaviour management	Positive behaviour management
	School failure	Opportunities for success and
	Member of deviant peer group	recognition of achievement
	Racism	Required responsibility/helpfulness
	Homophobia	Positive teacher-student relationships
	Poor attachment to school	Collaborative teaching strategies
	Family/parent disengagement in	Positive school climate
	school	Engagement in learning
Community	Isolation	Sense of belonging/ connectedness
	Social or cultural discrimination	Strong cultural identify
	Lack of support services	Participation within community groups

Note. Adapted from Bond et al., (2007); Catalano, Oesterle, Fleming, & Hawkins, (2004); Masten, (2009); Steinhardt & Dolbier, (2007); Stewart,

McWhirter, Rowe, Stewart, & Patterson, (2007); Werner, (2000).

Strengthening protective factors within an individual's micro and mesosystems while increasing the developmental resources within a person's socialecology, can enhance his or her capacity to cope with adversity (O'Dougherty Wright & Masten, 2015). Their adaptive capacity relies on their ability to alter or change their characteristics or behaviour to cope better with the existing or anticipated stressor (Brooks, 2003). Their competence is demonstrated by their adaptive use of personal or contextual resources to implement developmental tasks which are age-appropriate (O'Dougherty Wright & Masten, 2015).

Positive adaption can be expressed across systems as well as within the individual. As resilience is comprised of many attributes which are strongly associated with positive social interactions (e.g., self-esteem and self-efficacy) (Cason & Gillis, 1994; Miller & Allen-Craig, 2005; Neill & Richards, 1998), exposure to situations that involve supportive peers, positive teacher influences and opportunities for success, can improve resilience in young people (Olsson et al., 2003). Table 2.3 provides examples of the bio-psycho-social-cultural systems and processes that can positively impact a young person's capacity to adapt to challenges.

# Table 2.3

Examples of bio-psycho-social-cultural systems and processes that can implicate fostering resilience in young people

Domain	Systems	Examples of bio-psycho-social-cultural systems and
		processes
Within the	Microsystems	Genetic moderators and epigenetic processes
individual		Positive physical health and immune function
		Adaptive self-regulation system (physiological, emotional,
		cognitive, and behavioural)
		Adequacy of stress response systems
		Strong cognitive and problem-solving abilities
		Agency and an effective mastery motivation system
		Adaptive temperament and personality
Within the	Microsystems	Close attachment relationships
family	and	Positive extended family and kinship ties
	Mesosystems	Cohesiveness, structure, and support within the family
		Effectiveness of parenting in the cultural context
		Family rituals, values, and beliefs
Within the	Microsystems	Safety of the physical environment
community	and	Effective education system
	Mesosystems	Peer friendships with positive values and norms
		Presence of religious and spiritual communities
		Good public health care and social services
		Access to recreational facilities
Within the	Macrosystems	Belief systems that give life meaning and purpose
culture and		Protective child policies (child labour, child health and
society		welfare policies)
		Availability and adequacy of emergency response systems
		Access to material resources
		Global relationships with international community

Note. Adapted from O'Dougherty Wright & Masten (2015).

As demonstrated in the above tables, microsystems such as schools play a critical role in developing a young person (Bronfenbrenner, 1981). While some schools do not always operate in supportive ways for youth (Toland & Carrigan, 2011), they are an important part of promoting a young person's social competence and well-being by enhancing environmental protective factors and developmental assets such as fostering positive relationships and providing encouragement (Greenberg et al., 2003; R. D. Taylor & Dymnicki, 2007). Importance has been placed on school experiences and programs as they are seen as critical components to promoting resilience in young people (Toland & Carrigan, 2011).

The more adolescents are exposed to stressors and risk factors, the more they will benefit from developing protective factors, such as positive relationships with significant adults (e.g., mentors, coaches, teachers) (Lerner, Napolitano, Boyd, Mueller, & Callina, 2014; Ungar, 2011). Programs that include youth mentoring have been shown to develop the competence and character of adolescents by developing a personal relationship in which a caring individual provides consistent companionship, support, and guidance (Keller, 2007).

## **Defining Resilience**

Resilience is a multi-dimensional construct which integrates multiple levels of analysis and investigates the way systems interact to shape human development (Allan, McKenna, & Hind, 2012; Miller, 2015; Stokes, 2009). It is a dynamic process in which a person's capacity to positively adapt, expands beyond the internal skills and traits of the individual to his or her social connections and environment. An individual's capacity to demonstrate resilience is distributed across and connected with their biological, psychological and social-cultural systems (O'Dougherty Wright et al., 2013). Resilience should always be discussed in the context of adversity, as facing challenges illuminates people's responses which requires them to draw on resilience traits, assets and resources (Schoon, 2013). Masten and Coatsworth (1998) suggests that resilience is based on two fundamental judgments: (1) is a person 'doing ok'? and, (2) is there now, or has there been any significant risk or adversity to be overcome? Therefore, the construct is defined by two aspects:

- 1. Exposure to adversity, and
- Effective adaptation in the face of that risk (Luthar et al., 2000; Masten, 2001; Toland & Carrigan, 2011).

In Ungar's (2011) more recent explanation of resilience, he provides a definition which encompasses both the context of exposure to significant adversity as well as positive adaption. He defines resilience as "both the capacity of individuals to navigate their way to the psychological, social, cultural and physical resources that sustain their well-being, and their capacity individually and collectively to negotiate for these resources to be provided in culturally meaningful ways" (Ungar, 2008, p. 225). This definition acknowledges the complex interplay of the relationships, processes and protective mechanisms of resilience within a social-ecological framework. In summary, resilience is defined as an individual's capacity to utilise their internal and external assets to maintain or regain personal well-being when confronted with adversity.

49

#### **Related Constructs of Resilience**

As resilience is related to an individual's capacity to adapt to stressors, it is evident that resilience is closely related to many other constructs and concepts, including mental toughness (Booth & Neill, 2017), executive functioning (Bradley, 1990; Lee, Bull, & Ho, 2013), psychological flexibility (Kashdan & Rottenberg, 2010), self-efficacy (Bandura, 1977), grit (Duckworth & Eskreis-Winkler, 2013; Von Culin, Tsukayama, & Duckworth, 2014), hardiness (Bartone, 1995; Florian, Mikulincer, & Taubman, 1995) and self-determination (Hattie et al., 1997). These positive human traits are essential factors in building a 'pathway' to resilience (Bonanno, 2004), however researchers indicate that these constructs may not be included in the definition as it may cause confusion to the understating of resilience (Smith et al., 2008).

For instance, psychological resilience can also be interpreted as mental toughness (Booth & Neill, 2017). Mental toughness is used as an umbrella term to describe positive psychological resources that are crucial for mental health (Lin et al., 2017). As a cognitive process, mental toughness can be defined as disciplined thinking. The findings of a systematic review by Lin et al. (2017) indicates that mental toughness is associated positive with psychological traits, effective coping strategies and positive outcomes in both education and mental health. Similarly, hardiness has been defined as a combination of managing commitment, control and challenge, and to demonstrate the ability to turn a stressful situation into an

50

opportunity for growth (Bartone, 1995; Florian et al., 1995). The next section provides a brief summary of related resilience constructs considered in this study.

## **Executive Functioning**

Executive functioning is directly related to one's ability to cope with stressors as it refers to an individual's ability to regulate and apply purposeful control over emotional responses, behavioural impulses and cognitive processes, such as attentional control, working memory, and planning (Bradley, 1990; Lee et al., 2013). Executive functions are "an array of mental processes responsible for regulation of cognitive functioning during purposeful, goal directed problemsolving behaviour" (Dehn, 2014, p. 27).

## **Psychological Flexibility**

Closely related to executive function is psychological flexibility. Kashdan and Rottenberg (2010) defines psychological flexibility as the measure of how a person; (1) adapts to fluctuating situational demands, (2) reconfigures mental resources, (3) shifts perspective, and (4) balances competing desires, needs, and life domains.

Psychological flexibility has been found to increases levels of resilience, self-reliance, meaningful experiences, confidence, the ability to master challenges and also improve the time response in recovery from stressful events (Gjerde, Block, & Block, 1986). Conversely, individuals that score low in psychological flexibility experience higher levels of negative life outcomes, such as depression, anxiety, an increase in substance use and reduction in the ability to learn (Coulson, 2017).

## **Self-efficacy**

Another important human characteristic for all people, but especially adolescents is self-efficacy. Self-efficacy is directly linked to self-esteem and refers to an individual's personal expectation that they can achieve their desired goals or outcomes in specifically chosen areas (Bandura, 1977). A young person's perception of their self-efficacy can be enhanced by the development of selfreflection skills and self-regulatory processes. For instance, when a young person evaluates the success of a challenging goal achieved through perseverance and planning, their self-efficacy will be enhanced (Zimmerman & Cleary, 2006). An example of such a challenge that appears difficult but achievable may be a young person participating in a public speaking contest in front of the entire school. In contrast to this, if a young person has unrealistic expectations or is pushed into attempting unachievable goals, their self-esteem and self-efficacy may be negatively affected.

Human characteristics such as self-efficacy, resilience and self-esteem can increase if young people are given the opportunities to participate in challenging experiences which allow them to set and attempt authentic goals, where they can undertake concrete experiences, and also receive feedback from significant others (e.g., peers, teachers or parents) (Schunk & Meece, 2006; Zimmerman & Cleary, 2006). Even if the attempts at achieving the goals or overcoming the challenges are unsuccessful, opportunities for reflection where participants can identify the

52

reasons for failure can also support the growth of self-efficacy and self-esteem (Kidd & Shahar, 2008; Priest & Gass, 2005; Swarbrick, Eastwood, & Tutton, 2004).

#### **Internal Locus of Control**

Closely linked to self-efficacy is the concept of internal locus of control. This is a personality construct that determines how individuals perceive their success and failure of outcomes (Hans, 2000; Hong, Shull, & Haefner, 2011). Psychological resilience has been found to moderately to strongly correlate with higher internal locus of control (r = .4 to .7) (Jew, Green, & Kroger, 1999). The links between internal locus of control and other factors of resiliency are widely researched, including; locus of control, self-efficacy and motivation (A. Anderson, Hattie, & Hamilton, 2005); locus of control, self-efficacy, persistence, and commitment (Hong et al., 2011); locus of control and personal effectiveness (J. W. Johnson, 2012); and the effects of adventure programming on locus of control (Hans, 2000).

The meta-analysis conducted by Hans (2000) examined the effects of adventure therapy programming on locus of control, with data representing 24 studies (30 effects, 1,632 participants). The results demonstrated a small-moderate effect size of 0.38. Specifically, two significant factors were highlighted; (1) programs that included therapeutic goals as their primary purpose had higher effect sizes than other programs that had general goals, such as recreation; and (2) sessional programs that included adventure activities but did include overnight trips away from an unfamiliar environment, were less effective than programs of residential and semi-residential capacity, that removed participants from familiar environments overnight.

The study by Jew et al. (1999) also established that individuals who have higher levels of resilience also tend to demonstrate the following characteristics;

- higher internal locus of control,
- better academic skills,
- higher self-perceived competencies in athletic performances, friendships, scholastics and jobs,
- exhibit a more extensive repertoire of coping skills, and
- increase in a variety of personal beliefs that enable him/her to acquire and use more effective coping skills in times of stress (Jew et al., 1999).

In addition to the research presented here, Ungar (2011) suggests that locus of control is required to navigate both the internal assets and external resources for positive development and resiliency of an individual.

# **Measures of Resilience and Coping**

There are numerous instruments that are available to measure resilience with a range of target populations (e.g., The Connor-Davidson Resilience Scale (Connor & Davidson, 2003), California Healthy Kids Survey - The Resilience Scale of the Student Survey (Sun & Stewart, 2007), the Dispositional Resilience Scale (Bartone, 1995, 2007) and the Brief Resilience Scale (Smith et al., 2008). There are also various resilience instruments that have been used successfully with adolescent populations (e.g., The Child and Youth Resilience Measure (Ungar et al., 2008), Adolescent Resilience Scale (Oshio, Kaneko, Nagamine, & Nakaya, 2003), The Ego-Resiliency Scale (Block & Kremen, 1996) and Youth Resiliency: Assessing Developmental Strengths (T Donnon & Hammond, 2007; Tyrone Donnon, Hammond, & Charles, 2003). However, a review of the literature found the Resilience Scale (Wagnild & Young, 1993) to be the most appropriate and effective tool to measure resilience and resilience attributes with young people in outdoor education settings because:

- It has been successfully used to measure changes in adolescent resilience (Ahern, 2006; Gillespie & Allen-Craig, 2009; Skehill, 2001) and has provided data to support programs that potentially help youth at risk increase resilience (Fenemor et al., 2008).
- The Resilience Scale has been used to measure resilience in outdoor education contexts, which have shown positive changes in resilience scores (Neill & Dias, 2001).
- The Resilience Scale is psychometrically validated and has satisfactory psychometric properties (Ahern, 2006; McDonald, Jackson, Wilkes, & Vickers, 2012; Yoshino, 2008).
- The length of the Resilience Scale is an appropriate length for the attention span of adolescent's (Wagnild, 2009; Wagnild & Guinn, 2011).

- The language used in the scale is appropriate for the understanding and interpretation of questions with adolescent populations (Wagnild & Guinn, 2011).
- The conceptual framework and design of the measure aligns with outdoor education philosophies for personal growth and development (Ewert & Yoshino, 2008; McDonald et al., 2012).

Even though coping and resilience are similar and interrelated (Haggerty et al., 1997; Masten & Obradović, 2006), they require separate measurements as they are different constructs. Coping measures tend to focus on measuring specific cognitive and behavioural actions of an individual, whereas the resilience measures focus more on how the individual feels they are as a person. For example, the Resilience Scale uses statements such as 'I am determined' or 'I am resilient' (Wagnild & Guinn, 2011).

There are also various instruments available to measure coping such as the Life Events and Coping Inventory (LECI): Targeting Children (Schwarzer & Schwarzer, 1996), the Adolescent Coping Orientation for Problem Experiences Inventory (A-COPE) (Patterson & McCubbin, 1987) the Adolescent Coping Scale (Frydenberg & Lewis, 1993) and the original COPE Inventory (Carver et al., 1989).

The original COPE Inventory (Carver et al., 1989) was designed from a compilation of the Lazarus (1984) model of coping, relevant coping literature at the time and the model of behavioural self-regulation (Carver & Scheier, 1990; Carver & Scheier, 1981). However, upon review the original multidimensional 60

item COPE Inventory, the scale was shortened to a 28-item version, called the Brief COPE (Carver et al., 1989). The Brief COPE scale is a well-known instrument used to measure coping skills and has been implemented with a variety of target populations worldwide (Artinian et al., 2009; Badr, 2004; Glass, Flory, Hankin, Kloos, & Turecki, 2009; Hastings et al., 2005).

Research has shown the Brief COPE to be the most suitable instrument to measure coping skills with youth in outdoor education settings (Ewert & Yoshino, 2008), compared to the original 60-item COPE Inventory and other coping inventories for numerous reasons, such as:

- The Brief COPE did not include items that were deemed to be redundant or items that were proven not to be of value in previous research with the original 60-item COPE Inventory (Carver, 1997).
- The Brief COPE was deliberately designed for use in natural settings and therefore provides a suitable scale for outdoor education programs (Carver, 1997).
- The minimal length of the scale was favoured to cater for time constraints and attention spans of adolescent participants (Carver et al., 1993).
- 4. The language used in the scale was interpreted as appropriate for youth comprehension and has also been used successfully with adolescents in an outdoor education context (Ewert & Yoshino, 2008).

#### Approaches for the Development of Resilience and Coping

Numerous programs have targeted the development of resilience in young people, to enhance internal and external assets to assist them to manage difficulties and adversity in their lives (Beightol, Jevertson, Gray, Carter, & Gass, 2009; Ewert & Yoshino, 2011; Gillespie & Allen-Craig, 2009). As Ahern et al. (2008) suggests, resilience is developmental in nature, and with an active approach, it can be supported and promoted. However, even though resilience is a common process, not everyone has developed effective coping skills required to overcome the challenges of everyday life (Skehill, 2001). In order for young people to live at an optimal level of human functioning, it is important to support them in learning and practising coping skills and developmental assets to build their resilience capacity (Masten, 2001).

The terms 'therapy', 'treatment' and 'training' have been used to describe approaches to developing resilience, such as wilderness therapy (D. S. Berman & Davis-Berman, 1989), preventative treatment (Meichenbaum, 2017b) and stress inoculation training (Meichenbaum & Deffenbacher, 1988). However, Ballenger-Browning and Johnson (2010) suggests that renaming these approaches to 'resilience building' may help to reduce the stigma attached to the traditional approach to mental health treatment.

Resilience can be built through a variety of avenues, including through stress inoculation, setting attainable goals, decision making processes, building caring and supportive relationships within and outside the family and through

58

building social support in community groups (American Psychological Association, 2011; Benard, 1991; Werner & Smith, 2001).

Benard (2004) suggests three keys factors be included in frameworks to promote resilience in youth:

- 1. Caring relationships,
- 2. High expectations from adults, and
- 3. Meaningful ways to participate.

Along with caring relationships with adults, adults who have high expectations of youth and provide opportunities for meaningful participation, help to assist young people to develop resilience characteristics (Hanson & Kim, 2007). Other protective factors such as positive social relationships with peers and teachers have also been directly related to the enhancement of adolescent resilience (Olsson et al., 2003).

There are many resilience frameworks, models and interventions that aim to support student knowledge, skills and capacities to manage life challenges and maintain mental well-being across various target groups (Barrett et al., 2014; BoingBoing, 2012; MindMatters, 2018; Pathways Health and Research Centre, 2018; Scales, 2011). Successful resilience development frameworks appear to be underpinned by a number of overlapping theoretical approaches, including; social and emotional learning, stress inoculation, cognitive behavioural approaches and positive psychology. The next section will describe the approaches to building resilience that are specific to this research.

#### **Cognitive Behavioural Approaches**

Behaviours, thoughts and emotions all help to shape the learning process, therefore, when discussing personal growth and the concept of changing people and their behaviours, two widely accepted theories should be acknowledged; behaviouristic theory and cognitive behavioural change processes (also known as Cognitive Behaviour Therapy (CBT) when used in the context of therapy). Behaviouristic theory promotes that behaviour can be learned, and therefore, unlearned (Watson, 1913). This theory indicates that both positive and negative patterns of behaviour can be developed.

The basic premise of the cognitive behavioural change process is understanding that how a person thinks or feels will be the underlying reason for how that person reacts emotionally and behaviourally in variable circumstances (Long, 2011). Resilience development programs may support young people in becoming aware of thoughts, feelings and behaviours that may be influencing their personal development.

There are many cognitive behaviour modification techniques and somatophysiological coping strategies that are commonly used to aid in building resilience, and maintaining mental health and well-being (Lazarus, 1984), such as mindfulness (Lillis, Hayes, Bunting, & Masuda, 2009; Wilson & Baer, 2010), meditation (Miller, 2015), relaxed breathing techniques (O'Connell, 2005), journaling, progressive muscle relaxation (also known as body scanning), and positive reframing of thoughts (Antoni, Ironson, & Scheiderman, 2007; Bennett et al., 2013).

#### **Stress Inoculation as a Preventative Approach**

"Stress inoculation means acquiring sufficient knowledge, understanding and coping skills to facilitate better ways of handling expected stressful confrontations" (Lazarus & Folkman, 1984, p. 342). The concept of stress inoculation proposes that an individual's resistance or capacity to cope is enhanced by exposure to a stimulus that is strong enough to stimulate one's defences and coping mechanisms without being too intense that the stimulus becomes overwhelming for the individual (Meichenbaum, 2017b; Meichenbaum & Deffenbacher, 1988; Novaco, 1977).

To explain this concept, many researchers have used the analogy of immunisation to describe the relationship of how stress plays a vital role in enhancing resilience (Booth, 2015; Meichenbaum, 2017b; Rutter, 1993). For instance, young people are required to receive their immunisations against specific diseases to attend school in Australia, with the opinion that they will become immune to the disease through either small amounts or gradual exposure to a pathogen. Similarly, resilience can be developed through incremental exposure to challenges that simulate and test one's feelings of competence, comfort and safety (Masten & Reed, 2002). Researchers have termed this process the 'stress inoculation model' (Meichenbaum, 1996 ; Meichenbaum & Cameron, 1989; Meichenbaum & Deffenbacher, 1988; Neill, 2008).

Similarly, the principles of stress inoculation training (SIT) utilise a preventive approach build resilience (Meichenbaum, 2017b). SIT is a cognitive behaviour approach as it is a form of cognitive restructuring. It is used as a

method of changing an individual's thinking patterns about themselves and their lives (Antoni et al., 2007). SIT is used for a variety of types of people, including medical patients, psychiatric patients, individuals with performance anxiety, professional groups, as well individuals who have to deal with the stress of life transitions, such as coping with changes in adolescence and transitioning from junior school to secondary school (Meichenbaum, 2017b, p. 129).

SIT is designed to strengthen an individual's preparedness for adaption to challenges and help to develop a sense of mastery (Meichenbaum, 2017b, p. 122). The concept is based around helping people to firstly recognise their thoughts, behaviours and patterns, and secondly, to help them change negative thoughts that have been influencing their behaviour (Meichenbaum, 1977, 2017a).

The SIT model has three phases (Meichenbaum, 2017b, pp. 131-132). The first phase is the education phase; this involves understanding the way distressing emotions are generated, emphasising the cognitive factors and selfstatements (self-talk) that are involved in these emotions (Meichenbaum, 2017b; Novaco, 1977). Through the education process, individuals expand their awareness of their existing coping skills they use in response to challenges.

The second phase, known as the rehearsal phase, is where participants practice coping skills, such as using alternative self-statements while under emotional distress. This stage requires the participants to be provided, educated and coached about other ways to apply positive self-talk (Meichenbaum, 2017b; Novaco, 1977). This phase includes developing psychological tools and cognitive restructuring techniques, such as breathing, relaxation techniques and stopping negative thoughts. The strategies are designed to help participants become aware of their responses to challenging situations, help them to control unwanted, maladaptive thoughts and emotions, to help moderate emotional responses and evaluate their own performance (Meichenbaum, 1977).

The last phase of the stress inoculation model is the application and follow-through phase. In essence, this is the part of the programming where participants are 'inoculated' and have opportunities to test new learning and practice coping strategies Meichenbaum (2017b, pp. 131-132). This type of cognitive behaviour training may help participants to shift old habits and mindsets. For example, SIT has been used to influence people to utilise a 'growth mindset' where they can move from having a 'victim' mindset to having a 'growth', 'survivor' or 'thriver' mindset (Bennett et al., 2013; Dweck, 2006 ; Jelalian, Mehlenbeck, Lloyd-Richardson, Birmaher, & Wing, 2005; Payne, Youngcourt, & Beaubien, 2007).

#### **Positive Psychology**

The rapidly increasing domain of positive psychology is a strength-based approach in which the underpinning theory is based on resilience development and understanding adaptive mental, social, and physical health and well-being. According to Seligman and Csikszentmihalyi (2000, p. 5) positive psychology is "the scientific study of optimal human functioning (that) aims to discover and promote the factors that allow individuals and communities to thrive". As a means to address mental health and develop resilience attributes, this strengths-based approach includes the study of positive subjective experiences (i.e. well-being, flow, etc.), positive individual traits (i.e. optimism, forgiveness etc.), and positive institutions (Carr, 2011; Ewert & Yoshino, 2008; Seligman & Csikszentmihalyi, 2000; Snyder & Lopez, 2002).

Positive psychology aims to help young people to flourish by applying the six domains of the positive education model including; positive relationships, positive emotions, positive engagement, positive purpose, positive health and positive accomplishment (Norrish, Robinson, & Williams, 2011). Additionally, positive psychology focusses on developing one's character strengths and virtues (Park, Peterson, & Seligman, 2004), mindfulness, sense of life meaning and purpose, resilience attributes, and psychological well-being and mental fitness (Seligman, 2011).

Developing character strengths can have a significant positive impact on lives, including improved relationships, enhanced health and overall well-being and working as a buffer to help one manage and overcome problems (Park et al., 2004; Peterson, Ruch, Beermann, Park, & Seligman, 2007; Wood, Linley, Maltby, Kashdan, & Hurling, 2011). Character strengths are a universally recognised subset of personality traits that are morally valued (Park et al., 2004). Within the domain of positive psychology, there are several ways to understand character strengths. One of the most commonly used frameworks is the Values in Action (VIA) Framework (Park et al., 2004). Table 2.4 presents the 24 character strengths under six virtues within VIA framework (Park et al., 2004). Everyone possesses all 24 of the identified character strengths to some capacity; however, each individual's strengths range in the degree to which their unique character profile is operationalised.

Table 2.4Values in Action (VIA) framework of character strengths

The 6 character strength virtues	The 24 character strengths
Wisdom and knowledge	Perspective, creativity, curiosity, open- mindedness, love of learning
Courage	Persistence, honesty, bravery, and zest
Humanity	Social intelligence, kindness, love
Justice	Teamwork, leadership, fairness
Temperance	Self-regulation, forgiveness, modesty, prudence
Transcendence	Gratitude, appreciation of beauty and excellence, hope, humour, religiousness/spirituality

*Note*. Adapted from Park, Peterson, and Seligman (2004)

By supporting a young person to develop their emotional intelligence, they are more likely to be able to recognise and manage their emotional responses to stressors, solve conflicting situations and apply other resilience attributes, such as perseverance and self-reliance. Using positive psychology practices and helping young people to feel positive emotions, has the opportunity to increase selfconfidence, help them to perform and try new tasks, and assist them with processing new information (L. D. Hammond, Austin, Orcutt, & Rosso, 2001). On the contrary, negative emotions can elicit stress responses, such as fear or anxiety, which can distract the young person and have an impact on their ability to focus and learn.

#### **Understanding Outdoor Education**

Since the start of contemporary outdoor education in the early 20th century (T. James, 1990; Priest & Gass, 2005), the field has been presented with numerous simplistic definitions, such as that outdoor education is "education which takes place in the outdoors" (Hammerman, Hammerman, & Hammerman, 2001, p. 5) or that it is "education 'in', 'about', and 'for' the out-of-doors" (Donaldson & Donaldson, 1958, p. 63). To this day, the term 'outdoor education' is still yet to be clearly defined. It seems that educational professionals have avoided stringent definitions of outdoor education, as the meaning not only varies according to time, place and culture but also differs depending upon the philosophy of a company, educational institute or an individual facilitating the outdoor program (Brookes, 2004; Ford, 1986; Lugg, 2004; Neill, 2008).

Outdoor education is the combination of the interrelationships between humans, and humans and the nature environment, making it complex and diverse in nature (Priest, 1986). It is a holistic, interactive and dynamic learning process that engages an individual's physiological, cognitive, behavioural, emotional, spiritual, social, cultural, and environmental systems (Neill, 2008). Outdoor education programs provide opportunities for young people to explore how their actions influence themselves, others and the natural world. This type of learning has the potential to develop their capacity for healthy decision making around risk and increasing an understanding of self.

Specifically, journey style outdoor education programs take a threepronged approach that focuses on the development of an individual's relationship

66

with themselves (intrapersonal skills); an individual's relationships with others (interpersonal skills); and an individual's relationship with the natural environment (human-nature relationships) (Burridge, 2003; Cooper, 1994; Higgins & Loynes, 1997; Martin, 2010; Martin & Hewison, 2010; Priest, 1986). Researchers identify seven key elements of journey style outdoor education programs:

- The group sizes are small groups of ideally 12-14 participants with a maximum of 16-18 participants.
- 2. The program occurs in a semi-wilderness environment;
- Participants are isolated from many or all forms of technology and human impact or development;
- 4. There is the presence of a group leader and program facilitator;
- 5. There is continual contact with the learning activities and a structured curriculum;
- There are appropriate levels and forms of challenge for participants (physical, emotional, spiritual) and;
- There is an aim to transfer learning to and from the program. (Neill, 2008; Priest & Gass, 2005).

These elements are maintained by similar fields that use adventure and the natural environment as a catalyst for personal development with participants, such as adventure therapy and wilderness therapy (Gass, 1993a; Nadler, 1993; Norton, 2008; Norton et al., 2014; Tucker et al., 2018).

#### **Adventure and Wilderness Therapy**

The overarching goals of Wilderness Therapy are for the individual to increase their self-awareness across a variety of domains, by allowing participants to experience concrete examples of dysfunctional behaviour, and then be shown alternative behavioural responses and interpersonal choices which lead to success (Newes & Bandoroff, 2004). Similarly, with the intent to develop and transfer resilience attributes and positive coping skills, the goal of mainstream outdoor education programs is for participants to become self-aware and increase intrapersonal and interpersonal development.

Gass (1993a) has modified experiential learning principles and proposed how these elements can not only be applied for personal growth, but also for therapeutic purposes. Based on Gass's (1993a) adventure therapy theoretical framework, Kimball and Bacon (1993) identify 14 elements that are significant in the adventure therapy process. The framework includes: (1) multiple treatment formats, (2) group focus, (3) processing, (4) applicability to multi-model treatment, (5) sequencing of activities, (6) perceived risk, (7) unfamiliar environment, (8) challenge by choice, (9) provision of concrete consequences, (10) goal-setting, (11) trust-building, (12) enjoyment, (13) peak experience, and (14) therapeutic relationship (Kimball & Bacon, 1993; Nadler, 1993; Newes & Bandoroff, 2004). These elements are consistent with the seven key elements of outdoor education programs as outlined above.

Wilderness therapy programs are generally used as an adventure intervention for therapy with minority groups (Somervell & Lambie, 2009),

especially youth at risk (Bedard, Rosen, & Vacha-Haase, 2003; Gillespie & Allen-Craig, 2009; Nadler, 1993; Vissell, 2005). However, the program design elements of wilderness therapy are consistent with programming for mainstream outdoor education that aims for personal growth and relationship development through the application of adventure, challenge, risk and experiential learning.

Wilderness therapy programs involving adolescents have been found to have significant positive benefits, including increased levels of resilience (Gillespie & Allen-Craig, 2009) and reducing stress associated with interpersonal and mental health issues (Bettmann, 2012; Staunton, 2003). For example, Bowen and Neill's (2013) meta-analysis of adventure therapy outcomes compared overall results of wilderness therapy interventions with no treatment or alternative treatment groups and found an overall moderate effect size (ES = .47). The findings of Bowen and Neill (2013) study that examined the effects of programs on 17,728 participants indicated there to be a small positive (.05) overall effect on the long-term benefits of participants learning post-program. The results support the proposition that the program benefits, such as social development, selfconcept and family development, may be transferred into other areas of life and maintained long-term (Bowen & Neill, 2015). In addition, Bettmann (2012) examined 34 studies that included a total of 2,767 clients. Their data also demonstrated medium-sized (effect size, ES = .43) overall effects, revealing that wilderness therapy programs are effective interventions for adolescents. This is further supported by Cason and Gillis's (1994) meta-analysis examining the effects of adventure-based programs, considering 43 studies and including results

from 2,291 adolescents. The results of the adventure therapy studies consistently demonstrated moderate effect sizes in the .42 range.

### **Outdoor Education Program Design**

Depending on the aims, outdoor education programs can range in length from short hourly blocks, to single day experiences, to multi-day or expedition type experiences. Outdoor education programs can take place in various types of locations from residential style camps to journey style programs where small selfcontained groups of participants travel through natural environments.

Journey style programs primarily focus on the development of relationships through exposure to challenge and risk within an outdoor setting. This outdoor education approach appears to have the potential to develop resilience attributes (Gillespie & Allen-Craig, 2009) and coping skills (Yoshino, 2008). However, regardless of the length and location of programs, Hopkins (1985) suggests that for an outdoor education program to be considered effective, they should include the following characteristics:

- Experiential learning with high impact, problem-solving experiences;
- Adaptation to the physical, social, and cultural needs of participants;
- High expectations and focus on individual achievement;
- An empathic climate;
- An emphasis on group process, social skills, cooperation, and effective communication;

- Incorporation of environmental awareness and creative appreciation; and
- Regard adventure as a metaphor for life (Hopkins, 1985).

### **Concepts of Outdoor Education**

Outdoor education philosophies and practices are typically based on the premise that personal growth and development of the participants occurs through the process of inoculation to stress and development- by-challenge (Neill & Dias, 2001). This process is also known as the stress inoculation model (Meichenbaum, 1996 ; Meichenbaum & Cameron, 1989; Meichenbaum & Deffenbacher, 1988; Neill, 2008).

Fundamentally, outdoor education is built upon a pragmatic approach to learning, whereby, "pragmatism is the belief that the value of any learning experience is determined by the degree of learning that occurs from the actions and consequences of the experience" (Priest & Gass, 2018b, p. 34). Pragmatic philosophy details that the learning experiences, learning processes and theories are only valuable if they help the student learn and then apply their new learning back to their everyday lives (James, 1900; Kraft & Sakofs, 1988). Depending on the intended aims and outcomes, outdoor education programs are embedded with various behavioural, cognitive and experiential philosophies, theories, frameworks and constructs. According to Bowen et al. (2016, p. 32) who examined online survey responses from 98 program leaders and 24 program managers, the most commonly used theoretical frameworks on their outdoor education programs were "experiential learning (73%), challenge by choice (54%), facilitated reflection (52%), natural consequence (51%), and social learning (30%)". The next section provides a brief review of the learning theories relevant to this research.

## Constructivist theory.

Constructivism is one of the most commonly used theories in contemporary outdoor education (Gilbertson, Bates, McLaughlin, & Ewert, 2006). Social constructivists such as John Dewey, Lev Vygotsky and Jean Piaget have influenced contemporary outdoor education with their emphasis on the importance of understanding the culture and the context of learning within the environment in which it is experienced (Denzin & Lincoln, 2011; Von Glasersfeld, 1989). Outdoor education is a social constructivist way of learning and understanding the social and physical environment through personal experience. The constructivist approach to learning implies that students come to the program with an abundance of prior knowledge, skills and experiences, which, therefore, influences their perceptions (Gilbertson et al., 2006, p. 29).

# **Experiential learning.**

The central theory that underpins outdoor education practices is learning via direct experience (Kolb, 1984, 2007; Miles & Priest, 1999). Typically, outdoor education curriculum uses experiential learning practices as a tool to facilitate personal growth and learning outcomes (Higgins & Loynes, 1997). Experiential education is commonly known as 'learning by doing' (Ford, 1986) as all learning is through experience, even if it is not planned or learnt in a formal educational setting (Warren, Sakofs, & Hunt Jr, 1995). For deep learning and understanding to

occur from these experiences, some form of reflection must take place (Kolb, 1984; Kraft & Sakofs, 1988). Processes such as Kolb's (1984) 'Experiential Learning Cycle' or Warren, Sakofs, and Hunt's (1995) 'Action-Reflection Cycle' have provided models of teaching practice to support teachers facilitating students' reflection and understanding. Kolb's 'Experiential Learning Cycle' model uses four stages to connect the concrete experience of 'doing' with the intellectual process of abstract conceptualisation and learning transfer, through observation and reflection (see Figure 2.5).

The father of experiential education John Dewey (1859-1952) believed that education occurs as an experiential continuum which comes about through participating in quality experiences throughout one's lifetime (Dewey, 1916, 1929, 1938). Conceptually, learning is a seamless continuation of educational experiences which runs back and forth along a continuum, and therefore, experiences should not be viewed in terms of outcomes (Kolb, 1984).



Note. Sourced from Exeter (2001); adapted from Kolb (1984)

Figure 2.5. The four-stage experiential learning cycle.

Similarly, Priest and Gass (2018b, p. 42) describe a variance of the fourphases of their experiential learning cycle. Their model describes the cycle in relation to outdoor and adventure education specifically:

- 1. Action: referring to directly experiencing adventure activities with novel task completion and problem-solving.
- Reflection: referring to debriefing or similar facilitation methods used to look back on the experience in order to highlight the lessons learned, identify new behaviours and clarify existing concepts.
- 3. Integration: referring to the phase in which the lessons learned transfer into participants' daily lives as changes in feelings, thinking or behaving. It is suggested that this phase is where transference is strengthened by the conscious use of metaphors.
- 4. Continuation: referring to the ability for the participant to maintain and sustain these daily life changes in the face of external forces (e.g., lack of follow-up, lack of resources, peer pressure, avoidance), and therefore reverts back to old habits, ways of thinking, ideas or emotions.

Both Priest and Gass (2018b) and Kolb's (1984) experiential learning cycles are affected by many different factors. Including the role of the social group, the effectiveness of the leader and the role of the natural environment in creating a platform for participants to be exposed to different types of challenges. The first phase of the 'Experiential Learning Cycle' (see Figure 2.5) demonstrates the process of the individual and/or the social group experiencing a challenge or learning activity. During this phase, the widely accepted concept of 'challenge by choice' may be introduced when the participants are presented with the challenge (Priest & Gass, 2005; Wallia, 2008; Zinc, 2004). 'Challenge by choice' does not relate the 'choice' of engaging with the challenge or not, but rather, it relates to the individuals and/or the group having the 'choice' of choosing the level of challenge they are willing to commit to, as well as choosing how they are going to deal with the challenges presented (Rohnke, 1989; Wallia, 2008).

An effective outdoor leader would facilitate this process of group development by creating trust, a supportive environment and providing a safe space for participants to have the right to choose their level of engagement with the challenge (Priest & Gass, 2005; Wallia, 2008). The leader should understand how a participant is feeling and their reasons for wanting to engage more or less with the challenge. If the participant chooses not to fully engage, real alternatives should be provided (Wallia, 2008). Once group trust has been established, the leader provides guidance and choices of how to go about adapting and overcoming the challenges presented to them, empowering the participants to take control.

The second phase involves reviewing and reflecting on the experience. This process may or may not be guided by the leader. However, the leader can enhance this process by facilitating the group to observe what happened, what they did, how they felt and how it affected themselves and others (Priest & Gass, 2005). In this phase, opportunities for reflection of self, as well as opportunities to provide feedback for the social group should be provided.

76

In the third phase, the individual starts to draw conclusions and begins conceptualising the general principles gained from the learnings. Again, this process may be assisted by the leader to help participants draw conclusions linking to theories and models to support understanding. Leaders are important facilitators within this stage of the cycle to assist in forming new understandings about self and group interaction.

The fourth phase involves applying and practising the learnings in new environments, situations or contexts. At this final stage, it is suggested that the transference and application learning into various contexts occurs (Gass, 2003; Higgins, 1997). This four-stage experiential learning cycle draws together the process of engaging in the concrete experience, reflecting upon that experience, making connections with the experience and practising the learnings, to inform developmental changes and create new attitudes and ways of thinking (Kolb, 1984; Kraft & Sakofs, 1988).

## Transfer of learning.

'Transfer of learning' refers to the effect that a particular experience has on future experiences (Gass, 1993b). Typically, outdoor education programs promote the transfer of learning as a fundamental element of programs (Brown, 2010; Priest & Gass, 2005), yet transference still remains an assumption and a concept that is misunderstood (Gass & Buell, 1986; Miles & Priest, 1999; Priest & Gass, 2005; Wolfe & Samdahl, 2005).

'Transfer literature' suggests three main factors that are responsible for facilitating the transfer of learning in participants:

77

- 1. learner characteristics,
- 2. intervention design and delivery, and
- application/workplace environment (Baldwin & Ford, 1988; L. A. Burke & Hutchins, 2007).

In addition to these three factors, there are many different types of 'learning transfer' that are discussed in the literature (Baldwin & Ford, 1988; Cooley, Burns, & Cumming, 2016; Cooley, Cumming, Holland, & Burns, 2015; Sibthorp, Furman, Paisley, Gookin, & Schumann, 2011). However, the concept of transference in outdoor education has two arguments: The first argument outlines that a critical component of outdoor education programs, is that skills and knowledge learnt during the programs can be transferred into 'real life' contexts (Gass, 1995; Priest & Gass, 2005).

It is proposed that life skills learnt during outdoor education programs have the potential to transfer to various areas of a participants life, such as school, home or sport (Holman & McAvoy, 2005; Sibthorp, 2003). For example, Jim Sibthorp (2003) identified that students perceived the life skills they learnt during an outdoor education program (e.g., leadership skills, personal awareness, social skills, communication skills, tolerance and appreciation of others) have the highest probability of transferring into their lives at home compared to other hard skills learnt during the program (e.g., rock climbing, sailing and kayaking etc.).

Whereas, the second argument maintains the contrary where other researchers question the validity, lack of evidence and effectiveness of the transference of these skills and knowledge into other contexts (Brookes, 2003a, 2003b; Brown, 2010). Even though some research indicates that learning can be transferable (Hayhurst et al., 2015), there are still concerns about whether learning is context specific. For example, if the learning that takes place in the wilderness setting, there is question if the learning can be transferred and applied in different contexts such as the classroom.

Despite these conflicting views, Svinicki (2004, p. 99) suggests that there are two types of transfer that are most important when discussing skill development and the application of these skills across contexts. The first type of transfer is positive-versus-negative transfer, where positive transfer relates to the learner being able to make positive connections with the concept or skill. If this occurs positive transfer assists the learner to understand and integrate new learnings (Svinicki, 2004). A method to assist in positive transfer is metaphoric transfer, where learners make the link or connection between two situations or metaphors (Priest & Gass, 2018a). For example, positive transfer may occur when the learner understands that the leadership skills they learnt on camp can also be used when captaining their basketball team. On the other hand, if the learner is not able to make connections with the concept or skill, confusion and mistakes may occur, resulting in what is referred to as negative transfer (Svinicki, 2004). In this instance, negative transfer exists if the transference into other contexts has not occurred.

The second type of transfer is near-versus-far transfer, where near-transfer tasks, are tasks that are very much alike and follow the same sets of rules. In outdoor education literature, this is also known as specific transfer, where the tasks learnt are highly similar (Priest & Gass, 2018a). For example, near transfer tasks may include tasks such as pitching a tent in the bush compared to pitching a tent in a back yard. Even though the learner is in a different context, the learner would be using the same set of skills and rules. On the contrary, far transfer involves tasks that follow the same rules, but do not appear to be as similar, making the transference of the learning into other contexts more difficult. Far transfer is also known as non-specific transfer, where there may be interchangeable transfer of principles and attitudes (Priest & Gass, 2018a), but it requires more cognitive thinking and understanding by the learner.

There appear to be many factors that affect the transference of learning across settings, including;

- the similarity and difference between tasks and skills;
- the association and understanding made by the learner; and
- the context and degree of prior knowledge and skill sets (Barkley, 2010).

In addition, contemporary learning theories recognise that both reinforcement and practice play a role in the development and transference of skills across settings. Researchers have suggested that reinforcement theory plays a critical role in the outdoor education process, especially when the programs are designed with the intention for personal growth and to change individuals thoughts, attitudes and behaviours (Davidson, 2016; Ewert & Davidson, 2017a, p. 58; 2017b).

Baldwin and Ford (1988, p. 63) states that "for transfer to have occurred, learned behaviour must be generalised to the job context and maintained over a
period of time on the job". In this same way, it seems that for transference to occur during outdoor education programs, the participant must learn the skill and have opportunities to practice the skill in order for it to be transferred and maintained over a period of time.

## **Key Elements of Outdoor Education Programs**

The next section outlines four key elements that moderate experiential learning in journey style outdoor education: (a) the role of the leader, (b) the role of the natural environment, (c) the role of challenge, and (d) the role of the social group.

## The role of the leader.

Due to the nature of journey style outdoor education programs, young people spend significant amounts of time with their leaders, in close proximity and at times, under stressful conditions. This allows opportunities for leaders to develop supportive environments, build caring relationships, set boundaries and expectations, and provide meaningful ways for students to engage, which are all seen to be factors capable of promoting resilience in youth (Benard, 2004; Hanson & Kim, 2007).

Leaders assume many key roles during outdoor education programs. These responsibilities range from teaching practical outdoor skills, organising logistics and managing risk, to caring directly for the student's welfare, such as delivering medical treatment, pastoral care and providing holistic well-being for their group (Priest & Gass, 2005). Priest and Gass (2018b, p. 40) advises that such key roles also include the leader assuming the role of the:

- Translator: which helps the student interpret and reflect on experiences.
- Initiator: which engineers experiences to suit the participants' needs.
- Trainer: teaches skills and conditioning students for difficulties ahead.
- Maintainer: keeping the energy and motivation levels high.
- Authority: holding influence with the group.
- Guardian: being responsible for group safety (including, physical, social and emotional well-being).
- Exemplar: modelling behavioural patterns expected of the group.

In addition, one of the fundamental roles of leaders is to facilitate personal development and guide the experiential learning process. It is the role of the leader to choose when and how to implement the many different ways of presenting and facilitating challenges and to provide different ways of evaluating and reflecting upon them to create learning and personal development opportunities (Brown, 2002).

Outdoor education uses many different tools to facilitate the experiential learning processes of learning, reflection and the application of learning (Luckner & Nadler, 1997). Priest and Gass (2005) identified six types of facilitation techniques that are used in outdoor education practices. They refer to these as the six generations of facilitation. The first generation is called 'letting the experience speak for itself'. Also known as 'let the mountains speak for themselves'. This method is most commonly used after the participant's experience, whereby the leader assumes that learning has occurred from the experience.

The second generation is known as 'speaking for the experience'. This method relies on the leader interpreting the experience for the participants by being explicit and pointing out what they saw occurring. The third generation is called 'debriefing' or 'funnelling the experience'. This technique requires the leader to ask leading questions during guided discussions to help participants reflect and learn from their experiences. The leader can choose what types of questions they ask and what techniques to apply. This includes the use of open-ended questions, closed- questions, probing questions, clarifying questions, reflective questions, example questions, encouragements, acknowledgments and appreciations (Jordan, 2007).

The fourth generation is referred to as 'directly frontloading the experience', also known as 'pre-briefing'. This is one of the two most common facilitation techniques that occurs before the outdoor education experience. Frontloading involves the leader briefing the group and asking them questions before they engage in the experience. This is a beneficial technique to use when the leader is facilitating group dynamics and getting the group to focus on specific elements such as teamwork or communication. Frontloading an experience using a metaphor also affects the way the experience is processed and reflected upon by the participants (Priest & Gass, 2005).

The fifth generation of facilitation is called 'framing the experience'. This technique requires leaders to translate the experience into real-world contexts after

the fact. Leaders can assist in the transference process by facilitating information assimilation, such as the use of metaphors or analogies to help the participants link and cognitively understand how the experience relates back to their personal lives (e.g., cognitive learning processes) (Piaget, 1964; Priest & Gass, 2018b). This type of transfer can occur in any generation which uses metaphors as a tool for learning and reflection to aid in the experiential learning process

The sixth and final generation of facilitation that a leader can choose is referred to as 'indirectly frontloading the experience' This technique is usually used as a last resort and is the least used technique of facilitation used by outdoor professionals (Jordan, 2007; B. Martin, Cashel, Wagstaff, & Breunig, 2006). It is generally used when frontloading, and the use of metaphors is not working with the group (Priest & Gass, 2005).

Leaders presume the role of 'the gatekeeper' by having control of the experiential learning process Brown (2002), where the leader facilitates the participant being actively engaged in posing questions, investigating, experimenting, being curious, solving problems, assuming responsibility, being creative and constructing their own meaning (Association for Experiential Education, 2018). Leaders have the ability to promote and encourage the development of protective factors such as positive relationships with their participants or providing a supportive environment, which can directly impact a young person's ability to advance their resilience and coping skills (Olsson et al., 2003).

However, as leaders have a direct impact on individuals, experiences and the debriefing process, Brown (2002) called for experiential educators to "critically reflect on [their] practice with a view to exploring other avenues [rather than debrief talking circles] for facilitating learning" (Brown, 2002, p. 293). Brown (2002) also suggested that the literature should re-evaluate how powerbased relationships between students and leaders are built and sustained during facilitation of outdoor education programs.

#### The role of the natural environment.

Outdoor education programs generally take place in natural or semiwilderness environments away from urban areas (Priest & Gass, 2005). This provides a contrasting setting to that of an individual's daily urban life, where the built-up environment insulates people from the natural world and is often overstimulating with crowded cities, traffic, noise and constant advertising. Programs in natural environments can provide participants with opportunities for unique experiences, reflection and contemplation about their actions and feelings (D'Amato & Krasny, 2011; Kaplan & Talbot, 1983). This corresponds with the findings from the Hattie, et al. (1997) meta-analysis which examines 20 years of research on adventure programs. The study points out that "the most striking common denominator of adventure programs is that they involve doing physical things away from the person's normal environment" (Hattie et al., 1997, p. 44).

However, at present young people are spending more time living in urban areas, participating in passive indoor activities and experiencing an alarming amount of 'screen time' such as watching television, using smartphones and playing video games (Ginsburg, 2007; Juster, Ono, & Stafford, 2004; Louv, 2013; McCurdy, Winterbottom, Mehta, & Roberts, 2010). This is no surprise as over 50% of our population lives in urbanised areas, with the number expected to rise to 70% by the year 2050 (Heilig, 2012). Coincidently, evidence highlights that urbanisation has had a negative effect on people's mental health, with the growing amount of research linking decreased exposure to nature with changes in psychological functioning (Bronzaft, 2003; Lederbogen et al., 2011).

In contrast to urbanisation, nature continues to be referenced as a tool to promote health, prevent illness and provide restorative benefits (Annerstedt & Wahrborg, 2011; Kaplan, 2001; Kaplan, 1995; Pryor, Townsend, Maller, & Field, 2006). This has highlighted an increasing need for young people to reconnect with nature and to develop human-nature relationships (Louv, 2005, 2007; Martin, 2004, 2010; Martin & Hewison, 2010). Exposure to nature has been shown to provide a myriad of cognitive, social, physical, and emotional benefits including stress relief (Hansmann, Hug, & Seeland, 2007), coping (Doucette, 2004), happiness (Howell & Passmore, 2013), knowledge transfer (Basile & Copley, 1997), self-esteem (Swarbrick et al., 2004) and skill acquisition (Fjørtoft, 2001).

D'Amato and Krasny (2011) reported that 21 out of the 23 participants evaluated after outdoor adventure education programs reported feeling many psychological benefits after spending time in nature. This included feeling a stronger sense of self, feeling more balanced, comfortable and attaining "mental quiet" (Kaplan, 1983). Because natural environments tend to be more peaceful than other environments, and they generally remove modern life distractions, such as participation in 'screen time' (Higgins & Loynes, 1997; Lederbogen et al., 2011), spending time in nature can result in restorative benefits and can provide more opportunities for individuals to identify internal and external assets to utilise in difficult situations (Berman, Jonides, & Kaplan, 2008). This highlights the importance of and need for youth to spend more time in the outdoors.

Natural environments provide a unique setting for outdoor education programs. It is unpredictable and can provide a sense of inspiration, wonder and awe (Burridge, 2003; D'Amato & Krasny, 2011; Darbor, Lench, Davis, & Hicks, 2016). When immersed in wilderness settings, people may experience positive emotions such as 'awe' and 'wonder', as they observe the world around them and as they try to understand their place in the world (Darbor et al., 2016). This may be caused by being submerged in a natural setting, surrounded by information-rich environmental stimuli (Darbor et al., 2016). This emotional connection with nature often leads to further thoughts and inquiry into their own feelings, thoughts and emotions, triggering them to feel calmer, more at peace with themselves and feel a sense of inner beauty (Kaplan & Talbot, 1983, p. 178).

The natural environment is also used as a tool during outdoor education programs, to provide a context for challenge which supports the experiential learning process. Being placed in an environment that is out of the normal context of one's life, provides opportunities for participants to be challenged (Walsh & Golins, 1976). The environment creates a sense of perceived risk, fear of the unknown and situational stress (Kaplan & Talbot, 1983, p. 179). This can cause a state of disequilibrium when young people are stretched out of their 'comfort

zones' (Fletcher & Hinkle, 2002; Luckner & Nadler, 1997; Nadler, 1995; Senninger, 2000). "Disequilibrium refers to an individual's awareness that the previous way of processing information no longer applies in this new experience" (Gass, 1993a, p. 59).

Experiencing states if disequilibrium requires individuals to mediate the perceived stress/ors, and adapt to their unfamiliar surroundings (Brown, 2008; Cross, 2002; Panicucci, 2007), resulting in short- and long-term benefits for participants (Fletcher & Hinkle, 2002; McKenzie, 2000, 2003). Therefore, creating a space for participants to apply and practice healthy new habits and effective coping strategies in a supportive environment is vital (D'Amato & Krasny, 2011).

#### The role of challenge.

Through exposing an individual to potential risk, challenge and/or adventure during outdoor education, programs have the ability to enhance an individual's personal and interpersonal skills (McKenzie, 2000), including group development (Priest & Gass, 2005), self-esteem, self-reliance and moral character (Ewert & Garvey, 2007). The challenges that arise from participants being placed in an unfamiliar environment, away from their normal setting and through participating in new adventurous activities, allows for personal growth opportunities and for group cohesion to occur which may not be experienced in other settings, such as the classroom.

If personal growth is the intended learning outcome, it is important for programs to have an intentional focus on personal development. For instance, the meta-analysis conducted by Marsh (1999a, 1999b), found that camp programs that had a philosophy based on personal development (0.41) had considerably stronger effects than camps that did not have personal development philosophies, demonstrating negligible effects.

During participation in outdoor education programs, people may experience different types of challenges which can vary from social, emotional, mental and physical challenges. The different physiological responses a person experiences is also relevant to the level of difficulty of the challenge (Csikszentmihalyi, 1996) and the level of arousal that is perceived by the individual (Ellis, 1973; Yerkes & Dodson, 1908).

The 'optimal arousal theory of play', (Ellis, 1973) outlines that there is a distinctive point of optimal arousal for each individual, where people may be over or under stimulated by the environment around them. The original findings of Yerkes and Dodson (1908) study indicates that there is an empirical relationship between levels of arousal and performance. This is also known as the Yerkes-Dodson law (see Figure 2.6).



*Note*. Reprinted from Diamond, Campbell, Park, Halonen, and Zoladz (2007, p. 3) based on Yerkes and Dodson (1908) original relationship between performance and arousal.

Figure 2.6. Yerkes-Dodson law of optimal arousal.

Similarly, the Adventure Experience Paradigm details that when participants are exposed to an increased level of challenge, it is expected they will psychologically experience five distinct states as their degree of arousal increases in relation to the risk (i.e. the environment) and competence (i.e. the person):

- 1. Exploration and experimentation (low risk, high competence)
- 2. Adventure (competence is only marginally high than the risk)
- 3. Peak adventure (competence and risk levels match)
- 4. Misadventure (risk marginally higher than competence)
- Devastation and disaster (competence is low, and risk is high) (Martin & Priest, 1986; Priest, 1999; Priest & Baillie, 1987).

As the Adventure Experience Paradigm theory suggests, it is the level of challenge that is the critical element in the design and implementation of outdoor education programs (Martin, P & Priest, 1986; Priest, 1999; Priest & Baillie, 1987). An ideal situation would be for individuals to have a peak experience. The peak experience of engagement is when individuals are fully immersed, focused, and energised through engagement with the activity (Shernoff, Csikszentmihalyi, Shneider, & Shernoff, 2003). This can be conceptualised as a state of flow (Bakker, 2005; Csikszentmihalyi, 1996, 2008; Nakamura & Csikszentmihalyi, 2005). This is the realm where people can gain insights into themselves, leading to the most positive benefits of meeting a challenge (Priest & Gass, 2005, p. 46).

Overcoming challenges can lead to young people feeling a sense of accomplishment and empowerment, while also providing opportunities for personal growth such as attaining a greater sense of self-confidence. (D'Amato & Krasny, 2011; Shellman, 2009). When young people experience continual and repeated exposure to low-moderate levels of risk, they experience inoculation which in turn helps to prepare them to overcome further significant risks in the future (Fergus & Zimmerman, 2005; Werner, 2000; Windle, 2011).

Challenge is an essential factor in supporting the growth and development of resilience in participants. "Challenge cannot exist without both situational risk and personal competence engaged in an effort to resolve uncertainty" (Priest & Gass, 2018c, p. 201). This is supported by the definition of resilience and Toland and Carrigan (2011, p. 97) who states that "individuals are not considered to have displayed resilience if there has never been a significant threat to their development". Furthermore, young people that have the opportunity to practise coping skills and become more proficient at overcoming low to moderate level

risks will be more prepared to face harder challenges in the future (Fergus & Zimmerman, 2005). Therefore, the level of challenge presented to groups should be considered as an integral part of program, intervention, curriculum and design.

## The role of the social group.

Positive social relationships are key developmental assets required for a young person to be a successful contributor to society (Scales & Leffert, 2004). The social group is an aspect of outdoor education and is especially important for young people as they develop through adolescence (Jostad, Sibthorp, Pohja, & Gookin, 2015). Previous outdoor education research has provided evidence that the social group impacts student learning and potentially their entire outdoor education experience (Ewert & McAvoy, 2000; McKenzie, 2000; Sibthorp, Paisley, & Gookin, 2007). By developing positive social relationships, young people have the ability to strengthen their development of internal and external assets, and their understanding of self (Shaffer, 2005), as well as creating a sense of belonging through being part of a small community (Cross, 2002). Strong positive relationships are also shown to buffer young people from "developmental disruptions and help them to develop resilience" (Shonkoff, 2015).

A number of theorists have suggested the group size impacts participant outcomes (McKenzie, 2000; Nadler, 1993; Walsh & Golins, 1976). For instance, small-group theory suggests that the personal development of an individual should be greater because is the potential for increased opportunities for the program aims and objectives to be individualised. However, the outcomes of medium-sized groups of more than 12 participants should be even stronger due to providing an 'ideal' balance between individual engagement and group diversity (Neill, 2008).

Outdoor education programs provide a platform for group interaction that is different from social group interactions that young people may experience in their everyday lives. The nature of journey style outdoor education programs means that groups are small and that the groups spend a lot of time together in a close-knit environment (e.g., eating meals together and sharing tents for sleeping). The remoteness of the setting, unfamiliar physical environment and the various challenges that participants experience focuses on the importance and influence of the group (Jostad et al., 2015). For example, D'Amato and Krasny (2011) conducted a qualitative study that examined the experiences and perceptions of 23 former outdoor education participants who were aged 15 to 24 years. The trip length ranged from 16 and 78 days, with an average of most trips running for 28 days. The study found there to be four significant factors that contributed to personal transformations and changes in behaviour post-program: (1) living in pristine nature, (2) experiencing a different lifestyle, (3) being part of the course community, and (4) dealing with the intensity and challenges of the course (D'Amato & Krasny, 2011). D'Amato and Krasny (2011, p. 246) specifically commented on the importance of being part of a "tight-knit, supportive community that helped individuals as they undertook challenges, set high standards for environmental behaviour, and facilitated learning in the outdoors".

Contrary to the development of positive social relationships and a supportive environment, negative experiences may occur if the individual does

not develop positive social connections with the group (Jostad et al., 2015). This may result in feelings of isolation and abandonment with the potential effects on the person being detrimental and long-lasting (Goossens, 2006). Effective leaders will be able to identify the importance of group development and adjust for the factors affecting the group to support the social cohesion (Priest & Gass, 2005, p. 73).

During participation in effective outdoor education programs, students have opportunities to develop positive social relationships with their peers, teachers and other adults in contexts that are different to their daily lives (Australian Curriculum and Assessment Authority, 2012; Victorian Curriculum and Assessment Authority, 2009). Having the opportunity to build these positive relationships with themselves, others and the natural world has not only been linked to assisting students with academic achievement (Hattie, 2003), but these relationships also act as protective factors to foster resilience and resilience attributes (O'Dougherty Wright & Masten, 2015).

## **Outdoor Education as a Resilience and Coping Intervention**

By adopting a stress inoculation approach, resilience can be developed during outdoor education programs through incremental exposure to challenging experiences which require participants to apply various coping strategies to adapt to the situation (Booth, 2015; Masten & Reed, 2002; Neill, 2008). Whilst various outdoor education studies have reported improvements in either resilience as an entire construct (Booth, 2015; Gillespie & Allen-Craig, 2009; Neill & Dias, 2001; Shellman, 2009) or increases in individual components of resilience such as selfesteem (Cooper, 2004), social competence (Miller & Allen-Craig, 2005) and internal locus of control (Cason & Gillis, 1994; Hans, 2000; Neill & Richards, 1998).

From the year 2000 to 2016, there have been 12 international studies that have specifically investigated whether levels of psychological resilience in adolescents were affected during or after participation in outdoor education programs. An overview of these 12 studies is presented in Table 2.5. However, evidence indicates that not all outdoor education programs have positive effects on their participants (Allin & Humberstone, 2010; Brookes, 2003a, 2003b, 2004; Dewey, 1938; Neill, 1997, 2002, 2008; Skehill, 2001).

Remarkably, there is limited research investigating the impact of coping skills in outdoor education programs (Booth, 2015; Yoshino, 2008). The recent studies by (Booth, 2015) and (Yoshino, 2008), appear to be the only two studies that have investigated the effects of coping skills in outdoor education settings. Considering the direct link between coping strategies as a means for adaption and resilience capacity during stressful circumstances, it is surprising that Booth's (2015) study appears to be the only study of its kind that has examined both the development of psychological resilience and the application of specific coping strategies in the context of outdoor education programs with adolescents. While Yoshino's (2008) study measured how stress appraisal, coping response, and perceived success influenced perceived psychological growth, it did not specifically measure resilience as a construct.

# Summary of research on resilience and adolescents in outdoor education

Study	Place	Intervention Group	Control Group	Resilience Measure	Intervention	Results
Green, Kleiber,	United States	Youth at risk adolescents.	Youth at risk	Adapted version of the Protective	Summer camp program	Mixed results:
and Tarrant (2000)		Ropes course treatment	adolescents. Summer	Factors Scale	compared with challenge	Some protective factors evaluated had
		group:	camp program	(Witt, Baker, & Scott, 1996).	ropes course.	significant changes, such as
		(M = 11.6  years)	comparison group:			improvements in interested and caring
		( <i>n</i> = 25)	(M = 12.9  years)			adults, and neighbourhood resources.
			(n = 95)			
			Local high school			
			comparison group:			
			(M = 12.6  years)			
			( <i>n</i> = 57)			
Skehill (2001)	Australia	Year 9 adolescents:	NIL	The Resilience Scale (Wagnild &	Extended Stay Outdoor	No changes:
	(Sydney &	(M = 14.3  years)		Young, 1993) and the General	Education Program	No significant increase in resilience
	Melbourne)	( <i>N</i> = 99)		Well-Being Scale derived from	(ESOEP).	and no reported increases in levels of
		(n = 71  male)		Mental Health Index (Veit & Ware,		well-being or decreases in distress.
				1983).		
Neill and Dias	Australia	University students:	University students:	The Resilience Scale	22-day multi-element	Increased resilience:
(2001)		(M = 21 + 3.1  years)	(M = 24 + 7.1  years)	(Wagnild & Young, 1993).	Outward Bound programs in	All Outward Bound participants
		( <i>n</i> = 49)	(n = 31)		Australia.	reported positive changes in their
						resilience.
Buckner et al.	United States	Adolescents from a	NIL	Blackburn Resilience Scale	3-day residential camp.	Increased resilience:
(2005)		Children's Hospital		(Blackburn & Satcher, 1998).	Activities: swimming,	Increased resilience 6 weeks after the
		Asthma Clinic:			canoeing, horse-back riding,	camp, however at six months the
		(M = 12.4  years)			ropes, course crafts, and	mean scores dropped back but still
		(N = 12) ( <i>n</i> = 4 male)			games.	remained higher than baseline levels.

# Table 2.5 (continued)

Study	Place	Intervention Group	Control Group	Resilience Measure	Intervention	Results
Ewert and	United States	University students:	University students:	The Resilience Scale	3-week adventure based	Mixed results:
Yoshino (2008)		(M = 20.5  years)	(M = 20.5  years)	(Wagnild & Young, 1993).	expedition.	Both groups increased resilience
		(n = 17)	(n = 20)			scores at the post-test. The control
						group scored higher at post-test.
Gillespie and	Australia	Youth at risk male	NIL	The Resilience Scale (Wagnild &	5-week Wilderness Therapy	Increased resilience:
Allen-Craig	(Victoria)	adolescents:		Young, 1993) and the Life	Program.	Increase in overall resilience.
(2009)		Age range:14-16		effectiveness questionnaire- Youth		Individual attributes of resilience,
		(N = 19)		at-risk version (YARPET) (Neill,		such as meaningfulness, persistence
				Marsh, & Richards, 2003).		was not reported.
Beightol,	Santa Fe,	Grade 5 students:	Grade 5 students:	Anti-Bullying Initiative Survey	3-days of high ropes and 10	Mixed results:
Jevertson, Gray,	New Mexico.	( <i>n</i> = 51	( <i>n</i> = 54)	(Carter & Jevertson, 2006).	problem-solving and role	Males had no change, females
Carter, and Gass		(n = 26  male)	( <i>n</i> = 29 male)		pay sessions in school.	increased in resilience attributes, such
(2009)						self-efficacy, and goals and
						aspirations.
Shellman (2009)	United States	Outward Bound course	University students:	The Resilience Scale (Wagnild &	14- to 30-day Outward	Increased resilience:
		participants:	( <i>n</i> = 69)	Young, 1993), The Empowerment	Bound expedition.	A significant positive increase in both
		( <i>n</i> = 86)	Age range:14-30	Scale adapted from (Spreitzer,		psychological empowerment and
		Age range:14-30	(M = 20.1  years)	1995), the Characteristics of the		resilience was found for the Outward-
		(M = 17.3  years)		Experience Scale (CES-I and CES-		Bound group. No significant
				II) adapted from (Sibthorp, 2000)		differences were obtained for the
				and semi-structured interviews.		comparison group.

Table 2.5	(continued)	)
-----------	-------------	---

Study	Place	Intervention Group	Control Group	Resilience Measure	Intervention	Results
Ewert and	United States	University students:	University students:	37-item self-report instrument	3-week adventure based	Mixed results:
Yoshino (2011)		( <i>n</i> = 28)	( <i>n</i> = 27)	derived from the Resilience Scale	expedition including 3-day	Treatment group increased resilience
				(Wagnild & Young, 1993), Ego-	solo.	significantly, but the control group
				Resiliency (Block & Kremen,		had a higher level of resilience pre-
				1996), and the Connor-Davidson		and post-test.
				Resilience Scale (Connor &		
				Davidson, 2003).		
Hayhurst, Hunter,	New Zealand	Study 1: Trainee students	University students:	The Resilience Scale (Neill & Dias,	Study 1: 10-day	Increased resilience:
Kafka, and Boyes		( <i>n</i> = 126)	( <i>n</i> = 63)	2001; Wagnild & Young, 1993)	developmental sailing	Study 1 Results: Increased levels of
(2015)		(M = 16.55  years)	(M = 16.55  years)	15 item version.	voyage.	resilience over the course of the
						voyage.
		Study 2: Trainee students	High school students:	The Resilience Scale (Neill & Dias,	Study 2: 10-day	Study 2 Results: The increased levels
		( <i>n</i> = 146)	( <i>n</i> = 74)	2001; Wagnild & Young, 1993)	developmental sailing	of resilience were maintained five
		(M = 16.51  years)	(M = 16.43  years)	15 item version.	voyage.	months following the voyage.
Booth (2015)	Australia	Year 10 students:	NIL	The Brief Resilience Scale (Smith	12 day Outward Bound	Increased resilience:
		( <i>N</i> = 49)		et al., 2008), adapted version of the	expedition.	Small significant increase in
		( <i>n</i> = 24 male)		60-item COPE Inventory (Carver,		resilience scores. Positive correlations
		(n = 25  female)		Scheier, & Weintraub, 1989) and		between higher levels of resilience
				semi-structured interviews.		and the application of coping skills.
Davidson (2016)	United States	Outward Bound course	NIL	The Resilience Scales for Children	Outward Bound expeditions	Increased resilience:
		participants:		and Adolescents (Prince-Embury,	ranging from 5 days to 21	Positively increased levels of grit and
		( <i>N</i> = 350)		2008) and the Grit Inventory	days.	resilience.
		Age range: 12-22 years		(Duckworth & Quinn, 2009).		

The study conducted by Neill and Dias (2001) used the Resilience Scale to measure increased levels of resilience after participation in a 22-day multielement journey style outdoor education program with Outward Bound in Australia. The study used two groups; the experimental group (N = 49 and the control group (N = 31) which consisted of young adults. While the study demonstrated strength by using a comparison group, one of the limitations of the study was its small sample size.

Neill and Dias (2001) used a mixed design ANOVA for both the experimental and control groups. Using the 15-item version of the original 25 item Resilience Scale (Wagnild & Young, 1993), the author reported that all participants showed positive changes in their resilience and their overall change in effect size was very large. The data revealed a significant interaction between time and group, where the experimental group (F(1,68) = 6.39, p = .01) reported a greater change in resilience compared to the control group. An interaction between time, group and gender was also reported which indicated that the changes in resilience was consistent across gender (F(2,68) = 1.86, p = .16) (Neill & Dias, 2001).

Another program that used challenging expedition type experiences to enhance levels of resilience was the Gillespie and Allen-Craig (2009) study on atrisk male adolescents. This study also used a smaller 15-item version of the Resilience Scale to measure changes in resilience scores after participation in a five-week wilderness therapy program. The study indicated a moderate to large positive effect size of resilience scores (ES =0.60). The results demonstrate that participation in the wilderness therapy program did help youth at-risk increase their resilience.

A more recent study by Ewert and Yoshino (2011) also supports the notion that outdoor education programs enhance levels of resilience. Their study used a mix of the Resilience Scale (Wagnild & Young, 1993), Ego-Resiliency (Block & Kremen, 1996), and the Connor-Davidson Resilience Scale (Connor & Davidson, 2003) to make up a 37-item questionnaire to measure changes in resilience scores in university students who participated in a 3-week adventure-based expedition. Based on their results, the authors suggested that short-term expeditions may enhance participants levels of resilience (Ewert & Yoshino, 2011).

There is however, contradictory evidence that indicates that not all studies that have investigated in the impacts of outdoor education programs on levels of resilience and coping have had positive effects (Sheard & Golby, 2006; Skehill, 2001). Skehill's (2001) similar study that used the 15-item version of Resilience Scale to measure resilience levels in Year 9 adolescents, during an Extended Stay Outdoor Education Program, reported no significant increase in resilience, as well as no reported increases in well-being or in the reduction of distress in participants.

Whilst some studies support the belief that outdoor education programs do in fact enhance psychological resilience (Buckner et al., 2005; Gillespie & Allen-Craig, 2009; Hayhurst et al., 2015; Neill & Dias, 2001), Table 2.5 still demonstrates the lack of consistency in results to draw this conclusion (Beightol et al., 2009; Ewert & Yoshino, 2008, 2011; Green, Kleiber, & Tarrant, 2000).

Findings, as set out in Table 2.5 show that there is the potential for outdoor educations programs to be an effective approach to increase resilience and effective coping skills. Nevertheless, the table identifies that more research and evaluation is needed to understand the psychological and social processes of outdoor education (Burke & Collins, 2004; Burridge, 2003; Neill & Richards, 1998).

#### **Chapter Summary**

The review of literature in this chapter has explored the four key themes of this research; adolescents, resilience, coping and outdoor education. These elements provide the background for examining how outdoor education programs can be used as an effective approach to develop resilience attributes and coping skills in young people.

Initially, this review provided an overview of adolescence. This critical stage of human development that occurs between childhood and adulthood involves young people developing physically, socially and emotionally. In addition, young people are cognitively developing, and as their brains continue to mature they develop the capacity for formal operation thinking. This cognitive capacity allows them to think abstractly, self-reflect and take ownership of their actions and decisions. Even though young people experience many changes and challenges as they transition from being children to adults, and while some may experience turmoil, adolescence provides a period of opportunity for positive growth and learning.

Existing data reveals a large number of young people reported having experienced serious mental health issues during adolescence (Australian Institute of Health and Welfare, 2016; Bullot et al., 2017; Lawrence et al., 2015; The Department of Health). Young people who are undergoing these mental health problems are often feeling low self-esteem, increased social pressures of the 21<sup>st</sup> century and pressures to conform to unrealistic expectations, with the main contributing factors being violence, abuse, neglect, and bullying (UNICEF, 2011). Assisting young people to develop positive human characteristics, such as selfefficacy, self-esteem, resilience and coping skills, can minimise the risk and severity of adverse mental health conditions that may be developed during this vital developmental period.

The chapter then outlines the issues and difficulties in researchers agreeing on defining resilience as either a trait (individual level) or as a dynamic process (systems level). Consequently, to clarify the definition of resilience in relation to this research, resilience is conceptualised as the capacity of an individual to utilise their internal and external supportive systems to positively adapt to adversity, whilst maintaining and sustaining their personal well-being.

Resilience is the result of an individual's ability interact with their environment (i.e. mirco-, meso-, exo- and macro-systems.) to assist them in promoting or maintaining their well-being, protecting them from risk factors and helping them to be able to cope with adverse or challenging situations. The development of a young person is the direct result of the relationships and interactions of between themselves and their environments. An individual's resilience capacity is composed of numerous factors that cumulatively attribute to their ability to cope with difficulties. Protective factors and processes, also known as developmental assets, such as developing positive, supportive relationships with peers and teachers and developing positive selfconcept are crucial in the development of an individual's resilience and can bolster their ability to use effective coping methods. Ultimately, developmental assets can assist young people in coping with a variety of stress levels and types of challenges.

Resilience is not rare, in fact, it is a capacity that can be seen and utilised by most individuals. The skills and factors that make up a resilient individual are not necessarily inherent; they can be fostered, learned and developed. Therefore, supporting the development of internal and external assets is crucial for the future success of young people and their capacity to live an optimal level of human functioning. Young people who have assimilated developmental assets are more likely to positively adapt to adversity and have a higher chance of coping with challenges and perceived stress.

Following a review of the psychometric properties of existing instruments and literature, the Resilience Scale and the Brief COPE Scale were deemed to have the best reliability and validity to measure resilience and coping with young people, within an outdoor education setting. As demonstrated above, this research is timely in nature as it will become part of the current discourse about the importance of developing resilience in youth through Australian school systems

by using outdoor education programs as a platform for facilitate personal development (Passarelli et al., 2010).

Finally, this chapter presented research promoting journey style outdoor education programs as an effective approach to enhance resilience, resilience attributes and coping skills through using stress inoculation and experiential learning models. The role of the leader, the natural environment, challenge and the role of the social group are presented as four critical elements of the outdoor education process that occurs during journey style programs.

As journey style programs are of an extended length, they provide participants with the opportunities to be consistently exposed to stressors or challenges that require the application of coping skills to overcome the stress or achieve desired goals (Davidson, 2016). The length of the program allows more opportunities for participants to practice and develop assets over the duration of the program.

Leaders also play a critical role in this process as they guide the participant through reflection and experiential learning processes, which are pivotal to these types of programs. As Grümme (2011, p. 114) advocates "without conscious reflection, no experience is gained; events remain but events". Therefore, an individual's experience may be accordingly influenced by the effectiveness and facilitation methods used by the leader.

Previous research evidence indicates that journey style outdoor education programs do enhance resilience (Davidson, 2016; Gillespie & Allen-Craig, 2009; Hayhurst et al., 2015; Neill & Dias, 2001) and that there is potential for outdoor

educations programs to be used as an effective tool to increase both resilience and coping skills (Booth, 2015; Cooper, 2004; Miller & Allen-Craig, 2005; Yoshino, 2008). However, other studies present conclusions that not all outdoor education programs have significant positive effects on their participants (Allin & Humberstone, 2010; Brookes, 2003a, 2003b, 2004; Neill, 1997, 2002, 2008; Sheard & Golby, 2006; Skehill, 2001). The lack of continuity in results reinforces that further research is required to investigate which particular attributes of resilience and coping skills are strengthened by outdoor education programs (Neill, 2001; Neill & Richards, 1998). Furthermore, supporting evidence is lacking on how resilience and coping skills can be best developed, enhanced, and transferred into the everyday lives of young people (Ahern et al., 2008; Booth, 2015; Neill & Dias, 2001; Neill & Richards, 1998).

The primary purpose of this literature review was to provide a current and comprehensive review of the core concepts of this research. The review identified gaps in the research and provided an analysis of methods used in past research that aimed to identify levels of resilience and coping in adolescents within an outdoor education setting. Based on the research and findings considered in this chapter, it is logical to assume that exposure to challenging experiences during outdoor education programs will enhance participants' capacity for adaptation, and therefore, increase levels of resilience and strengthen their repertoire of coping skills. This chapter provided the basis for the next chapter which outlines the methods selected and subsequently applied in this study.

### **CHAPTER 3: METHODS**

'Outdoor education fits comfortably under the school's vision statement of providing for the 'academic, physical, emotional, psychological, social and spiritual dimensions of student life'



- Yarra College Statement

Figure 3.1. Watching the sunrise.

Firstly, this chapter introduces the research paradigm and explains the connection between the research questions and the methodological approach. A pragmatic paradigm was used to capture the phenomena of an outdoor education program's potential influence on young people's resilience and coping skills. The chapter then provides an overview of the sample selection, the outdoor education program and the methods applied throughout the study.

The data collection process is presented next, detailing the three phases of data collection and analysis. Data was collected via three separate means; pre- and post-surveys, program observations and participant interviews. Phase I outlines the administration of the survey instruments and the subsequent analysis of the data. Phase II describes the collection and analysis of observations made during the outdoor education program and Phase III details how participant interviews were conducted and how themes were explored. The chapter concludes with an overview of the ethical issues and considerations of the research.

In Chapter 2, the literature review provided evidence that outdoor education programs are promoted as an efficient method for strengthening resilience and coping skills (Allen-Craig & Miller, 2007; Cooper, 2004; Neill, 2008). However, as Neill and Dias (2001) alluded, it is unclear which attributes are strengthened through participation in outdoor education programs.

The overarching aim of this research is to investigate if resilience is context specific. The research aims to identify the particular attributes of resilience and coping that are strengthened through participation in an outdoor education program and examine whether these attributes are transferred into other contexts of their school lives. The study also investigates how resilience and coping can be best supported, developed and transferred into other contexts. The four research questions that this study aims to address are:

## **Research Question 1**

Which attributes of resilience and coping are enhanced through participation in an extended secondary school outdoor education program?

## **Research Question 2**

If attributes were enhanced, what supported and fostered the development of these resilience attributes and coping skills?

## **Research Question 3**

If attributes were enhanced, what attributes of resilience and coping were transferred and applied in contexts other than an outdoor education setting?

## **Research Question 4**

How can resilience and coping skills be best developed and transferred into other contexts of learning?

This chapter is separated into the following sections: (a) research paradigm; (b) sample selection; (c) outdoor education program; (d) data collection process; (e) Phase I; (f) Phase II; (g) Phase III; and (h) ethical issues and considerations. The structure corresponds to the research process developed for this study which is presented in the flowchart in Figure *3.2*.

#### 1. Literature search & review

Review of all relevant literature on resilience, adolescents and outdoor education

2. Survey instruments chosen

Based on findings from the literature review

3. Phase I: Pre-survey data collected

Questionnaires collected from both sample groups before the outdoor education program

4. Phase II: Observation data collected

Researcher observed and recorded qualitative data during the 21-day outdoor education program

5. Phase I: Post-survey data collected

Questionnaires collected from both sample groups after completion of the program

6. Phase III: First set of interview data collected

Small group interviews conducted with the program group immediately after the program

7. Phase I: Data compared & analysed

Quantitative data from pre- and post-surveys are analysed and reported

8. Phase III: Second set of interview data collected

Small group interviews conducted with the program group six months after the program

9. Phase II & Phase III: Data compared & analysed

Qualitative data is analysed and reported

10. Phase I, II & III: Discussion of findings & relationships between data sets

Quantitative and qualitative data is analysed and discussed

Figure 3.2. Research process flowchart.

#### **Research Paradigm**

In deciding the theory underlying this thesis, in the context of outdoor education and human research, the researcher endeavoured to find appropriate measures between the subject area and her personal philosophies (Allen, 1994). Due to the complexities of studying the development of resilience and coping in young people who are participating in outdoor adventure education activities, a mixed methods approach was utilised to quantify the outdoor education program outcomes and qualify the participants' rich meanings of the experience. It was therefore, deemed appropriate to use a pragmatic approach (Brantlinger, Klingner, Richardson, & Taylor, 2005; Creswell, Plano Clark, Gutmann, & Hanson, 2003; Ihuah & Eaton, 2013; Onwuegbuzie, Johnson, & Collins, 2009).

The pragmatist approach was chosen for this research as it utilises both qualitative and quantitative methods for data collection (Creswell, 2013; Ihuah & Eaton, 2013) and thus has the capacity to explore the four research questions. By utilising both subjective and objective reasoning, this approach allows the statistics from the survey data in Phase I to play a fundamental role in interpreting and giving meaning to the observation data from Phase II and the interview data from Phase III of the research.

This approach uses a variety of data collection and analysis techniques to assist with interpreting two separate styles of data. The data was collected concurrently with an aim to answer the research questions by forming conclusive meanings and confirming findings. This was achieved by comparing the survey data, observation data and interview data (Creswell, 2013; Creswell et al., 2003). In this research, the pragmatist approach combines two paradigms: postpositivist theory to analyse the quantitative data, and constructivist theory to analyse the qualitative data. Using these two approaches helps to strengthen the findings of both methods as it acknowledges that human experiences are complex and benefit from multiple forms of analysis (Guba & Lincoln, 1988a). It allows the survey data to be examined from both a scientific inquiry that emphasises a structured, singular stance and by considering qualitative data from a constructivist approach that focuses on understanding the phenomena from the participants' perspective (Guba & Lincoln, 1988a).

#### **Post-Positivist Paradigm**

Based on previous literature, it was hypothesised that participation in the outdoor education intervention would affect the development of participants' resilience and coping attributes (Cooper, 2004; Miller & Allen-Craig, 2005; Neill, 2008). To answer the first research question, survey data was collected to indicate any statistical changes in participants' resilience and coping scores. A postpositivist paradigm was used to analyse the survey data as this approach uses systematic, controlled and empirical methods (Ryan, 2006; Wildemuth, 1993). Within this paradigm, a quasi-scientific approach was used to identify changes in levels of resilience and coping by comparing the program group and control group participants. The quasi-scientific approach acknowledges that with human research there are many variables that cannot be accounted for in the same way that mathematical or laboratory scientific research can be controlled (Guba & Lincoln, 1988c).

The purpose of Phase I was to measure and compare the participants' resilience and coping scores pre-program and post-program to establish whether the outdoor education program affected the participants' levels of resilience and coping skills. Using the computer-based Statistical Package for the Social Sciences (SPSS) version 22 allowed the researcher to critically investigate the survey data and assess how attributes of resilience and coping were affected by participation in an outdoor education program (Ryan, 2006; Wildemuth, 1993).

To identify participants' self-reported changes in resilience and coping, the researcher distributed the surveys across a larger sample size compared to using only small group interviews. Using surveys may be less subjective as they have the ability to obtain larger sets of data across sample groups, whereas small group interviews provide a limited number of responses. Implementation of the surveys also allowed the researcher to remain external and independent from Phase I of the study (Ihuah & Eaton, 2013). A point of comparison was provided by having one group that experienced the outdoor education program, and one group that did not. Statistical findings from the survey data also identified the creation of new knowledge, understanding of the program, and gave further meaning to the interview data in Phase III (Ryan, 2006).

The pre- and post-survey data identified changes that occurred to participants through their experiences in the 'City to Summit' program. However, on its own, this data does not give insights into why the observed changes occurred. For this reason, a constructivist paradigm was utilised to better

understand the changes in survey data from the participants' and the researcher's perspectives.

## **Constructivist Paradigm**

The foundation of outdoor education is facilitating learning through the reflection of personal experiences in the social and natural world (Allan et al., 2012; McKenzie, 2000), therefore, taking a constructivist approach to researching outcomes of outdoor education experiences enables understanding of the phenomena from the participants' point of view (Savery & Duffy, 1995, 2001). It allows them to explain their personal insights and describe how these experiences have influenced their understandings of their own resilience and coping strategies.

Naturalistic inquiry was the constructivist approach selected, which enabled the researcher to be immersed in the research setting as an assistant on the program. In addition, as a professional outdoor educator, the researcher had a deep understanding of outdoor education and program design. The researcher's professional prior knowledge and experience supported immersion into the outdoor setting, while her role as the assistant group leader allowed for the adoption of an insider's view of the program without becoming a member of the observed group (Guba & Lincoln, 1988a).

The naturalistic inquiry approach promoted understanding of the complex nature of experiences from the participants' point of view. It also enhanced an understanding of how the construct of resilience and coping may be affected by participating in an outdoor education program. This was achieved through an open and exploratory view during the small group interviews (Guba & Lincoln, 1988a, 1988c). During interviews, participants were able to expand upon their selfidentified changes in coping and resilience, which allowed for rich descriptions of their experience and explanations of possible links to the changes identified. The insights from the participants gave meaning to the results of the survey data and strengthened the research analysis process.

Savery and Duffy's (1995, 2001) constructivist philosophy informed the identification process of the relationships between the participant's experiences during the extended journey style outdoor education program, and the development of their understandings. The following three aspects of their philosophy were used to inform and guide the qualitative data analysis process by understanding that:

- Our interactions with the environment can impact our construction of knowledge and affect our personal awareness or understandings;
- 2. Cognitive conflict or puzzlement is the stimulus for learning and determines the organisation and nature of what is learned; and
- Knowledge evolves through social negotiation and through the evaluation of the viability of individual understandings (Savery & Duffy, 1995, 2001).

Chapter 5 discusses the key themes, concepts and findings of Phase II and Phase III data, and considers the data and theory associated with research questions two, three and four.

## **Sample Selection**

### **Selection of Schools**

A broad range of Australian schools across the state of Victoria was examined to see which schools matched the selection criteria as outlined in Table 3.1. Sixteen Victorian schools were identified and were approached to participate in this study. Two private boys' schools from Melbourne, Victoria matched the research selection criteria and agreed to take part in this study (see Table 3.1).<sup>1</sup> The first school has been given the pseudonym Yarra College and will be referred to as the program group throughout this research. Yarra College was chosen because it matched the selection criteria, and because the outdoor education program is a compulsory for their entire Year 10 cohort. Specifically, this Year 10 cohort participated in an extended journey-style outdoor education program, which focused on the personal growth of students, including the development of resilience and coping skills. The extended length of the outdoor education program on offer and the potentially large sample size were also reasons for its selection. The second school, referred to by the pseudonym Beachside College, served as the control group throughout this research. Beachside College also matched the selection criteria but did not offer any outdoor education programs to their Year 10 cohort. This sample group was chosen to provide the comparison data set in the study.

<sup>&</sup>lt;sup>1</sup> Private schools in Melbourne are non-government schools. Along with government schools, these schools all follow the Australian Curriculum, Assessment and Reporting Authority. (ACARA). However, the non-government schools may be governed by a Catholic, private or independent organisation.

## Table 3.1

## Sample selection criteria and results for both sample groups

Selection Criteria	Yarra College	Beachside College	
Sample groups must be	School Location:	School Location:	
sourced from Victorian	Melbourne, Victoria	Melbourne, Victoria	
secondary schools			
Schools must be sourced	Non-government,	Non-government,	
from the same school sector	Private Boys School	Private Boys School	
Schools must provide a	Junior and secondary	Secondary only	
secondary school education			
Sample groups are of a	Year 10 all boys cohort	Year 10 all boys cohort	
similar age and gender			
Secondary school size and	Enrolments in 2012: 798	Enrolments in 2012: 492	
size of year levels must be	(For Junior and Secondary)	(For Secondary only)	
similar	Year Level size:	Year Level size:	
	Between 150-200 boys	Between 80-100 boys	
Sample Groups have	See Table 3.2 for details	See Table 3.2 for details	
similarities of societal			
demographics			
Delivers similar school	Follows the Australian	Follows the Australian	
curriculum	Curriculum, Assessment and	Curriculum, Assessment and	
	Reporting Authority	Reporting Authority	
	(ACARA)	(ACARA)	
Similar school focus	Educating the whole person,	Educating the whole person,	
	school values and life skills.	school values and life skills.	
Similar curriculum offerings	Compulsory outdoor	Compulsory outdoor	
for outdoor education	education program from	education program from	
	Prep to Year 12	Year 7 to Year 9	
One sample group offers an	Delivers a compulsory 21-	No outdoor education	
extended journey style	day outdoor education	offered to the Year 10 cohort	
outdoor education program	program ('City to Summit')		
to Year 10 cohort			

Note. Information presented in this table was provided by the sample groups themselves
As Table 3.1 demonstrates, both sample groups were sourced from private boys' schools which are based in inner-city Melbourne. The Index of Community Socio-Educational Advantage (ICSEA) has been reported for each school. ICSEA is a scale of socio-educational advantage, which provides a comparison of the two schools that have very similar demographics as far as social, cultural and school environments. Each school's ICSEA values are calculated on a scale which has a median of 1000 and a standard deviation of 100. ICSEA values typically range from approximately 500, which represents schools that have students who are classified as coming from extremely educationally disadvantaged backgrounds to a score of approximately 1300, which represents schools with students with very educationally advantaged backgrounds. Schools rating over 1000 on the scale are considered to have students with educationally advantaged backgrounds (Australian Curriculum and Assessment Authority, 2015).

Similarities and differences in the socio-educational factors for both schools are outlined in Table 3.2. These key factors are based on a student's family background such as their parents' occupation, and school or non-school education. These factors are pertinent because they may influence the students' educational outcomes at school. The school's geographical location and the proportion of non-English speaking and Indigenous students are also considered when determining their level of educational advantage or disadvantage (Australian Curriculum and Assessment Authority, 2015).

Table 3.2 shows that both schools are classified as having students from socio-educational advantaged backgrounds, as their ICSEA ratings are over 1000

117

(Yarra College: 1187; Beachside: 1078) and of similar median values. The main difference between the schools is that Beachside does not have a Junior School program and only runs compulsory outdoor education programs during Year 7 to Year 9. Their Year 10 cohort does not participate in any organised outdoor education at school during their Year 10 studies. Due to the high number of similarities between the schools, and the fact that the control group did not participate in any outdoor education programs, Beachside College provides a suitable comparison with Yarra College.

## Table 3.2

Factor	Yarra College	Beachside College	
School sector	Non-government	Non-government	
School type	Junior and secondary	Secondary only	
Location	Metropolitan	Metropolitan	
Total enrolments	798	492	
School ICSEA value	1187	1078	
Indigenous students	1%	0%	
Student attendance rate	91%	97%	
Whole school curriculum	ACARA	ACARA	
Religion	Anglican	Catholic	
Language background other	27%	41%	
than English			

## Socio-educational rating factors for both private boys' schools

Note. Data presented in this table is retrieved from Australian Curriculum and Assessment Authority (2012)

## **Participants**

The sample for this study was comprised of two groups: the program group and the control group (N = 300). Participants in both groups were Year 10 boys who were aged between 14 and 17 years. The researcher presented information about this study to Yarra College's students and parents at the 'City to Summit' program information night held at the school. Detailed research information and consent forms for 'City to Summit' were distributed to both parents and students. Originally, all Year 10 students (n = 200) from the program group agreed to take part in Phase I of the study. All participants in the observation group gave consent for photographs to be published (see Appendix V). This high rate of returned signed consent forms for participation in Phase I, may have been due to the fact that participation in 'City to Summit' is compulsory for the program group. Therefore, most documentation regarding 'City to Summit' is signed and returned by participants.

There were only 20 Year 10 boys from the program group (n = 200) who consented to participate in both the questionnaires in Phase I and the small group interviews in Phase III. These 20 participants from different activity groups were invited to participate in the small group semi-structured interviews immediately post-trip and six months after the completion of 'City to Summit'. However, two participants did not attend the small group interviews, therefore, bringing the total number of participants to eighteen for Phase III.

During Phase I, all participants were given an opportunity to complete the same questionnaires at Time 1 (T1) and Time 2 (T2). Initially, a smaller number of students from the program group (n = 95) completed the questionnaires at T1. However, at T2 the numbers of students completing questionnaires increased (n = 154) (see Table 3.3). The final number of completed questionnaires for the program group was 69. This number was reduced due to a high number of incomplete questionnaires and questionnaire data that did not match at T1 and T2 (refer to Table 3.3).

The entire Year 10 cohort (n = 100) from the control group were offered to participate in Phase I section of the study. Students and parents were supplied with written information and consent forms, which outlined the research in detail and the potential risks involved. The control group did not participate in any outdoor education experiences during their Year 10 studies at school as the school

120

did not offer any program options. However, it should be noted that the school does have a compulsory outdoor education program which runs from Year 7 to Year 9.

Table 3.3

Number of participants who completed the questionnaire data

	Time 1	Time 2	Matched	Number of	Final
	(T1)	(T2)	results	participants	participants
			from	removed	included
			T1 & T2		
Group	п	п	п	п	п
Program	95	154	73	$4^{*}$	69
group					
Control	62	67	50	$8^*$	42
group					

Note. \* = Number of participants removed from study due to 10% or more of the questionnaire being unanswered.

## **Outdoor Education Program**

## **Overview**

As part of the school curriculum, participants from Yarra College take part in a sequential, compulsory outdoor education program which operates from Prep to Year 12. Their outdoor education program design ensures that students in Years 7 to 10 are exposed to more challenging and complex outdoor activities as they progress through the year levels to the culmination outdoor education experience, 'City to Summit'. The Principal of Yarra College (2012) provided a summary of the aims and attributes that the school perceives to be crucial in the development of young people on their path towards adulthood through participation in outdoor education programs:

- 1. The outdoor education program presents challenges.
- 2. Builds aspects of leadership training.
- 3. Builds perseverance, resilience and teamwork skills.
- 4. Promotes an appreciation of the importance of the natural world.

Along with other aims and outcomes specific to each year level's school curriculum, the program's length increases each year and the outdoor activities become more challenging and complex as students' progress through the year levels. All scheduled activities are considered to be well within the capabilities of every student and the staff representatives from the school.

Being progressive in nature, the programs are designed to prepare the boys with skills and experiences required for their culminating flagship experience in the Year 10 'City to Summit' program. The Principal of Yarra College has identified that:

"[m] any (of the Year 10 students') lack clear goals for academic study, at a time when both positive and negative peer-related influences grow very strong in their lives. Some boys merely 'tread water' and become disengaged in the educative process. My hope is that the 'City to Summit' program will positively address such issues. I am excited by the opportunity, the learning experiences and the transferability of skills which will emerge from the unique program" (Yarra College Documentation, 2015).

## 'City to Summit' Outdoor Education Program

In Term IV 2012, the program group took part in 'City to Summit'. The intervention is a compulsory, extended, journey-style outdoor education program and is regarded as an integral part of the leadership, pastoral, personal development and community service programs at the Yarra College. The program was presented as an opportunity for students to challenge themselves and understand their capacities in responding to these challenges, take risks, learn new skills, to work together with their teammates and strengthen their developmental assets. During the program participants' will experience:

- A wilderness environment.
- A variety of 'hard' and 'soft' skill-learning experiences.
- Relative isolation from the comfort zone of family and school.
- A reflective personal and group experience.
- A celebration of the journey upon their return to school

(Yarra College Documentation, 2015).

An external not-for-profit outdoor education company was assigned to coordinate the three-week program. Groups travelled from outside their school gates in Melbourne and continued for 21 days until they reached the highest peak in Australia, Mt Kosciusko, which stands at 2,228m above sea level. The cohort of 200 Year 10 boys were divided into 17 smaller groups consisting of 12-14 participants, a group leader, a teacher or school staff representative and an adventure activity specialist. Two groups left at the same time following different routes that paralleled one another. Two days later, another two groups departed. This process continued until all 17 groups had embarked on the 'City to Summit' program (see Appendix M for the Program Schedule and Outline).

The boys were required to use a range of challenging adventure activities, such as hiking, whitewater rafting and mountain biking as modes of transport to reach their final destination. With the aim of building personal and interpersonal skills, City to Summit's program design also included group development activities, outdoor living and travel skills, initiative tasks, reflection, and individual and group feedback sessions. These curriculum elements occurred sporadically throughout the program. Through participation in 'City to Summit', it is proposed that participants will naturally develop the inherent practical skills of outdoor education, such as camp craft, camp hygiene, navigation, bush safety, and activity-specific tasks.

Set out by Yarra College, the six guiding principles that bind the 21 day outdoor learning experience are as follows:

**Journey:** Exploring the concept of journey and all that it entails. This includes the willingness to take risks, make decisions and live with the consequences. A journey is personal and continuous. It involves travelling to 'hard places' outside of our comfort zones whilst holding the emotional, social, physical and spiritual dimensions of our person together.

**Discovery:** To search, explore, discover, learn and grow. Insightful observations for the actions, events, phenomena, people, stories and places we do

not know or have not yet heard. Transforming these observations into new ways of thinking and being.

**Identity:** What does it mean to be me? Who I am and who I aspire to be. To explore one's personality in the context of social and cultural identity as a young male growing up in a rapidly changing society. 'City to Summit' provides an opportunity to explore notions around identity and masculinity in an environment free from media influence and some of the subsequent conflicting and contradictory messaging. This provides the opportunity for some reflection and thinking free from these often persuasive societal elements.

**Spirit:** "The non-quantifiable energy present in all living things"-Anonymous. The holistic connection between mind, body and soul as represented by the notion of the whole person entity. Taking this entity and exploring the connection with self, others and the environment and how these three dimensions are interdependent on each other.

**Collaboration:** In its most simple form, working together with others to realise shared and individual goals. The expression of leadership through the nurturing of creativity, knowledge sharing, care of others and personal reflection.

**Challenge:** Social, physical, mental, intellectual, spiritual and emotional. Challenge is many different things to many different people. At the heart of challenge is the willingness of the individual/group to embrace the opportunity that the challenge represents and continue to strive towards their individual/shared goals (Yarra College Documentation, 2012).

#### **Group Leaders and Teaching Staff**

Each group had one leader, plus another adult as a school representative. The school representatives were made up of teaching staff, other qualified outdoor education instructors or 'old boys' who previously attended the school. Each adventure activity also had an additional specialist staff member that would join the group during activity sessions. This group of staff were specifically qualified in the activity specialisation area.

As an integral part of the 'City to Summit' curriculum and program design, the group leader was responsible for delivering different aspects of the curriculum, including the importance of school culture, values and content specifically related to the program objectives. It is the role of the group leader to facilitate the sequentially challenging experiences in consultation with the school representative and their individual groups.

The outdoor education company's staff are responsible for running the program effectively and making sure all adventure activities and safety elements remain within the company's risk management policies and procedures. The group leaders also facilitate daily feedback, reflection, discussion and debriefing sessions, which focus on encouraging healthy group dynamics and awareness of self, others and the natural environment.

It is the role of the school representatives to provide pastoral care and supervision. As all school rules apply on camp, they also provide behaviour management and discipline in line with the school's expectations. These staff members often have a background and contextual knowledge of students which can greatly assist with group and individual learning throughout the program.

## **Program Outline**

'City to Summit' is guided by experiential learning processes which was facilitated by the group leader from an external outdoor education provider. The program design of 'City to Summit' was delivered in three phases. In this thesis, this will be known as:

- Phase 1) Frontloading Phase: Pre-activity outdoor education
- Phase 2) Immersion Phase: The outdoor education program
- Phase 3) Integration Phase: Post-activity outdoor education

## Phase 1: Frontloading phase.

The day before City to Summit's Immersion Phase, the boys' participated in a pre-trip training day (Day 0) which was held at the school to frontload the boys with important trip information before departure. Day 0 was facilitated by a senior leading teacher, who focused on delivering leadership training, trip briefings, goal setting, teamwork and initiative activities in an effort to set the scene for 'City to Summit'.

## Phase 2: Immersion phase.

The Immersion Phase started on day one and saw the boys leave on buses from the school gates before arriving at Mount Beauty to prepare for their journey ahead. Groups started by hiking from Windy Corner at Falls Creek until they reached the banks of the Mitta Mitta River (hiking example provided in Figure *3.3*). The next leg of the program involved whitewater rafting down the Mitta Mitta River from Jokers Flat to Taylors Crossing where they start their mountain biking leg. After the mountain biking section, the boys embarked on a 24-hour solo experience. Their last stretch of the program consisted of hiking their way to Mt Kosciusko to complete their journey (see Appendix M for the detailed 17 group program outline).



Figure 3.3. The journey begins: Group 9, Day 1 hiking leg.

Around the halfway mark of 'City to Summit', all participants engaged in a 24 hour solo experience, where they slept overnight in a wilderness setting away from any other participants or group leaders. By this stage of the program, participants were expected to be able to set up their own shelter, organise food and reflect on their experiences so far. The opportunity also required the boys to manage themselves and make their own decisions for a 24 hour period without the assistance from others. Participants are thoroughly briefed on the safety and protocol of the solo activity and are placed in a specific location that is known to group leaders. The group leaders and the school representatives are located in a safe area where participants can return to in case of an emergency. Day 21 was the last day of the journey section and required travelling back to school on a bus from Mt Kosciusko.

#### **Phase 3: Integration phase.**

Day 22 comprised of a debriefing day back at the school which was aimed at summarising program outcomes, linking the boys' experience back to the goals they had set on day zero and help them to integrate learnings from the program into their school lives (see Appendix M for the Program Schedule and Outline).

## **Ethical Issues and Considerations**

This research project was classified as high risk as it involved human participants who were minors in a school setting. It gained Human Research Ethics Approval (Approval Code HREC 12/141/HRETH 12/198) (see Appendix A) from the Arts, Education and Human Development Human Research Ethics Committee (AEHD HREC) of Victoria University. The researcher had meetings with the Principals and relevant teaching staff from both Yarra College and Beachside College to discuss and explain the nature of the research before gaining verbal and written consent (see Appendix B for the invitation letter to the program group Principal and Appendix C for the invitation letter to the control group Principal). Parents and students from Yarra College were informed about the research and the potential risks involved during a parent information night held at the school three months prior to 'City to Summit'. This was accompanied by the school's camp documentation along with the written information and consent forms outlining the research (see Appendix H for the information letter and Appendix J & K for the consent forms provided to the program groups' parents and participants). Parents and students from Beachside College received detailed written information and consent forms which outlined this research and any potential risks (see Appendix I for the information letter and Appendix L for the consent forms provided to the parents and participants of the control group).

It was made clear that participation in the research was voluntary for all parties involved. Given the participants' age bracket (14 -17 years) and their ability to make decisions, consent forms were received from both participants and parents/guardians who agreed to take part in the study (Fontana & Frey, 2003; Punch, 1994). Confidentiality and privacy implications were explained to the participants in the consent forms and verbally during interviews. It was made clear that the data collected and presented would not be associated with the participants' names and would remain confidential. All boys who participated in the semi-structured interviews were given pseudonyms to ensure confidentiality (Fontana & Frey, 2003; Punch, 1994).

Following approval from the school, parents, and participants, Phase I questionnaires were administered to the program group on the pre-trip planning and packing day of the program (see Appendix D for the information letter and Appendix F for the consent form provided to Yarra College and see Appendix E for information letter and Appendix G for consent form provided to Beachside College). Participants and staff in Group 9 that were involved in Phase II gave permission for photos collected during the program to be used and printed throughout this thesis (see Appendix V).

To ensure the participants' voluntary participation, the researcher was onsite at the school's during all data collection phases to answer any questions and reiterate that participants could withdraw from the study at any time, without feeling any type of obligation. Before each data collection phase, participants were reminded of the confidentiality and privacy implications (written and verbal) and were asked to respond to questions honestly.

During Phase III, further explanation was given about the research and how the data would be collected and used. All information provided to participants was in plain language statements. It was made clear to the participants that no personal names would be used and all recordings and responses would be kept confidential. Participants understood that counselling services at Yarra College or via Victoria University were available to them if they did experience any form of anxiety or alternative emotional responses as a result of answering the questions.

## **Data Collection Process**

The data collection process is described in three phases which identify the data collection methods, procedures and analysis. The first phase involves the collection of survey data from both sample groups. The second phase includes the

collection of researcher observations during 'City to Summit', and the final phase is the implementation of the small group interviews with the program group only. In order to respond to the research questions of this study, each phase in the research process was used to inform subsequent phases. The research timeline is presented in Table 3.4, and *Figure 3.2* presents the research process flow chart. A conceptual overview of the research process is outlined below:

#### Phase I: Survey Data

Pre- and post-survey data from both groups are included in this phase.

#### **Phase II: Program Observation Data**

The researcher acted as an adult support staff for one of the 17 activity groups. She recorded observations of daily activities and events occurring during Group 9's experience of 'City to Summit'. Group 9 consisted of 13 boys, one group leader, activity support staff and the researcher.

## **Phase III: Interview Data**

Small group, semi-structured interview sets are recorded upon the completion of the outdoor education program with eighteen boys. The first set of interviews is followed up with a second set of interviews six months post-program.

# Table 3.4

Research				
Process	Action		Phase Description	
Phases				
Planning &	Ongoing		Used to inform the instruments used in Phase I.	
research	literature		Guided practice for recording observation diary's in	
	search and		Phase II.	
	review		Guided the development of the semi-structured	
			interviews guides in Phase III.	
Planning &	Survey		Guided practice for recording observation diary's in	
research	instruments		Phase II.	
	chosen		Guided the development of semi-structured interviews in Phase III.	
Phase I	Pre-survey		Program group data was collected at school on 21 &	
	data		22 September 2012 during Day 0, the training day	
			before the program group started participation in 'City	
		_	to Summit .	
			Control group pre-survey data was collected on 11 October 2012.	
			The first set of data collected in Phase I was used as	
			the baseline scores for a comparative data set to see if	
			resilience and coping scores changed over time and	
			after the program group's participation in 'City to	
			Summit'.	
Phase II	Observation		Group observations and notes were recorded from 27	
	uata		October to 18 November 2012 by the researcher while	
			as an Assistant Group Leader with Activity Group 9	
			Guided the development of semi-structured interviews	
			in Phase III.	
			Used to compare findings and relationships in	
			qualitative data.	
Phase I	Post-survey		Throughout November 2012 the program group	
	data		completed post-survey's immediately after completion	
			of 'City to Summit', during the debrief day at school.	
			On the 20 November 2012, the control group	
			completed the post-survey data at school which was	
			within a similar timeframe as the program group data	
			collection.	
			Data used as a comparison to the first data set to find	
			examine if levels of resilience and coping had changed	
			over time and between groups after the program	
		_	groups participation in City to Summit'.	
			Findings were used to inform the interview guides.	
			Quantitative data used to compare the findings of qualitative data sets	
			quantative uata sets.	

# Data collection timeline with descriptions of each phase

Phase III	First set of	From 20 to 28 November 2012, four separate small
	interview	group interviews were conducted with eighteen
	data	students from the program group after participation in
		'City to Summit'.
		The four small group interviews ranged in size from
		three to five participants in each interview group.
		The eighteen participants consisted of boys from seven
		different activity groups.
		Used to inform the second semi-structured interviews.
		Used to compare findings and relationships to find out
		what types of resilience attributes and coping skills the
		boys applied during the 'City to Summit' program,
		what were the most challenging experiences and how
		they responded to them, and to see if their self-reported
		levels of resilience and coping had changed.
Data Analysis		Pre- and post-survey data was compared and analysed.
		The observation data in Phase II and the first set of
		interview data in Phase III was compared and
		analysed.
		Results from the data analysis was used to inform the
		second semi-structured interview guide in Phase III.
Phase III	Second set	From 27 to 30 May 2012, four separate small group
	of Interview	interviews were conducted with the same eighteen
	data	students from the program group six months after
		participation in 'City to Summit'.
		Interview data was used as a comparison to the first set
		of interview data.
		Used to compare themes, findings and relationships to
		examine if;
		• The resilience attributes and coping skills that
		were used were context-specific to the setting
		in which they were developed;
		• These skills transferred into the lives of the
		young people back at school; and
		• Participants were able to draw upon these
		resilience and coping attributes when
		confronted with adversity in the context of
		their school lives.
Data Analysis	Quantitative	Results from pre- and post-survey data from Phase I
	and	was reported and compared with the findings from
	qualitative	qualitative data sets in Phase II and Phase III.
	data	Discussion of relationships, themes and findings are
	analysis	reported in Chapter 6.

## Table 3.4 (continued)

Note. Table sequence is in order of data collection dates. See Appendix P: Data Collection Phases and Interview Schedule

## **Phase I: Survey Data Collection and Procedures**

This section describes the quantitative research methods, the selection process for the survey instruments and the attributes the instruments measure. This is followed by a description of the procedures for pre- and postadministration of the instruments, data collection, data analysis and reliability analysis.

## Instruments.

In order to measure changes in resilience and coping attributes after participation in an extended journey style outdoor education program, the Resilience Scale (Wagnild & Young, 1993) (see Appendix N) and the Brief COPE scale (Carver, 1997) (see Appendix O) were administered to participants in the program group pre- and post-participation in the outdoor education program. It is also important to note that the Resilience Scale and Brief COPE were combined into one questionnaire and completed by participants at the same time, at both preand post-collection of survey data. To evaluate the effect of the program, the control group also completed two sets of questionnaires within a similar time frame as the program group.

## The Resilience Scale.

Resilience was measured using the validated Resilience Scale (Wagnild, 2009; Wagnild & Young, 1993). The Resilience Scale has been identified as an effective tool to measure resilience in many populations worldwide (Ahern, 2006; Hunter & Chandler, 1999; Leppert, Gunzelmann, Schumacher, Strauss, &

Brähler, 2005; Wagnild, 2009) and has previously been used effectively in outdoor education settings with young people (Neill & Dias, 2001; Skehill, 2001) (Cronbach's alpha = 0.72; Hunter & Chandler, 1999).

The Resilience Scale measures five core components of resilience: having a purposeful life, perseverance, equanimity, self-reliance and existential aloneness (Wagnild, 2009, 2010; Wagnild & Young, 1993). This scale has been shown to have strong reliability and validity (Wagnild, 2009; Wagnild & Young, 1993; Windle, Bennett, & Noyes, 2011) with original internal consistency reported as high (Cronbach's alpha > 0.8; Wagnild & Young, 1993). A review of the Resilience Scale was conducted by Wagnild (2009) and analysed 12 studies which used the Resilience Scale from 1993 to 2007. Internal consistency reliability of the Resilience Scale was regularly high in 11 of 12 reviewed studies (Cronbach's alpha coefficient ranged from .85 to .94; Wagnild, 2009). The 12 studies analysed in the review of the Resilience Scale provided clear evidence of instrument validity and construct validity (Wagnild, 2009).

The 25 items reflecting the five elements of resilience in the scale are positively worded, and responses are presented on a Likert scale ranging from 1 (agree) to 7 (disagree) (Wagnild, 2009; Wagnild & Young, 1993).

The five interrelated psychological attributes that encompass the multifaceted construct of resilience measured by the Resilience Scale are:

 Purposeful Life: Having an understanding of one's purpose in life. This involves understanding one's life goals and how these can be achieved in challenging times and adverse conditions. For example,

136

being able to identify personal life goals such as achieving high results on an exam or winning a football grand final, and then actively working towards achieving these goals (Wagnild, 2009, 2010; Wagnild & Young, 1993).

- Perseverance: Having the determination to persist in an activity or goal pursuit despite adversities or disappointment. Perseverance allows someone to continue to be involved in a situation through selfdiscipline and determination, despite being discouraged by a difficult situation. For example, a student may demonstrate perseverance if they revise and study for upcoming examinations despite receiving poor results in previous tests (Wagnild, 2009, 2010; Wagnild & Young, 1993).
- Self-reliance: Refers to understanding your personal capabilities and limitations yet maintaining an inherently positive self-belief. This may be demonstrated by a student who cooks a meal for themselves for the first time on an outdoor education program (Wagnild, 2009, 2010; Wagnild & Young, 1993).
- 4. Equanimity: Having an optimistic outlook, which brings balance and harmony to one's life. Equanimity is having a balanced perspective by being able to moderate extreme responses to difficult situations. For example, being able to view the situation from a positive perspective even if a mistake has been made or one has experienced a bad day (Wagnild, 2009, 2010; Wagnild & Young, 1993).

5. Existential Aloneness: Otherwise known as coming home to yourself. This is related to knowing one's self and being "comfortable in your own skin" (Wagnild, 2010). Even though experiences may be shared with others, some experiences in life must be faced alone. Understanding that it is the individual who makes their choices and life decisions. An example of this is being able to feel comfortable on a solo section of an outdoor education program or feeling content being alone without feeling the need to have other people around to make decisions for you (Wagnild, 2009, 2010; Wagnild & Young, 1993).

#### The Brief COPE.

The Brief COPE scale was implemented in this study to measure the construct of coping, as it has been used successfully in previous research with adolescents in an outdoor education context (Ewert & Yoshino, 2008) (see Appendix O for the Brief COPE). The scale has demonstrated both reliability and validity. Carver (1997) conducted a reliability analysis to evaluate the internal structure of the Brief COPE. The internal consistency alphas ranged from .52 to .90, however the Acceptance, Denial and Venting scales had alpha values lower than .60 (Carver, 1997). These statistics are deemed by researchers as acceptable internal reliabilities (Yusoff, Low, & Yip, 2010), as .50 is an accepted minimal value for internal reliability (Nunnaly, 1978). Carver (1997) also conducted an exploratory factor analysis (EFA) to ensure validity of the internal structure of the Brief COPE. The EFA produced nine factors of which eight factors were greater than 1.0 and accounted for 72.4% of the variance.

Participants were asked to respond to a coping statement based on a fourpoint scale, with answers ranging from one: 'I don't do this at all' to four: 'I do this a lot'. Five represented 'I prefer not to answer' (see Appendix O). Questions were neither positively nor negatively worded. The scale measured 28 items with 14 subscales of different coping skills (summarised in Table 3.5). These 14 different subscales measured a variety of both emotion-focused and problemfocused coping skills, which also measured various responses in relation to known effective and ineffective coping methods (Carver, 1997). Caver (2007) explicitly points out that each subscale should be looked at separately as he has not labelled the subscales as either adaptive or maladaptive (Carver, 2007). This supports the research by Lazarus and Folkman (1984) who suggest that coping responses should not be labelled as inherently good or bad, but rather should be viewed as emotion-focused or problem-focused coping skills. Henceforth, this research will treat coping skills within each factor as emotion-focused and problem-focused skills, rather than labelling them as maladaptive or adaptive coping skills (summarised in Table 3.5).

## Table 3.5

Emotion-focused and problem-focused examples of coping within the 14 subscales of the Brief COPE

Emotion-focused coping dimensions					
Item	Description	Example			
Self-distraction, items 1 and 19	Keeping yourself busy to take mind off the issue.	Watching TV.			
Active Coping, items 2 and 7	Taking action to help yourself.	Gratitude writing or journaling.			
Denial, items 3 and 8	Suppressing negative thoughts or emotions.	Pretend the problem doesn't exist.			
Substance Use, items 4 and 11	Using substances to cope.	Drinking alcohol or using drugs.			
Behavioural Disengagement, items 6 and 16	Another form of distraction or active avoidance	Arriving late to class or intentionally not paying attention.			
Venting, items 9 and 21	Emotional disclosure. A two-way process: the person venting and the person hearing the vent.	Expressing strong emotions to a friend.			
Humour, items 18 and 28	Can suppress negative thoughts and emotions through distractions. Can relive tension and anxiety.	Making a joke out of a difficult situation. Laughing in the face of adversity.			
Religion, items 22 and 27	Fining strength from personal belief systems.	Praying for guidance and strength.			
Self-blame, items 13 and 26	Cognitive reappraisal of self	Feelings and excessive thoughts of real or perceived failure.			

Table 3.5 (continued)

Problem-focused coping dimensions		
Item	Description	Example
Use of Emotional Support, items 5 and 15	Gaining emotional assistance externally to cope.	Seeking support from a counsellor.
Use of Instrumental Support, items 10 and 23	Tangible help that others provide.	Seeking support from friends and family to drive you to school.
Positive Reframing, items 12 and 17	Problem-solving with positive thoughts.	Use of positive affirmations and visualisations.
Planning, items 14 and 25	Problem-solving or time management.	Planning you day or week in your diary.
Acceptance, items 20 and 24	Awareness and acknowledgment of circumstances.	Staying optimistic regardless of current hardship.

## **Survey Data Collection and Procedure**

Consent to participate in the research was obtained from both the participants themselves and their parent or primary caregiver. The questionnaire administered included both the Resilience Scale and the Brief COPE Scale and was completed online by participants using Qualtrics online survey software. During the pre-trip preparation day (Day 0), students from the program group (N = 200) completed the questionnaire before the program commenced at their school in private computer booths. This dataset is referred to as Time 1 (T1). The same students repeated the questionnaires at school, in the same computer booths after completion of the program during the post-trip debrief on day 22. This dataset is referred to as Time 2 (T2). See Appendix P for detailed information of data collection phases and the interview schedule.

Participants from the control group completed the same questionnaire at school at both T1 and T2 within a similar three-week timeframe. The questionnaires at T1 and T2 were matched to individual students through a birthday code to maintain the privacy of all students.

#### Survey Data Analysis

As a comparison group, this study utilised data from a control group that was not exposed to any sort of outdoor education programs during their Year 10 studies. The two data sets (pre- and post-) from the control group were used as comparative data sets against the two data sets (pre- and post-) from the program group. Having the control group of students provided an opportunity to compare the data sets in order to identify if the control groups results remained the same with no intervention and to see if the program group showed any changes in resilience attributes and coping skills after participation in 'City to Summit'. The pre-survey results provided a baseline data set to see if both groups showed similar results before participation in 'City to Summit'. The second set of data allowed the researcher to identify if participation in 'City to Summit' affected levels of resilience and coping attributes compared to the control group who did not participate in any outdoor education programs within the same timeframe.

Questionnaire data was analysed to provide both descriptive and inferential statistics. The survey data collected from the Resilience Scale and the Brief COPE Scale were analysed and statistically described using the computerbased SPSS version 22. ANOVA was used to evaluate the influence of outdoor education program on the five resilience attributes and the 14 coping subscale scores over time and within groups (pre- and post-).

The primary goal of the data analysis was to compare the group's results and determine if either group had developed resilience and coping skills over time. Based on results from previous research (e.g., Ewert & Yoshino, 2011; Gillespie & Allen-Craig, 2009; Neill & Dias, 2001) and because of the exposure to risk, challenges and new supportive environments, which require the application of resilience and coping skills, the researcher predicted that the outdoor education program would be more beneficial in helping students to develop resilience and coping skills compared to other experiences in their daily school lives. Statistical analyses involved the following procedures:

- 1. Factor structure was evaluated using Exploratory Factor Analysis.
- 2. The internal consistency of the test was assessed by calculating the Cronbach's alpha coefficient for each scale. Correlations were calculated between the Resilience Scale and the revised factor structure of the Brief COPE scale to demonstrate the relationships between the factor variables of each scale.
- Descriptive statistics were calculated for both groups with each scale, including means, standard deviations, and skewness. Frequencies are also presented.
- 4. ANOVA and repeated measures t-tests were used to compare means between groups at both time points and within groups over time.
  Significance was assumed at p < .05.</li>

#### **Exploratory Factor Analysis Procedure**

Consistent with previous studies, a procedure of factor structure refinement of the Brief COPE was adopted (Carver, 1997; Ewert & Yoshino, 2008; Hastings et al., 2005; Kimemia, Asner-Self, & Daire, 2011) due to the 14 subscales not being previously determined as either problem-focused or emotionfocused skills (Carver, 1997; Ewert & Yoshino, 2008).

This research uses an Exploratory Factor Analysis (EFA) to determine a revised factor structure, with an aim to refine complex patterns by exploring the relationships of variables and factors in the Brief COPE (Ferguson & Cox, 1993). The researcher analysed the participant data to examine the factor structure of the self-report measures using the principal axis factoring extraction with oblique rotation (direct oblimin) (Child, 2006). Principal axis factoring extraction was utilised as it acknowledges communality estimates and includes an error term within its model, consequently, avoiding the assumption within principal components analysis that the variables will correlate perfectly (Ferguson & Cox, 1993; Gorsuch, 1997).

Oblique rotation was selected instead of orthogonal rotation due to the increased possibility of a correlation between the factors being explored in the Brief COPE scale. The EFA used mathematical procedures for the interpretation of interrelated measures to discover patterns in the set of coping dimensions (Gorsuch, 1997; Yong & Pearce, 2013). Following consideration of the statistical results, factor patterns will also be assessed and considered in relation to logical analysis (Boros et al., 2000).

## **Reliability Analysis**

Internal consistencies for the measures were calculated for both scales using Cronbach's alpha ( $\alpha$ ). The original values published in the Resilience Scale's Manual (Wagnild & Guinn, 2011), were consistent with the reliability values in this research. For different types of research, such as group research purposes, the typical expected minimum reliability is .70 (Cortina, 1993; Gliem & Gliem, 2003; Nunnaly, 1978), however, other authors (Salvia, Ysseldyke, & Bolt, 2007; Santos, 1999) have indicated, .60 is also considered to be a lower limit of acceptability for the reliability coefficient.

145

## **Phase II: Program Observation Data**

The researcher attended the outdoor education program as an assistant guide to observe one group's experience. The researcher's participation in the program provided an insight into the goals and activities of 'City to Summit', which helped to inform the design of the semi-structured interview guides in Phase III.

During Phase II, the researcher observed the activities during 'City to Summit' and the students' responses to these experiences. Recordings of the researcher observations were kept in 3 parts:

- 1. **Descriptive Diary:** Recording of the facts about daily activities of the group and their reactions. This was a clinical account of the program.
- Observation Diary: Detailing observations about the group, individual's personal development and the participant's emotional responses.
- 3. **Personal Diary:** Reflections presenting a personal account of the researcher's feelings, thoughts, emotions and experiences.

Findings from the diaries, including participants' comments and specific examples are presented in Chapter 5.

## **Descriptive Diary**

The descriptive diary was used to record specific facts and did not analyse or portray any preconceived ideas or notions about the observations. The descriptions provide accounts, clear explanations and examples of what occurred during the program. It also provides actual quotes provided by the participants. The descriptive diary followed the following format:

- Day, date, time.
- Weather.
- Description of the day (e.g., activities, locations, landmarks, challenges etc.).
- Student responses to the activities and program challenges.

## **Observation Diary**

The information recorded in the observation diary was guided by the literature and survey instruments used in Phase I. The observation diary took the form of a narrative, as the researcher was looking for specific themes and examples of participants' experiences and responses observed during 'City to Summit'. The information recorded included identification of the participants displaying examples of:

- Resilience attributes (purposeful life, equanimity, perseverance, self-reliance, existential aloneness).
- Coping skills (problem-focused or emotion-focused).
- Application of other possible constructs of resilience (e.g., selfefficacy, grit, hardiness, mental-toughness, self-concept, internal locus of control, restored functioning, self-actualisation, adaption etc.).
- Other potential contributing factors (e.g., challenge level, supportive environment, social needs, social support, well-being, psychological state etc.).

#### **Personal Diary**

Prior to the research taking place, the researcher had been a professional outdoor educator for over a decade with an extensive understanding of skills and knowledge in the field of outdoor education. This allowed her to step back from the group dynamics and consider the interaction of participants during the program. A personal diary of the researcher's conceptual thoughts, emotions and experiences were kept separately from the other two diaries. This diary outlined the researcher's personal feelings, emotions and interactions with the group. These accounts allowed for the ability to monitor any biases the researcher may have had.

#### **Program Observation Analysis**

The researcher's observational diaries recorded during 'City to Summit' were analysed for emerging themes and then compared with the relevant literature and survey instruments used, which helped to inform the coding processes used in Phase III. To maintain confidentiality, all observations and participant comments presented in this research include pseudonyms. Recorded themes of observational data were categorised and used to inform the questions for the first semistructured interview guide (see Appendix Q). The emergent themes from the diaries were analysed and refined four times to help guide and inform the interview questions used in Phase III. The findings from these observations may help to give further meaning to participants' comments, personal understandings, behaviour and perceptions during the interviews. Phase II data was compared with findings from both the survey data and interview data, and the relationships are discussed in Chapter 6.

## **Trustworthiness of Phase II Data**

Due to the nature of collecting Phase II data through personal observations and interactions, the trustworthiness of the data has the potential to be undermined by personal bias, misinterpretation or misrepresentation by the researcher (Shenton, 2004). To overcome this, the trustworthiness of the observation data is supported in several ways by the researcher:

- Recording the personal diary which outlines the researcher's own emotions and involvement with the group. This allowed for assessment of potential researcher influences and bias, including the possible effect on the participants' experiences, responses and behaviours during 'City to Summit' (Brymer, 2002).
- Taking the methodological stance of being a peripheral member of the group as much as possible. The researcher was involved enough with the group to establish an insider's identity without having an overbearing input or effect on group decision making (Fontana & Frey, 1994).
- Recording the context of the research environment in the descriptive diary. It was important that the person collecting the data kept the observations in the context of the research environment that was being observed, ensuring reliability and validity of the data in the context of observations in natural environments (S. J. Taylor & Bodgan, 1984).

• Having transparent diaries that were cross-reviewed and validated by two separate researchers (Shenton, 2004).

#### **Phase III: Interview Data Collection and Procedures**

The previous section summarised the methods and procedures of survey data collection from Phase I and outlined the researcher's program observations from Phase II. This next section describes the methods for data collection processes and procedures for the small group semi-structured interviews in Phase III.

## **Semi-Structured Interviews**

Interviews are an effective data collection method that can be used within a naturalistic inquiry approach for investigating phenomena because participants articulate the experience from their own perspective (Denzin & Lincoln, 1994, 2011, 2003; Fontana & Frey, 2003; Guba & Lincoln, 1988a). Interviews generate rich descriptions to explore the context and experiences of the participants in relation to their resilience and coping strategies (Denzin & Lincoln, 1994; Fontana & Frey, 2003; Herrman et al., 2011; Sofaer, 1999) which would not be possible without face-to-face verbal interaction (Fontana & Frey, 2003; Guba & Lincoln, 1988b). Interviews were conducted with eighteen participants immediately after 'City to Summit' and then again six months post-program.

The aim of conducting semi-structured interviews immediately after 'City to Summit' was to explore the students' initial responses to spending 21 days travelling as a small self-contained group through semi-wilderness areas, as well as understanding how these experiences may have influenced their development of resilience attributes and coping skills. Conducting a second set of interviews six months post-program, provided the opportunity to investigate any changes that had occurred during that period and to investigate whether the participants had applied any identified change in resilience or coping skills beyond the scope of the outdoor education program.

#### Semi-structured interview guide 1.

The first interview guide was informed by the literature review, survey instruments and the researcher's observations during the outdoor education program. Themes of perseverance, self-reliance, existential aloneness, equanimity and meaningfulness (purposeful life) as outlined in the Resilience Scale were explored. The problem-focused and emotion-focused subscales as categorised in the Brief COPE were also investigated. The examination of themes commenced with general comments about the boy's experiences and transitioned to focussing on specific examples of their actions, behaviours and emotional responses.

The first set of four small group interviews were conducted straight after the experience. The initial interviews focused on the participants' personal experiences during the outdoor education program. The focus of these interviews was to investigate how the participants responded to program challenges and to see if they were able to identify their use of, or change in, resilience attributes and coping skills.

To allow for the natural flow of conversation, questions were open-ended and not specifically set in order. Participants were asked to conceptualise their understating of resilience in their own words and were questioned about any changes they noticed within themselves, their behaviours or relationships with others during or immediately after 'City to Summit'. If the boys started talking about a topic, the researcher was able to facilitate the flow, then bring the awareness back to the interview guide. In addition, the researcher's observations from the outdoor education program guided the discussion examining the role of supportive environments, staff impacts, group dynamics, leadership, personal learnings, self-concept and self-control (see Appendix Q for Interview Guide 1). Responses then informed the process and questions for exploring how resilience and coping skills are potentially transferred across settings in the second set of interviews.

#### Semi-structured interview guide 2.

The focus of the second interview was to explore any changes in resilience attributes and coping skills that participants identified in their first interview. These interviews were intended to answer research questions two, three and four:

### **Research Question 2**

If attributes were enhanced, what supported and fostered the development of these resilience attributes and coping skills?

#### **Research Question 3**

If attributes were enhanced, what attributes of resilience and coping were transferred and applied in contexts other than an outdoor education setting?

## **Research Question 4**
How can resilience and coping skills be best developed and transferred into other contexts of learning?

The second interview guide was comprised of two parts. The first part investigated if the boys recognised or recalled their personal learnings outlined in the first set of interviews. Specifically, the questions were related the changes they identified, their emotional responses and personal perceptions on the way they dealt with program challenges. The second part was intended to clarify if they had applied the identified changes or key personal learnings into other contexts of their lives.

As the groups included the same participants from the initial interview, this guide was structured based on the participants' previous comments about 'City to Summit'. Participants were asked whether they could recall their responses to personal learnings or memories they discussed in the first interview. If they were unable to recall their responses, the researcher reminded them of their initial response and investigated whether they still felt the same way. For example, one of the themes that emerged in the first set of interviews was appreciation, which helped to inform specific questions such as:

Researcher: Now that you guys are back to 'reality' as you called it, how have you gone at remembering the things you appreciated most?

Why do you think that is?

From the initial interview, do you remember anything you learnt that you said you will take from the trip and utilise it back to your school life? Do you think you have done that? Can you give me any examples?

Responses were then followed by an inquiry into the participant's experience of transferring their learning into other contexts as well as an enquiry as to why this may or may not have occurred (see Appendix R for Interview Guide 2).

## **Interview Data Collection and Procedure**

The first set of small group interviews were conducted the day after completion of their 'City to Summit' experience. Initially, the 20 consenting participants were divided into four groups of five boys. Two students did not attend; therefore, the total number of participants was eighteen. The final small groups ranged in size from 3-5 participants in each group. The eighteen participants were sourced from seven different activity groups from within the program group, with some participants experiencing 'City to Summit' together (see Table 3.4). The second set of small group interviews was conducted six months after the conclusion of the program with the same eighteen boys in the same four groups. All pre- and post-interviews typically took an hour and were conducted in a private room at the school to insure confidentiality (see Appendix P for the detailed Interview Schedule).

#### **Interview Data Analysis**

Interviews were conducted by the researcher and were digitally audiorecorded and transcribed by the researcher and another professional transcriber. All boys who participated in the small group interviews were given pseudonyms to protect their identity and keep their responses confidential. The researcher analysed the interview data in three stages:

**Stage 1:** The first set of interviews were transcribed and analysed for emerging themes using Nvivo qualitative data analysis software (Creswell et al., 2003). The emerging themes were compared with the themes from the observational data in Phase II, along with relevant literature, which then informed the development of second semi-structured interview guide. This comparison process allowed the researcher to cross-check whether the themes found in observational data were similar to findings in other studies as well as findings from the interview data, giving further credibility to the findings. Recurrent themes for each participant were also coded (Fossey, Harvey, McDermott, & Davidson, 2002). After the first set of interview data was collected and coded, common themes from the observational dairies were compared with the first set of interview data, which helped to inform the second semi-structured interview guide (see Appendix R).

**Stage 2:** The second set of interviews were transcribed and analysed for emerging themes using Nvivo qualitative data analysis software. As the analysis progressed, theories emerged which provided an understanding of the influences of the outdoor education program on the development of resilience attributes and

coping skills, as well as the strengths and weaknesses of the program over time (Schwandt, 2007). Themes were analysed and coded based on the literature, survey data, observation data and the first set of interview data.

**Stage 3:** The final stage of the qualitative data analysis compared the findings from Stage 1 with the findings from Stage 2. The researcher's coding analysis was independently validated by two separate researchers. Identified themes were then brought together into meaningful relation with each other, and a structural synthesis of the core elements of participants' experiences described in the interviews was formed. From the initial large set of codes identified, a reduction stage included three separate cycles of analysis (Fossey et al., 2002).

- Cycle 1: The first cycle of qualitative data analysis involved the identification of all higher order parent themes and child themes within Phase II and Phase III.
- Cycle 2: Key relationships were analysed, and themes are combined.
  Structural synthesis of the core elements in both Phase II and Phase III are compared and analysed.
- 3. Cycle 3: Final higher order themes were identified.

The findings of qualitative data sets were used to address the aims and answer the research questions of this study. This provided the platform for the researcher to understand the relationships between the changes in participants levels of resilience and coping attributes in the context of their outdoor education experiences. Qualitative results and analysis of Phase II and Phase III are presented in Chapter 5 (see *Figure 3.2* for the flowchart of data analysis).

## **Trustworthiness of Phase III Data**

The terms validity, reliability and generality in qualitative research are up for debate (Onwuegbuzie & Leech, 2007). These scientific terms and concepts apply more to quantitative research (Kvale, 1995) and may be deemed as being inappropriate for qualitative research due to external manipulating factors, such as the potential for influence or error caused by the researcher (Yardley, 2008). However, other research (Brantlinger et al., 2005; Mack, Woodsong, MacQueen, Guest, & Namey, 2005) has shown the advantages of using qualitative methods, which include allowing the participants to express their views freely, allowing for topics of discussion to be explored in greater depth (Mack et al., 2005).

The semi-structured small group interviews aimed to investigate individual experiences and their responses to participation in 'City to Summit', and to provide context for the values reported in the survey data (Yardley, 2008). Whereas, the observation data collected was implemented to provide real-life descriptive accounts of observations made during the program from the researcher's point of view. The qualitative data collected during Phase II and Phase III provided rich descriptions of complex phenomena to give meaning and understanding to the multifaceted dimensions of human interaction including, biological, psychological, social and cultural influences of the participant's experiences (Fossey et al., 2002; Sofaer, 1999).

The qualitative methods for data collection and analysis used in this research aim to be considered plausible, credible, trustworthy, and therefore defensible (J. R. Burke, 1997). Prior to the collection of qualitative data, the

following four elements were taken into consideration to determine the validity, reliability and trustworthiness of data: Sensitivity to context; commitment and rigour; coherence and transparency; and impact and importance (Yardley, 2008).

Sensitivity to context was demonstrated in a number of ways. Firstly, the participants were made aware that the researcher was an outdoor education teacher who had also completed 'City to Summit' with another group. This allowed the participants and researcher to easily relate to program experiences, whilst also allowing the researcher to be sensitive towards their comments and feelings.

Secondly, the researcher was acutely aware of the sensitivity to various contexts, and therefore accordingly facilitated the flow of communication during interviews. This facilitation style set a relaxed environment for the participants to respond to questions. Having participants feel as though they are being listened to may contribute to a therapeutic effect during interviews (Poggenpoel & Myburgh, 2003).

Lastly, the researcher ensured all participants understood the risks involved in the research and that they had the ability to leave the interviews at any stage without question. Participants understood that their personal interview responses would only be shared with the researchers and that their privacy would be protected by allocating pseudonyms and following confidentiality procedures. Counselling services were also offered to support participants if they deemed it necessary.

Commitment and rigour was determined through a number of ways. Initially, the researcher completed the 21 day 'City to Summit' program herself so that she could have a first-hand experience of what the participants experienced throughout the entire program, how they felt, what the environment was like and to experience the same challenges the students experienced from start to finish. She then kept a series of three observational diaries to correlate and cross-check if any themes arose. The descriptive diary recorded factual accounts of daily activities and the student's responses to the challenges and experiences, the observational diary demonstrated participants emotional responses, and the personal diary was used to monitor the researcher's personal responses to the data collection. Upon completion of the program, the three observational diaries were cross-checked in a cyclic analysis of the data to identify themes in Phase II and Phase III.

The researcher also showed commitment and rigour through conducting and rehearsing interviewing techniques before starting Phase III. Interview formats allowed for an authentic two-way communication, which then permitted in-depth descriptions and discussions within the small groups. In an effort to minimise influences on participants' comments during the interviews, the researcher remained neutral and tried not offer any personal opinions. The researcher demonstrated commitment and rigour by attending all interviews on time and allowing time for questions before and after the recordings.

Coherence and transparency were established by conducting interviews that were all digitally recorded and transcribed verbatim. The same semistructured interview guides were followed, with the same groups of participants during both interview sets. This ensured that the group was comfortable and allowed the researcher to build on questions and answers from the previous

159

interviews. Transparency was demonstrated clearly as interview data was crossvalidated by another two researchers and all results were made available for participants to access.

Impact and importance were demonstrated by comparing findings from interview data sets with both Phase I and Phase II. The survey data results were able to be put into context, further explained and understood through the individual responses recorded during the interviews. This provided further insights into the effects and impacts of participation in 'City to Summit' (Denzin & Lincoln, 1994, 2008, 2003; Fontana & Frey, 2003). In addition, collecting two sets of qualitative data not only demonstrated the impact and importance of Phase III but was also imperative to answering the research questions as it enabled a comparative data set of participant responses.

## **Chapter Summary**

This chapter outlined the pragmatist approach applied to this research. It detailed the mix of methods selected, the procedures and the data analysis used in the three data collection phases. A pragmatic approach, which included both postpositivist and constructivist paradigms, were used to guide the data collection and analysis in order to give meaning to the multidimensional and complex phenomena of an outdoor education program's effect on teenage boys' resilience and coping.

In Phase I, two groups of school students were surveyed to collect data about their self-reported levels resilience and coping attributes (N = 111). The program group (n = 69) participated in the extended journey style outdoor education program and answered both the Resilience Scale and the Brief COPE scale questionnaires. A control group (n = 42) who did not participate in any outdoor education programs also completed the survey measures. Both groups could be compared as they had similar demographics, as far as social, cultural and school environments. Data from Phase I was statistically analysed using SPSS version 22. The results of the descriptive statistics, frequencies, ANOVA, repeated measures t-tests, internal consistencies, EFA, and correlations are presented in Chapter 4.

Qualitative Data was collected during Phase II and Phase III and only involved the program group. The researcher collected program observation data throughout Phase II. She observed students in one group participating in the outdoor education program and kept observational recordings in three parts; (1) a descriptive diary; (2) an observation diary; and (3) a personal diary.

Phase III involved collecting data from two sets of small group semistructured interviews with a total of eighteen boys from seven different activity groups. This first set of interviews was conducted in the week post-program and followed up by a second set of interviews 6 months post-program with the same participants. Results from Phase I are presented in the next Chapter, and the results from Phase II and Phase III are presented in Chapter 5. The overall findings and the discussion of the relationships between the data and are presented in Chapter 6.

## **CHAPTER 4: QUANTITATIVE RESULTS- PHASE I**

A mind that is stretched by a new experience can never

go back to its old dimensions

-Oliver Wendell Holmes



Figure 4.1. Self-guided raft on the Mitta Mitta River.

#### Impact of an Outdoor Education Program on Resilience and Coping skills

The previous Chapter outlined the methodology, the premise behind this study, and presented the implementation and justification of this research using a mixed methods approach. This study used two psychometrically validated questionnaires, combined with small group interviews to assess any immediate and short-term changes in the participants' resilience and coping attributes after participation in a journey style outdoor education program. The approach provided a solid framework from which to examine the trends and themes that emerged from the mixed data collected, enabling a richer understanding of participants' actions, behaviours and experiences.

As outlined in the methodology, the mixed methods approach enabled empirical testing in Phase I to support the qualitative data collected in Phase II and Phase III. The qualitative data provided detailed accounts from the participants to complement the findings of the quantitative data presented in this chapter (Driscoll, Appiah-Yeboah, Salib, & Rupert, 2007). The results of Phase II and Phase III are presented in Chapter 5 and Chapter 6 provides a discussion on how the detailed annotations and theoretical underpinnings in Phase II and III support and interact with the quantitative findings from Phase I. This chapter details Phase I of the research that examines the data collected from the questionnaires and provides the results of the quantitative data sets of the research.

## Aims

This research aimed to identify the particular attributes of resilience and coping which were strengthened through participation in a journey style outdoor education program. Specifically, Phase I aimed to answer the first research question:

## **Research Question 1**

Which attributes of resilience and coping are enhanced through participation in an extended secondary school outdoor education program? This was specifically demonstrated through the examination of pre-and postprogram questionnaires of the Resilience Scale and the Brief COPE Scale.

### **Phase I: Quantitative Results**

Statistical analyses and results from the pre- and post-program questionnaire administration are presented in the following sections: (a) Resilience Scale descriptive and inferential statistics, (b) Resilience Scale internal consistencies, (c) Brief COPE descriptive and inferential statistics, (d) Brief COPE exploratory factor analysis, (e) Brief COPE internal consistencies, and (f) correlations. The chapter concludes with a discussion of the quantitative results.

#### **Resilience Scale Descriptive and Inferential Statistics**

Descriptive statistics for each dependent variable in the Resilience Scale are presented in Table 4.1. Data pertaining to the inferential outcomes associated with t-test comparisons are also detailed in Table 4.2. Descriptive statistics for each dependent variable in the Resilience Scale are also presented graphically in Table 4.1. These results include pre-test at Time 1 (T1) and post-test at Time 2 (T2) scores for both the program group and control group (N = 111). Results from the program group (n = 69) were compared to the control group (n = 42) at Pre-(Time 1) and Post-test (Time 2), which satisfied the minimum number of participants required for a large effect.

Resilience Scale means, standard deviations and t-tests

Resilience Attributes	Group	Time	1	Time	2	Repeated measures t-test
	-	М	SD	М	SD	Т(р)
Overall	PG	136.00	16.13	140.65	13.59	-3.16 (.002*)
Resilience	CG	135. 29	16.50	134.69	16.80	0.25 (.808)
Parsavaranca	PG	27.97	3.75	29.06	3.48	-2.82 (.006*)
Perseverance	CG	27.26	4.49	27.31	4.04	-0.07 (.942)
Self Reliance	PG	27.86	4.03	28.33	3.45	-1.31 (.193)
Sen-Renance	CG	26.07	3.86	26.40	4.08	-0.67 (.508)
Existential	PG	27.90	3.32	29.12	3.11	-3.18 (.003*)
Aloneness	CG	28.43	3.74	27.50	4.40	1.45 (.154)
Fauanimity	PG	25.54	4.32	26.01	3.15	-1.16 (.250)
Lquaininity	CG	25.69	4.89	25.60	3.65	0.13 (.900)
Purposeful Life	PG	26.74	4.26	28.13	3.46	-2.73 (.008*)
i urposetui Elle	CG	27.83	4.29	27.88	3.64	-0.08 (.936)

*Note.* \* = p < .05, PG: program group, CG: control group

All initial t-test scores reported probability values of p = >.05 in Levene's test for equality of variances and the assumption of homogeneity of variance was met. Therefore, repeated measures t-tests were performed with equal variances assumed and no significant differences at baseline. Repeated measures t-tests showed significant differences for Overall Resilience (p = .002), Perseverance (p = .006) Existential Aloneness (p = .003), and Purposeful Life (p = .008) as highlighted in Table 4.1. These follow-up t-tests also indicated that despite similar

levels of Overall Resilience at T1, the groups differed at T2 (p = .043) with the program group reporting greater resilience (M = 140.65, SD = 13.59). Existential Aloneness also had a significant interaction effect between time and group (p = .003). Differences at T2 (p = .026) were also found with the program group (M = 29.12, SD = 3.11) reporting higher Existential Aloneness than the control group (M = 27.50, SD = 4.40).

A mixed-design analysis of variance model (ANOVA) was then used to test for differences between the program group and control group, as well as to test for any differences over time. The ANOVA design consisted of a 'between subjects' (x 2 levels) and a 'within subjects' variable (x 2 levels), giving a total of 4 cells in the ANOVA design. A main effect was found for group and Self-Reliance, F(1, 109) = 7.31, p = .008. An interaction effect for both time and group was also found for Overall Resilience, F(1, 109) = 3.86, p = .043 and Existential Aloneness, F(1, 109) = 9.40, p = .003. All other ANOVA results for the Resilience Scale were non-significant (refer to Table 4.2).

Attribute	Main effect 1	Main effect 2	Time x Group
	Timo	Group	Internation
		Gloup	
	F(1, 109)	F(1, 109)	F(1, 109)
Overall Resilience	2.31	1.48	3.86*
Perseverance	2.55	3.36	2.14
Self-Reliance	1.80	7.31**	0.06
Existential Aloneness	0.17	0.80	9.40***
Equanimity	0.24	0.04	0.53
Purposeful Life	3.25	0.42	2.84

ANOVA results for the Resilience Scale

*Note.* \* p < .05, \*\* p < .01, \*\*\* p < .005

## **Resilience Scale Frequencies**

In all cases, the program group reported a higher percentage increase in scores for the student cohort with an increase in resilience attributes after participation in 'City to Summit' compared to the control group. The percentage of the students who had no change was calculated with a minimal change score by less than 3% in either direction. The program group (N = 69) showed 46.4% of students had an increase in Overall Resilience scores compared to the control group (N = 42) who had 33.3% of students with an increase. At T2, 40.6% of participants in the program group reported an increase in Perseverance score, while 28.6% of the control group reported an increase in Perseverance score. Purposeful Life showed an increase with 53.6% of students from the program group who reported

only 33.3% of students' increasing their score. Almost half the program group (47.8%) registered an increase in their Existential Aloneness scores, whereas the control group only has 23.8% of participants with an increase. See Table 4.3 for all percentage changes in the Resilience Scale.

#### Table 4.3

Attributes	Group	% change	% no	% change
		decrease.	change	increase
Overall Resilience	PG	17.4	36.2	46.4
	CG	35.7	31.0	33.3
Perseverance	PG	17.4	42.0	40.6
	CG	31.0	40.5	28.6
Self-Reliance	PG	26.1	36.2	37.7
	CG	21.4	57.1	21.4
Existential Aloneness	PG	24.6	27.5	47.8
	CG	40.5	37.5	23.8
Equanimity	PG	30.4	29.0	40.6
	CG	38.0	31.0	31.0
Purposeful Life	PG	23.2	23.2	53.6
	CG	28.6	38.1	33.3

Resilience Scale percentage scores for program and control groups

## **Resilience Scale Internal Consistencies**

The Resilience Scale Cronbach's Alpha value was  $\alpha = .88$ . Individual subscales scores were also calculated for the Resilience Scale which had 26 items in total. All five subscales included five items each and the 26<sup>th</sup> item measured Overall Resilience. Subscales with moderate coefficients were Purposeful Life ( $\alpha$ = .63), Self-Reliance ( $\alpha$  = .69) and Equanimity ( $\alpha$  = .61). Perseverance ( $\alpha$  = 0.70) was the only subscale that indicated strong coefficients, while Existential Aloneness ( $\alpha$  = .43) was the only subscale to measure a low to moderate coefficient between subscale items.

### **Brief COPE Descriptive and Inferential Statistics**

The means, standard deviations and repeated measures t-test for each subscale in the Brief COPE scale are presented in Table 4.4 and report both pre-(T1) and post-test (T2) scores for each group. Participants all showed varying results for the 14 coping subscales. Participants in the program group increased their mean scores at T2 for Active Coping (M = 3.06; 3.20, SD = 0.61; 0.54) and Planning (M = 2.98; 3.17, SD = 0.71; 0.55), and the control group also increased at T2 in the same subscales; Active Coping (M = 2.95; 3.17, SD = 0.75; 0.54) and Planning (M = 3.01; 3.07 SD = 0.71; 0.65). The program group decreased their scores for Substance Use (M = 1.30; 1.15, SD = 0.74; 0.45) and Behavioural Disengagement (M = 1.70; 1.55, SD = 0.69; 0.60) at T2, whereas the control group increased their scores in Substance Use (M = 1.49; 1.68, SD = 0.96; 0.97) and Behavioural Disengagement (M = 2.02; 2.15, SD = 0.91; 0.75) within the same time frame.

Attributes	Group	Pre-	Pre-test		-test	Repeated measures t-test
		М	SD	М	SD	<i>t(p)</i>
Self-distraction	PG	2.80	0.63	2.89	0.57	-0.95 (.348)
	CG	3.16	0.79	3.11	0.66	0.54 (.593)
Active coping	PG	3.06	0.61	3.20	0.54	-1.85 (.068)
	CG	2.95	0.75	3.17	0.54	-1.79 (.081)
Denial	PG	1.49	0.67	1.48	0.66	0.15 (.881)
	CG	2.10	1.02	2.15	0.95	-0.43 (.673)
Substance use	PG	1.30	0.74	1.15	0.45	1.69 (.096)
	CG	1.49	0.96	1.68	0.97	-1.56 (.128)
Use of Emotional	PG	2.44	0.80	2.62	0.76	-1.85 (.068)
Support	CG	2.50	0.83	2.46	0.89	0.34 (.734)
Behavioural	PG	1.70	0.69	1.55	0.60	1.54 (.128)
Disengagement	CG	2.02	0.91	2.15	0.75	-1.17 (.251)
Venting	PG	2.37	0.79	2.33	0.65	0.46 (.649)
	CG	2.64	1.03	2.69	0.90	-0.42 (.675)
Use of Instrumental Support	PG	2.75	0.80	2.81	0.75	-0.65 (.520)
	CG	2.60	0.88	2.65	0.78	-0.40 (.692)
Positive Reframing	PG	2.82	0.71	2.90	0.65	-0.88 (.381)
	CG	3.04	0.79	3.05	0.65	-0.22 (.831)
Planning	PG	2.98	0.71	3.17	0.55	-2.07 (.042*)
	CG	3.01	0.71	3.07	0.65	-0.58 (.566)
Self-blame	PG	2.57	0.70	2.59	0.69	-0.25 (.807)
	CG	2.63	0.86	2.55	0.77	0.58 (.567)
Humour	PG	2.84	0.91	2.74	0.93	0.97 (.338)
	CG	2.75	0.91	2.89	0.92	-1.11 (.274)
Religion	PG	1.70	0.93	1.69	0.89	0.15 (.879)
	CG	1.88	0.88	1.86	0.85	0.8 (.940)
Acceptance	PG	3.24	0.62	3.26	0.56	-0.73 (.728)
	CG	3.07	0.65	3.27	0.61	-2.26 (.030*)

# Brief COPE means, standard deviations and repeated measures t-tests

Note. \* = p < .05, PG: program group, CG: control group

Follow-up t-tests results (see Table 4.4) indicated that despite no significant differences between groups at T1, at T2 the program group reported coping less (p = .001) through Substance Use (M = 1.30; 1.15, SD = 0.74; 0.45) compared to the control group which increased their scores at T2 (M = 1.49; 1.68, SD = 0.96; 0.97). This trend was repeated for Behavioural Disengagement where a near significant interaction effect (p = .060) was caused by a decrease in Behavioural Disengagement at T2 for the program group (M = 1.70; 1.55, SD =0.69; 0.60) compared to the control group which increased at T2 (M = 2.02; 2.15; SD = 0.91; 0.75). Using repeated measures t-test for the Brief COPE Scale, the only significant difference for the program group to be in planning (p = .042) and the control group showed only one significant difference in acceptance (p = .030) which is highlighted in Table 4.4.

Repeated measures ANOVA presented in Table 4.5 was also conducted for all attributes of the Brief COPE Scale and revealed significant main effects for both time and group. There was a significant interaction effect between time and group for Substance Use (p = .026). Active coping (p = .01) showed a significant main effect for time. Several main effects for group were also exposed for selfdistraction (p = .01), Denial (p = .00), Substance Use (p = .01), Behavioural Disengagement (p = .00) and Venting (p = .03).

## ANOVA results for the Brief COPE scale

Attribute	Main effect 1	Main effect 2	Time x Group
	Time	Group	Interaction
	F(1, 102)	F(1, 102)	F(1, 102)
Self-distraction	0.05	6.45**	0.97
Active Coping	7.09**	0.80	0.45
Denial	0.08	19.96****	0.20
Substance Use	0.05	6.63**	5.12*
Use of Emotional Support	0.81	0.12	1.99
Behavioural Disengagement	0.03	15.16****	3.63
Venting	0.00	4.66*	0.37
Use of Instrumental Support	0.51	1.11	0.00
Positive Reframing	0.50	1.90	0.13
Planning	0.93	0.12	0.66
Self-blame	0.12	0.01	0.40
Humour	0.07	0.02	2.11
Religion	0.02	1.25	0.00
Acceptance	3.57	0.60	2.37

## **Brief COPE Frequencies**

Table 4.6 presents subscale percentages and reveals that both the program group and the control group presented various increases and decreases in subscale scores for the Brief COPE. As with the Resilience Scale, the percentage of the students who had no change was calculated with a minimal change score by less than 3% in either direction. The program group (N = 69) showed that 15.6% of students reported a decrease in Substance Use after participation in 'City to Summit'. Whereas, the control group (N = 42) had 24.3% of students reporting an increase in Substance Use and the program group only reported a 4.7% increase during the same time period. The percentage of scores which decreased for Behaviour Disengagement was relatively the same for the program group (39.4%) and the control group (36.6%), however, the percentage of participants in the control group (43.9%) reported a greater increase in Behavioural Disengagement compared to the program group (25.8%). Almost half of the program group (44.8%) registered a decrease in their Venting scores, whereas the control group only had a 30% decrease in Venting. The program group showed an increase of 42.4% in Planning skills, compared to the control group who reported a 32.4% of an increase.

Percentages for both Humour and Acceptance increased more in the control group. Humour scores in the program group decreased by 40.9%, while participants from the control group only reported a 35.9 % increase in using Humour at T2, compared to only 21.2% for the program group. Acceptance showed a higher increase for the control group (41.5%) compared to the program group (31.3%) after participation in 'City to Summit'.

Brief COPE percentage scores for program and control groups

Attributes	Group	%	%	%
	_	change	no	change
		decrease	change	increase
Self-distraction	PG	28.8	31.8	39.4
	CG	39.5	31.6	28.9
Active Coping	PG	23.9	38.8	37.3
	CG	22.5	40.0	37.5
Denial	PG	25.4	52.4	22.2
	CG	31.6	34.2	34.2
Substance Use	PG	15.6	79.7	4.7
	CG	10.8	64.9	24.3
Use of Emotional	PG	26.9	29.9	43.3
Support	CG	42.1	23.7	34.2
Behavioural	PG	39.4	34.8	25.8
Disengagement	CG	36.6	19.5	43.9
Venting	PG	44.8	22.4	32.8
	CG	30.0	40.0	30.0
Use of Instrumental	PG	35.8	25.4	38.8
Support	CG	38.5	28.2	33.3
Positive Reframing	PG	27.9	36.8	35.3
	CG	33.3	43.6	23.1
Planning	PG	25.8	31.8	42.4
	CG	32.4	31.5	32.4
Self-blame	PG	40.3	19.4	40.3
	CG	41.0	30.8	28.2
Humour	PG	40.9	37.9	21.2
	CG	25.6	38.5	35.9
Religion	PG	32.8	46.9	20.3
	CG	25.0	50.0	25.0
Acceptance	PG	32.8	35.8	31.3
	CG	17.1	41.5	41.5

#### **Brief COPE Exploratory Factor Analysis**

The original 14 subscales in the Brief COPE were designed to be analysed as separate subscales as (Carver, 1997) did not label them to be either adaptive or maladaptive coping skills. Therefore, an Exploratory Factor Analysis (EFA) was used to examine the relationships of each item and subscale in the questionnaire and to revise the existing factor structure. The EFA was based on T1 for both the program group and control group as the sample. Other research studies that implemented the Brief COPE as an instrument to measure coping have also designed alternative factor structures to evaluate the 14 subscales of the Brief COPE (Ewert & Yoshino, 2008; Hastings et al., 2005; Hussein, 2009; Kimemia et al., 2011). Previous research also supports and demonstrates the construct validity of the EFA (Duffy; Onwuegbuzie & Johnson, 2006).

#### Initial analysis of the Exploratory Factor Analysis.

Principal axis factoring in the EFA originally resulted in the extraction of an eight-factor structure for the 28 items, with factor loadings ranging between -.815 (Use of instrumental support) to .791 (Denial). Direct oblimin rotation converged in thirty-one iterations. Variables with loadings greater than .30 were used to interpret the factors. The EFA revealed that several of the factors did not load in direct alignment with the item structure of the measure, or as expected, therefore, logical analysis was applied to refine the EFA 8 factor structure to seven more coherent factors (Boros et al., 2000; Gorsuch, 1997). Refer to Table 4.7 for factor loading results of the EFA. Factor's 3, 6 and 7 all retained their original structure produced by the EFA. After further analysis, Acceptance and Positive Reframing were included in Factor 8 as it appeared to relate these items. Factor 2 originally contained Acceptance as a negative score (-.392). Acceptance also scored in Factor 4 (-.303) and Factor 8 (-.384).

Although Positive Reframing loaded on Factor 8 (-.805), it also loaded on Factor 5 (-.330). The researcher used the strongest weighted score from Positive Reframing and therefore was subsequently included in Factor 8. As Factor 5 showed a lower score (-.330), it was felt that this was acceptable Acceptance and Positive Reframing items did not fit logically with the other items in other factors so were included in Factor 8.

## Factor loadings for the Exploratory Factor Analysis with oblimin rotation for the Brief COPE

Brief COPE Items				EFA Facto	or Loadings			
	Factor 1	Factor 2	Factor 3	Factor 4	Factor 5	Factor 6	Factor 7	Factor 8
Self-Distraction. Item 1: I turn to work or other activities to take my mind off things.	.357							
Active Coping. Item 2: I concentrate my efforts on doing something about the situation I'm in.	.670							
Denial: Item 3: I say to myself "this situation isn't real."		.791						
Substance Use. Item 4: I use alcohol or other drugs to make myself feel better.							788	
Use of emotional support. Item 5: I get emotional support from others.			661					
Behavioural disengagement: Item 6: I just give up trying to deal with the situation.	539							
Active Coping, Item 7: I take action to try to make the situation better.	.627							
Denial. Item 8: I refuse to believe the situation has happened.		.687						
Venting: Item 9: I say things to let my unpleasant feelings escape.					.342			
Use of instrumental support. Item 10: I get help and advice from other people.			815					
Substance use. Item 11: I use alcohol or other drugs to help me get through it.							756	
Positive reframing. Item 12: I try to see things in a different light, to make it seem more positive.	.342							
Self-Blame. Item 13: I criticise myself.					.735			
Planning. Item 14: I try to come up with a strategy about what to do.	.603							311
Use of emotional support. Item 15: I get comfort and understanding from someone.			582					
Behavioural disengagement. Item 16: I give up the attempt to cope.		.425						
Positive Reframing. Item 17: I look for something good in what is happening.					330			805
Humour. Item 18: I make jokes about the situation.				935				
Self-Distraction. Item 19: I do something to think about it less, such as going to movies, watching TV, reading, daydreaming, sleeping, or shopping.		.406						
Acceptance. Item 20: I accept the reality of the fact that the situation has happened.		392		303				
Venting. Item 21: I express my negative feelings.					.583			
Religion. Item 22: I try to find comfort in my religion or spiritual beliefs.						.691		
Use of instrumental support. Item 23: I try to get advice or help from other people about what to do.			803					
Acceptance. Item 24: I learn to live with the situation.								384
Planning. Item 25: I think hard about what steps to take.	.488							
Self-Blame. Item 26: I blame myself for things that happened.					.419			
Religion. Item 27: I pray or meditate.						.783		
Humour. Item 28: I make fun of the situation.				815				

Note. Extraction Method: Principal Axis Factoring. Rotation Method: Oblimin with Kaiser Normalization. a. Rotation converged in 24 iterations. Variables with loadings greater than .30 were used to interpret the factors.

## Coping dimension categories.

As an outcome of logical analysis and results of the EFA, each of the subscales' in the Brief COPE has been categorised as an emotion-focused or problem-focused coping skills (see Table 4.8). This is in line with Lazarus (1984)'s interpretation that coping skills that suggests coping should not be labelled either as fundamentally 'good or bad', but rather as emotion-focused or problem-focused.

## Table 4.8

<b>-</b> · · · · ·		<u> </u>			0 1	
Emotion_tocused	and proble	pm_tocused	coning	dimensions	nt the l	Rriet ('APF'
Linonon jocuscu	unu provie	m jocuscu	coping	amensions	0] 1110 1	

Coping Dimension	Subscale and item number									
Emotion-focused	Self-distraction, items 1 and 19									
	Active Coping, items 2 and 7									
	Denial, items 3 and 8									
	Substance Use, items 4 and 11									
	Behavioural Disengagement, items 6 and 16									
	Venting, items 9 and 21									
	Humour, items 18 and 28									
	Religion, items 22 and 27									
	Self-blame, items 13 and 26									
Problem-focused	Use of Emotional Support, items 5 and 15									
	Use of Instrumental Support, items 10 and 23									
	Positive Reframing, items 12 and 17									
	Planning, items 14 and 25									
	Acceptance, items 20 and 24									

### Final analysis of the Exploratory Factor Analysis.

Further logical analysis inferred that both Factor 1 and 2 should be combined to allow for similar item loading scores within Factor 1 (Boros et al., 2000). Table 4.7 outlines the original factor loadings and Table 4.9 presents the new factor structure. All other items retained their original item pattern structure produced by the EFA. Following the combining of Factor 1 and 2, the EFA resulted in a seven-factor structure in total. Therefore, all factors were constructed and moved down one numerical place. See Table 4.9 for the new factor structure labels for the Brief COPE, including the items which comprise each of the seven factors.

## New factor structure, labels and coping dimensions of the Brief COPE

Factor	Factor	Items	Coping Dimension
Number	Label	Included	
Factor 1	Active Coping	Self-distraction, items 1 and 19 Active coping, items 2 and 7 Denial, items 3 and 8 Behavioural disengagement, items 6 and 16 Planning, items 14 and 25	Emotion-focused (apart from planning which is categorised as problem-focused skill)
Factor 2	Problem-focused	Use of emotional support, items 5 and 15 Use of instrumental support, items 10 and 23	Problem-focused
Factor 3	Humour	Humour, items 18 and 28	Emotion-focused
Factor 4	Emotion-focused	Venting, items 9 and 21 Self-blame, items 13 and 26	Emotion-focused
Factor 5	Religion	Religion, items 22 and 27	Emotion-focused
Factor 6	Substance Use	Substance use, items 4 and 11	Emotion-focused
Factor 7	Positive Coping	Positive Reframing, items 12 and 17 Acceptance, items 20 and 24	Problem-focused

### **Brief COPE Internal Consistencies**

The overall Cronbach's alpha for the Brief COPE Scale was ( $\alpha = .78$ ). Reliability was calculated for the EFA's seven factors of coping. Subscales with moderate coefficients were Active Coping ( $\alpha = .67$ ), Emotion-focused ( $\alpha = .66$ ) and Positive Coping ( $\alpha = .65$ ), while Problem-focused ( $\alpha = .83$ ), Humour ( $\alpha =$ .79), Religion ( $\alpha = .79$ ), and Substance Use ( $\alpha = .89$ ) all revealed strong coefficients.

## Correlations

Correlations between the five subscales of resilience in the Resilience Scale were calculated and presented in Table 4.10. As expected, all attributes of resilience had strong significant correlations to one another as well as to Overall Resilience. In terms of convergent validity, all five Resilience Scale attributes correlated significantly and positively with each other.

Correlations of the new factor structure for the Brief COPE identified through the EFA (Active Coping, Problem-focused, Humour, Emotion-focused, Religion, Substance Use and Positive Coping) were also calculated and presented in Table 4.10. Convergent and divergent validity was demonstrated through a range of correlations found between the factors identified in the Brief COPE EFA. Strong positive correlations were found between Emotion-focused and Active Coping (.51) factors. Three moderate positive correlations were found for Positive Coping with three other factors; Active Coping (.47), Problem-focused (.38) and Humour (.30). Active Coping and Problem-focused (.31) factors, also revealed a moderate positive correlation. Substance Use demonstrated a small positive correlation with both Emotion-focused (.27) and Religion (.26) coping factors. Some evidence of a relationship was found between Humour and Active Coping (.27), Emotion-focused and Problem-focused (.20), and Positive Coping and Emotion-focused (.27) factors which demonstrated small positive relationships.

Convergent and divergent validity between the subscales of the two measures was demonstrated through a range of correlations found between the factors identified. A moderate positive correlation was found between Positive Coping and Perseverance (.30). All other correlations found between the Brief COPE factors and the Resilience Scale subscales revealed small positive relationships, apart from the Emotion-focused factors, which showed a small negative correlation with Self Reliance (-.20) and Equanimity (-.27). These small negative relationships between the Brief COPE and the resilience subscales demonstrate the divergent validity.

# Correlations between the Resilience Scale subscales and EFA factors of the Brief COPE

Correlations														
		Perseverance	Self-reliance	Existential Aloneness	Equanimity	Purposeful Life	Resilience	Active Coping	Problem-focused	Humour	Emotion-focused	Religion	Substance Use	Positive Coping
Perseverance	=r	1												
Self-reliance	=r	.71**	1											
Existential Aloneness	=r	.53**	.56**	1										
Equanimity	=r	.61**	.53**	.41**	1									
Purposeful Life	=r	.60**	.55**	.51**	.55**	1								
Resilience	=r	.65**	.53**	.54**	.44**	.52**	1							
Active Coping	=r	.21*	.07	.17	.16	.10	.17	1						
Problem-focused	=r	.24*	.08	07	.15	.14	.12	.31**	1					
Humour	=r	.04	.12	.07	.05	.23*	.10	.27**	.15	1				
Emotion-focused	=r	18	20*	11	27**	20*	13	.51**	$.20^{*}$	.15	1			
Religion	=r	.01	04	.13	.06	03	.09	.20	04	01	.03	1		
Substance Use	=r	16	00	.04	04	.03	22*	.22*	.09	.14	.27**	.26**	1	
Positive Coping	=r	.30**	.23*	$.20^{*}$	.14	.26**	.34**	.47**	.38**	.30**	.27**	05	09	1

#### **Quantitative Discussion**

In response to Research Question 1, there were several resilience attributes and coping skills that were enhanced after participation in the extended secondary school outdoor education program, 'City to Summit'. In terms of Overall Resilience, an interaction effect was observed between time and group. The repeated measures t-tests revealed significant differences for Overall Resilience. The follow-up t-tests indicated that despite similar levels of Overall Resilience at T1, the groups differed at T2 where the program group reported greater resilience than the control group. These results indicate that participation in 'City to Summit' had a significant impact on the program group as their levels of Overall Resilience increased immediately after participation in the program.

These results are consistent with the findings of Booth's (2015) study who also reported a small, positive increases in participants levels of resilience (d = 0.20). However, while the average levels of resilience increased in participants at T1 (M = 3.42) compared to T2 (M = 3.56), participants initial scores indicated that they were already moderately resilient before the program. It is also important to note that this study did not use a comparison group to compare data.

In addition, these findings are also supported by Neill and Dias's (2001) study, which reported all participants showed positive changes in their resilience with a large effect size observed. Similarly, their study reported a significant interaction between time and group, where the experimental group reported a greater change in resilience compared to the control group. An interaction between time, group and gender was also reported which indicated that the changes in resilience were consistent across gender (Neill & Dias, 2001). However, the main difference was that this study did not use gender as a variable and was only focusing on males. In contrast, not all participants in the current study showed positive changes in their scores in Overall Resilience and the resilience attributes, but rather as a whole the program group increased in Overall Resilience scores compared to the control group.

In terms of separate resilience attributes, the results of the repeated measures t-tests revealed significant interaction effects between time and group in three out of the five resilience attributes: Perseverance, Existential Aloneness, and Purposeful Life. This was an anticipated result as the program challenges presented to the participants during 'City to Summit' required them to apply and practice these assets in order for them to manage the difficulties during the program.

The process of facing these challenges required individuals to have the determination to persist with the challenge despite adversities or disappointment. The challenges presented required the participants to practice and develop Perseverance, allowing them to continue to be engaged in the challenge through showing self-discipline and determination, despite being discouraged by challenging situations that were presented to them (Wagnild, 2009, 2010; Wagnild & Young, 1993). The increases in Perseverance scores in this study are also consistent with the findings of Ewert and Yoshino's (2011) study, that also

185

indicate short-term expeditions may enhance participants' levels of Perseverance and Overall Resilience.

It was an anticipated result that the program group showed increases in scores for Purposeful Life, as the experiential learning processes experienced during the program allowed opportunities for solo time, reflection, journal writing and critical questioning. This allowed opportunities for participants to further understand their purpose in life, their life goals and how these can be achieved in challenging times and adverse conditions (Wagnild, 2009, 2010; Wagnild & Young, 1993).

The follow-up t-tests showed differences after 'City to Summit' with the program group reporting higher Existential Aloneness than the control group. These results may have been impacted upon by the program group participating in an overnight 24-hour solo experience during 'City to Summit'. As Wagnild (2010) describes, Existential Aloneness is about knowing one's self and being 'comfortable in your own skin'. The solo experiences during 'City to Summit' gave the program group an opportunity to be alone and reflect that it is them who makes your choices and life decisions (Wagnild, 2009, 2010; Wagnild & Young, 1993). Even though experiences may be shared with others, some experiences in life must be faced alone.

A comparable program that used challenging expedition type experiences to enhance levels of resilience was detailed in Gillespie and Allen-Craig's (2009) study involving at-risk male adolescents. Similar to (Neill & Dias, 2001), their investigation also used the 15-item Resilience Scale to measure changes in

186

resilience scores after participation in a 5-week wilderness therapy program. Gillespie and Allen-Craig (2009) reported a moderate to large positive effect size of resilience scores. Similar to the results of this study which indicate that extended journey style outdoor education programs have a positive effect on levels of resilience, their results also indicate that participation in extended wilderness therapy programs can help youth at risk increase their resilience.

It is important to note that there is also contradicting evidence which indicates that not all studies investigating the impacts of outdoor education programs on levels of resilience and coping have yielded positive effects. Skehill's (2001) similar study using the 15-item version of Resilience Scale to measure resilience levels in Year 9 adolescents during an Extended Stay Outdoor Education Program found no effects on resilience, nor on participants' levels of well-being or distress.

Comparable to this study, the findings of the literature review presented in the previous chapter, reveal that Yoshino (2008) and Booth's (2015) studies are the only two of its kind that used have used both the Resilience Scale (Wagnild & Young, 1993) and a similar coping measure concurrently in their research to measure the impacts of outdoor education programs as the intervention to foster resilience and coping with young people. In addition to the Resilience Scale, Booth (2015) used a modified version of the original COPE Inventory (Carver et al., 1989), whereas Yoshino (2008) used a modified version Brief COPE Scale (C. L. Park & Blumberg, 2002), and this study used the original version of the Brief COPE Scale (Carver, 1997). However, in addition to these two measures, Yoshino's (2003) study also used the Stressor Inventory (Robinson & Stevens, 1990), Stress Appraisal Measurement (Peacock & Wong, 1990), perceived success (Skehill, 2001), and Posttraumatic Growth Inventory (Tedeschi & Calhoun, 1996) to measure how a hypothetical model explained the impacts and influences of stress appraisal, coping responses, and perceived success of participants levels of psychological growth.

The results from the repeated measures ANOVA derived from the students' completion of the Brief COPE Scale revealed significant main effects for both time and group. Active coping showed a significant main effect for time, indicating that participation in 'City to Summit' may have influenced the development and application of Active Coping skills in order for the program group to manage the difficulties presented to them during the program. Several main effects for group were also exposed for Self-Distraction, Denial, Substance Use, Behavioural Disengagement and Venting. However, only two coping skills changed over time and between groups; Substance Use and Behavioural Disengagement.

There was a significant interaction between time and group for Substance Use. Follow-up t-tests indicated that despite no differences between groups at T1, at T2 the program group reported coping less through substance use compared to the control group. This may have been influenced by the difficulty for participants in the program group to access substances to use between T1 and T2 as they were immersed in a school controlled program in a wilderness setting. It is also important to note that while the program group was participating in 'City to
Summit', the control group was being administered their end of year exams. This may have put further stress on the control group which could have impacted the coping skills they used during this period (Misra et al., 2003).

This trend was repeated for Behavioural Disengagement. Although the interaction was not significant, the pattern of descriptive scores inferred a decrease in Behavioural Disengagement at T2 for the program group compared to the control group which increased at T2. The decrease may have been achieved due to the program group participating in new, exciting, challenging and adventurous experiences during 'City to Summit'. While the increase in Behavioural Disengagement with the control group may have been negatively influenced by the distraction of requirements of their end of year exams.

The only significant difference for the Brief COPE Scale was disclosed in by the repeated measures t-test which revealed that the program group showed a significant difference in Planning and the control group showed only one significant difference in Acceptance. Interestingly, the whole program group increased their scores in Active Coping, and Planning, and decreased their scores in two emotion-focused coping skills; Substance Use, and Behavioural Disengagement. This may have been influenced by the program group participating in daily skills, tasks and challenging situations requiring them to develop and practice skills such as planning and activity coping to help them to manage difficulties consistently over the course of three weeks. It is also important to note that while it is suggested by Carver (1997) not to label the coping skills measured in the Brief COPE scale as either maladaptive or adaptive, both Substance Use, and Behavioural Disengagement have the potential to undermine well-being, and Active Coping, and Planning can be viewed as positive effective coping skills which have the potential to enhance well-being Lazarus and Folkman (1984).

Consistent with these results, Booth (2015) also reported the application of problem-focused coping skills, such as Active Coping (M = 2.66; SD =.57) and Planning (M = 2.54; SD =.66) to be in the top 5 coping strategies used by participants. A reduction in scores of two emotion-focused coping strategies (acceptance and venting of emotions during the program) was revealed and was found to be associated with a 13% positive change in psychological resilience. These findings support the results of this study that also showed a significant reduction of two emotion-focused coping skills; Substance Use and Behavioural Disengagement for time and group. However, while the boys in the program group reported coping less (p = .001) through Substance Use (M = 1.30; 1.15, SD = 0.74; 0.45), this reduction may have been influenced by the lack of accessibility to substances available to them when they are immersed in a wilderness setting.

With regards to the control group, it is important to point out that the boys within the control group were sitting their school exams at the time the quantitative data was collected. Results showed the control group reported a 33.3% increase, 35.7% decrease and 31% showing no change in resilience scores at T2, indicating that some individuals were able to adapt to the stressors, while others found it more difficult. The variance in these results may have been contributed to by increased academic pressures which could further perpetuate

levels of stress due to self-expectations and standards of performance deemed necessary to succeed; testing their resilience capacity and requiring them to apply various coping strategies in responses to the stressors.

This investigation used EFA to reveal seven factors within the Brief COPE. While three factors in this EFA retained their original structure, there were four factors still requiring further logical analysis. This may have been due to several different influences, including the students' age (14-17 years) and the variance in how the participants comprehended the detail of 'items' within the Brief COPE, such as 'Item 20'. In comparison to other studies using the Brief COPE, Yoshino (2008) used a confirmatory factor analysis (CFA) with Principal Axis Factoring within a modified version of the scale. Compared to the seven factor solution revealed in this study, Yoshino's study resulted in four factors named 'social support', 'problem-focused', 'self-distractive', and 'lessconstructive', and accounted for 46% of the variance in the sample group.

In terms of correlations, it was expected and reported that all attributes of resilience had strong relationships with one another as well as with Overall Resilience. These results are also indicative of convergent validity in that all five Resilience Scale attributes correlated significantly and positively with each other. These strong relationships may be related to the participants from the program group demonstrating various other internal assets that resilient people have been identified to acquire, such as self-esteem (Dumont & Provost, 1999), self- efficacy (Bandura, 1977), self-acceptance, adaptability, responsibility or their ability to regulate their emotions (Miller, 2016). The researcher also predicted strong positive relationships between the Emotion-focused and Active Coping factors. These strong correlations may have been impacted by the fact that all the items in the two factors consisted of strategies from the emotion-focused coping dimension (see Table 4.9). This also may have been influenced by participants demonstrating other emotion-focused coping strategies such as humour, demonstrating empathy or having faith (Miller, 2016).

Conversely, instead of consisting of strategies from the emotion-focused coping dimension, all the items in the Positive Coping and Problem-focused factors consisted of strategies from the problem-focused coping dimension (refer to Table 4.9). These two factors revealed a moderate positive relationship which may have been affected by participants demonstrating and developing other problem-focused coping skills such as their ability to regulate their behaviours, solve problems, apply communication skills and show self-awareness (Bandura, 1986; Benard, 2004; Masten, 2001; Miller, 2016). Problem-focused coping skills are seen to be more effective than emotion-focused coping for long-term stressors or challenges that exist over an extended time frame, such as managing emotional stressors of being away from home for three weeks for the first time. This is because problem-focused strategies generally address the causes of challenge or stress in practical ways and aim to deal with the challenge and reduce the stress (Breinbauer & Maddaleno, 2005).

Positive Coping also revealed moderate positive relationships with two other factors; Active Coping and Humour. These factors primarily involve

192

strategies from the emotion-focused coping dimension (see Table 4.9) and may have been influenced by the development and application of other internal assets such as confidence, optimism and hope (Benard, 1991, 1995, 2004).

Positive Coping and Perseverance revealed a moderate positive correlation and these factors conversely involved strategies from the problem-focused coping dimension (see Table 4.9). This may have been influenced by participants needing to apply a broad range of other practical, problem-focused strategies such as problem-solving or time management skills to manage challenges. This moderate relationship may have also been affected by participants' in the program group being required to apply other resilient characteristics such as determination or persistence (Miller, 2016).

In terms of the Brief COPE subscales and the Resilience Scale attributes, there were small negative relationships demonstrating the divergent validity. This was an expected divergent relationship as the Emotion-focused factor includes items from the Venting and Self-Blame subscales, which would be predicted to negatively correlate with items such as Self-reliance and Equanimity as these are factors which increase a participant's likelihood to demonstrate resilience.

### **Chapter Summary**

This chapter presented the quantitative results from Phase I and addressed the research question investigating which attributes of resilience and coping were enhanced through participation in an extended secondary school outdoor education program. Based on the results provided in this chapter, it was

193

determined that Overall Resilience increased in the program group immediately after participation in 'City to Summit', compared with the control group who did not report any significant changes. Existential Aloneness, Perseverance and Purposeful Life were the three resilience attributes that showed significant increases for the program group after participation in 'City to Summit'. The program group showed reductions in coping skills that have the potential to undermine well-being in youth, such as Substance Use and Behavioural Disengagement, while increasing in problem-focused coping skills, such as Active Coping, and Planning after participation in 'City to Summit'.

The findings from Phase I reinforce that participation in an extended journey style outdoor education program does in fact increase scores in Overall Resilience. These findings are also supported by previous research which has shown that resilience and coping can be enhanced through participation in outdoor education programs (Gillespie & Allen-Craig, 2009; Neill & Dias, 2001). However, existing research is yet to identify if secondary school students draw upon the enhanced resilience attributes and coping strategies achieved through participation in outdoor education programs in other contexts of their lives, or if these attributes are context specific, and only drawn upon in an outdoor setting. These questions will be addressed in the next Chapter which presents the results of Phase II and Phase III and provides a discussion of the interrelationships of the main themes. Chapter 6 will present the final discussion linking the quantitative and qualitative results.

### CHAPTER 5: QUALITATIVE RESULTS- PHASE II & PHASE III

A journey of a thousand miles begins with one step

– Lao Tzu



*Figure 5.1.* The last day of the journey.

As noted in Chapter 4, this study employs a mixed methods approach, using both quantitative and qualitative methods. The previous chapter summarised the results from Phase I's quantitative section of the study. This chapter presents Phase II and Phase III's qualitative results from the analysis of the participant interviews and the field observations, using the active voice of the participants to support the analysis. Images of the observation group's experience are dispersed throughout this chapter to give context of the program, the group, and the natural environment. The images also support the participants comments about their program experiences. The next Chapter discusses the insights and relationships between the quantitative data presented in the previous Chapter and qualitative themes outlined in this chapter.

### Phase II and Phase III: Qualitative Data Results

Semi-structured interviews were chosen for this research as it allowed the researcher to 'tap into the experience' using the participants' normal language (Guba & Lincoln, 1988c). The interviews also enabled exploration of the participant's perspectives and provided a first-hand understanding of their application of resilience attributes and coping skills to the program challenges (Ungar, 2003). These findings also provide insights about the changes identified through the quantitative data (Denzin & Lincoln, 1994; Guba & Lincoln, 1988a, 1988c; Windle et al., 2011).

The major themes, concepts and findings of Phases II and III are discussed and analysed in consideration with the following three philosophical principles:

- Our interactions with the environment can impact our construction of knowledge and influence our personal awareness or understandings;
- 2. That cognitive conflict or puzzlement is the stimulus for learning and determines the organisation and nature of what is learned and;
- That knowledge evolves through social negotiation and through the evaluation of the viability of individual understandings (Savery & Duffy, 2001).

Data from one observation group and four interview groups were analysed in the qualitative phase of this research. Table 5.1 provides the pseudonyms for the thirteen students from Group 9 who were observed by the researcher in Phase II.

Table 5.1

Group 9 participant pseudonyms

Observation Group Pseudonyms				
Syed	Bob	Tan		
Scott	Ahmed	Joshua		
Conrad	Isaac	Cooper		
Jack	Dale	Noah		
Ian	Group leader	Researcher		

Four small groups of students participated in the semi-structured interviewed and are labelled as Group A, B, C or D. The eighteen students interviewed came from seven of the 17 activity groups that completed the program (see Table 5.2). These groups were interviewed twice. The first set of interviews were conducted the week after completion of 'City to Summit' and the second set of interviews were conducted six months later (see Appendix P for the Interview Schedule).

## Table 5.2

Participant pseudonyms: Interview and activity group breakdowns

Small Group	Activity	Interview
Interview	Activity	Crown
Pseudonyms	Group	Group
Brodie	10	А
Matthew	10	А
Elliot	10	А
James	11	В
Tristan	11	В
Reece	11	В
Robert	11	В
Greg	12	В
Zac	13	С
Chris	13	С
Jake	13	С
Lachlan	14	С
Oliver	14	С
Mark	15	D
Frank	15	D
Kevin	16	D
Jason	16	D
Cody	16	D

Three cycles of qualitative data analysis were conducted to identify themes relating to the student's program experience, their resilience capacity and coping skills. Nvivo computer analysis program was used to explore the data with the first cycle revealing 54 parent codes and 33 child codes (see Appendix S). During the second cycle, key relationships were analysed, and all parent and child codes were broken down into 50 first-order themes and seven second-order themes (see Appendix T). The final cycle of the analysis of the data identified five major themes (see Table 5.3).

Table 5.3

Theme		Overview	
1.	Challenge	Common events, activities or situations that the students perceived to be difficult.	
2.	Response to challenge	This theme is associated with the participants responses, and their application of resilience attributes, problem-focused coping skills and	
3.	Personal development	emotion-focused coping skills to the challenges they identified during 'City to Summit'. The personal development theme focusses the on the participants' understandings of what they learnt, the self-reported increases of internal developmental assets and how they personally changed during the program. The theme also	
4.	Relationships	uncovers the participants heightened awareness and appreciation of one's circumstances, privileges and relationships with family. Relationships theme is about the development of external developmental assets, such as, the participants connection with their peers, their group, the teachers on the program, the group leaders and the natural environment	
5.	Transfer of learning	This theme relates to the transfer of learning and the student's insights six months post-program. It involves the students recognising if learning identified from the program was maintained and if they were able to apply these skills in other contexts of their lives six months after completing 'City to Summit'	

Results and overview of the five major qualitative themes

The presentation of the data in this chapter is divided into two sections. Firstly, the five major themes are presented which focus on what occurred during and after the program. This section provides detailed descriptions of each theme and specifies the results identified from the data analysis in Phase II and Phase III. The second section expands the themes and discusses the interrelationships between the themes. Lastly, a chapter summary is presented which provides an overall compilation of the significant qualitative results.

### Theme 1: Challenge

As described, the 'City to Summit' program was included in Yarra School's Year 10 curriculum as a platform to provide students with personal challenges. The program design presented targeted activities over the course of the journey for the boys to encounter. These challenging activities ranged from adventure activities that were used for modes of transport to move the groups to different campsites, such as hiking, mountain biking and rafting; to everyday challenges that the wilderness environment presented, such as setting up camp each day or dealing with adverse weather conditions.

The school's objectives of presenting challenges throughout the program appears to have provided the boys with opportunities for personal growth. It was observed that exposure to program challenges assisted the boys in strengthening their developmental assets, such as resilience and independence. Not surprisingly, each person engaged with these challenging situations differently. Some students found the level of the program challenges to be more difficult, while others found the same program challenges to be quite manageable. The program challenges varied in type, length and level of difficulty which required the boys to draw upon on different aspects of their personal attributes and resources. The types of challenges have been identified and collated into three key areas; physical, emotional and social challenges.

Table 5.4 presents a summary of the three types of challenge areas identified. It is important to note that although the challenges can be classified into three distinct types, the program challenges influenced participants more holistically than just drawing on a specific attribute or type of challenge. For example, the challenge presented may be physically demanding, but it may also require the participant to draw on their emotional capacities or social resources to manage the difficulties presented. The following sections describe and provide supporting evidence of the physical, emotional and social challenges that the program presented to the participants.

# Table 5.4

# Types of challenges, definitions and participant comments

Challenge Type	Description of Challenge Type	Examples of Participant's Comments
Physical challenges	Physical challenges represented activities that drew on qualities associated with a person's physical capacities, including strength, stamina and endurance. These challenges were connected with a	Conrad: I feel weak and I have no energy (Descriptive Diary: Day 3). Ian: Everything was hurting on the hike. My
	person's body, rather than with their mind. Physical challenges were tangible or concrete goals such as getting to the top of a mountain or riding a bike for 8 hours to get to the next camp.	hips, my shoulders and especially my feet (Descriptive Diary: Day 3).
Emotional challenges	Emotional challenges were identified by situations where participants felt or outwardly demonstrated emotional responses to the program challenge, such as crying, anger or frustration. Participants identified emotional challenges to be mental challenges that required cognitive capacities such as acts of thinking, discussing approaches, and wrestling with ideas rather than physical action. Emotional challenges required participants to apply mental processes to manage their emotional responses to the challenge.	Reece: I think it was like six hours of just flat water [when rafting in inflatable kayaks in pairs] and I got really frustrated and bored. And I got a bit angry and decided, huh, I'm having a bit of a, what's the word? Greg: Sook. Reece: Yeah. A bit of a sook. A bit of a tantrum I don't even know why I got so angry (Group B: Interview 1).
Social challenges	Living in close proximity with a small group of peers in an environment different to their urban lifestyles provided social challenges. These included managing the dynamics of the group, dealing with conflict and situations requiring tolerance.	Lachlan: Well, the thing I found most challenging was being around the same amount, like the same group of people of a long period of time (Group C: Interview 1).

### **Physical Challenges**

Participants identified physical challenges as something that impacted on their bodies and caused some sort of pain or feeling physical discomfort. The boys agreed that certain experiences were more physically demanding than others. During the mountain biking, students were observed to be physically struggling the most (Observation Diary: Day 10). This observation was consistent with comments from all eighteen boys interviewed, revealing that the mountain bike riding section was perceived to be the most physically demanding experience of the program. This was not a surprise as the mountain biking section consisted of cycling over 45km predominately uphill. Figure *5.2* shows the boys in Group 9 preparing to cycle the 45km uphill section of the mountain biking leg. Matthew's comment summarised how the boys responded to the bike riding:

The most physically challenging thing would have to be the bike riding. Um, I was sorta in the same position that Brodie was. I just really struggled, and then I sorta kept telling myself just push it a little bit, push it a little bit and then yeah, I finally got up the top (Group A: Interview 1).

The researcher observed most participants in Group 9 struggling, being out of breath and complaining of muscle pain in their legs. Many of these boys gave up and walked up the steep hills as they found the challenge of riding too difficult and the pain unbearable (Observation Diary: Day 9 & 10). For instance, Joshua comments that: I remember the pain in my legs when I was mountain biking. Supposedly you can't remember pain, but I'm telling you, you can! (Descriptive Diary: Day 15).





The researcher also observed students physically struggling during some of the hiking sections, especially the steep uphills and downhills (Observation Diary: Day 13). In the 6km uphill hiking section to the top of 'Mount Misery', most students commented that this section was physically demanding. For example:

Zac: To be honest the...what I found difficult, like in a physical sense, was that one day where it was just 6km nonstop uphill and like no one was talking cause it was literally like we were really tired... Researcher: Was that up Mount Misery?

Zac: Yeah. And yeah, everyone wasn't talking, and everyone sort of just got really depressed (Group C: Interview 1).



Figure 5.3. Students resting after hiking to the top of Mount Misery.

While hiking participants were observed complaining about pain in their shoulders, feet and legs (see Figure 5.3). Most students commented about feeling constant pain in their backs from carrying heavy backpacks (Observation Diary: Day 13). However, whilst the mountain biking and steep hiking sections were perceived as the most physically challenging experiences of 'City to Summit', these same experiences also created emotional and social challenges. These challenges are discussed in the following sections.

### **Emotional Challenges**

Most of the time, emotional challenges occurred when either the participants found that a challenge was physically demanding (e.g., when they were feeling pain), when they could not control the situation (e.g., they could not just leave the program or control what another person was doing) or if the mental challenge of managing their discomfort was too much for their mind to handle (e.g., fear responses or continual negative self-talk). When participants spoke about emotional challenges, they related this to anything coming from the mind.

Both emotional and mental challenges required the participants to cognitively manage internal emotional responses or external responses that others could see or identify. Many students outwardly displayed emotional responses such as frustration and anger when confronted with a challenging situation. For example, Conrad explained *"yeah, I was pretty angry. I was frustrated with the leaders as we had missed two campsites and we were only supposed to walk two kilometres. And I was frustrated with myself because I kept walking and I didn't care about the group. I really didn't deal with it" (Descriptive Diary: Day 18). Interestingly, 17 of the eighteen participants interviewed identified the whole experience as being more mentally demanding than physically demanding. For instance, Chris stated:* 

Well, it was a good experience. Something I probably won't do ever again at the same level. But it was definitely mentally challenging more than physically (Group B: Interview 1).

As the above comment indicates, the boys seemed to use the term mentally challenging when they were referring to both emotional and cognitive responses to the challenges presented. For instance, even though Kevin describes the program to be more of a mental challenge, his comment below indicates that it wasn't that the program challenges were cognitively demanding for him, but rather that the program challenges elicited emotional responses, such as anxiety, fear and frustration.

Kevin: It wasn't actually that physically challenging if you think about it. It was like more a mental challenge.

Researcher: What sort of things did you find mentally challenging?

Kevin: I reckon just the time of being away from everything...I remember saying to myself at the start 'you have three days down, there are however more many to go'. And just the monotony of it, if you don't focus on what's actually going on. That was my main challenge (Group D: Interview 2).

The length of 'City to Summit' and time away from home was a common emotionally challenging aspect for most students observed and interviewed. Not only were participants immersed in an unfamiliar environment, but they were also surrounded by a group of people they did not know very well. In each group, some students knew each other well, however, the Yarra College teachers handpicked the groups to ensure groups consisted of a mix of students who did and did not know one another. Many of the participants had experienced being away from their home environment for long periods, such as when holidaying with their families, but most participants had not spent this length of time away from their families. Sixteen out of the eighteen boys commented on how the length of 'City to Summit' challenged them throughout their experience. As Ian put it, This whole camp is challenging to be honest. Everything about it is challenging. Being away for this long is the longest I've ever been away. The longest I've been away is like five days. I also hate camps, so everything about this experience is challenging (Descriptive Diary: Day 14).

Occasionally these emotional challenges would elicit a physiological response such as feeling, crying or not understanding why their bodies, feelings or emotions were responding in certain ways. Certain challenges and situations were new experiences and participants didn't know how they were going to respond. For instance, when Ian commented, *"I'm not feeling well today. I feel homesick. I'm really missing my mum and just being indoors"* (Descriptive Diary: Day 11).

At times participants demonstrated that emotions got the better of them and they found it difficult to control their responses. This is illustrated by Scott's apology below, following an emotional outburst at camp. Scott was nominated as one of the leaders for the last day of mountain biking. He was observed yelling and swearing in frustration at four boys who were laying around camp and not contributing to the daily group tasks when the group arrived at camp (Observation Diary: Day 10):

Scott: I want to apologise for the way I behaved this afternoon. I was just really tired and emotional. Then I just snapped. I'm sorry. I'm not normally like that. That's not me. I'm really sorry (Descriptive Diary: Day 10). Interestingly, this situation occurred at the end of the mountain biking section day, which was identified by most participants as the most physically challenging activity of 'City to Summit' program. This example provides evidence that some of the program challenges were more demanding experiences than the participants have encountered in the past, thus stimulating strong emotional responses.

The 24-hour solo activity proved to be an emotional challenge for most boys. When a group leader asked their group how comfortable they felt about going out into the bush and spending the night alone during the 24-hour solo experience, the boys replied with a range of responses with over two thirds of the group commenting that they were feeling a bit anxious to very anxious (Observation Diary: Day 11). Before the experience, the boys appeared to be very apprehensive about the challenge, as most students had never spent that amount of time alone and especially not in a wilderness setting (Observation Diary: Day 11). The issues that appeared to scare the boys the most was being alone in an isolated, unknown environment, not being able to ask someone else for advice and having to make decisions on their own. As Chris explained, one of his greatest difficulties was having to rely on his own judgements and decisions:

I found it [the solo] pretty hard, and there were times that you realised you really needed someone for advice (Group B: Interview 1).

The boys seemed most worried about how they were going to manage through a whole 24-hours making their own decisions and completely relying on

209

their own abilities to support themselves. The solo experience was a novel situation for participants, as normally the boys would have access to ask peers, family or teachers for advice, but in this case, they were forced to make decisions on their own.

Whilst Zac felt that the solo was a confronting experience that made him very nervous, he also saw the value of solo time in developing his independence and self-reliance:

Personally, I was kind of nervous... Well, I didn't realise you were completely isolated, so obviously I was a bit nervous [about the solo]...But I think 'City to Summit' was kind of character building for me. It was the main thing we were there for... Kind of getting a new experience of what it's like being by yourself for that long...Uh, personally, I think it just made me independent.... Having to not like always get things your way (Group C: Interview 1).

Many of the boys commented that the solo was a mentally challenging experience (Observation Diary: Day 12). However, their comments seemed to refer to the emotional challenges that the experience provided, such as the fear, anxiety and self-doubt. Some students felt increases in levels of anxiety as they were also worried about being alone with 'wild animals'. Such as Cody who had a fear of animals and snakes: I think everyone was a bit scared, but I was really scared cause there were wild animals all around... Solo time was really hard for me because I'm shit scared of snakes. I saw four snakes. That was really scary, and you know, I was away from everyone else. I usually like to be close to people instead of um away... (Group D: interview 2).

Cody's comment indicates that his major fears were focused on worrying about how he would deal with a situation if he was confronted with wild animals or if something was to go wrong. An example of the wild animals Cody is referring to is demonstrated in Figure 5.4 which shows the wild brumbies that were found around campsites in the high plains. These Brumbies were very protective of their areas and had previously stampeded through Group 9's campsite.



Figure 5.4. Wild Brumbies in the high plains.

#### **Social Challenges**

Outdoor education programs provide a perfect platform for a social development (Ewert & McAvoy, 2000; McKenzie, 2000; Sibthorp et al., 2007). Placing a small group of Year 10 boys in an unknown environment, living together in close proximity with people they don't know well, whilst undertaking new and challenging experiences, provided groups with many difficulties to manage. Each of the 17 activity groups was made up of 12-14 boys, which created their social group for their 'City to Summit' experience. The mix of the boy's personal backgrounds, personalities and past experiences influenced how the group dynamics developed, how they dealt with conflict and responded as a group to the program challenges presented. Group dynamics were also influenced by the individual's responses to the challenges presented to them (Observation Diary: Day 10, 15, 20).

A common social challenge identified by all interview groups was the challenge of having to live and work together in the same small group for three weeks. Living and working together in small groups, formed the foundation of 'City to Summit' and required the boys to develop trust with one another over the course of the program (see Figure 5.5). Although 'Day 0' involved the groups participating in initiative and teamwork activities, that were designed to commence building trust amongst the group, it was clear that groups would take time to develop their capacity to trust one another (Observation Diary: Day 1). This was evident early in the program, as the boys struggled to trust one another

with certain tasks and activities such as navigation and setting up camp (Observation Diary: Day 1).



*Figure 5.5.* Group 9 undertaking teamwork and initiative activities designed to build trust.

In the comment below, Greg indicates how he struggled to trust the group at the start of the program. However, as the program continued, he found that the more the group shared new experiences together, the more his levels of trust increased:

One big one [issue] for me was learning to trust people a lot more. Like um at the start, I didn't trust people with jobs to do. Like, mainly cleaning... Yeah and by the end [of 'City to Summit'] I think I was a bit more lax about, you know, letting people um clean and stuff. I think we all learned how to stay in the routines set, like washing your hands before cooking and all of that and I learned to trust everyone a bit more (Group B: Interview 2).

Social challenges became apparent when group trust was required but not displayed, such as when the group needed to trust the boy who had the role of being the navigator to direct the group to the correct path during hiking sections. This was apparent in the first week when boys displayed signs of distrust in each other, leading to group conflict (Observation Diary: Day 4). This was seen on the first day of the program when Syed had the lead role of navigating and giving the group hiking directions up a steep hill from Windy Corner to the Aqua Duct in the Victorian high country. The group's behaviour made it obvious that the group did not trust Syed as they were yelling at him and questioning his ability to make the right decision (Observation Diary: Day 1). The group questioned his navigation skills, yet no one else was willing to share the responsibility of navigating or take on a leadership role. Some of the participants in Group 9 would say things like, *"Syed, if we are walking up this hill for no reason, I'm going to kill you. If we get lost, it's your fault"* (Descriptive Diary: Day 1).

Syed tried showing the group where they were on the map, and he tried to explain what the group needed to do, but nobody wanted to follow him. The group became completely separated, with some of the group running to the top of the hill, while others were walking at a 'snail's pace'. The entire group was complaining, yelling at one another, looking for other people to blame, and no one else was taking ownership or responsibility for the situation. It wasn't until the group reached the top of the hill and made it to the correct track junction, that everyone stopped complaining (Observation Diary: Day 1). After Syed had successfully guided the group on the right path, the group started to trust him. Figure *5.6* shows the group feeling very happy with Syed for navigating their group to the correct track junction at the top of the hill. Over the next few days, Syed was asked by the group to help them with their navigation skills when it was their turn to lead the group. Nobody in Group 9 questioned Syed about his navigation ability again (Observation Diary: Day 5).



Figure 5.6. Day 1: Arriving at the first track junction.

Group B also struggled with navigation issues, which caused social challenges amongst the group. Their comments indicate how factors such as physical challenges of walking uphill, the strength of peer relationships and the level of trust towards one another can influence the level of difficulty a challenging activity presents.

Kevin: The first day of hiking after the solo, was our worst day [for group issues].

Jason: Was it the day we got lost?

Kevin: It was the day that we had to go through that forest and then we got lost, and it was all uphill, and everyone was just fighting with each other. That was pretty bad. Everyone got frustrated... (Group B: Interview 1).

It seemed that individuals who took on leadership and navigation tasks also faced social pressures to perform and lead their peers on the right path. Figure 5.7 shows a student from Group 9 seeking navigation assistance from the group leader. As Tristan explained, some participants that were in a leadership role identified that they were scared of failing when the group needed to rely on them. His comment below indicates that as the program unfolded and boys within their activity groups started to develop trust between each other, it seemed to become easier for individuals to assume a leadership role:

Um, I definitely found it hard when I was on nav. It was like the second day or third day, and I'd never really done any big navigating with the whole group before. And of course, like it's probably easier for other people, but I had to navigate through like hills and mountains off track through like, some marshlands, up like a mountain, like uphill. And it was horrible. Like, every step would be like 'slosh, slosh, slosh' and different like gradients and stuff. It was just really annoying... And I kind of had to rely on myself. And in the end, it paid off. Like, we got to our destination. But at that time, I was really scared cause the group was just like following me and to be honest, I didn't really have a clue where I was going (Group C: Interview 1).



Figure 5.7. Seeking navigation assistance from the group leader.

The compulsory element of the program influenced group dynamics as the students were not able to choose which participants were in their groups. This meant that students faced the challenge of living with people they did not necessarily know in an environment that was not familiar for 21 days. Living in close proximity with each other, tested participants tolerance levels and understanding of one another. Figure *5.8* provides an example of Group 9's normal campsite set up, whereby a central group tarp would be located with four tents placed in close proximity around the group area. Participants would generally share one tent between two or three students.

Many of the boys observed and interviewed voiced wanting to be more tolerant of one another. This is demonstrated by Dale's comment who states, "*I need to work on being more tolerant when I'm leading and not say sarcastic comments to people who are asking me questions*" (Descriptive Diary: Day 5). Twelve out of the eighteen boys interviewed had similar insights, identifying that their lack of tolerance was an issue. For instance, Elliot found the program to be socially challenging as he was surrounded by the same group of people for threeweeks:

I probably learnt about my ability to work with others. Definitely, my tolerance level is a big one... Yeah, there is definitely a lot of tolerance involved there [dealing with others]. It's hard not to get angry at them. It's hard to stay positive when it's taking longer than you think it should (Group A: Interview 1).



Figure 5.8. A normal campsite setup

Even though most participants interviewed felt that 'City to Summit' was more 'mentally challenging' than physically challenging, some students also found the social challenge of living with a small group for an extended period of time to be one of the most difficult parts of the program. As Oliver comments:

I wasn't really worried about the physical side of it. I thought that it wasn't going to be too much of an issue, and it wasn't. Um, I was more concerned about like mentally being away with and being surrounded by people the entire time (Group C: Interview 1).

All students in Group 9 agreed that one of the most challenging situations of 'City to Summit' was being around the same people; living together, sleeping together and eating together for three weeks (Observation Diary: Day 18). Living together in close environments whilst being exposed to various stressors often led to conflict amongst the group and between individual participants. Forming 'cliques' and talking behind other's backs also appeared to be a common occurrence amongst all the groups interviewed. This was consistent with observations of Group 9. For instance, participants in Group 9 were observed making personal judgements about one other and doing so behind each other's backs rather than during open discussions. The group started forming smaller 'cliques' of approximately 3 - 4 people that gathered together to talk about people in the other 'cliques' and not talking with one another. The 'cliques' were complaining to one another about issues that were causing them to feel annoyed, frustrated and angry about certain people in the other 'cliques' (Observation

Diary: Day 10). An example of this was when Syed's 'clique' was saying, "how annoying is it when they keep swearing and singing?", and Jack's 'clique' was complaining about other people, especially Tan. "I am getting so over Tan singing. His voice is so frustrating. Don't ya reckon? It's getting to the point where it's making me angry", said Jack to three other students (Descriptive Diary: Day 10). It became apparent that the continuous actions and behaviours of specific group members was affecting some individuals more than others (Observation Diary: Day 5, 10, 15, 20).

This formation of cliques in Group 9 was causing obvious conflict within the group which was leading to further segregation and undermining of group trust. Some duty groups were not working well together to complete tasks when at camp. There was a general 'feel of negativity' amongst the group and when the entire team was surrounded by one another at camp. No one was talking or communicating as they had been (Observation Diary: Day 10). It appeared that participants needed to vent to others about their feelings, but no one wanted to create open conflict amongst the whole group (Observation Diary: Day 10).



*Figure 5.9.* An example of students not talking with one another.

Jack, one of the students in Group 9, was concerned that the group conflict was causing big problems and felt that the problems would get bigger if they were not dealt with. Therefore, during the nightly debrief and he said to his group, "come on guys, let's bring everything out in the open now. I know people are saying things behind other people's backs and I don't like it, so yeah" (Descriptive Diary: Day 10). After Jack opened up this conversation to the group, everyone had a chance to comment on what was bothering them in an open and safe

Similarly, all four interview groups commented about difficulties with group tasks such as cooking dinner, cleaning, sleeping together in confined spaces in the tents and setting up a group tarp each night. As Figure *5.10* demonstrates, group tasks, such as cooking, required the boys to work together, trust one

another, communicate and rely on each other's skills to achieve the task at hand. Often, these group tasks appeared to cause conflict and social challenges amongst the groups:

Tristan: I got annoyed with people that didn't really help out. It would kind of grind my gears a bit. Uh, it's when you're kind of washing up and stuff. And kids just go off and don't really do much (Group B: Interview 1).



Figure 5.10. An example of a duty group preparing food.

Another example of a socially challenging situation was when Group 9 experienced a one and a half hour conflict situation during the process of setting up a group tarp on night seven. Each night during 'City to Summit', all 17 groups were required to set up a group shelter at their campsites to prepare for inclement weather and to practice the skill. On nights one to six, the staff and boys with prior experience in Group 9 helped to teach and guide the other boys with less experience on how to set up a tarp, learn knots and the skills required to achieve setting up a shelter. This was in an aim to practice their skills as they would be required to set up their own shelter during the 24-hour solo experience. On night seven, the group leader set the challenge for the boys with less experience to set up the group tarp without any assistance from the staff, or the boys who had prior experience and were competent in tying knots and putting up tarps. The group challenge was to get the group tarp set up before nightfall.

Figure 5.11 demonstrates how the boys were struggling to put the tarp up in the wind. The boys in Group 9 were struggling with the task. Individuals were yelling, blaming, procrastinating, standing around and talking to each other in an inappropriate manner (Observation Diary: Day 7).

During the conflict, the boys became very frustrated with one another, struggling to work together as a team, communicate respectfully or trust one another with jobs. All the boys appeared to very tired and hungry after a long day of whitewater rafting. However, while the boys were not as efficient at setting up the tarp as they were on nights one to six and they were not able to get the tarp up before nightfall, they still managed to set up a group shelter without assistance from the experienced people.



*Figure 5.11*. Day 7: Group 9 struggling to step up their group tarp in the wind.

This example highlights how the boy's responses to challenges changed throughout the program. Even though the boys may have been presented with the same difficult task each day, such as putting up the group tarp, their responses to the challenges may have been different each time. This could be caused by other contributing factors such the level of tiredness, team members feeling other stressors, adverse weather conditions, social pressures, hunger, level of daylight or be influenced by the people that were involved, which can alter a situation dramatically. It also demonstrates how the boys were able to learn new skills during 'City to Summit' and applying their learnings each day. Having opportunities to practice and apply these skills each day appeared to make it easier for the boys to manage the difficulty as the time went on. This didn't alter the
level of difficulty of the task but rather increased the boys' ability to manage the difficulty.

#### **Theme Summary**

The challenge theme comprised of three identifiable types of challenges; physical, emotional and social challenges. The boys responded differently to the different types of challenges presented during the program. The challenges were dynamic with these different types of challenges overlapping and interacting. For instance, the bike riding was identified as a physical challenge; however, it also proved to be a catalyst for increasing social and emotional challenges for some participants.

Overall, the eighteen participants interviewed felt that 'City to Summit' was more mentally challenging than physically challenging. The main contributing factor leading to social challenges was the various difficulties that were provided by living with a small group, in a novel environment for an extended period of time. Providing opportunities to practice skills and apply their learnings throughout the program, appeared to increase the boys' ability to manage program challenges. For example, each day the boys were required to set up a group shelter. At the start of the program, the boys found this task challenging. However, throughout the program, the boys learnt and applied the skills require to execute this task. The level of difficulty of the task did not change. However, the boys' ability to manage the task increased due to their increased levels of competence, thus altering the perceived level of difficulty.

225

#### **Theme 2: Response to Challenge**

As previously identified in the literature review, participants' responses to stress and challenges can be physiological, emotional, cognitive, behavioural and/or physical (Bradley, 1990; Csikszentmihalyi, 1996; Lee et al., 2013). There was a wide variety of coping strategies that were observed and identified by participants that were applied in response to the challenges and stressful situations presented during 'City to Summit' program.

The response to challenge theme has been classified into two methods of coping; emotion-focused and problem-focused strategies. This is consistent with the literature on coping, as Lazarus and Folkman (1984) suggests, coping skills should be identified as emotion-focused or problem-focused coping skills, rather than being seen as adaptive or maladaptive, or seen as inherently good or bad. Problem-focused coping aims to deal with stressors or difficulties in practical ways, including strategies such as planning how to manage the situation or overcome the stress through problem-solving. Whilst emotion-focused coping involves an individual aiming to reduce the effect of the negative emotional response to a difficulty, such as the emotional responses from fear, anxiety, embarrassment, excitement or frustration.

Problem-focused coping strategies were identified when participants accepted the challenge and took action to tackle the problem itself. Whereas, emotion-focused coping typically occurred when participants made a change in relation to the situation. Table 5.5 outlines the most common problem-focused

226

and emotion-focused coping strategies that were identified and applied by the participants during 'City to Summit'.

Table 5.5

Common problem-focused and emotion-focused coping skills applied

Problem-focused Coping		Emotion-focused Coping
1.	Putting Things into Perspective	1. Distraction
2.	Removing Oneself from the Stressor	2. Avoidance
3.	Ability to Accept Social Support	
4.	Addressing the Issue	
5.	Chunking	
6.	Coming to Terms with Difficulties	
7.	Cognitive Reframing and Applying	
	Positive Thinking	

# **Problem-Focused Coping Strategies**

The sections below describe the seven problem-focused coping strategies identified by the boys. Each section is explained in detail and supported by examples of the participants' comments and the researcher's observations.

# Putting things into perspective.

The most common problem-focused coping strategy identified amongst the participants was 'putting things into perspective'. Many students used the term 'putting things into perspective' as a way to identify the coping strategy they applied to help them manage challenging situations presented to them during the program. Although there was a range of approaches of 'putting things into perspective', Table 5.6 describes the three approaches participants used as problem-focused coping strategies:

# Table 5.6

# Description of the three approaches to 'put things into perspective'

'Putting things into perspective'		Explanation
1.	Compares past experiences	The participant compares and relates
		their current situation with their past
		experiences.
2.	Compares level of challenge	Evaluates and compares their current
		level of difficulties with other levels of
		difficulty they have experienced in the
		past.
3.	Assistance from others to compare the level of challenge	Participants look to others to assist and
		guide them in making a comparison with
		the level of other various challenging
		situations.

These three approaches of 'putting things into perspective' also highlighted two common factors;

- 1. using other situations as a point of comparison and,
- 2. using the feeling or the memory of the comparison as a motivator to help them manage in their current challenging situation.

This coping strategy appeared to provide a general framework to assist the boys in controlling their emotional responses to a situation, by providing a logical and rational comparison to their current situation with past experiences. At times the boys were able to 'put things into perspective' themselves, and at other times they required facilitation from peers or the leaders to identify the point of comparison. It did not seem to matter if the students perceived their past experience to be positive or negative to be useful as a comparison. Even if the student felt they were unsuccessful in their past difficult experiences or they had feelings of disappointment, they were still able to use the comparison of the difficult situation as a motivator. Table 5.7 provides an overview of the three approaches of 'putting things into perspective' and provides examples from the participants' comments.

# Table 5.7

Overview and Description	Examples of Dortiginant's Comments		
of the Approach	Examples of Participant's Comments		
Compares past experiences:	Compares past experiences:		
Jack compares past experiences to help him push through the barriers of feeling physical pain during challenging situations	Jake: Like, I'd push through to the point where it would hurt I was just kind of thinking about past decisions, I had made and that kind of helped me get through like the long hikes and in pain and stuff (Group C: Interview 1).		
Jack compares the level of challenge with another level of challenge he felt he had successfully achieved in the past	Jack: this is way harder and way wetter than it was on our other outdoor ed camp, you know, the one that everyone was saying was their worst outdoor experience. And this isn't even that bad (Descriptive Dairy: Day 5).		
Compares level of challenge: Matthew compares the creature comforts of life at	Matthew: Just putting everything into perspective, sort of learning that we might be a bit too comfortable in our lifestyle now and that um, yeah, we should just take everything um, respect everything and yeah (Group A: Interview 1).		
home, to the challenges of life on program			
Greg compares the disappointment of failing a challenging situation and uses it motivate him to do better next time	Greg: You just sort of get over the disappointment [of failing] and sort of, use it to motivate you. Which I think is a skill you, a skill you have to learn and to do that that you have to keep going and just get through the, the tough times (Group B: Interview 2).		
Assistance from others to compare the level of challenge:			
Group Leader assisted others to point out a comparison of the level of challenge and apply it as a motivator	Cody: I remember in our group, we all sat down halfway through and our leader said this is the hump. We just got past, we just got past the first half [of 'City to Summit'] so you can easily do the second half. And we were just like we are halfway through the trip, just do the rest and you'll be sweet (Group D: Interview 2).		
Group members assisted others to point out a comparison of the level of challenge and apply it as a motivator	Tan: Just think Cooper, as soon as we get to the river you won't have to hike with a pack on for like 6 days (Descriptive Dairy: Day 5).		

Various approaches and participant comments of 'putting things into perspective'

#### Removing oneself from the stressor.

Students demonstrated that this strategy involved the boys being able to identify what was triggering their stress and take positive actions to remove themselves from what was causing their emotional response. On many occasions, the researcher observed most of the students in Group 9 managing emotional and social challenges through walking away from the stressful environment, allowing themselves time to calm down, then come back to the situation and deal with things in a rational manner when the felt they were better equipped to manage their emotional responses (Observation Diary: Day 20).

For instance, in Lachlan's comment below, he describes how he was able to manage his response to the social challenges of living in close proximity with his peers by removing himself from the conflict or the challenging situation that was causing the issue:

Well for me, if I felt like someone was getting on my nerves, I'd just walk away and like at one stage ... I was getting frustrated because the people in my tent didn't seem to care, so I just sort of walked away and um took, took a 'breather'. And then came back and I just felt a lot better and just had a little bit of time to myself. And then after that, I was just back to normal, like all was chill (Group C: Interview 1).

Lachlan uses the term taking a 'breather' to demonstrate taking a small break or respite from the stress when his tent partners were not supportive of his difficulty. In many other cases, the researcher observed boys 'literally breathing' and taking in deep breaths to help them to calm down after they had removed themselves from the stressor (Observation Diary: Day 20).

In another example, Isaac reports how he was able to manage social challenges by removing himself from the stressor. In this case, the stressor was people in his group. Instead of having to remove people in his group from the current environment or situation, he took the practical approach and removed himself from the situation. He was able to understand that the people in his group were triggering his frustration and the best way for him to manage it at the time was to manage himself and not others:

Just being around everyone for like 21 days can be frustrating, so normally I just remove myself and get some alone time. I'd walk at the back of the group and just get away from whatever is frustrating me at the time (Descriptive Diary: Day 18).

#### Addressing the issue.

To assist in managing social challenges, that were identified as one of the major difficulties experienced during 'City to Summit', the boys pinpointed that a common approach to deal with these difficulties was to address the issue with the parties involved. In the following comment, James provides an example of how the social challenges of sharing a tent with two other group members for an extended period of time caused conflict amongst the three of them. His example provides evidence of how some students tried to cope with conflict by bringing issues to light and discussing their related frustrations, however, the comment also suggests that this approach did not work for everyone. The behaviour of one of the students in this account indicates that he wanted to avoid dealing with the conflict and used denial to create a sense that there were not any issues that needed to be dealt with:

Uh, going to go back to the tent partner problem... When we had issues, my tent partner would express his points of views, and I'd express mine. And then we like got into this one argument and my buddy and I tried to explain the situation to him. At this point, he just went into denial and just cracked it. He just pretended like nothing had happened (Group B: Interview 1).

All groups commented that 'addressing this issue' was a common response when participants were feeling strong emotional responses to stressors. 'Addressing the issue' may also be seen as a form of venting, where participants addressed issues by freely discussing their feelings with an individual or their entire group to help them manage their emotional responses to a stressor (Observation Diary: Day 4, 7, 10, 15, 19).

Occasionally the boys were able to recognise that for change to occur, some situations required action and for the issues to be addressed amongst the group. For example:

Frank: We did have a bit of conflict, but we got over it quickly...We all just sat down one night, and we just talked about everyone

and what they had to fix. Everyone just went around [the circle] and basically just spoke up.

Jason: The leaders weren't involved, and we just went around, and everyone said explicitly, what they felt wasn't working, what they felt needed to change for us to be more effective... And so, people had sort of things directed at them that people weren't liking what they were doing. And no one really took offence to it. Everyone was just taking it as advice, listening and taking it in...And there was a drastic change after that (Group D: Interview 1).

During a nightly debrief, the researcher observed changes in group dynamics between the Group 9 boys (Observation Diary: Day 10). The group was experiencing a stage of group development where participants were becoming very frustrated with one another's actions and behaviours. Yet the boys seemed to realise that the problems needed to be addressed as they understood they were required to spend another 11 days together. The conversation below outlines a situation of how particular students aimed to address the issue. This conversation was initiated and facilitated by the students, without any guidance from the group leaders:

*Tan: I just wanna know if I annoy anyone. Please tell me so I know and I have the chance to change it...* 

Jack: Yeah stop singing 'morale booster'... it's so annoying. And Joshua, can you please stop singing that annoying 'baby song'?

Joshua: I'm sorry for singing and being annoying... Also, I was annoyed at everyone too. When you were killing spiders and ants. Can you please stop that? Or if you are going to do it, please don't do it around me. I was brought up to believe that every living thing is equal, so please don't (Descriptive Diary Day 10).

This snippet of conversation demonstrates how the group was able to open up to one another and tell each other how they really felt. The researcher observed the students not attributing blame but showing a willingness to accept everyone's comments, even if they didn't agree. The next day, the group actively responded to everyone's wishes. It became obvious that this conversation had a positive effect on the dynamics of the group. The boys seemed to have developed a deeper understanding of each other through the discussion and were experiencing less conflicting situations (Observation Diary: Day 10, 11).

It is important to note that not all conflicting situations that were addressed resulted in positive outcomes. As Lachlan's recounts, his group tried to address a group conflict, but it seemed to have negative repercussions for a particular individual as he did not respond well to the group's feedback:

Lachlan: There's just this one person who was just negative, didn't want to play and sort of wrecked it for everyone else... Like everyone said at the end, he's so annoying.

Researcher: How did your group cope with that then?

Lachlan: We voiced our concerns and opinions. And he didn't take it well. No one wanted to sort of surround themselves with such a negative person. (Group C: Interview 1).

#### Ability to accept social support.

There appeared to be two types of social support which assisted participants to manage the various challenges presented during the program. The first type of social support was provided by their peers, and the second provided by adults in the group. The common factor of social support was whether or not students had the ability or willingness to accept the assistance that was being offered to them.

Groups and individuals were continually observed giving and receiving support from their peers to help manage the challenges that were presented during the program (Observation Diary: Day 4, 6, 8, 10, 15, 17, 20). Whether it was helping out with group tasks at the campsite, providing words of encouragement or physically assisting in challenges experienced during the adventurous activities (Observation Diary: Day 18). For instance, when Ian's feet and shoulders were hurting during the hiking section, two other boys in his group, who seemed to be managing better with the challenge, offered to carry his tent and some of the group food to lighten his load. Ian understood that he was struggling with this program challenge and was willing to accept the support from his peers (Observation Diary: Day 3).

Will provides an example of how he not only seeks but also accepts support from his best friends. He acknowledges that when he is experiencing stressful situations throughout the program as well as in his home life he accepts assistance from others help him with difficulties:

*Will: Sometimes I talk to people [when I'm under stress] ... like my close friends (Group B: Interview 2).* 

About halfway down the whitewater rafting section, students were invited to accept a challenge to jump off a nine-meter high rock into the river (see *Figure 5.12*). The boys in all four interview groups identified the jump rock challenge to be one of the most memorable and fun moments experienced on the entire trip. This was one of the program challenges that the boys were able to 'choose' to engage with, rather than being forced into the challenge because of circumstances.

Some students were scared of being up at height, while others were worried about jumping into the freezing cold, snowmelt water of the Mitta Mitta River (Observation Diary: Day 6). To overcome their fear of heights and engage with the jump rock challenge, the boys used a range of coping strategies, such as positive self-talk, goal setting and breaking the challenge down into smaller, more achievable obstacles ('chunking') (Observation Diary: Day 6). Nevertheless, most students who were expressing emotional responses to the challenge, such as fear and anxiety, responded best when they received support from their peers either via encouragement or pressure. For example, as described by Matthew, who had his arm around Elliot during the interview when he was retelling the story:

Matthew: My highlight moment would be as Elliot said jumping off the rocks cause that took me some courage to get up there, but I *did it...I'm just scared of heights. Yeah, Elliot got me up there... He got me up there through peer pressure and support.* 

Elliot: I was calling him a wuss [said in a laughing, good-natured manner] (Group A: Interview 1).

During the interview, the boys appeared to show comradery towards one another, and the teasing appeared to be light-hearted fun rather than bullying. It was this sort of cheerful teasing and support exhibited by Elliot that helped Matthew to confront his fear of heights and push himself further than he originally thought was possible.



Figure 5.12. The jump rock challenge, Mitta Mitta River.

Each day of the program, two boys in each group were designated to be the 'student leaders' for that day. These leadership roles rotated daily and meant that the boys would have several opportunities throughout the program to take on the role as the student leader and be able to practice and apply skills they had learnt throughout the program. An example of this was when the student leaders showed an increase in empathy and understanding to their peers who were struggling with some program challenges. For instance, when people were struggling during the hiking sections, a common coping strategy applied by Group 9's student leaders was to move the fast walkers to the back of the group to provide support through encouragement, conversation or distractions techniques. This also ensured that the slower walkers could set the pace for the rest of the group and stopped the group from separating during the walk with the potential of causing conflict. These group management skills appeared to be learnt during the program and was initially facilitated by the adult group leader. These strategies applied by the student leaders provided a positive impact on the group as it supported their peers who were not managing as well as others with program challenges (Observation Diary: Day 3, 6, 7, 10, 15, 17, 20). However, the support offered by the student leaders was affected by the individual's 'ability to accept the support' or not.

Adult group leaders provided continual support to participants at both an individual level and group level (Observation Diary: Day 1, 3, 5, 7, 10, 14, 15, 18, 20). During adventurous activities and practical tasks presented to the groups, leaders provided safety guidelines and knowledge, taught skills and coached students through difficulties providing motivation and encouragement to succeed. This was demonstrated during the mountain biking section where group leaders and the mountain bike support staff were observed encouraging students who appeared to be struggling by riding next to them and commenting things like 'you can do it', 'just keep it in a low gear and peddle slowly'. The leaders were also observed coaching students about the best way to tackle the uphill sections by using a variety of different gears on their bikes (Observation Diary: Day 10).

During group conflict scenarios, group leaders facilitated games and activities to help bring the groups together (Observation Diary: Day 15). At times when participants were struggling with emotional challenges, they would listen to the student's concerns and demonstrate care and support for the students' wellbeing (Observation Diary: Day 19). Reece recalls how he accepted the support from his group leaders. He acknowledges that the group leaders supported him during difficulties, helped to keep their group motivated, provided knowledge to support their experience and safety and that this impacted his enjoyment of the experience:

Reece: Um, I enjoyed it a lot ['City to Summit'], and the main reason for that was because I had two really good group leaders. They helped us out when we were struggling. Yeah, I was always getting advice and stuff of them. And yeah, they just kept us motivated. (Group B: Interview 1).

An example of a situation where a student was not willing to accept support from his peers was when Isaac walked to the back of the group to start a conversation with Cooper while they were hiking. It was obvious that Cooper was struggling with the physical challenges of hiking, showing unenthusiastic body language, he looked physically tired and was demonstrating negative facial expressions. Isaac noticed this and tried to take his mind off the task. Isaac said, "so Cooper what are your hobbies?". Cooper replied "NO, NO! Just go away". Isaac kept trying to ask other questions, only to hear the reply, "shut up Isaac, or I will hit you!" Isaac continued trying to make conversation with Cooper for about 5 minutes before he walked off and stopped trying to help (Descriptive Diary: Day 5). In this instance, Cooper did not want to accept the help of Isaac. It appeared that Isaac was trying to help by distracting Cooper through having a conversation, however, for Cooper, having a conversation may have added an extra burden on top of the physical, social and emotional challenges triggered by the hiking (Observation Diary: Day 5). This highlights the importance of how the support or assistance is presented to someone. For instance, if Isaac had have said "hey Cooper, why don't we have a chat so that we can keep our minds off hiking and we will be there in no time?", Cooper's response may have been different, and he may have accepted the support.

#### Chunking.

Throughout the entire program, the boys were observed breaking a challenge down into smaller, more achievable steps. This was a form of goal setting where the tasks were broken into achievable chunks (Observation Diary: Day 3, 5, 6, 8, 15, 18, 20). This process of 'chunking' appeared to stop the boys becoming overwhelmed by the task that they perceived at first to be unachievable or too daunting. For instance, the boys would say things like, *"let's just focus on making it to here first", "we can do this, we just have to take it day by day"* (Descriptive Diary: Day 3, 8).

Jason recalls how he used 'chunking' as a method to breakdown tasks into smaller time slots so that he perceived the task to be more manageable:

Like on 'City to Summit', I sort of taught myself this thing where like if I am getting up in the morning and I'm like uh, 'I really don't wanna get up this morning'. I'll just sort of break it down to half an hour slots, so I'll be like, 'Oh, I'll get out of bed and then I'll, you know, have breakfast, and then I'll pack up the tent' (Group D: Interview 1).

Another example of 'chunking' was observed during the jump rock challenge. For example, as depicted in Figure 5.18 the boys got out of their rafts at water level. If they accepted the challenge to jump off the rock into the water, they each had to scale up the face of the rock, until they reached a height that they wanted to jump off the rock into the water. Some of the boys scaled the rock and went straight to the top of the 9-meter-high outcrop. Whereas, other boys used 'chunking' as a method to break down the challenge. Most boys in Group 9 who were scared of heights used 'chunking' as a method to overcome their fear of heights and jump off the rock (Observation Diary: Day 6). Feeling overwhelmed with the thought of climbing straight to the top of the cliff, the boys applying this method stopping at various flat points on the way to make the challenge appear more achievable. They set themselves small goals, such as reaching the 1-meter mark of the rock. Once they had achieved that, they would celebrate their achievements and reassess the situation. Next, they would set another smaller goal for themselves that felt achievable, such as reaching the 3-meter mark of the rock. They kept applying this process until they either reached the top of the rock or they reached a point where they felt their limit was. The method of 'chunking' appeared to make the challenge feel more achievable and less overwhelming for the boys who were scared of heights (Observation Diary: Day 6).

#### Coming to terms with difficulties.

All four interview groups commented on the need for them to come to terms with difficulties as a means of problem-solving to help them to overcome program challenges. As 'City to Summit' was a compulsory program for the Year 10 cohort, the level of choice to engage in certain challenges was limited. Most of the program challenges involved the boys having to come to terms with the fact that they found themselves confronted with a situation which was not by choice. For example, there were no taps or running water at most of the campsites. Therefore, when the boys needed to collect water, they would need to make their way to a water source. When the groups were rafting, the river was easily accessible. Whereas, when the groups were hiking or mountain biking, they would have to make their way to closest the water source, which could be up to 150 meters from their campsite. They would then need to collect the water from the water source (e.g., river, creek) and carry the water back to the campsite. A duty crew was normally assigned to collect the water for the group for that particular day.

The strategy of 'coming to terms with difficulties' appeared to involve the boys applying three steps. The first step was to acknowledge and understand the situation they found themselves in. Secondly, to become aware of the various obstacles or trigger to their emotional responses. Last, to plan and take actionable steps to manage the stressor or their emotional responses to the difficulties presented. The first step of acknowledgement and acceptance of the difficult situation seemed to strengthen the boys' ability to manage their initial emotional responses (Observation Diary: Day 20).

Oliver's comment below demonstrates his ability to first acknowledge the situation and accept the challenge. Second, he understood that he needed to complete the hike in order to make it to camp. Finally, his actions to help overcome the obstacle was is to apply his strategy to keep looking at his feet and get into a meditative rhythm:

When I was walking like I did this thing where I just didn't look up. I just looked at like my feet. And that really helped cause if you look up, you just say "ah". Another three kilometres of this crap. But if you just look at your feet, just keep going, and then your kind of like, it's like falling asleep. You kind of just like just get lost, get lost in the rhythm (Group C: Interview 1).

Some program challenges required participants to immediately accept the situation and therefore, take appropriate actions to overcome obstacles, such as setting up tents or a group shelter in a storm. Participants understood that if they didn't come to terms with the situation and take actionable steps to certain obstacles, that they may experience negative consequences for their actions or lack of action. For instance, it was raining one morning with no shelter to have breakfast. Figure *5.13* demonstrates how the group decided to wrap the tarp

around them to provide shelter from the rain, so they could eat breakfast, keep warm and stay out of the wind. As a group, had they decided not to take the actionable steps to set up a shelter, the consequence of doing nothing would be getting wet and becoming extremely cold (Observation Diary: Day 17).



Figure 5.13. Day 17: Group 9's taco wrap shelter.

# Cognitive reframing and applying positive thinking.

A common, everyday occurrence amongst the groups and individuals was reframing negative thoughts and attitudes, to a positive frame of mind (Observation Diary: Day 10, 15, 18, 21). This involved the boys identifying the problem, becoming aware of their thought processes and taking the actions required to maintain or change their thoughts to positive thinking. For instance, Tan commented that "we should focus on keeping a positive frame of mind and keep up our group morale while we are hiking" (Descriptive Diary: Day 3). The positive thinking techniques appeared to assist in keeping negative emotions at bay when confronted with program challenges. It appeared that the participants' thoughts and attitudes directly affected their feelings, behaviours and actions (Observation Diary: Day 3, 4, 5, 7, 10, 15). For example, Oliver recalls how instead of thinking pessimistic thoughts about the challenges of hiking long distances, he reframed his thoughts to congratulate himself and celebrate how far he had already come:

Oliver: One thing that um helped me get through all the tough bits were not looking at challenges negatively... not thinking 'how much further do we have to go' but thinking like 'how far we'd come' (Group C: Interview 1).

Using positive self-talk or keeping a positive mindset appeared to help reduce the stress that the program challenges provided (Observation Diary: Day 19). As the boys shared more experiences throughout the program, they started to become aware of how their negative attitudes could have a potentially negative impact on the group. Student leaders were observed addressing their group to maintain a positive mindset in order to help manage social challenges. For example, Lewis asked the group:

Can everyone please be aware of their bad moods and try not to bring everyone else down? Cause it can just go through the group. I know I'm like that sometimes. But just try to hide your bad mood cause everyone else is tired too (Descriptive Diary: Day 10). In the comment below, Lachlan's indicates how he felt it was very important for groups to maintain a positive mindset to assist with managing social group challenges:

I didn't mind my group because I had lots of guys that I knew really well, and that got you in a good mindset...And then there is just days, where like all it takes is one person just to not be in the right mindset and it just shuts you all down. And then after that everyone was just really down after that... No one wanted to surround themselves with negative people... You need to remind everyone that no one really likes a negative person. It's so important [to stay positive] (Group C: Interview 1).

#### **Emotion-Focused Coping Strategies**

The following sections describe the two common emotion-focused coping strategies highlighted by the boys. The coping strategies are clarified and reinforced by the explanations of the participants and researcher observations.

#### **Distraction.**

The most common emotion-focused strategy used to help the boys cope with physical, mental and emotional challenges was 'distraction'. All four interview groups commented that they used distraction as a method to keep their minds busy and divert their thinking from the difficult aspects that the program presented. Interestingly, the boys used a range of different methods of distraction, such as humour, singing songs, playing games, working on riddles or talking to one another to keep their minds busy. These various methods of distraction seemed to be focused on the coping with difficulties of the challenge, such as feeling tired, feeling pain, feeling bored, scared, or having feelings of anxiety, rather than using distraction as a means of avoidance. For example, Reece commented:

Yeah, the ascent [on the hikes] was really challenging. That was one of the hardest things... so we just, kind of made up games... that kept us going" (Group B: Interview 1).

Brodie also explained how he used self-distraction methods, such as singing to help him cope. He acknowledges that he tried his hardest to take his mind off the challenging tasks to help him manage the physical and emotional challenges he experienced during the mountain biking:

Brodie: Like Matthew said earlier; definitely, the hardest part for me was the bike riding on the second day. The first big hill, it was hard but I sorta just put myself in 1, 1 and swapped between 1, 1 and 1, 2 and I would set myself like a post or a tree in the distance and just make it to that and then make it to the next one. I just tried as hard as I could to get my mind off what I was doing. Cause like, you don't have to think that much about it if you're just peddling. You've just gotta do it, and I mean it took a long time, but when I got there, I was like that's not really that hard. I was sort of like singing just songs to myself and anything to keep, just in my head, anything to keep my mind off what I was doing (Group A:

Interview 1).



*Figure 5.14.* Day 10: A student resting after the last uphill leg of the mountain biking section.

Oliver recounts to others another situation where singing, as well as positive thinking, was used as a method of distraction to help him cope with all the difficulties presented to him during the program:

Oliver: I kind of had a Bob Marley song going through my head.

Lachlan: "Everything, is gonna be all right" [singing]

*Oliver: "It's gonna be all right" [singing]. I actually did, and it kind of got me through [the challenges].* 

*Researcher: And did you apply that [singing and positive attitude] in your head?*  Oliver: Yeah legit... That song was going through my head. It kind of just shows a bit of resilience, that song... (Group C: Interview 1).

Humour was identified by all interview groups as a common method of distraction. Throughout the experience, groups were observed using humour as a way to manage stress and help one another get through difficulties (Observation Diary: Day 19). Jake provides an example of where he and a friend used laughter as a way to cope with the challenge of having a broken tent. He believed their actions demonstrated their ability to be resilient:

I was camping with a friend, Paul. And we were in our tent, and at that time it [the situation] was really crap. We were just like swearing and everything like "what, what do we do", and that kind of stuff. Our tent's like broken. Then we looked at each other and like we were distressed, but we were laughing, and we both said after this, we will be laughing and like it'll be like, it'll be a funny story afterwards. And like a half an hour to an hour later, we'd got back to the hut, packed everything up and we just bounced back, and we were just all happy again. We were just laughing at the situation. But earlier [in the program], we would've been like really angry, frustrated, and screaming and yeah. It's just like little things like that; I think that's resilience (Group C: Interview 2). As demonstrated in *Figure 5.15*, the boys in Group 9 mainly used humour as a way to keep the group morale high, to help maintain a positive attitude or 'feel' within the group, and to keep their minds off the current difficulty (Observation Diary: Day 19). For instance, Ahmed recalls how they used humour and laughter to help get their group through tough times during the program:

Our group is just amazing, even in the so-called 'bad times', we are still laughing, cracking jokes and using humour to get us through (Descriptive Diary: Day 19).



Figure 5.15. Students joking around at camp after a long day of hiking.

# Avoidance.

The boys identified active avoidance as a common emotion-focused strategy that was used in response to difficult situations. Avoidance was also observed by the researcher on many occasions (Observation Diary: Day 3, 5, 10, 16, 20). Active avoidance was particularly applicable to situations that may have resulted in group conflicts.

An example provided by Chris demonstrates how he and one of his tent partners responded to the social challenges of building new relationships and dealing with conflict. His comment below indicates that his tent partner would avoid conflicting issues by walking off or not talking about it, while Chris demonstrated the use of harsh words to vent his frustrations:

One of my tent partners was a guy I didn't really know before. He wasn't really speaking to anyone in the group and then a few times he like really wouldn't help out or he'd be rude about it. Not reacting to him is hard cause he sort of gets on my nerves and he tries to get a reaction [from me] but um... And one day, he like broke my walking stick which I've had for a few days. I wasn't happy with that but um...I didn't like physically do anything. I did abuse him a little bit with words but... (Group B: Interview 1).

In another situation, Chris describes how he would avoid creating conflict between his peers by completing the tasks himself rather than confronting his peers about his frustrations:

Um, I found sometimes my group would sorta just avoid it [duty group tasks], like not all the time but a lot of the time I'd end up doing it by myself or like maybe packing the tent up by myself or putting it up by myself just cause no one could be bothered, and I

# *didn't want to keep hassling them to help out (Group B: Interview 1).*

A further example of avoiding conflict was when comments from Ian in Group 9 were obviously frustrating the group. Group 9 team members all seemed aware of the issues but did not show any willingness to take action to confront the situation. Figure 5.16 shows the group sitting around the campfire after a long and wet day hiking and, Ian said to the group, "someone should put more wood on the fire", while he sat back and did nothing. Everyone looked at each other, but no one said a word. After a while, Ian repeated the same thing, "someone should put more wood on the fire". After becoming frustrated, Dale replied, "yeah, and that someone is you". Ian slouched his head down in between his legs and mumbled something under his breath, but no one took any action (Descriptive Diary: Day 4). This example exhibits how Group 9 avoided conflicting social situations instead of managing or addressing the group's feelings and actions. While some types of active avoidance proved to have a positive effect, this type of avoidance was later observed to cause negative repercussions and further social challenges later in the program. This may have been a result of not addressing initial group frustrations (Observation Diary: Day 10).



Figure 5.16. Group 9 sitting around the campfire.

# **Theme Summary**

The response to challenge theme consisted of two approaches of coping; problem-focused and emotion-focused coping skills. Participants reported the application of seven common problem-focused coping strategies used to assist them to cope with program challenges; (1) 'putting things into perspective', (2) 'removing oneself from the stressor', (3) 'ability to accept social support', (4) 'addressing the issue', (5) 'chunking', (6) 'coming to terms with difficulties', and (7) 'cognitive reframing and applying positive thinking'. Only two emotionfocused coping strategies were identified and applied by the boys. These were labelled as; (1) distraction, and (2) avoidance.

The coping methods most often identified in the interviews were also the most common strategies observed by the researcher. These were the problemfocused coping strategy of 'putting things into perspective' and the emotionalfocused strategy of 'distraction'. Both these approaches for coping entailed various methods. For example, the boys who applied the strategy of 'putting things into perspective', used three various types of comparisons; (a) comparing past experiences, (b) comparing the level of challenge, or (c) requiring assistance from others to help the boys compare the level of challenge. Whereas, emotionfocused coping skills, such as distraction, involved using numerous strategies, such as conversations, singing songs, playing games, humour or using conundrums.

#### **Theme 3: Personal Development**

The personal development theme encompasses the participants' common thoughts, learnings and perceptions about the various experiences they encountered during and after 'City to Summit'. Specifically, this theme pinpoints and presents the four common areas that were found to impact the personal development, learning and growth that the participants identified. Each of these three key areas is broken down into sub-themes and are presented in further detail in the sections below. The first area presents the key sub-themes that influenced and impacted learning outcomes. The second area presents the self-identified internal developmental assets that were enhanced as a result of participation in the program. The third section, details evidence of unconscious learnings, and the final section of this personal development theme presents new insights and understandings from the follow-up interview data.

### **Impacts of Program Design.**

Although the boys were all presented with the same challenges during the 'City to Summit' program, the individual perceptions of the experiences and the learning achieved differed. The researcher's observations highlighted that the boy's perceptions of the experiences and their ability to learn and retain knowledge may have been affected by numerous variables, such as the individuals;

- previous experience,
- psychological/emotional state,
- level of tiredness,
- personality,
- fitness levels, and
- personal skills and capabilities (Observation Diary: 2, 5, 10, 19).

An example of contrasting perspectives from the same challenge was provided by Joshua and Scott when Joshua said, "*that hill today was nearly as bad as the first day*". Whereas, Scott contradicted Joshua's statement and said, "No, it wasn't. It was nowhere near as bad as the hill on the first day, that was a huge mountain" (Descriptive Diary: Day 4).

Six months after the program, each of the eighteen participants reported that they felt the overall experience of 'City to Summit' was positive. These results were consistent with the boy's initial perceptions of their experience immediately after the program. For example, Lachlan highlights how he and his teammates felt when they got back to school after the program:

I had a positive experience. I just remember everyone being really happy. Like, no one was in a bad mood after 'City to Summit'. Everyone was saying, "Oh, how was it?", "Oh, so good. How was yours?" And just everyone just had such a good time. Everyone was just so pumped up like, "Oh, yeah. 'City to Summit' was sick" (Group D: Interview 2).

All the boys identified three key elements that influenced their positive learning experiences due to the program design;

- 1. feeling a sense of achievement,
- 2. feeling a sense of adventure, or
- 3. feeling a sense of privilege when they became aware of positive circumstances or things they were grateful for in their daily lives.

The following sections detail these three elements sub-themes in further detail.

#### Sense of achievement.

The design of the 'City to Summit' provided a framework to support the boys in gaining a sense of achievement through the completion of daily program challenges and the entirety of the program. During the Frontloading Phase at school (day 0), the boys were asked to identify potential goals they would like to achieve through completing a form with questions, such as "what are your goals for this program?" or "what do you want to learn about yourself?" They put their responses to these questions in an envelopment; the envelopes were sealed and kept at school with the intention of the boys opening their envelope at the end of the program to see if they had achieved these goals. However, not all groups completed this goal-setting task, and not all groups were given their envelopes to open and reflect upon on the during the Integration Phase at school (day 22).

The goal setting questions and structure tended to guide boys to set goals of a very general nature. For example:

Matthew: I set self myself a goal to reach Kosciuszko safely and with as little dramas as possible (Group A: Interview 1).

Reece: My goal was to complete the program and then go home and appreciate life a lot more (Group B: Interview 1).

There were two common types of goals between the boys who completed the form. Firstly, some boys just wanted to survive and get through the experience. Secondly, all of the boys commented about wanting to develop better friendships. These goals were very broad, demonstrating that the boys were struggling to set specific goals that related to personal growth and understanding:

Robert: I can't really remember what mine was, but I think, maybe like trying to get through it without too much of a problem missing anything at home (Group B: Interview 1).

Even though the boys generally struggled to articulate or remember their goals, all interviewees commented on feeling empowered and feeling a sense of achievement at some point throughout the program. The program challenges provided opportunities for the boys to engage with challenges, resulting in feeling a sense of achievement. As Brodie's comments indicate, these situations often became opportunities for learning and provided insights into how persisting with difficulties and 'hard feelings can lead to feelings of elation and achievement:

I think it's definitely more rewarding when you finish it, like each [day]; but the whole thing ['City to Summit'] as well. Each different part which had the hard parts and hard feelings, come with like the best feelings after. Sorta no pain, no gain sort of thing... When we got to the end of the day [bike riding], it was probably like the best feeling I had all camp. I was like "you've done it', you've got through the day". Getting to the end for the ride and it was pissing down rain and everything, it was the hardest thing we did all camp, but I didn't really mind, cause you to know, I finished it (Group A: Interview 1).

All eighteen boys commented on how good it felt to reach the top of Australia after the challenge of 20 days of hiking, mountain biking or whitewater rafting. Cody's comment emphasises that the primary goal for the entire program was for the boys to complete their journey and reach the highest point of Australia. His comment summarises the boy's feelings of achievement once they had persisted with the challenges and made it to their destination:

Cody: The climb at the top of Kosciuszko was so awesome. Cause, you know, it doesn't seem like much, but the whole trips about Kosciuszko at the end of the day... When we're standing on top of the summit, and you're looking over the view, you're like, 'This is awesome. It was worth it', you know you've done it (Group D: Interview 1).

The conversation below between the boys in interview Group B also highlights how the beauty of the mountains, completing the challenge and being in a novel setting also contributed to their feelings of accomplishment:

Reece: My most memorable moment was definitely sunrise [On Mt Kosciuszko]. That was amazing.

Chris: Yep. My most memorable moment was just after we hiked up into the Thredbo mountains and the last four days up there was probably the best part. Cause, like, it was awesome just to be up at that altitude, waking up to awesome sunrises and scenery around us the whole time was pretty cool (Group B: Interview 1).


Figure 5.17. Sunrise on the top of Mt Kosciuszko.

### Sense of adventure.

The majority of boys from the four interview groups and the observation group connected feelings of adventure with experiencing either a highlight or a memorable moment. The outdoor education program created these opportunities through exposure to risk and difficulties that required the boys having to draw on all their resources to overcome the challenging, adventurous activity. For instance, Robert stated:

I really enjoyed the rafting. I didn't think I would as much but I just, I loved the thrills and stuff, going down, going down the rapids and stuff is pretty cool (Group B: Interview 1).

Feeling a sense of adventure appeared to be heightened by also having the option to choose to engage with these adventurous opportunities or not. The most

common example of this was when the boys were able to choose to engage with the 'jump rock challenge'. This involved the boys 'opting in or out' of jumping off a 9m high rocky outcrop on the banks of the river and landing in the water when they were rafting. Even though for some students, this was one of the most challenging situations on camp, most of the students felt that jumping off the rock was a fun type of challenging experience. It was a challenge that they were able to choose to participant in. For example, Cooper didn't think he could jump off the rock, but having the opportunity to choose to engage in the challenge, made him feel not only a sense of adventure but also a sense of achievement:

Cooper: My highlight was definitely the jump rock because I never thought I could do it, but when I did it, it just felt so good (Descriptive Diary: Day 9).

In addition, Zac points out how he was craving the feeling of adventure and having an adrenaline rush, and how this experience became his most memorable moment on the program:

Personally, my most memorable moment was the second day of rafting. We were surrounded by good people and awesome rapids, and then we got to this point where we were able to jump off this rock. It was like nine meters tall or something. And jumping into the water just gave me that adrenaline rush that I needed (Group C: Interview 1). As Figure 5.18 shows, the participants from group 9 were feeling very proud of themselves for overcoming their fears, which resulted in feeling a sense of adventure and achievement. The evidence provided in Figure 5.18 is also supported by Ian who said, "*I remember how good it felt jumping off the rock*. *The wind in your face for about 3 seconds and then the water slaps your face*" (Descriptive Diary: Day 15).



Figure 5.18. Group 9 after the jump rock challenge.

The conversation below from Interview Group C indicates how choosing to engage in a challenge that was something new, adventurous, and perceived to be risky and difficult, caused the boys to have fun, feel satisfied, feel a sense of adventure and accomplishment.

Lachlan: I had a couple of most memorable moments. One was when we jumped off that rock cause it was really high. And I got a bit scared as I was standing on top of it. When I jumped off it was a really good feeling. Oliver: Yeah, that was really good fun [jumping off the rock]. I enjoyed that and also seeing some of my mates who were pretty scared about it. You know, being able to show themselves that they can do it. I thought that was good. Um, including myself. I think it was pretty scary. I'm not gonna lie (Group C: Interview 1).

# Awareness and appreciation of one's circumstances, privileges and relationships.

During the interviews, the boys used the terms grateful, gratitude and appreciation interchangeably to describe their feelings when they had become aware and appreciative of the comforts and conveniences of modern urban life at home. All boys frequently commented on how being removed from their regular living environment heightened their awareness of their normal life circumstances and privileges of what they have access to in their daily lives. Every one of the boys interviewed and observed frequently discussed how being removed from urban settings and not having access to 'material things', such as hot water, transport, houses and comfortable beds, made them feel a greater sense of appreciation or gratitude for their life circumstances and privileges (Observation Diary: 2, 5, 7, 10, 12, 19):

Chris: You appreciate the things that are already given to you. Like hygiene for me is a big one... Like a shower, a toilet and stuff. Greg: Yeah, material stuff like a house, bed and all of that. I just really appreciated transport... (Group B: Interview 1). 264

Similarly, all eighteen boys commented on how they normally take things for granted when they are back at home, not realising how privileged they are until access to these privileges were taken away from during the length of the program. For example, Jake's comment below highlights how the program has caused him to not take things for granted like he used to:

Jake: You treasure the little things like running water, hot water, little chocolate bars and stuff.... And I just noticed that I'm not taking things for granted as much as I would have before. Like, if I use my phone, I don't abuse the privilege of having it... (Group C: Interview 1).

It was not only material items that the boys appreciated, but the boys also commented on becoming aware of their privileged family circumstances and how much people, such as their parents, do for them in their home lives. As Ian points out, through the actions of needing to be self-reliant and do tasks for himself during the program, he became more aware of his privileged circumstances in his home life:

I have realised just how much everything gets done for me at home. My parents do everything for me. And now I really appreciate it. They cook, they clean, they do the dishes. Here, I have to rely on myself to do everything like cook, clean and actually get to different places through rafting and cycling and stuff like that (Descriptive Diary: Day 9).

#### Appreciation of family.

Immediately after the program all eighteen boys interviewed commented on the fact that they felt their relationships with their families had changed and became stronger when they returned home from the program. As mentioned previously, one of the most significant program challenges for the boys was being away from home for an extended period of time. However, the challenge of being removed from their daily lives caused the boys to miss their families, resulting in a heightened awareness and greater understanding of the importance of home life and how much it means to them. It was clear that being away from family triggered the boys to question the value of family relationships, what it means to them and question their role within the family. Amongst all boys, there was a general 'feel of appreciation' for what their family members do to help and support them in their normal lives. These feelings combined with the absence of family members during the program, initiated the boys to alter their previous perception of their family relationships. For example:

Matthew: I now appreciate more what my parents do for me. And like whether it be cooking breakfast in the morning or something. You just really appreciate what your family do for you... (Group A: Interview 1).

Chris: Missing my family was a really big part [of learning] for me. Like, I missed my family, but I probably missed my smallest sister the most. And like, I usually like to call everyone [family] every one or two weeks and, see how she is going... Yeah, I just missed her a fair bit, so when I saw her again, it was pretty good cause I've been away from her for so long (Group B: Interview 1).

Tristan: When I got home, I realised, I was just like nonstop talking with my Mum and Dad. Like, when I got home, I just told Mum about every single part of the camp. I think I just really missed her (Group C: Interview 1).

Jake: I changed like my personal values as soon as I got home. Like, I haven't fought with my Mum at all or my brother. I've just been more respectful towards them cause I just treasure them so much more (Group C: Interview 1).

The comments above also highlight how the boys were able to identify specific significant others in their family group, such as Chris's little sister or Tristan's Mum, that impacted on them more than others. It appears that being away from their families with no means of contact increased the boy's awareness of the importance of these relationships.

As part of the program design and curriculum, there was a pre-planned activity where boys would receive a letter from their families during the program. This may have impacted the boy's beliefs and feelings of appreciation for their families, as it appeared that receiving the letters had a significant impact on the boy's perceptions of their family members and their appreciation of their relationships with their families. For instance, Chris comments that:

267

[Reading the letters] made you think that they care about you more than you know...what they said about you was like a big realisation. Like, how they actually felt about you. It was pretty good to read the good things they said about you and stuff so... (Group B: Interview 2).

#### Perceptions of technology.

Comments from the boys in all four interview groups indicate that technology is a distraction for them in their daily lives. The data reveals two aspects about the boys' perceptions of technology post-participation in the outdoor education program:

- Without technology, the boys were able to focus on the task at hand during the program.
- 2. The boys appeared not to miss having technology throughout the program.

Table 5.8 demonstrates the commonalities between the boy's comments across all four interview groups, and how being in the outdoor setting, without technology influenced their perceptions and feelings about home life.

# Table 5.8

# Examples of Group comments about the effects of nature on their perspectives about technology

Interview Group	Sample of Participant's Comments Post-Program			
Group A:	Brodie: Ah, that's another thing that camp helps you with because there			
Interview 1	is not technology or anything to distract you with when you're in nature.			
	You sort of just focus on what you are doing, but at home, you've just got			
	so much stuff you can start using.			
Group B:	Reece: I was kind of worried about not having a phone, technology and			
Interview 1	stuff, but on camp, I didn't care at all. I got back, saw my family. I didn't			
	really acknowledge it [technology] at allThat was surprising for me			
	I normally have it [my phone] glued to my hand. Yeah, I don't really			
	miss technology muchWhich Mum likes cause she used to think I was			
	an addict.			
Group C:	Jake: I think a lot of boys at home just use technology and rely on it all			
Interview 1	the time. So, getting away from that and being in the bush was a good			
	thing. It's good seeing what it's like without it [technology].			
	Tristan: And also, it gives everyone a good idea of what it's like outside			
	the city and what it's like outside without technology. Cause um these			
	days, there's so much technology involved in your life that it's hard to get			
	away from. So, going to New South Wales and the high plains in			
	Northern Victoria was really good to get away from that whole side of			
	things.			
Group D:	Mark: I like to push myself more now. Um, before the camp I was easily			
Interview 1	distracted by things I would get distracted by the uh computer, and I			
	would put my book away and play the computer. So, after being on			
	camp, in nature, um I learned to um stick to something more			

#### **Developing and Demonstrating Overall Resilience**

After completion of the program, the boys in all four interview groups were asked about what the meaning of resilience meant to them. They responded with a range of different answers, such as comprehending resilience as mental toughness, or the ability to think positively (See Table 5.9). When the boys were asked if they felt they had become more resilient as a result of participating in 'City to Summit', all boys observed and interviewed commented that they felt their resilience capacity had increased immediately after participation in the program (Observation Diary: Day 19).

In order for the participants to be able to apply resilience attributes and coping skills, they needed exposure to difficulties, challenge or adversity. As Brodie suggests, the reason the boys felt the program developed their overall resilience was because the boys were exposed to difficult challenges which extended them beyond their comfort zones, whilst still feeling like they could achieve or be successful in overcoming the challenge:

I reckon it [the program] does build resilience, but I reckon it's only because of the one or two hardest things you do on camp. Like, again with the cycling, you're trying your hardest, it's getting you out of your comfort zones and then you know you can achieve it (Group A: Interview 1). In addition, Oliver's comment below indicates how the boys felt the program not only developed their overall resilience but also influenced the enhancement of other, attributes such as determination and persistence:

I think it's not just overall resilience. I think there's a lot of other attributes that you bring back [home] like determination and persistence. So, a lot of those things that you've learned from the camp you bring back so easily so you'll start to persist in certain areas and start to be like more determined about just life in general, to reach your goals and stuff (Group C: Interview 1).

Lachlan also acknowledges that the program not only provided opportunities for him to demonstrate his resilience capacity, but it also helped to increase his confidence levels to push himself more mentally and physically:

I was really struggling, like a lot, but I just thought that if I keep on, keeping on, you know, like moving my feet, eventually I'd get to the end so, I think that's definitely resilience for me. The one big thing I've taken from the camp is knowing I have more confidence in myself to like push myself (Group C: Interview 1).

# Table 5.9

# Participants definitions of resilience

Group A: Interview 1	Group B: Interview 1	Group C: Interview 1	Group D: Interview 1
Resilience is	Resilience is	Resilience is	Resilience is
Brodie: From this camp, I now understand resilience to be how I can bounce back without hurting	<i>Reece: Self-reliance. Like relying on yourself more to do things.</i>	Lachlan: Bouncing back from something that's not good. It's sort of like, it's not about	Jason: The capacity to endure.
myself. Elliot: Resilience was like when	Greg: Bouncing back from adversity or something.	how hard you can get hit, but how much you get up and keep moving forward. It's like	Frank: Just coping with a situation.
I'd push through to the point where I would physically hurt I	Robert: To keen going when the	being mentally strong, headstrong and like being	Kevin: I can't remember. Isn't it like to make
can't really explain it but pretty much what helped me through was	going gets tough.	able to persist with something that you find difficult.	something out of a bad situation?
thinking about past experiences, past decisions and knowing I had people around me that could help	<i>Chris: Being able to rely on yourself basically.</i>	Knowing that if you keep pushing and keep going forward, eventually	Jason: Turning something apparently bad, sort of
me through.	James: Being able to bounce back from tough situations and being able to look on the	everything's gonna be all right.	around to become something good. Isn't it?
	positive side of it.	Oliver: Bouncing back from something that's not beneficial to you.	<i>Mark: Thinking in a positive mind.</i>

The common areas of personal development that were identified by the participants are summarised in the Venn diagram (Figure *5.19*). The visual representation presents the self-identified internal and external assets that were developed as a result of participation in 'City to Summit'. As Figure *5.19* exhibits, Overall Resilience appeared to be demonstrated when the boys internal and external assets overlapped. The analysis of Phase II and the first set of interview data identified four common internal assets to be either learnt or enhanced. These internal assets are presented in detail in the following sections. The external assets listed are discussed in the next theme that explores the development of relationships.



Figure 5.19. Self-identified developmental assets enhanced as a result of participation in 'City to Summit'.

#### **Internal Developmental Assets**

This section explores the boys' understandings of their development of internal assets, also known as developing a positive relationship with oneself (Benard, 1995, 2004; Wagnild, 2009). The sections provide examples and descriptions of the boys' application and understanding of the four sub-themes; developmental tasks, self-reliance and independence, mental strength and perseverance, in relation to the boys' initial perceptions of 'City to Summit' immediately after the program (see Figure *5.19*).

#### **Developmental tasks.**

As the literature identifies, developmental tasks are age-related standards of behaviour which affect an individual's human functioning (McCormick et al., 2011). These include tasks, such as self-sufficiency, learning to get along with others, leadership skills and acquiring a set of values to guide behaviour. The majority of boys across all four interview groups and all the boys in Group 9 consistently reported using five common developmental tasks which may have assisted their development of assets (Observation Diary: Day 2, 5, 9, 10, 13, 18, 20). These skills included the boys reporting:

- 1. Developing communication skills.
- Increasing their ability to show empathy and understanding towards others when they were struggling. (e.g., "Even though this may be easy for me, I can see that this situation is difficult for...").

- Increasing their levels of tolerance and understanding towards other's needs, values and opinions.
- 4. Learning how to work with others towards a common goal.
- 5. Learning skills of how to lead others without dominating.

For instance, Jason reports that *"we learnt effective ways of communicating with people"* (Group D: Interview 1). Lachlan also explains that he was able to practice developing his communication and interpersonal skills throughout the program:

My ability to communicate with people got better I thought... Like, it may not have seemed like that on the camp, but I think it's improved a lot. Just being able to like to read a situation, I think it's gotten a lot better for me so... (Group C: Interview 1).

Twelve out of the eighteen boys were able to provide examples of how their tolerance of others increased during the program, such as Elliot who states:

I learnt a kind of tolerance. Appreciation for just the self and focus on other people's opinions and stuff... Yeah, it ['City to Summit'] like opens your eyes a little bit. Like understanding the dynamics of other people (Group A: Interview 1).

Matthew provides another example of how he is able to manage his emotional responses by accepting the situation and demonstrating a level of tolerance towards his initial frustrations. His comment also indicates the use of problem-focused coping skills, such as 'removing oneself from the stressor': I think I've done well. Um, whenever there's someone that's irritating me or something like that, I just, I just let it slide. I don't really dwell on it too much. Just being accepting. If it gets too annoying, well, I'll remove myself or do something more useful (Group A: Interview 1).

Immersion into challenging environments clearly put the boys out of their comfort zones. Brodie's comment indicates how the program could bring out the worst behaviours in people in his group, yet it was the people in his group showing these problematic behaviours that caused him to require increasing his levels of tolerance towards others:

I think...'City to Summit', can see bring out the worst in people...so I guess it has made me a bit more tolerant of people...Group A: Interview 1).

#### Self-reliance and independence.

Self-reliance and independence were the most commonly mentioned resilience attributes that the boys identified to be improved due to participation in 'City to Summit'. This meant that the boys were able to demonstrate an understanding of their personal capabilities and limitations, whilst also being able to independently undertake and manage challenges within their own limitations.

The outdoor education program and the natural environment provided opportunities for students to be away from their daily urban lives. All boys interviewed and observed mentioned that having the opportunity to practice being self-reliant and independent which supported the development of these skills (Observation Diary: Day 5, 7, 10, 19). For example, Cody's comment below indicates a level of understanding that the program helped them to build their selfreliance by being away from family who would normally assist with daily tasks:

I kinda learnt to be more independent because my Mum kind of spoon feeds me a little bit [at home] ... Cause we are all spoon fed everything in Melbourne ... Like, phones and people cook our food and like teach us stuff. In the wild it's like, you have got to do all that yourself (Group D: Interview 1).

Researcher observations and participants' comments demonstrate that the group leaders appeared to have provided supportive learning environments in which participants were able to learn from their mistakes (Observation Diary: 2, 3, 5, 12, 15). For example, Jason acknowledges how having the opportunities to practice skills throughout the program helped to support the boys developing their sense of independence:

Jason: There's a lot of the teachers letting us make mistakes and making us sort things out ourselves to sort of develop that independence and decision-making skills... (Group D: Interview 1).

Cody also provides an example of how the leaders supported the students to become more independent by allowing them to learn through their direct experiences instead of telling them what to do: Like, our camp leader. He let us just go off the right track [hiking]...But he didn't tell us. Which I reckon was good because then we had to learn, you know, we have to keep checking the map, keeping us more independent, you know (Group D: Interview 1).

In addition to becoming more self-reliant, over two-thirds of the boys commented on increased other developmental tasks, such as cooking and cleaning. This demonstrates their ability to be more self-sufficient by taking actions requiring them to depend on themselves without the assistance of their parents or other adults. Brodie provides an example of how he felt as though the program forced him to develop these skills as there was no-one else there to do it for them:

Brodie: At home, I never really get asked to clean or cook that much. I've always thought that it wouldn't be fun at all and that it's a lot more difficult. But when you're actually like forced to do it. Obviously, it's a lot worse situation cause you're on camp, with a bucket of water, cooking with a stove and you're cooking like dehydrated vegetables and everything. But it was actually a lot easier than I thought it would be. And I thought I should probably try that at home and just try to help out a bit more (Group A: Interview 1).

During a nightly debriefing, students in Group 9 were asked if there was anything they had learnt over the past 10 days that they thought they would be able to apply back in their home lives. Syed's response shows his understandings of how the experience of 'City to Summit' helped to prepare the boys to be more self-reliant in their own lives:

The past 10 days have made me realise just how independent we can be at this stage in our lives. We have to rely on ourselves, and we can do it. The experience prepares us for going back home and being able to live by ourselves is we have to move out when we are eighteen. I think I will use my learnings back at the farm and at home (Descriptive Diary: Day 9).

Not surprisingly, the solo day appeared to be the prescribed activity that influenced the boys' levels of self-reliance the most. All the boys interviewed and observed provided multiple examples of how the solo day was the specific program challenge that improved their levels of self-reliance and independence. As Greg points out, being independent was essential as the boys were required to live alone and survive for a 24-hour period:

I think that I was pretty independent on the solo day cause you're by yourself... It felt good that you could know that you could put up your shelter, you could live with yourself for 24 hours, and be selfsufficient in a way, so that was good (Group B: Interview 1).

The boys were supported to cope with the solo challenges by gradually being exposed to more time alone each day. The leader facilitated daily reflection or solo sessions to gradually expose the boys to the stressor of being alone in a wilderness setting, with the amount of time increasing each day. The boys were taught the skills required for them to be self-sufficient, such as cooking and setting up shelter. Figure *5.20* shows a student practising setting up a tarp by himself to prepare for the 24-hour solo experience.



Figure 5.20. A student practising setting up a shelter.

For the majority of the boys, this was their first time spending an extended amount of time alone, in a wilderness setting, with no one else around to assist them. As Matthew emphasises, at home the boys have their family and friends that they can rely on for support, whereas during the solo day the boys need to rely on themselves, causing them to problem solve, take action and responsibility for their own well-being:

On solo day, it most definitely was the time I showed the most independence cause you're by yourself, no one around to help. You're really reliant on yourself and your well-being during the solo. At home, you've got someone else to help you. Your family or something or your friends might help you, but here it's all by yourself (Group A: Interview 1).

Brodie also recalls how he would normally rely on other team members to do things for him, and how being alone forced him to be persistent, self-reliant and apply skills himself:

My tarp kept falling down. I would put it up, and it would fall back down. Eventually, I got it to just at the right point to stand up. That was probably when I had to use my independence the most. Normally, it's sort of easy to just sit back when there is a big group and lots of people that are willing to do it. And in the end, they're the people that are better on solo day. They make the better tarps, but it's easier to sit back and just rely on everyone else to do stuff. When you're forced to do it by yourself, yeah, it's hard (Group A: Interview 1).

#### Mental strength.

Mental strength refers to an individual's capacity to effectively manage their psychological resources in response to stressors and challenges presented, while performing at their best, despite their current difficult circumstances. As all eighteen of boys identified the program to be more mentally demanding than physically demanding, it is no surprise that the boys commented on developing their mental strength. For example, Cody states: Sections of 'City to Summit' were more like mentally, a little bit physically but mostly mentally hard, you know, like especially solo time... (Group D: interview 2).

Mark also found that the program required him to push himself mentally, which is something he wouldn't usually do:

I really learnt to push myself more mentally. Like, frankly, I would normally stop something [that was hard], but while I was hiking, I just had to push myself to join the others and catch up with the others, so that's been good (Group D: Interview 1).

The boys experienced challenges requiring mental strength before the program actually started. For example, sixteen boys reported feeling apprehensive about attending 'City to Summit' prior to the program starting. This was a perceived level of challenge and difficulty that required the boys to apply mental strength to manage their emotional responses:

Zac: Obviously before it ['City to Summit'] you're a bit nervous and kind of tentative about going. But when your back [at school] and you look back, you just realise its kind of one of those once off experiences, lifetime experiences that you'll remember for the rest of your life (Group C: Interview 2).

Jake's response below also demonstrates the boys were able to apply mental strength to help them push through mental and physical pain barriers in order to achieve a desired outcome: I also learned how resilient I can be and how far I can actually push myself... I thought that was amazing cause I've never done anything like that before and that challenged me at times... Now I know how hard I can push myself. To the point where I will cry. And even after the mountain biking, I felt sick, and I vomited a little. I'd never done that before, and so I know how hard I can push myself before my body gives up... I just can't really comprehend it at the moment (Group C: Interview 1).

The program challenges provided numerous opportunities for the boys to test their strengths, weaknesses and limitations. Interestingly, over half of the boys commented on lacking the self-confidence in their ability to overcome program challenges. This is highlighted by Scott, when he said, *"camp has really shown me about my strengths and my weaknesses and my limitations. I didn't actually think I could do all of these things*" (Descriptive Diary: Day 19).

#### **Determination.**

As the literature confirms, determination can also be viewed as using persistence or perseverance (Wagnild, 2009, 2011). The boys in all four interview groups used these words interchangeably to describe their feelings when they felt they were extending their personal capabilities and found themselves out of their comfort zones. All boys observed an interviewed commented on developing their determination and their ability to persevere in the face of difficulties. The following conversation between Brodie and Jake provides an example of how they felt that being exposed to the program challenges, whether it be physical, social or emotional, helped them to increase their levels of determination:

Oliver: I definably had more determination...

Jake: I think I leant determination too. Like, I think in developing determination, it's the hills that were a major factor but I think it's also like the will to wanna get to the end... You're like so determined to get to the finish and like, I think, that was the main rewarding thing about Kozi, getting to the top, just like the feeling of relief, sort of. Like you pushed yourself... (Group C: Interview 1).

Oliver also highlights how having a goal and the 'will', want or need to achieve that goal, also helps in developing and applying determination.

Similarly, Brodie and Elliot's conversation indicates how when they were placed out of their comfort zones and confronted with challenges; they were able to demonstrate and practice the use of their personal capabilities to help them succeed:

Brodie: I did use a bit of determination to do the hard bits, so I didn't give up... I put myself through a pretty hard time, but it was worth it...

Elliot: Yeah, I think I learned self-discipline and determination. Like, especially determination when we were coming to the summit (Group A: Interview 1). The researcher also observed Group 9 demonstrating persistence, perseverance and determination throughout the entire program. Not surprisingly, these attributes were typically observed during days when students found the adventure activities particularly physically challenging, such as the mountain biking and hike up Mount Misery (Observation Diary: Day 1, 10, 13, 16).

### **Unconscious Learnings**

The term 'unconscious' was used frequently used by the boys throughout the interviews. When the boys found it difficult to articulate their feelings or put the program outcomes into words, they tended to identify that their learnings were at an unconscious level, highlighting a lack of awareness. For instance, Jason was able to recognise how the developmental task of 'communication' could be applied in different contexts, but he was unable to understand how he could practically use it:

Um, we could use communication skills and that back at school... I couldn't really say exactly what or how. It's more of a subconscious thing you develop of how you communicate with people effectively (Group D: Interview 1).

Over two-thirds of the boys felt that their learnings were unconscious. An example of this is provided by Elliot who indicates how almost all the boys commented on finding it difficult to firstly understand what they had learnt on program, and then be able to maintain the learnings after the program: I think that's the problem. You can't really mentally, make a conscious effort to maintain what you learned cause it's like you don't actually know what you learned; it probably affects you just unconsciously or something (Group A: Interview 1).

Lachlan was one of the students who felt he had maintained his resilience six months later. However, he found it difficult to provide examples of how he has applied his resilience capacity to overcome difficulties, feeling that his learnings have been on the subconscious level:

I think the program made me more resilient. But I feel like my resilience is subconscious though. Like, I know it ['City to Summit'] made me resilient but I can't really think of any specific examples about it (Group D: Interview 2).

#### Follow-Up Interview Insights and Understandings

In the follow-up interviews, the boys provided new insights into the program that they did not comment about during the first set of interviews. It appeared that allowing the boys six months' time to process the experience brought out different insights that the boys did not comment on initially. Brodie highlights the point that it was not until the boys were removed and had distance from the program, that they were able to make connections of what they learnt and have time to fully process the experience:

When you get back [you realise], but while you're there, you don't necessarily make the connections. I enjoyed it but, you might not

think it's the best thing at the time. Getting back, you can really see how different it is and why you get put on the program (Group A: Interview 2).

The boys' new insights six months after program specifically related to two conceptualisations; (1) the level of difficulty of the program challenges, and (2) the link between challenge and fun. The following sections describe these sub-themes in further detail.

#### Level of difficulty of the program challenges.

As outlined in the 'challenge' and 'response to challenge' themes, program challenges required the boys to negotiate difficulties by apply coping skills and psychosocial attributes. Students responded in many ways to program difficulties, such as crying, becoming frustrated, extreme joy and happiness or getting angry with the situation. Their responses indicate that they did, in fact, find the obstacles difficult. For example, Isaac states that *"I've only cried like five times. I learnt how much I can actually challenge myself and what I can achieve"* (Descriptive Diary: Day 19). However, after the boys were back at school and had experienced the achievement of completing 'City to Summit', more than half of the boys felt that the overall level of difficulty of the program could have been harder:

Jason: I didn't really find any aspect of it super challenging. I was expecting the solo to be more challenging (Group D: Interview 1).

The below conversation from the first set of interviews with Group A indicates that the boys felt the entire program could have been more challenging:

Elliot: Solo day should have been I think could have been longer. Brodie: Um yeah, I agree with Elliot, it was definitely worthwhile. But it [City to Summit] was a bit easier than I thought it would be...

Matthew: I found the biking physically challenging...But other than that I thought physically it was pretty easy.

Elliot: For some kids, it was a lot harder than others.

Brodie: Yeah, I think maybe they could make the whole camp a little more physically harder...Because like after the days where you haven't done much, although like you feel good, you sorta feel like it's been sort of a cheap feeling... Like, I cheated myself through this day (Group A: Interview 1).

However, six months later, the boys had changed their opinions about the program challenges needing to be more difficult. In the follow-up interviews, more than half of the boys' suggested that the program didn't need to be more difficult, but that the level of difficulty of the program challenges should be aimed at the correct level for individuals, not the entire group. For example, Elliot acknowledges that for his resilience to be further developed, that the challenge level needed to be more difficult, however, this would not be the same for everyone:

City to Summit was a lot harder for some people than for me. And I think that if it was a lot harder for me, I would've got more out of it (Group A: Interview 2).

The conversation below that occurred within Interview Group D highlights the importance of specific program design which actively aims to include tasks and challenges that are aimed at the right level for individual students:

Cody: I think if it was more physically challenging, that it would have been too hard... like if it was more physically challenging, it would have had repercussions...

Kevin: Yeah like, everyone would be more tired and cranky.

*Cody: Then you'd start becoming like mentally fatigued.* 

Jason: You'd be getting into camp late and...

Cody: And like be cranky, and the fat kids will be, you know, just struggling. They'd make it worse for everyone (Group D: Interview 2).

However, as Greg points out, "*I think maybe it's a little bit harder to just cater the challenge levels for everyone*" (Group B: Interview 2).

#### Link between challenge and fun.

When students were engaged in a challenge which was perceived to be a difficult, yet a fun activity, it appeared to have an increased lasting effect on their

ability to recall a situation six months later. This was especially applicable to activities or program challenges that they had had not done before 'City to Summit':

Frank: I think my best memory was just when we were rafting, and we got to run a solo boat. Cause it was kind of different and fun. I haven't done that before (Group D: Interview 2).

More than half the boys made comments on their new understandings and conceptualisations about the relationship between the level of challenge and the fun factor they experienced whilst participating in the program. For example, Tristan from Group C proposes that there is a relationship between challenge and fun:

You'd probably get the resilience factor increasing if you made it more challenging. And then the experience or fun factor decreasing... So, there's something... like there's a relationship between those two [fun and challenge] (Group C: Interview 2).

The data indicates that an activity is perceived to be 'fun' when it is difficult but achievable. If the boys consistently failed at the challenges presented due to a high level of difficulty, the 'fun' factor would decrease. Conversely, if the challenges were perceived to be too easy, the boys became disengaged and felt they had 'cheated' their way through the day (Brodie: Group A: Interview 1).

#### **Theme Summary**

This theme outlined the participant's self-reported understandings of their personal development and increases in their overall resilience capacity, including the development of numerous internal developmental assets such as self-reliance and independence, mental strength, perseverance and developmental tasks (e.g., tolerance and communication). Specifically, this theme highlighted how the boy's psychological state influenced their ability to learn, retain information and apply skills. For instance, when the boys were feeling over-aroused (extremely anxious or fearful), it appeared that this was not the best mental state to be experiencing for supporting their ability to learn. Conversely, when the boys were experiencing levels of under-arousal (boredom), the boys were less engaged in the learning process and program challenges.

Being immersed in a wilderness environment away from their normal urban setting, provided a comparison point for the boys' everyday lives, resulting in the boys' being more appreciative and showing gratitude for their normal life circumstances at home and at school (e.g., realising the amount specific family members do for them at home). This resulted in the boys being more appreciative and grateful for their relationships with family. The boys' also experienced feelings of gratitude and appreciation when they became aware of one's privileges, such as their access to technology, shelter, material items, food and running water.

Overall, the boys generally felt that the program impacted them in positive ways. For instance, all boys reported positive learning outcomes when they felt either a sense of achievement or a sense of adventure. However, on the whole, the boys struggled to articulate the direct effect the program had on them, describing their development as 'unconscious'. The boys felt that their learning had just been 'ingrained' in them throughout the experience.

The six months distance between the program's completion and the postprogram interviews allowed the boys' time for reflection, providing new insights that the boys did not conceptualise originally in the first set of interviews. Six months after the program, the boys' felt that there was an association between the level of challenge and the level of fun experienced during 'City to Summit'. It was no surprise that all the boys' found the program challenges difficult, however, in the first set of interviews, the majority of boys felt that program challenges needed to be more difficult in order to increase levels of resilience. Whereas, in the follow-up interviews, more than half the boys suggested that the level of difficulty of the program challenges should not be changed and that the program challenges should also cater for individuals specifically.

#### **Theme 4: Relationships**

#### **External Developmental Assets**

In contrast to the internal assets presented in the previous personal development theme, the relationships theme captured three significant sub-themes in relation to the participants developing their external assets; (1) relationships with the natural environment, (2) relationships with leaders, and (3) relationships with peers (see Figure 5.19). Throughout this next section, supportive accounts are provided, including researcher observations, participant reflections and

evidence of the boy's experiences. The overall relationship theme will close with a summary of the major contributing factors in developing these relationships.

#### Relationship with the natural environment.

The boys from all four interview groups were easily able to articulate how immersion in the wilderness setting impacted both their perceptions of the natural environment and the positive experiences they had in nature. They were also able to explain in detail how being away from urban life gave them a heightened sense of appreciation for family, technology and modern appliances (see Table 5.8). For example, Bob stated:

I'm like Dale. I really learnt how much I like the outdoors and how much I really like nature and just being out here (Descriptive Diary: Day 19).

The boys also became aware of and appreciated the beauty and aesthetics of nature. Their comments indicate that they developed a closer relationship with nature and became more respectful and aware of environmental impacts and the sensitivities of the environment:

Scott: It's [the environment] just so fragile, so we need to look after it. That's why we use the paths and toilets and stuff that Parks [Victoria] make so we can keep it beautiful for future generations (Descriptive Diary: Day 19).

Dale: The environment is just so diverse. We have been through so many different environments. From the Mitta Mitta River to farmland, to bush and now we are in the Alps above the trees (Descriptive Diary: Day 19).

It was clear that when the students started to develop an understanding of the environmental resources we rely on as humans, consequently, they started increasing their levels of respect and appreciation of the environment. An example of this is when Isaac highlighted his change in perceptions and an increase in respect for nature:

Water is something I respect way more now. We have travelled on it, drank it, washed within, and it's just so beautiful and crystal clear up here (Descriptive Diary: Day 19).

Solitude in the wilderness also provided the boys with new, exciting experiences that offered opportunities to experience a connection with wildlife that they would not usually experience at home. For instance, Tan said, "*I remember the sound that the wombat made during my solo, I will never forget that*" (*Descriptive Diary: Day 15*). Bob also loved spending time in the environment, exploring and "looking at the insects and the spiders out in the bush. They are so different and cool; you don't get that in the city" (Descriptive Diary: Day 15). Syed also appreciated moments spent in nature during the program:

I love listening to the sound of wild brumbies around your tent. That was so awesome (Descriptive Diary: Day 15). 295

However, while all boys interviewed and observed were able to explain the positive experiences that occurred when they were in the natural environment, they found it difficult to express their feelings about their connection with the environment, their relationship to it and what the environment meant to them. The data from the observation diary implies that the boys in Group 9 clearly demonstrated a connection and positive relationship with the environment due to their comments and behaviour, yet they just struggled to articulate their feelings and put it into words (Observation Diary: 7, 9, 10, 18). For example, the researcher comments:

During the debrief today, the boys were talking about how much the loved their solo time each day and just hanging out in the bush, but they couldn't really explain how they felt about the environment (Observation Diary: Day 18).

#### **Relationships with leaders.**

Each of the 17 groups had a different group leader for their 'City to Summit' experience. The relationships that the participants developed with their group leaders appeared to have a significant impact on the participant's experience during and after the program. Interestingly, while each group had different leaders, who had varying skill sets, qualifications, levels of experience, personalities, leadership skills, and facilitation styles, sixteen of the eighteen boys commented on how the impacts of their group leaders improved their experience. Similarly, all boys in Group 9 commented on the positive impact of their leader,
such as Jack who expressed his gratitude in a debrief during one of the last nights of the program:

Thank you so much [group leader's name], this has been one of the best things I've ever done in my life thanks to you (Descriptive Diary: Day 19).

# Leaders role in relationship development.

Leaders played a critical role in group development and the development of relationships during outdoor education programs. It appeared that the primary factor contributing to the development of meaningful relationships with leaders, was the ability for group leaders to develop trusting, positive connections with the students early in the program. Figure *5.21* shows an example of Group 9's leader playing games with his students on Day 3 to assist in developing relationships with the boys early in the program.

Once the group leaders had developed trust with individuals and their groups, the boys' relationships with their group leaders appeared to be influenced by two other factors: (1) the group leader's ability to read and manage group dynamics, and consequently, facilitate the group accordingly, and (2) the group leader's ability to apply leadership and facilitation skills when required. For example, Reece recalls how his group leaders had a positive impact on his experience. This seemed to be due to his group leader's ability to apply leadership skills, such as supporting him during difficulties, keep up the group's motivation and providing knowledge to support the boy's experience: I enjoyed it a lot ['City to Summit'], and the main reason for that was because I had two really good group leaders. They helped us out when we were struggling. Yeah, I was always getting advice and stuff of them. And yeah, they just kept us motivated (Group B: Interview 1).



Figure 5.21. Day 3: Leader's and participant's playing games.

One of the key roles of group leaders was to use a range of facilitation techniques to assist students to manage issues and conflicting situations within the groups. The next example demonstrates how a leader applied facilitation and conflict resolution strategies by implementing a problem-focused coping activity that helped their group to express their feelings. This is an example of how the group leader was able to recognise group dynamic issues and apply strategies that not only helped to resolve group conflicts but also influenced the participant's relationships and perceptions of the group leader. Frank: We often had these things called 'summits' where we'd all write down on a piece of paper our problems. We just had a piece of paper and pen, and we would hand it around. And we'd write down any problems we had. And then it would stay anonymous and then our group leader would read them out to the group and then chuck them into the fire once we've resolved it. That was awesome how he [group leader] did that (Group D: Interview 1).

In another example, Lachlan interprets a quote often used by his group leader, which demonstrates how his group leader provided support, knowledge and advice during the program. His comment also shows how his leader was able to help their group to refrain from using negative attitudes:

Lachlan: Um, as my camp leader said, "it only takes one drop of poison to poison a well". And it only takes one negative person to speak or say something to just ruin the group (Group C: interview 1).

Facilitation methods, such as highlighting a point or using memorable quotes as teachable moments, appeared to have positively impacted their group dynamics and may have also reduced the likelihood of team conflicts occurring due to negativity, which ultimately impacts the team's relationships with others.

An example of exceptional facilitation and group management by Group 9's leader was observed during the hiking leg to the top of Mount Misery. In this section, the boys were required to hike up a steep 6km uphill incline. It was at this point that Group 9's morale was very low (see Figure 5.22). The group leader recognised that the group was struggling with managing their emotional and social responses to the physical challenge, as there was lots of complaining and negative comments occurring amongst the group (Observation Diary: Day 13).



Figure 5.22. Group 9 complaining to one another at the top of Mt Misery.

The leader was observed initiating a discussion with the entire group in which he outlined his 'positive thinking model' and how the positive thinking cycle was impacted upon by their habits, beliefs in spirituality, attitude and current environment (see *Figure 5.23* below) (Descriptive Diary: Day 13). During this group discussion, he explained the model in detail and told the boys about the power of choice, and how they all had the power to choose how they felt in any given situation (Observation Diary: Day 13). The tool was applicable for

individual use, as well as within the group to help them manage their responses to difficulties not only in this current challenge but also in future challenges.



*Figure 5.23.* The 'positive thinking model' from the leader of Group 9.

Immediately after this discussion, the boys displayed changes in behaviour, reverting to positive attitudes and behaviours (Observation Diary: Day 13). This conversation and learning experience appeared to have a considerable impact on Group 9. It influenced how they coped with the upcoming physical, mental, social and emotional challenges by causing the boys to become more aware of their current negative thoughts, attitudes or behaviours, and recommending that they reframe their negative thoughts to ones that were positive. In this instance, the group leader had developed his relationships with the group in a particular way. He chose to introduce this model on Day 13, once he had already built trust and positive relationships with the group. His actions demonstrate his leadership abilities by firstly acknowledging that the group was experiencing difficulties and secondly by presenting the 'positive thinking model' (see *Figure 5.23*) at a moment where they needed it the most (Descriptive Diary: Day 13). The group leader recognised the need for positive reframing to assist the boys in managing the difficulties they were experiencing individually as well as a group and facilitated the group accordingly. Because of the prior relationships he had developed with the students, they listened to him during the crux of the challenge. Whereas, if this model had been delivered to the boys in a different context, or at a different time during the program, the model may not have resulted in the same positive impact or significance on his group (Observation Diary: Day 13).

The above situation demonstrates how one leader can impact an entire group by influencing their ability to cope with challenges through support, guidance and the application of facilitation and learning tools which altered the boys' perception of difficult situations through reflection and awareness. However, it is essential to acknowledge that the facilitation of this model only occurred with Group 9. This particular group leader was very experienced and well prepared, bringing along a laminated paper copy of the model to deliver to the boys at a time when there was an influential 'teachable moment' (Observation Diary: Day 13). Interestingly, after this discussion, the positive behaviours and attitudes of Group 9 tended to continue over the course of the program (Observation Diary: Day 14, 18, 20). If a student started demonstrating any negativity, the boys initiated a conversation about the importance of the 'positive thinking model' (Observation Diary: Day 14, 17, 19). However, there was one instance after the facilitation of the 'positive thinking model' where Group 9's attitudes and behaviours became quite negative. The researcher then observed Group 9's leader facilitate take a different approach with an activity called 'it makes me mad when...' During the activity, each participant was required to pour water into a bowl, while saying a comment about something that makes them feel 'mad' or 'frustrated'. The more the participant felt a strong or heightened sense of negative emotions, the more water they were required to pour into the bowl. As they poured the water into the bowl, participants would say, ''it makes me mad when...''. For example:

Scott: It makes me mad when people intentionally try and piss people off.

Dale: It makes me mad when people are ignorant. Cooper: It makes me mad when people don't value silence. Syed: It makes me mad when we do these meetings, we shouldn't have to do it guys. Group Leader: It makes me mad when people have a lack of tolerance and understanding towards one another (Descriptive Diary: Day16).

After the group leader's final comment, he proceeded to overflow the bowl of water and throw the bowl of water up in the air, spilling water on everyone as he said, *"so, guys what do you think the purpose of this activity was?* Most of the boys stood there wet and in shock. Jack and Dale were the only ones to respond. Jack believed the purpose of the activity was *"to get everyone's opinion out in the open"*, while Dale said *"Na, the bowl kinda represents the group and all of us have our things that make us mad, but then it overflows or exploded like we did. We need to get back to that positive thinking"* (Descriptive Diary: Day 16). The above examples demonstrate how leaders' can positively influence a group's development, manage conflict and create opportunities for relationships to become stronger, by applying specific facilitation styles, leadership skills and techniques.

#### **Relationships with peers.**

All participants interviewed and observed commented on how the relationships with their peers developed throughout the program, and how they felt their relationship development with their peers was one of the most important elements of 'City to Summit'. As James identified the program provided a platform for students to 'really' get to know one another, *"you really learn how to work together as a team and make new friendships that last quite a lot longer than while you're just school" (Group B: Interview 1).* 

Even though the boys' may have spent their past four years of schooling with the same cohort of students, it seemed that across the board, the cohort did not know each other well. A common theme advocated by all the boys was that they tended to associate with the same people and friendship groups back at school before they participated in 'City to Summit'. Yarra College seemed to be aware of this as they strategically chose to place students in groups with boys they either didn't know well or wouldn't commonly associate with. The students had no choice of allocating who was in their group or whom they spent their 21-days. This strategic group planning undoubtedly influenced the development of the peer to peer relationships. As Syed, James and Lachlan acknowledge, the fact that they were not placed in a group with their mates, allowed them to develop relationships with people in their year level that they usually wouldn't associate with:

Syed: I like the camaraderie-ship that we have built and the relationships I have developed with people I wouldn't normally talk to (Descriptive Diary: Day 8).

James: I developed friendships. I didn't know all the people in my group that well. And then once on camp, then they became all pretty good friends. Some [people] were pretty funny and all that so I had a lot of fun (Group B: Interview 2).

Lachlan: A main one [goal] for me was making new friends...I had six other people in my group that I've never interacted with. Or had but just hadn't really talked to and had a proper conversation with so it was really good getting to know those different people (Group C: Interview 1).

Another key factor contributing to the development of the boy's relationships was the program design and the extended length of the program. The additional time spent together, immersed in nature with the same small group who were travelling independently of other groups appeared to support this group development process. As Chris explained,

Being around the same people for two or three weeks. You don't really focus on it [making friends], but you just become friends with everyone. You may not be friends with someone but just cause you're around them a lot, you learn about them, and you learn to appreciate them. I just developed friendships, I guess (Group B: Interview 2).

Participation in the vast array of shared experiences and everyday group tasks, such as cooking, cleaning, lighting a fire or setting up the group shelter, provided opportunities for students to develop their teamwork skills and get to know each other more (see *Figure 5.24*). For instance, Syed said "*I appreciate having Ian in my duty group. I have gotten to know Ian through working together, and he is a really good guy*" (*Descriptive Diary: Day 9*).



Figure 5.24. A duty group working together.

The adventure activities also seemed to provide a platform for relationship development. Exposure to risk and adventure during the program challenges required the boys to apply and develop a range of skills, including initiative, teamwork, communication, support, leadership and conflict resolution. For instance, Ahmed mentioned that "*during all the activities, you just learn that people are people, not just some guy at school, but you really get to know their personalities (Descriptive Diary: Day 19).* 

Figure 5.25 shows Group 9 playing around and forming relationships throughout the shared experience of hiking during the first week of the program. This is supported by Noah who explains, hiking on past camps, as well as 'City to Summit' has been the catalyst for him making best friends: I actually love the feeling of hiking because it makes me remember that on all of my hiking camps, I have made some of my best mates just by chatting (Descriptive Diary: Day 15).



Figure 5.25. Peers having fun during hiking.

These program challenges and shared experiences also required the boys to develop trust for one another. An example of a situation requiring group trust was when the boys were required to support and hold onto each other while they were crossing the river so that no one would get swept down the fast current. As the program challenges increased in difficulty over the course of the trip, the need for the boys to trust one another also increased. Tan describes how his group developed skills throughout the program and how they learnt to demonstrate positive, caring behaviour towards one another: I really learnt how compassionate and caring everyone is. I think it's really great that our group is so caring and we all look after each other (Descriptive Diary: Day 19).

However, it is important to note that this type of behaviour did not occur at the start of the program. In the beginning, it was rare for the researcher to observe caring or compassionate behaviour amongst Group 9. Group members appeared to have an individual mindset, only focusing on themselves and what they were experiencing at the time (Observation Diary: Day 1, 2, 3).

As the program progressed, however, and the boys had more shared experiences, groups and individuals appeared to develop more trust for one another and form stronger bonds between group members. The development of these peer to peer relationships seemed to provide a safe, supportive, trustworthy environment which increased the likelihood of the boys talking openly to one another about their feelings. The nightly debriefs became a common, safe and supportive forum for the boys to express themselves without judgement (Observation Diary: 7, 8, 9, 10, 11, 13, 14, 19). An example of a student developing trust throughout the course of the program was provided by Ian from Group 9. During the first week of the program, Ian demonstrated reserved behaviour and did not form many relationships (Observation Diary: Day 9). He would rarely contribute to any of the group discussions, however, one particular night during a debrief, Ian opened up and expressed his feelings to the group like he never had before. He said, *"I need to work on my self-confidence. I don't know if any of you have noticed but I ask way too many questions, and I am not very*  self-confident. I have stayed in a guided boat the whole way down [the river], so yeah I need to work on my self-confidence" (Descriptive Diary: Day 9).

The researcher observed the boys in Group 9 starting to understand how expressing their needs and concerns to the group, made it more likely for the group to be able to help or address any issues. As the program increased in time, the boys became more likely to share their feelings, express caring and empathetic behaviour, and to provide support to one another. This is supported by Isaac's comment below that demonstrates how the boys have changed over the course of the program. From initially having an individualistic approach, to forming an awareness of each other and understanding the strengths of acting in the interests of the group:

It's amazing to see how much everyone has changed in my eyes and first impressions don't always last. Before the trip, I would have said the Year 10 boys at Yarra College are just selfish, immature boys. But now I learned that they are not all like that, and I've proved myself wrong. They're actually caring and really compassionate and actually really care about one another. So that's been awesome to see (Descriptive Diary: Day 19).

The explanation of the following incident observed by the researcher highlights the influence that the environment had on group development and behaviours. By day nine, Group 9 was working exceptionally well together, they had developed strong bonds, and the group demonstrated trust of one another. (Observation Diary: Day 4, 5, 7). By this stage, they had shared many challenging

310

experiences in which the boys became more aware of the importance of group trust, supporting one another and their relationships in the wilderness environment.

However, on day nine, the researcher observed a noticeable change in Group 9's behaviour when they were camped at the football oval in Benambra after the rafting section (Observation Diary: Day 9). This was the only location during the trip where the students camped in 'civilisation' as they called it. As soon as the boy's entered back into an 'urban' setting, they reverted back to their individualistic urban approach that was similar to their behaviour at the start of the program. Specifically, once they arrived in the township, they immediately started yelling, showing sarcastic attitudes and were talking over one another. They became distracted, lazy and were not at all interested in getting their group tasks completed. It was not until their group leader made them aware of their actions that their negative behaviour ceased. Once the students were immersed back in a semi-wilderness environment, their behaviour and group bonds were restored to how they were behaving prior to the arriving in the town of Benambra (Observation Diary: Day 10, 11).

# **Theme Summary**

The relationships theme consisted of three distinct areas: (1) the participants' relationships with the natural environment; (2) the participant's relationships with their leaders, and the roles and impacts of group leaders; and (3) the participant's relationships with their peers. The three main contributing factors that influenced these relationships were:

- The program design, curriculum and learning objectives;
- Being immersed in a novel, wilderness setting away from modern life distractions and;
- The impact of how the group leader applied facilitation and leadership skills throughout the program.

# **Theme 5: Transfer of Learning**

The 'transfer of learning' theme focusses on whether or not the boys were able to maintain the psychosocial attributes and skills that they developed during the program. The theme presents evidence of the application of the various resilience attributes and coping skills that the boys used in various contexts during the six months post-program.

The results of the post-program interviews identified a range of outcomes. Some boys had a strong sense of what they learned from the experience and could describe influences on their current lives, while others were less clear on the skills and knowledge they had learned. However, all boys agreed that 'City to Summit' felt like a stand-alone experience, that everyone was just 'required to participate in' as part of their Year 10 curriculum.

Even though there was a general consensus that boys felt their levels of resilience had increased immediately after "City to Summit', in the post-program interviews sixteen out of the eighteen boys commented that they were not able to maintain their levels of resilience six months later. For example, Will and Tristan comment that: Will: I was similar to Greg. When I got back, we were sort of like, 'yea I'm more resilient!'... And then pretty quickly it went back down to normal...it just went back to just the same. Like, nothing's really changed [after 'City to Summit] (Group B: Interview 2).

Tristan: I don't think 'City to Summit' has made me more resilient. I've seen it as more of an experience rather than something that's like built my resilience... To an extent, it built my confidence, but not significantly (Group C: Interview 2).

Their feelings of not maintaining levels of resilience may have been influenced by the boys not feeling they had been exposed to adversity in the six months post-program. When the boys were asked if they had experienced any significant challenges that they have had to deal with over the last few months at school, none of the boys reported being faced with a situation which had a level of difficulty where they felt as challenged and out of their depth as they were during 'City to Summit'. Such as Elliot, who reports, *"I'd probably say I'm more resilient overall, but I haven't used it" (Group A: Interview 2)*.

In the comment below, Brodie, suggests that the exams at school were the only significant challenge they faced in the six months post-program, and how exam challenges are nothing like the level of challenge that 'City to Summit' presented. His comment also reveals a common finding, that all the boys found it very difficult to associate how they could apply the skills they developed in the outdoor education setting in their everyday lives. It is therefore difficult to report if the boys' levels of overall resilience have been maintained after the program because it appears that they haven't had the opportunities to be tested or required to demonstrate the use of assets and coping skills to demonstrate their resilience capacity:

The exams have been pretty stressful but probably nothing as big as a challenge as City to Summit. Definitely not as physically [challenging] as City to Summit. Mentally, it's a different sort of mental challenge too... Doing homework is a different sort of mental challenge than say getting a big hike done (Group A: Interview 2).

In reference to the boys' personal growth, the most common concept that the boys across all interview groups agreed on, was how difficult it was to maintain the learnings and their feelings of appreciation they initially felt after the program. As Elliot describes,

I definitely agree that the hardest part is to maintain it [what you have learnt]. It takes a really special kind of person to be able to maintain how you feel and what you learnt after the program (Group A: Interview 2).

More than half of the students were able to provide examples of how they felt their feelings had dissipated six months after the program. For instance, James explains how his levels of general appreciation had decreased, and how he had returned to the way he felt and behaved before the 'City to Summit' experience: I sort of appreciated a shower a lot when I got back cause we didn't have one for like three weeks...Like, I had a shower for a ridiculously long time when I got back... But now, I shower normally every day, and it's not something I really have as much like respect or appreciation for, I guess (Group B: Interview 2).

His comment indicates how being in a unique setting, removed from the comforts and privileges of everyday urban life that people become accustomed to, such as access to shelter and hot running water, can cause various levels of appreciation and gratitude. However, it also indicates how these feelings can be lost once the individual is immersed back into their urban environment.

The sections that follow elaborate on and provide examples of the skills and learning the boys transferred from 'City to Summit' and were then able to apply these skills in the lives post-program. The data presented demonstrates that both the learnings that dissipated after the program, as well as the learnings and skills the boys were able to identify that they were continuing to use in their daily lives. The sections are divided into three parts; coping skills, internal assets and external assets.

#### **Transference of Coping Skills**

Participants reported common approaches to managing program challenges as described in Theme 2: Response to Challenge. Their comments identified specific problem-focused and emotion-focused coping strategies developed during the program (see Table 5.5 for common problem-focused and emotion-focused coping skills used). When the boys were asked if they could provide any specific examples of coping skills they used to overcome challenges or difficulties they encountered in the six months post-program, the boys identified five problem-focused coping skills. These were the same coping skills the boys used during the program. These strategies included; 'putting things into perspective', 'removing oneself from the stressor', 'ability to accept (and seek) social support', 'coming to terms with difficulties', and 'chunking'. Intriguingly, not one of the participants reported the use of any emotional-focused coping skills such as 'distraction' and 'avoidance' in the six months after the program.

Evidence of how the boys were able to apply the problem-focused coping skills in their everyday lives is provided in Table 5.10. The boys' examples provide confirmation of how they were able to maintain the use of these five skills in the six months after 'City to Summit'. A common association that over half of the boys made was linking feelings of gratitude and appreciation with the coping strategy 'putting things into perspective' (see Elliot's comment in Table 5.10). This was particularly applicable when the boys were comparing the comforts of their home life, to the ruggedness of living outdoors. Over one-third of the boys identified using other specific coping tools such as, planning or writing in a journal, which could be seen as additional ways to assist people in 'coming to terms with difficulties'. This is highlighted in the comments below.

316

# Table 5.10

Examples of participant's reporting the application of problem-focused coping strategies

Problem-Focused Coping Strategy	Examples of Participant's Comments
Putting things into perspective:	Elliot: There's a thought in your head that we are extremely fortunate and it's like 90% of people
Gratitude and appreciation	or more like 99% of people are a lot less fortunate than us. I'm a person who can get up each day
	and stay positive because I know how fortunate I am. A very small per cent of people might think
	that. But it helped me. It helped me to tell myself that. So, since City to Summit, I have a said that to
	myself a few times, just when you're feeling down or something. Feeling a bit deflated that you kind
	of you realise how fortunate you are (Group A: Interview 2).
Removing oneself from the stressor:	_ James: Yeah, the way I handle thingslike just going into my room and lay down and just thinking
Ability to accept (and seek) social support:	about it. Or maybe writing it [my problems] down, and then I think about it and go over it again.
Coming to terms with difficulties:	And once I'm more comfortable with it, I'll probably talk to someone about it. Like, my close friends
Uses writing as a tool for reflection	and see what they think. I used to think to myself before, but not as much as I do now. Like I did it a hit before, but now I do it a lot more (Crown Pr. Interview 2)
	bil bejore, bul now 1 ub ll d tol more (Group B. Interview 2).
Coming to terms with difficulties:	Cody: Yeah definitely planning cause like I hate homework but I just think like I'll do one-hour
Planning	homework tonight even if it's like on a Friday night. So I'll do one hour then I'll have the whole
1 14111118	weekend. So just get it done and you'll be fine, and you just smash it out, and you don't have to
	worry Before I'd probably sit down for 10 minutes and then be like 'oh. I'll do it on Sunday'.
	Sunday comes, and I'll be up till midnight, 1:00 am, and then I'll be really tired for Monday so,
	yeah (Group D: Interview 2).
Removing oneself from the stressor:	Will: I sort of like I go to my room and or like to go for a bike ride. I like just to get away from it.
C	Chris: [When I was challenged] I went and ran like 10 K's. I justI sort of had to just get away
	from everyone else (Group B: Interview 2).
Chunking:	Iason: I sort of do that now with school [chunking to manage challenges] I ike I'll wake up in the
chunking.	morning and I'd be like 'oh I don't really want to go to school today' And I'll just think to myself
	<i>Well if I just get out of hed, have breakfast, you know, and then get out of the house, get on the</i>
	tram, go to school'. You just sort of break it down. It makes it a lot easier That's one thing that
	'City to Summit' has taught me (Group C: Interview 2).

# **Transference of Internal Assets**

After the first set of interviews, almost all boys predicted that the skills they had developed during the program would be useful and they would be able to apply these skills in their lives back at home. For instance, Oliver suggests that:

It's not just resilience. I think there's a lot of other attributes that you bring back, like determination and persistence. So a lot of those things that you've learned from the camp you bring back so easily so you'll start to persist in certain areas and start to be like more determined about just life in general, to reach your goals and stuff...(Group C: Interview 1).

However, while the boys predicted that these skills would transfer into other contexts, when they were asked if they had 'put any skills that you learnt from 'City to Summit' into practice at school', the boys had difficulty identifying examples, as Elliot summarised, *'I think 'City to Summit' would've definitely built mental resilience but I can't really think of any examples*... (Group C: Interview 2).

There was a general feeling across all the interview groups that they had not been exposed to challenges of the same difficulty as 'City to Summit' and may explain the lack of comments. It suggests that the boys either had not been exposed to challenges requiring the application of these skills or the boys did not recognise how to use these skills in relation to other aspects of their lives. There were two internal assets that were used by the boys in the six months post-program; self-reliance (also referred to as independence) and tolerance. These internal assets are detailed in the following sections.

#### Transfer of self-reliance and independence.

About half of the boys were able to provide examples of how they demonstrated being more independent and self-reliant in their home lives after the program. Whilst, the other half of the boys, did not have feelings of being more independent and felt that their learnings had dissipated in the six months postprogram.

During the program, the researcher observed a spectrum of interactions with the program challenges. At one end of the spectrum, some of the boys relied heavily on their peers and leaders to overcome obstacles, whereas on the other end there were boys who demonstrated high levels of independence and initiative undertaking obstacles on their own (Observation Diary: Day 2, 5, 7, 9, 19). The boys' comments throughout the interviews and researcher observations revealed that there was a continuum of experiences, as well as a continuum of learning (Observation Diary: Day 2, 5, 7, 9, 19). For instance, during the program, some of the boys, especially those who boarded at the school during the term, found tasks requiring self-reliance and independence, easier than others. Whereas, other boys found these daily tasks more difficult, especially the boys who had lived at home (Observation Diary: Day 2, 5, 7, 9, 19). This suggests that the boys in the latter group may have had less experience or opportunities to be self-reliant and independent in their home lives. An illustration of the learning and experience continuum is provided by Cody's comment below. His comment highlights how the program design of 'City to Summit', required the boys to take action and practice daily tasks themselves without the assistance of parents or other adults to help them. He also acknowledges how in his normal life; his parents or cleaner's do a lot of daily tasks for him, therefore decreasing the opportunities for him to develop independence and the ability to learn essential life skills:

I was spoon-fed [before 'City to Summit']. I definitely would say that because, you know, Mum usually makes the bed. You're used to them making the bed, cooking food and stuff like that. But after 'City to Summit', I make my bed every morning... Even when our cleaner comes, and she tries to do it for us, I still do it anyway, just by default. And I appreciate all that stuff, you know. I just do it. I do all the chores without having to be asked... (Group D: Interview 2).

Interestingly, Cody points out that he still makes his bed even if the cleaner or his Mum are there to do it for him. His comment indicates that he has made a conscious decision to do more for himself. Suggesting that this may be because he wants to demonstrate feelings of competency, adding value to his family, which is demonstrated by his actions. He appeared to want to maintain the feelings of self-reliance, independence and sense of competency that he had developed during 'City to Summit'.

For some boys, such as Cody, their experiences in developing independence and self-reliance during 'City to Summit', appeared to have influenced aspects of their daily life after the program. These skills appeared to have been developed through the boys engaging in simple daily tasks requiring them to be more responsible for themselves, such as making beds, cooking food or helping out with chores. Whereas, for other boys who had already been exposed to responsibilities requiring self-care, such as students who make their own lunch and beds each day, it didn't appear to make an impact in their daily living practices. This may indicate that these boys may have had higher levels of self-reliance and independence before the program.

# Transfer of tolerance.

Unlike mental strength, when the boys were asked how their levels of tolerance were since they had been back in their school environment, more than half the boys were able to articulate how the felt their tolerance levels had been transferred into their daily lives and how they maintain their increased tolerance in the six months post-program. Elliot found that:

I've definitely improved tolerating people and their viewpoints and perspectives since 'City to Summit'. I've found that learning tolerance and accepting others opinions or whatever, is like really important, especially in the Bordo (School boarding house) (Group A: Interview 2). In another example, Chris discusses how he has become more tolerant of people in general. Instead of using aggressive and defensive behaviour, he has now learnt to listen to others opinions and take time to process information before acting:

I wouldn't really take anything [conflict or attitude] from anyone before 'City to Summit'. I'd probably just go off at the person, and I just wouldn't take it. So, I learned to just cop it on the chin. And sort it out later, I guess, instead of acting in the moment (Group B: Interview 2).

#### **Transference of External Assets**

The external assets that the boys identified they had maintained were; positive relationships with peers, positive relationships with the environment, appreciation of relationships with family. The next section explores the boys' application and understandings of these external assets.

# Transference of positive relationships.

Boys in all interview groups felt that they maintained the relationships they developed with the people in their group. Reece, for example, says that "every time I walk past one of my group members we always have a little chuckle or a giggle about something that happened on camp" (Group B: Interview 2). These positive relationships seemed to extend to other boys who participated in 'City to Summit', and had the same shared experience, even if they were not in the same group. For instance, Greg feels that "there has definitely been a shift in the way I communicate with them and the way I see them [people in my year level](Group B: Interview 2).

However, there was no indication that the boys were able to apply the skills of relationship building more broadly into their school setting. For instance, James and Chris imply that they would be more likely to speak to the students that did attend 'City to Summit' compared to the ones that didn't attend the program. Their comments indicate that having shared experiences was important for the cohort to not only make friendships but also maintain them:

James: Like, the people who did go [on 'City to Summit'], you can talk to them a lot more easily than the ones who didn't.

Chris: Yeah, we socialise with the kids that we went away with cause you sort of shared something special with them, that you won't share with anyone else... You socialise more with those people than you would've before you left (Group B: Interview 2).

Throughout the interviews, the boys discussed getting to know people and developing relationships with others that they wouldn't usually spend time with. However, this concept of building relationships, awareness of others and being more open, appears not to have extended beyond the program. The boys discussed developing positive relationships with other adults (i.e. leaders and teachers) during the program and described how they are utilising them as an external asset and supportive resource. However, in the six months post-program, none of the boy's commented on developing positive relationships with their teachers once they returned to school or other adults that could be a potential source of support.

# Transference of positive relationships with the environment.

Almost all of the boys in the interview groups commented on how they liked to spend more time outside after 'City to Summit'. Such as Brodie, whose comment indicates that be had previously visited the same place many times before, but after 'City to Summit', his stronger relationship with the environment, had resulted in a heightened sense of awareness and appreciation of nature:

I have a better appreciation of nature now. That's one thing 'City to Summit' taught me. You just look around and take in the stuff around you more. Like, in terms of nature, when I was out at my Grandma's farm walking around. I noticed a lot more, and I took in a lot more in after being on the 'City to Summit' (Group A: Interview 2).

Over half the boys commented on feeling the calming and restorative effects of nature in the six months post-program. Seven of these boys mentioned that they didn't just go outside more because they liked spending time in nature, but they also used it as a method to manage stress and help them to cope with the challenges of everyday life. For instance, Kevin recalls:

I go outside more and play the guitar and things like that to cope. And yeah. I'm still doing that now...I know that's a link to 'City to Summit' like I remember thinking that. And I still do it now (Group D: Interview 2).

#### Appreciation of relationships with family.

The boys' comments varied as to whether they were able to maintain their feelings of appreciation that they immediately felt after the program. A common theme for appreciation of relationships with their families was that they had 'slipped back into a routine' after the program and lacked the awareness of their previous feelings.

Matthew: At the start, I was sort of keen on making a change around the house and like helping my family out and stuff. I did for like the first month or so. I was like helping with the dishes after dinner or something like that, but then as school sort of goes on, you sort of don't have time, and then I was sort of like falling back into my old routine (Group 1: Interview 2).

All eighteen boys commented on the fact that being immersed back into their normal urban environment impacted on their ability to maintain their learnings and feelings of appreciation. Specifically, two-thirds of the boys commented on how difficult it was to maintain their initial feelings of appreciation of their family members. Their comments emphasise the importance of maintaining conscious awareness of learnings. For instance, Matthew commented: I still have like appreciation for what they do [family], but it's subconscious. It's not like I think about it every time my Mum's cleaning or something like that (Group 1: Interview 2).

The boys in the other third felt they had maintained an appreciation of their family as Cody described:

I really appreciate my Dad. Cause like we've always had a really, like really strong relationship ... But when I got back [from 'City to Summit'], I just started helping him out a lot more around the house. Our relationship is getting stronger and stronger by the day I would say (Group D: Interview 2).

#### Participant's Struggle to Make Links Between Contexts

The boys in all interview groups found it easy to provide examples of the various skills they applied and developed in the outdoor environment when they were faced with challenges that were presented on the program (see Theme 3: Personal Development section). Across the board, they demonstrated applying a range of approaches to deal with the difficulties associated with these challenges (see Theme 2: Response to Challenge section). However, the main finding that came out of the 'transfer of learning' theme was how all eighteen boys struggled to make links between the contexts of learning in the program environment and their urban life. On the whole, the boys appeared to view the skills and concepts they learnt during 'City to Summit' in very concrete terms and they weren't able to apply the concepts more abstractly to other contexts of their lives. For example,

Elliot makes a comment immediately after the program, in which he highlights the attributes and skills he felt had been developed during the program. He also acknowledges that he feels these skills are specifically relevant to the outdoor environment and how it would be difficult for him to use these skills in his daily life:

'City to Summit' definitely made me more resilient...That's probably just toward hiking and stuff, cause those bags are very heavy. I don't know if it would affect me in life here [at school] because there's not a huge physical aspect... It's hard to bring that back to urban life I guess (Group A: Interview 1).

He believes the skills learnt during 'City to Summit' consisted more of a physical nature, such as carrying heavy packs. Like most of the other boys, Elliot wasn't able to make links between connecting the concepts of specific coping skills and psychosocial attributes, with the potential of applying the same skills and assets in their daily life. Even though the boys were able to identify that they used specific skills, such as persistence by carrying their heavy pack up the hill, they couldn't see how they could apply persistence approaches to other aspects and challenges in their lives. Six months after the program, Elliot thinks that the skills he learnt during the program will definitely help to build resilience, but he struggled to see how these skills could be used in other areas of life, apart from the context in which he learnt the skills.

Elliot's views are consistent with the rest of the boys' opinions which indicate they feel the skills developed during the program are only relevant to the

327

context in which they learnt them. For example, even though in the first set of interview data, the boys identified the program as mentally challenging, statements made by Interview Group B below indicate that the boys linked the skills learnt to the physical aspects of the program. While Will acknowledges that challenges back at school may require mental strength, the boys' comments emphasise how they struggled to make links between the environmental differences, the different contexts of learning, and understand how they could apply the skills they learnt on the program back in their normal lives:

Will: I feel that was a bit more physical out there [on 'City to Summit'] and not as much down here [at school]. But I guess you'd say there's some correlation like mentally when you say plan out things.

Greg: Yeah, I think the environment we're in there [wilderness context], and the environment here [school context] makes it two very different places...It's hard to relate them [school and the outdoor environment] to each other when they are completely different.

James: Yeah, I agree with the environment thing. It's hard to relate, sort of out there is in the wilderness, compared to back here in the city. It's completely different environments (Group B: Interview 2).

#### **Theme Summary**

During 'City to Summit', it was evident that the transfer of learning occurred with the skills and assets that were developed during the program. Five problem-focus coping skills that the boys used during the program, were also reported to be used to manage difficulties in their urban/home lives after completion of the program; (1) 'putting things into perspective, (2) 'removing oneself from the stressor', (3) 'ability to accept (and seek) social support', (4) 'coming to terms with difficulties', and (5) 'chunking'.

In terms of resilience attributes, the boys were able to provide examples of where they used internal assets such as self-reliance, independence and tolerance in their lives after the program. All the boys commented on maintaining two external assets; positive relationships with their peers, and a positive relationship with the natural environment. In addition, one-third of the boys provided examples of how their feelings of appreciation of their family members flourished, while the other two-thirds commented on how difficult it was to maintain these initial feelings of appreciation of their family members six months after the program.

It was unclear if the boys maintained their initial self-reported levels of increased Overall Resilience, as none of the boys recounted being confronted with any significant events or experiences where they had to draw upon the assets and strategies they had developed during the program. The primary contributing factor affecting the boys' ability to transfer learning from the outdoor education program into other areas of their lives, was gaining an undersatdning the different copcepts of resilience and the contexts of the learning. Developing the skills in the novel program environment appeared to make it difficult to identify how to apply these same skills in a different context. They struggled to see how the skills they were using in the outdoor environment could also be used back at in their urban home environment. Their comments demonstrate how they thought in concrete terms about the skills and concepts they learnt in the outdoor environment, and how they viewed the learning environment of the outdoors and their home life as completely different. They were unable to identify the similarities and differences of the underlying premises and concepts of the coping skills and resilience attributes. Instead, their comments illustrate how they are unable to think more abstractly and identify how they may be able to apply these skills in different contexts or to different problems. The section that follows provides a detailed discussion of the interrelationships between the five major themes of the qualitative data (see Table 5.3).

# Factors Impacting the Development of Resilience and Coping Skills and the Transfer of Learning

The previous sections of this chapter presented the themes that emerged from the qualitative results in Phase II and Phase III. A thematic analysis of this data revealed four elements that directly impacted the participant's capacity to develop resilience and coping skills during the outdoor education program, as well as influencing the participant's ability to transfer their learning into other contexts. The following discussion explores the interrelationships of the five major themes and discusses how the participants engaged with the challenges and interacted with the social and physical context during the program.

# **Transactional Processes of Stress Appraisal**

To present the four key elements that emerged from the data, Figure 5.26 provides a visual summary of the qualitative results, displaying the processes of the participants' responses to program challenges during the outdoor education program. This model uses elements of Lazarus's 'Transactional Model of Stress and Coping' (1984) to compare and contrast the findings of this research (refer to Figure 2.3). Figure 5.26 presents the four elements that have influenced the process of stress appraisal and the participant's response to the program challenges and demonstrates the transactional process of stress appraisal from the first instance of when the participants are presented with the program challenges during the program. This produced the stress(ors) which was followed by their decision-making processes of how they approached the challenges. The process continues into the participant's responses when they were experiencing stressors and program challenges. Finally, the process results in two possible outcomes; either (1) a participant's ability to adapt, or (2) a participant's inability to adapt to the program challenges These four factors have been presented in the model as filters: (1) environment filter; (2) the individual perception filter; (3) the application or non-application of skills and assets filter; and (4) the possible outcome filter. Each of the four stages in the transactional stress appraisal process filters down and influences the next stage of the transactional process. For

instance, a stressor presented in the environment filter will influence the stress appraisal in the individual perception filter.

The results emphasise that individuals reacted differently to the various program challenges, demonstrating that what may be perceived as an unachievable obstacle for one young person, may be the opportunity for personal growth for another. To demonstrate the difference of individuals perspectives, each of the stages in the process has been labelled as a 'filter' to represent the potential differences in an individual's perception. Just as filters are used in photography and on social media applications to show a different lens, view or perspective, the four 'filters' represent an individual's perspective at each of the four stages in the transactional process, demonstrating that each person will have a different frame of reference and experience during each stage.

The four filters are colour coded to clearly show the contributing factors in each filter. To explain this model in further detail, the next sections provide breakdowns of the model using three insets of the filters (see Figure 5.28, Figure 5.29, Figure 5.30). Each of the four filters will be discussed in detail in effects to the impacts and relationships between of the four filters and the five major's themes of the qualitative results.

332


Note. This model incorporates elements of Lazarus & Folkman's (1984, 1987) transactional model of stress and coping.

Figure 5.26. Transactional process of stress appraisal in response to program challenges

### **Environment filter.**

The environment refers to the outdoor education program and context that was responsible for producing the stressors or stimuli, caused by the natural environment and the program challenges to which the boys reacted and responded. The environment filter encompasses the complex interplay between the physical, social and emotional challenges that the environment and program design provided. While participants were able to identify the specific challenges they faced, comments also indicated they often experienced more than one type of challenge at the same time. For instance, all eighteen boys found mountain biking to be the most physically challenging adventure activity during the program. However, the mountain biking also provided social challenges, where the boys were required to show tolerance of others who were less fit or physically able. The physical and social challenges often invoked emotional challenges, which required the application of resilience attributes and character strengths, such as mental strength, determination and perseverance.

The development of these character strengths and assets was observed throughout the program (see Figure 5.19), with data showing the participants varying responses to program challenges, and the variance in individual's capacity to apply various coping strategies. As highlighted in Table 2.4, all 24 character strengths are typically possessed by most individuals to some degree; however, the level of which the strength is utilised to its capacity can be developed (Park et al., 2004; Peterson et al., 2007; Wood et al., 2011). Supported by experimental and correlation investigations, research suggests that individuals who apply their character strengths, such as persistence, perseverance, gratitude and appreciation have reported an increase in their resilience after stressful incidences it (Peterson, Park, & Seligman, 2005. As mentioned in the literature review, prior studies have noted the importance developing these character strengths {Park, 2004), resilience attributes (Wagnild, 2009) and developmental assets (Benson, 2007) to improve the likelihood for young people to experience enhanced health and overall well-being and work as a buffer to assist young people in overcoming problems and managing the stressors of daily life (Park et al., 2004; Peterson et al., 2007; Wood et al., 2011).

Immersion in a wilderness environment (novel setting) away from modern life distractions and normal urban life, provided opportunities for the developmental of internal and external assets. For instance, the unfamiliar environment appeared to provide many benefits, including providing opportunities to compare and contrast circumstances to help them to cope with difficulties. This is supported by Yoshino (2008) who's study showed that over half of participants indicated the main stressors of the program were the activities, the intensity of the natural environment and the novelty of the new experience.

The social, mental, emotional and physical challenges provided by living with a small group in a novel environment while being exposed to challenges, created an ideal platform to form positive bonds and relationships with peers and leaders. These elements are supported by other models and frameworks, such as the adventure therapy theoretical framework that also highlights factors such as

disequilibrium, novel setting, cooperative environment, feelings of accomplishment, processing the experience and transfer as being critical elements to personal growth and the learning process (Gass, 1993a; Nadler, 1993).

One of the most emotionally challenging experience, as highlighted by the boys, was being away from home for three weeks. The time away from their families highlighted how important their relationships with their family members were and how much their parents did for them on a daily basis. It also played an integral role in highlighting and understanding how privileged they were to have access to material things, such as hot running water, a comfortable home and transport, which they usually take for granted. The extended period of time in the novel setting also allowed opportunities for the boys to practice coping strategies, such as 'putting things into perspective', which was the most commonly applied problem-solving coping strategy used by the boys. For instance, when the boys did not have instant access to running water or a home cooked meal, they were required to become self-reliant and source the water or make the meal for themselves. Hence, causing them to 'put things into perspective', to compare and contrast familiar environments with the unfamiliar environment, and therefore, reflect on how lucky they are in their normal lives.

# Level of challenge.

The program design of this extended journey style program supported the development of the boy's levels of resilience through a 'stress inoculation' process which involved creating challenges that gradually increased in the level of difficulty throughout their journey (See Appendix W for the progressive levels of difficulties of adventure activities). This type of programming has been a common element in program design, that aims to provide participants with increasing exposure to challenges and increasing task complexity, which promotes a sense of mastery (Walsh & Golins, 1976). This is supported by Walsh and Golins (1976, p. 7) who state that outdoor education programs that increase "incrementally in terms of complexity and consequence", prove to be more successful as they are progressively sequenced and address the group's needs, skill levels, level of challenge and difficulty.

'City to Summit' created a heightened sense of arousal for participants by utilising the novelty and sense of disequilibrium created by being placed in a semi-wilderness environment. It became apparent that the disequilibrium caused tension between the boys' current reality and their established view of the world and personal capabilities. For instance, as illustrated by the boy's comments they experienced feelings of apprehension and nervousness about the experience before the program started.

Typically, the lessons learned throughout the program encompassed growth in either an individual's behaviour, group behaviour and/or the environment, and occur during the participant's adaption to the state of dissonance presented by the program challenges (Priest & Gass, 2018b). This state of dissonance is the 'anxious difference' between the individual's perception of their current state of being, and their perceived reality of where they would like to be once the problem or challenge has been solved (Priest & Gass, 2018b). For instance, Cooper (Group 9) felt disequilibrium when he was presented with the

jump rock challenge. Cooper didn't feel as though he could jump off the 9m high cliff due to his fears and emotional responses to the challenge. He was stretched beyond his confront zone; however, he had the motivation and support from his peers and leaders and chose to engage in the challenge. The sense of achievement in overcoming his fears and a challenge that he didn't think he could do, *"just felt so good"* (Descriptive Diary: Day 9).

As Yerkes and Dodson (1908) noted, one way to stimulate change in the behaviour of participants is to create situations that produce strong adaptive dissonance. Through repeated exposure to the challenge, as a result, a sense of mastery may occur through the participants successfully adapting to the dissonance (caused by challenging circumstances), which then motivates behavioural change (Long, 2011; Meichenbaum, 2017b, p. 122). Therefore, the gap between the perceived state of dissonance and a feeling of a sense of mastery or competence, is adaption. Adaption was demonstrated by the boys throughout the program when they applied psychosocial attributes and coping skills to help them adapt to stressors. As Brodie suggests, the reason the boys felt the program developed their overall resilience was because the boys were exposed to difficult challenges which extended them beyond their comfort zones, whilst still feeling like they could achieve or be successful in overcoming the challenge:

I reckon it [the program] does build resilience, but I reckon it's only because of the one or two hardest things you do on camp. Like, again with the cycling, you're trying your hardest, it's getting

you out of your comfort zones, but then you know you can achieve it (Group A: Interview 1).

The findings of this research indicate that the level of arousal is a key indicator for participants to experience optimal levels of learning. For instance, the boys frequently spoke about the relationship between the level of challenge and the level of fun, referred to as the 'fun factor', that the boys experienced during the program. This may also be explained as having a 'peak experience' or feeling 'optimal levels of arousal' (Ellis, 1973; Martin & Priest, 1986; Priest, 1999; Priest & Baillie, 1987; Yerkes & Dodson, 1908). This was evident in the data when the boys demonstrated complete engagement and focus on tasks that were appropriately challenging. For example, Greg from Interview Group B suggests there is a sweet spot for individuals, where the 'fun factor' and 'challenge levels' intersect:

There would've been a spot, like, in the challenge level, where there is the difficulty that is just right, and then you would also get as much fun out of is as you can... Like, where the two cross [fun and challenge (Group B: Interview 2).

If the challenge or perceived risk presented to the participants was too high, there is a danger that the participant will experience misadventure, or devastation and disaster (Martin & Priest, 1986; Priest, 1999; Priest & Baillie, 1987) (see the adventure experience paradigm theory on page 88). On the contrary, if programs are designed with the right level of challenge, the 'sweet

spot', that matches the individuals needs in terms of risk and competence, the challenge will require the participant to fully engage, which deepens their experience, ability to learn from the experience and requires them to focus on the present moment (D'Amato & Krasny, 2011).

As Figure 2.6 in the literature review indicates, optimal arousal occurs when a participant's level of arousal and performance are at their peaks (Yerkes & Dodson, 1908). If the levels of arousal are high (over-arousal) and the participants level of performance (competency) is low, programs challenges may create anxieties and lead to misadventure or devastation and disaster. Conversely, if the levels of arousal are low (under-arousal) and the participants' level of performance (competency) is high, this may result in feelings of boredom and disengagement. Therefore, the optimal level of arousal is the midpoint between boredom and anxiety (Ellis, 1973; Yerkes & Dodson, 1908). For example, when the boys felt they went challenged enough and got into camp early, they felt bored and 'cheated' because they hadn't been pushed enough (Brodie, Group A: Interview 1).

Based on the findings of this study, a clear link can be made between the Adventure Experience Paradigm (Martin & Priest, 1986; Priest, 1999), the Yerkes-Dodson law (Yerkes & Dodson, 1908) and the Goldilocks principle. Similar to Yerkes-Dodson law (Yerkes & Dodson, 1908) the Goldilocks principal uses the analogy from the classic children's storybook, whereby the program challenges are designed at a level that is 'just right' for individuals (see Figure *5.27*). To elaborate on this analogy, program challenges throughout the entire program should be designed where individuals are experiencing challenges that are 'not too difficult' (with risk of putting participants in a state of fear or anxiety), 'not too little' (which increases the likelihood of apathy and boredom), but challenges that are 'just right' for individuals, known as the 'Goldilocks Zone' or 'fun factor'. Providing program challenges that sit within the 'Goldilocks Zone', where the difficulty of challenges are perceived to be 'manageable' to 'just manageable', increased the likelihood of participants experiencing peak adventure and optimal levels of arousal. For example, as Brodie's comment above indicates, he felt that it was a combination of getting out of your comfort zone, engaging in challenges that were perceived to be difficult, yet achievable, that helped him to build resilience (Group A: Interview 1).



Level of difficulty of challenge

Figure 5.27. Goldilocks approach to program design.

The current findings that emphasise the importance of challenges that are aimed at the right level of difficulty to promote optimal levels of arousal and peak experience, resulting in participants feeling a sense of achievement, whilst also fostering the development of psychosocial attributes and coping skills, is supported by the results of Shellman's (2009) study. Her study revealed that participants who were tested to their limits or close to their limits (Goldilocks Zone), also experienced a sense of achievement and positive feelings from knowing that they could achieve more than they thought was previously possible. Shellman (2009) also found that resilience was highly correlated with factors including the responsibilities participants had, the successes they felt they achieved, and the decisions they had made. The boys in the current study also felt that when they were persisting with the difficulties of program challenges and when leaders 'let us make mistakes' and allowed the boys to make their own decisions, this helped them to feel a sense of achievement and develop psychosocial attributes and coping skills, such as self-reliance and healthy decision-making skills (Jason, Group D: Interview 1; Brodie, Group A: Interview 1).

### Individual perception filter.

Mainstream outdoor education programs typically apply the concepts of eustress and stress inoculation to challenge participants physically, spiritually, mentally, emotionally and socially with the intent of intrapersonal and interpersonal development of participants (Meichenbaum & Cameron, 1989; Neill, 2008; Selye, 1974).The factors described in the environment filter above contribute to an individual's perception of the stressors and stimuli caused by the introduction of the program challenges. The individual perception filter explains the process of primary and secondary appraisal of the boy's personal perceptions and decision-making process once they were exposed to the program challenges. This is the stage in the process where the participants know that they will be experiencing these challenges in the future but are not currently engaged in physically experiencing the program challenge yet.

In accordance with Transactional Theory (Lazarus, 1984; Wong et al., 2006), when participants are presented with program challenges, their initial interpretation of the stressors (primary appraisal) results in determining what the problem is, whether it is in their control and what internal and external assets (resources) they have to manage themselves to adapt to the stressor (Wong et al., 2006, p. 236). While Lazarus's 'Transactional Model of Stress and Coping' model (1984)(see Figure 2.3) presents the secondary appraisal as a linear process which occurs in response to the initial primary appraisal, the findings of this research indicates the that transactional process of primary and secondary appraisal is not linear, but rather that the participants analysis of their internal and external assets (secondary appraisal) and their interpretation of the stressors (primary appraisal) both interact and influence one another (refer to Figure 5.26). For instance, initially, the boys determined whether the stressor was positive, had relevance or was perceived as dangerous (challenge, threat, harm/loss) during the primary appraisal, which was followed by the secondary appraisal process. However, the secondary appraisal and identification of internal and external assets may change

the evaluation of the stressors presented, resulting in a further secondary appraisal which determined the boy's response options.

When the boys interpreted the difficulty as positive, they tended to express pleasurable emotions, such as happiness, exhilaration and excitement (Lazarus & Folkman, p. 32). As Lazarus and Folkman (1984) suggest, this is when benignpositive appraisals occur. On the other hand, when participants perceived the difficulty as a perceived challenge, threat or something that may cause harm or loss, the boys typically showed emotional and behavioural responses as a result of their fear response, such as anxiety, worry and frustration.

Being presented with the program challenges provided the boys both eustress and distress. This required the boys to go through the transactional process of stress appraisal in responses to the program challenge. Eustress is directly associated with the concept of adaptive dissonance (Walsh & Golins, 1976). This is where the participants are required to adjust their behaviours to resolve their feelings of disequilibrium, the gap between their perceived present and future states. The disequilibrium was caused by eustress, and therefore, to acquire their desired state of equilibrium, participants are required to adapt to the challenging circumstances (Priest & Gass, 2018b). For example, the boys commented on experiencing 'positive stress' (eustress) when they could choose to engage in program challenges that felt like it was 'something new', adventurous and perceived to be risky and difficult, yet just manageable (within the Goldilocks Zone). It was these types of stressors that resulted in optimal arousal, having fun, feeling satisfied, feeling a sense of adventure, accomplishment, and achievement:

Lachlan: I had a couple of most memorable moments. One was when we jumped off that rock cause it was really high. And I got a bit scared as I was standing on top of it. When I jumped off it was a really good feeling (Group C: Interview 1).

Meichenbaum (2017b) uses the analogy, that just like an individual's perception of beauty is in 'the eye of the beholder', so too in an individual's perception of stress. When the boys were determining their resources to approach program challenges, appraisal of internal assets may have included the participants self-esteem, self-efficacy, personal responsibility, physical capacities, decision-making skills, internal locus of control, existential beliefs, interpersonal trust, goals and commitments (Hans, 2000; Hong et al., 2011; Jew et al., 1999). In addition, appraisal of external assets may have included their level of support from peers, their group, the school and other significant adults (Benson, 2007; Benson et al., 2012; Benson et al., 1999; Bronfenbrenner, 1979; Lazarus & Folkman, 1984).

Figure 5.28 presents two feedback loops that occurred during the primary and secondary appraisal of program challenges; (1) the distress loop and, (2) the eustress loop. The distress feedback loop was initiated by an individual's perception that they do not currently have the personal capacity to cope with the demands of the stressors caused by the program challenges. Their perception in their inability to cope is influenced by a range of factors that determine whether they have the insufficient resources of internal assets (e.g., self-efficacy, internal locus of control) and external assets (e.g., peer or adult support). Conversely, the eustress feedback loop is initiated by the range of factors that leads them to perceive they have skills and sufficient resources (internal and external assets) to cope with the demands of stressors (see Figure *5.28*).

As a response to the program challenge, three outcomes were observed in their decision-making process. Participants either chose to; (1) engage in the challenge, (2) change the perceived level of risk or level of challenge or (3) chose not to engage in the challenge. This process was followed by their responses to the challenges, demonstrating their actions and resulting in the outcome of either showing failure or success in adapting to the program challenges.



Interestingly, even if the boys weren't completely sure if they had the skills or the capacity to accomplish the challenge, if they appeared to have a growth mindset and positive attitude, they were more likely to have a positive outcome. This is reinforced by Dweck's (2006) research in growth mindset that maintains young people are more likely to have positive learning experiences when they have a growth mindset. Nussbaum and Dweck (2008) also suggest that individuals with a growth mindset are more willing to try new things and step outside their comfort zones.

In addition, the length of the program supported the development of resilience attributes and coping skills by providing more opportunities to practice skills and develop a sense of mastery of the internal and external assets and coping skills developed throughout the program. The sense of mastery is directly related to the concept of self-efficacy (Bandura, 1977), which suggests that by mastering skills during an outdoor education program, one's perception of their own ability increases, thereby influencing an individual's self-efficacy. This is supported by Davidson (2016) who suggests that programs longer than five days in length can be effective in enhancing personal attributes, such as resilience and grit.

When the boys felt successful and experienced a sense of achievement, this appeared to influence their perception of their personal capabilities, which ultimately strengthened their self-esteem and enhanced self-efficacy. This finding is supported by Zimmerman and Cleary (2006) who posits that when young

people have a positive sense of personal agency and evaluate their success in achieving a challenging goal, their self-esteem and self-efficacy will be enhanced.

Following the primary and secondary appraisal interaction, participants typically had either one of two responses; a

- perceived sense of potential failure due to insufficient resources of internal and external assets, or
- 2. a perceived sense of potential success due to sufficient resources of internal and external assets (see the Inset 1 in Figure *5.28* for further explanation).

The participants perceived failure or success in their own abilities to adapt to the program challenges, therefore impacted the boy's decision-making process.

### Application or non-application of skills and assets filter.

As presented in the Individual Perception Filter (see Figure 5.28), the cyclic processes of the primary and secondary appraisals of stressors/stimuli continue as an ongoing process when participants are exposed to new challenges or situations where they perceive something to be difficult or a threat. Once the boys had decided on their initial response to being presented with the program challenges, and they had decided to engage in the challenge, the boys would then physically engage in the program challenges (stressors). This filter demonstrates the stage of the process in which the boys responded to the program challenges, showing either the application or non-application of skills and assets.

When the boys were experiencing the stressors caused by the program challenges, they responded in a myriad of ways. Figure 5.29 presents the range of

responses the boys identified as skills and assets that they used in response to the program challenges, which resulted in demonstrating their resilience capacity and ability to adapt to the challenges.



*Figure 5.29.* Inset 2: Application or non-application of skills assets filter.

## Application of coping skills.

Copings skills were demonstrated by the behaviours, thoughts and actions taken to assist participants in adapting to the program challenges. The process of continual exposure to stressors allowed the boys to practice these skills through the stress inoculation process. Interestingly, the results indicate that the boys found 'City to Summit' to be more mentally challenging than physically challenging. While most of the adventure activities presented physical challenges, these challenges caused varying emotional and behavioural responses which required the boys to apply problem-focused and emotion-focused coping skills (see Figure *5.29*).

Yoshino (2008) revealed that certain types of stress appraisal predicted certain types of coping strategies employed by participants. For example, the higher the level of difficulty of the challenge, the more participants would apply problem-focused coping skills. Yoshino's (2008) results are consistent with this study, revealing that boys predominantly applied problem-focused coping strategies, such as active coping. 'Active coping' involves either cognitive or behavioural responses that either change how one thinks about the stressor or the strategy changes the nature of the stressor itself (Baker & Berenbaum, 2007; Carpenter, 1992). People who are exposed to more personal and environmental resources are more likely to rely on active coping skills and less likely to use avoidance coping strategies (Holahan & Moos, 1987).

Active coping required participants to recognise the problem and take specific targeted action to help minimise the stress that was caused by the challenging situation. These specific efforts helped the boys to tolerate, reduce, or minimise the stressful circumstances. For example, James talks about his writing in a journal and reflecting on his thoughts as an active method of coping to help manage stress. He also recognises the use of other coping strategies, such as seeking social support from his friends and awareness of removing oneself from the stressor. In this case, he removes himself from the stressful circumstances and seeks solitude. This response is supported by Yoshino (2008), who found that social support and problem-focused coping were the most effective coping mechanisms. Her study also indicated that neither gender or perceived outdoor experience levels were significant factors in determining the individual types of coping styles that participants used. Yoshino (2008) also found that the types of coping strategies that were most used in responses to challenges were social support (31%), followed by problem-focused (16%), less-constructive (12%), and self-distractive (5%) coping skills.

Opposite to active coping strategies are avoidant coping strategies. Avoidant strategies impacted the boys directly by addressing the circumstances that were causing the stress. Distraction, a coping strategy used frequently by the boys, may be seen as a type of avoidant strategy. For instance, Holahan and Moos (1987) suggests that avoidant coping strategies are markers for adverse responses to stressful life events which may cause issues in the future. However, in the context of short-term difficulties, the boys have demonstrated that distraction, such as singing or talking to the person next to you while hiking, was a healthy

and productive method of coping and helped the boys manage the challenges they were faced with during the program.

It was evident that when the boys were continually experiencing phases of disequilibrium due to the physical, social and emotional challenges that were experienced daily, their emotional responses triggered them to search for and apply more effective coping strategies to manage their responses to the difficulties. For example, when the boys were observed applying the coping strategy of 'coming to terms with difficulties', the boys firstly acknowledged the situation they found themselves in. Secondly, they became aware of the stressors that were triggering their emotional responses, and then they would take actionable steps to manage the stressor or their emotional responses to the difficulties presented. Such as when Group 9 understood that they could not remove themselves from the inclement weather, they addressed the groups emotional responses (e.g., fear, worry, negative attitudes) and took action to cover themselves with a tarp to protect them from the wind and rain (Observation Diary: Day 17).

Participants used a range of ways to 'put things into perspective', such as comparing past experiences and levels of difficulties of challenges, using time as a comparison, comparing environments or past failure as a motivator to help them manage their responses to program challenges. For example, participants utilised their 'sense of achievement' from achieving past program challenges as a comparison point for other challenges (e.g., feeling the sense of achievement of achieving their goal of hiking to the top of Mount Misery, motivated the boys to

persevere and climb to the top of Mt Kosciusko). It is critical to point out that if the participant had not been exposed to prior challenges, they did not have a personal comparison point to draw on, thus making it very difficult or impossible to apply this type of problem-focused coping. This demonstrates the importance that the role of past experiences has on helping the participants to put things into perspective and compare and contrast their past responses and lessons learnt in previous challenging experiences. This highlights the need for young people to be continually provided with opportunities to be challenged, in a variety of challenging situations, with opportunities to acquire and practice skills, and be able to reflect and refine their learnings in a supportive, collaborative environment.

The coping skill of 'cognitive reframing and applying positive thinking' that was applied by the boys is also consistent with having a 'growth mindset'. When the boys engaged in a 'growth mindset' and having a positive attitude, they were able to move from the perception of thinking "why me, why should I have to do this" ('victim' mindset) to having a positive learning experience through engagement in and becoming successful in adapting to the challenge. For instance, when Oliver (Group C: Interview 1) notes the importance of keeping a positive attitude and having a growth mindset to manage his emotional responses to program challenges. This is supported by researchers in the area of 'growth mindset' (e.g., Bennett et al., 2013; Dweck, 2006; Nussbaum & Dweck, 2008; Payne et al., 2007) who support the notion that people who have a 'growth mindset' are more willing to engage in challenges, have new experiences and

venture outside their comfort zones, and as a result they experience personal development. However, while both emotion-focused and problem-focused coping skills played an important role in the participants effective responses to the program challenges (Lee et al., 2017), the use of problem-solving skills and problem-focused coping strategies should be promoted for effective, long term coping solutions that can be transferred and used in other contexts (Breinbauer & Maddaleno, 2005).

The results of this study demonstrate that the boys' coping mechanisms provided an advantage when responding to program challenges, by acting as a buffer between the individual's perceived stress (state of dissonance) and the personal development of the individual (development of resilience capacity and other skills). These results are supported by findings from Yoshino's (2008) study indicate that in most cases coping responses act as an 'asset' and can also be used as 'buffer' to assist individuals in the process of negotiating initial perceived stress and challenge, to moving towards an individual perceiving psychological growth.

# Application of internal and external assets.

During the program, the boy's developmental assets acted as protective factors, which increased the likelihood of positive adaption to the program challenges (Benson, 1997, 2007; Benson et al., 2012; Benson et al., 1999). For instance, some of the interviewee's pointed out the importance of being 'mentally tough' or having 'mental toughness', which is also known as psychological resilience (Booth & Neill, 2017; Lin et al., 2017; Neill & Dias, 2001). The participant's ability to cope with the program challenges was directly linked with their ability to draw upon their internal and external assets. In particular, positive supportive relationships, including the relationship with themselves, their peers and the group leaders. It was evident that these relationships provided the support, trust and collaboration required to assist in overcoming program challenges. For instance, when Matthew relied on the support and trust of Elliot to help him overcome his fear of heights during the jump rock challenge (Group A: Interview 1).

One of the three key contributing factors that influenced the boy's abilities to develop positive relationships with; their peers, leaders, and the natural environment, was the immersion in a novel, wilderness setting away from modern life distractions (e.g., smartphones, computers). As the boys highlighted, society relies so much on technology and it's difficult to get away from it (Group C: Interview 1). The substantial distance from the urban setting of their everyday lives meant that instead of using their time being distracted by technology (e.g., engaging in social media applications on their smartphones), they had the time and the circumstances to make real, face to face connections and focus on what they were doing (Brodie, Group A: Interview 1). This result is not surprising, as one of the goals of the program was to immerse participants in new, unique and challenging environments that help to create disequilibrium and foster the development of interpersonal and intrapersonal relationships.

Central to the development of positive relationships, was the leader's ability to create a safe, cooperative and supportive learning environment. The

study revealed that the group leader's role was significant in this learning process, as it was their responsibility to guide the learner through experiential learning processes. In supporting this notion, Roehlkepartain et al. (2017) found that when young people experienced strong developmental relationships with leaders, participants were more likely to show a clear sense of community identity, social responsibility and conservation leadership. In addition, Roehlkepartain et al. (2017) also found that adolescents who reported high levels of developmental relationships with caring adults (e.g., teachers and leaders), were eight times more likely to persevere with tasks that were challenging. The participants also enjoyed the hard work of challenges and felt that it was ok to make mistakes, compared to students who reported low levels of positive developmental relationships (Roehlkepartain et al., 2017).

All interviewees felt the cooperative and supportive learning environment created by peers and leaders, supported the development of positive relationships with one another. The boys striving together to overcome the challenges presented during the program with the support of the leader helped to develop trust in one another. The boys identified that through this process, they valued different aspects of each other which they had not previously noticed back in their school environment. They began to rely on one another in new ways and support one another through difficulties. These types of supportive relationships are crucial developmental assets and help young people to nurture the desire and capacity to thrive (The Search Institute, 2014, 2016).

It was clear that the function of the social group played a key role in participants' responses to challenges. The challenges tested participants and required individuals and groups to work together to achieve their desired outcomes. They were required to rely on each other to help them overcome program challenges (e.g., whitewater rafting down the Mitta Mitta River or setting up group shelter), or to help them 'put things into perspective', by comparing past experiences they have had together. Having these shared experiences where they were able to help one another, develop trust, and overcome programs challenges together, strengthen the bonds between the group members. Priest (1998) supports the importance of fostering group trust in the development of relationships, stating that "the acquisition and maintenance of trust is critical to the success of a team and goes hand in hand with cooperation and communication" (p.31).

It was the social interactions with peers, and the leader's ability to help facilitate reflection and awareness activities, that assisted the boys in making connections between the group and their environment. This is consistent with constructivist philosophies that advocate that an individual's knowledge and learning evolves through social negotiations with their interactions with others and their environments (e.g., meso-systems interactions) (Bronfenbrenner, 1981; Savery & Duffy, 2001; Ungar, 2012). This is also supported by outdoor education philosophies and program design which has long followed this constructivist approach aiming to create an environment that is intentionally supportive, collaborative and open to support the participants learning experiences and

promote the personal development of the participant (Sibthorp, 2003; Walsh & Golins, 1976).

'Shared experiences' became a key contributing factor to developing positive relationships in a supportive environment to foster the development of resilience. For example, if a participant had a shared experience with someone else, regardless of the fact that the other person was in their group or not, they were more likely to engage in a positive relationship back at school. This finding is reinforced by MindMatters (2018a) who suggest that resilience develops best in supportive environments that provide opportunities for students to develop, have shared experiences and generalise their skills.

Developing positive relationships with the environment was also an outcome relevant to having shared experiences. For instance, if an individual shared a positive experience with the environment, felt a sense of gratitude through understanding and comparing their privileges of urban life or seeing the beauty in nature, they were more likely to develop respect and care for nature.

### Unable to apply skills and assets.

Not all the boys showed the ability to adapt to the program challenges at all times. Distress was the result of the level of difficulty being so far along the spectrum of challenges that some boys perceived the difficulty to be out of reach of their current capabilities. At times boys demonstrated an inability to cope with the challenges that were perceived to exceed their level of competence. This resulted in some boys experiencing misadventure. The outcome was that some participants were evacuated off the program because they were experiencing psychological issues, including homesickness and high levels of anxiety (Observation Diary: Day 21). In this case, these boys were unable to cope with the program challenges, reverting back to their comfort zone by being removed from the program and taken home.

Examples of boys not coping, but the situation not leading to misadventure also occurred. This was seen during the first week of the program with participants in Group 9 displaying countless examples of not applying coping skills to manage the social and emotional challenges of working together in a small group setting, leading to group conflicts (Observation Diary: Day 1, 2, 3, 4). Although these conflicts were the result of poor coping, opportunities also arose for learning when the program was effectively managed by the leader. These results are in line with findings by Yoshino (2008) who found that participants who did not apply active coping skills and felt a greater sense of perceived stress and threats during participation an outdoor education program, were at less likely to report psychological growth. The results of the current study, taken with the results from Yoshino's (2008) study indicate the importance of the leader's role in tailoring the program challenges to be aimed at the right level of difficulty for optimal arousal and to foster and support individual's development of psychosocial assets (see Figure 5.29).

## Possible outcomes filter.

The last filter in the transactional process of stress appraisal in response to the program challenges is the possible outcomes filter presented in Figure *5.30*. This filter discusses the process of how the actions taken by participants to apply

or not apply skills and assets influenced their responses to the program challenges, which results in individuals experiencing either one of two outcomes; (1) success in adaption to program challenges or, (2) failure to adapt to program challenges. For instance, some boys demonstrated success in adapting to the jump rock challenge, by choosing to engage in the challenge and applying coping strategies, such as chunking to manage their response to the stressors. Whereas, other boys demonstrated a failure to adapt to the same challenge by choosing not to engage in the challenge, as their perceived risk outweighed their perceived competence, causing them to revert back to their comfort zone.

### Exploring the interaction of elements that lead to adaption.

To explain the process from the initial stage of when the boys were presented with the program challenges, to the final stage that represents the possible outcomes of either perceived success or failure to adapt to the program challenges; the transactional process of stress appraisals has been separated into layers within concentric circles. The process firstly starts at the individual's 'core' layer (inner circle). The subsequent layers expand out from the inner circle and represent the different layers that impact the individual's possible outcomes in response to the program challenges. The subsequent layers are represented in order as the (a) conditioning layer, (b) feelings layer, (c) choices layer and (d) actions layer. These layers are permeable and build upon and influence one another. At each of the layers, there is an option for participants to continue in the progress to the next layer, or there is an option at each layer for individuals to hit their 'limit line'. If the individual does not have the skills, personal attributes or

support required to continue through the stress appraisal process, the reaction and consequence is that they generally hit their 'limit line and' retreat back to their comfort zone. To explain this transactional process in detail, the following sections will discuss each of the influencing layers of the process.

## The core.

The core inner circle of the process, represents an individual's state of being, encompassing their core beliefs, current knowledge and skill sets (assets and competencies). The core layer represents an individual's level of functioning which impacts their psychological, neurobiological and behavioural responses to challenges (Garmezy, 1985; Werner, 1995, 2000; Windle, 2011). The results showed that the following variables within the core layer influenced the student's ability to learn and retain knowledge;

- previous experience,
- psychological/emotional state,
- level of tiredness,
- personality,
- fitness levels, and
- personal skills and capabilities (Observation Diary: 2, 5, 10, 19).

This is supported by a constructivist approach to learning, that suggests participants come to the program with an abundance of prior knowledge, skills and experiences, consequently, affecting their approach to program challenges (Gilbertson et al., 2006, p. 29).

The conditioning layer.

Expanding out from the 'core', the next layer represents the 'conditioning layer'. This layer demonstrates that an individual's perception and approach to a challenge is influenced by their personal conditioning. Their conditioning is the impact of their environmental microsystems, including peers, family, school, culture and their community (Bronfenbrenner, 1977, 1981). In accordance with Bronfenbrenner's (1977) ecological model, the program process demonstrates the multifaceted interactions between an individual's psychological and social-cultural systems when they are confronted with a challenge and how each of the different layers and factors interacts (see Figure 2.4).

All individuals have a different 'core' and 'conditioning' layers, based on personal circumstances. The core and conditioning layers also represent a participant's comfort zone. This is the area in which an individual feels comfortable, safe and unchallenged (Luckner & Nadler, 1997; Senninger, 2000). Only after the boys were exposed to a stressor or challenge, did the feelings layer come into play in the process.

# The feelings layer.

The feelings layer represents the potential psychological state of the participant. In other words, this layer represents how the boys were potentially feeling emotionally and mentally in response to being presented with the program challenges. This layer involves participants distinguishing how they 'feel' about the challenging circumstances, in order for them to make a decision about how they will respond. It is important to note that each individual will perceive the

challenge level differently depending upon various factors which make up their core and conditioning layers.

The boys' psychological states were found to affect their ability to apply or learn skills and were influenced by a variety of social, biological, environmental and psychological factors (Friedli, 2009; Herrman et al., 2011). For example, when Reece threw a tantrum because he was angry, frustrated and bored, he was unable to focus on the needs of the group when he was in a leadership role (Group B: Interview 1).

The challenges required the boys to cognitively manage internal emotional and behavioural responses. Depending on the circumstances of the challenges and the individual, the stressors caused a range of feelings and elicited emotional responses, such as fear, anxiety, excitement, anger or frustration. For instance, Isaac's emotional response was crying five times in response to the perceived difficulties (Descriptive Diary: Day 19). This is supported by Csikszentmihalyi (1996) who suggests that an individual may experience a variety of eight different mental states when encountering a challenging situation. These psychological states include anxiety, apathy, arousal, boredom, control, relaxation, worry and flow.

Participants who were able to regulate and apply purposeful control over their emotional responses, behavioural impulses and cognitive processes within the feelings layer, were seen to be demonstrating executive functioning (Bradley, 1990; Lee et al., 2013). For instance, when Jake (Group C: Interview 2) acknowledged that at the start of the program his emotional and behavioural response to their broken tent would have been feeling '*really angry, frustrated, and screaming*', whereas, he recognised that his actual responses (e.g., laughing and seeing the positive side to the difficulty) were due to developing resilience and coping skills throughout the program.

Experiencing feelings of 'disequilibrium' caused some of the boy's feelings of fear and uncertainty about how they would cope with the various program challenges they would encounter, causing a momentary internal state of conflict. However, as Gass (1993a, p. 59) suggests this type of eustress provokes healthy levels disequilibrium, and it must be present for change to occur through adventure experiences.

As previously discussed, the level of challenge is critical in the decisionmaking process. If the individual is feeling over-aroused, they may feel emotions associated with distress. On the other hand, feelings of boredom may be associated with feeling under-aroused. Again, this reinforces the importance of the level of difficulty of challenges being set at an optimal level of arousal (Goldilocks Zone), resulting in eustress.

# The choices layer.

Both the thoughts and emotions of participants shaped their learning process. These feelings ultimately impacted their choices and decisions of whether to confront or retreat from the challenge. Based on their feelings, the boys made decisions. This is known as the choices layer. As Figure *5.28* demonstrates, the boys decided one of three responses to the program challenges;

1. engage in the challenge,

- 2. change the perceived level of risk or challenge, or
- 3. not to engage in the challenge.

The boy's choices were impacted by a range of different underlying thoughts, beliefs and values, including self-image, self-efficacy, self-talk and their perception of the risk. As discussed, their choices are also influenced by determining whether or not they have sufficient internal and external assets to overcome the difficulty. The choices layer is also influenced by an individual's awareness and conscious decision-making processes, which can also be affected by their goals and intentions.

## The actions layer.

Actions and behaviours were typically modified by the choices the boys made. It is at this stage in the transactional process of stress appraisal the boys take action (action layer) based on their decisions through the interactive process of the primary and secondary appraisals. This process of transactional stress appraisal is highlighted in both the model demonstrating the transactional process of stress appraisal in response to program challenges, as well as the application or non-application of skills and assets filter (see Figure *5.29*)(Carver et al., 1989; Lazarus, 1984; Lazarus & Folkman, 1987; Wong et al., 2006).

Once the boys had made the choice to take action, it is in this layer that the boys experienced exposure to the actual challenge. This is the area that required the application of character strengths, resilience attributes, coping skills, and specific actions and behaviours that assisted individuals in the process of adapting to the program challenges (see Figure *5.29*)

The feelings, choices and actions layers often moved individuals from their comfort zone (core and conditioning layer), into the groan and stretch zones where they experience being stretched and challenged beyond levels of comfort, resulting in the potential for personal growth and development of assets and coping skills (Luckner & Nadler, 1997; Senninger, 2000). The growth zone is where participants move from the comfort zone, through the groan and stretch zones, into the growth zone where personal growth occurred (Luckner & Nadler, 1997; Senninger, 2000).

### The breakthrough line vs. edge line.

To clarify the boy's transactional process of stress appraisals during 'City to Summit', Figure 5.30 presents the entire process; demonstrating the possible outcomes filter and explaining the concept of the breakthrough line vs. edge line. In Figure 5.30, the edge of the 'actions layer' is the areas where an individual will either hit their 'edge line' or their 'breakthrough line'. Past the breakthrough line is an outer layer which is broken up into four quadrants (refer to Figure 5.30). These four quadrants represent the 'possible outcomes'. The top two quadrants represent the area that demonstrates success in adaption to the challenges, while the bottom two quadrants represent failure in adapting to the challenge. These outcomes were largely dependent on the following factors:

- 1. Level of difficulty of the challenge.
- The willingness for an individual to engage in the challenge and not be stopped by their limit line, causing them to retreat back to their comfort zone.
- 3. Whether the individual felt supported by their peers and leaders.
- Their current state of being (core): knowledge, beliefs and skill sets (competencies and assets).

When a participant was able to draw upon their own skills and assets to confront and overcome the challenge, they generally passed through their 'breakthrough line'. The 'breakthrough line' represents the potential positive consequences which may result in outcomes, such as personal growth, the development of skills and assets, and experiencing the success of desired outcomes. It is at this point where the participant typically experienced a sense of achievement and a sense of adventure.

On the contrary, participants' personal growth may also be hindered by the challenge being too difficult, resulting in individuals hitting the 'edge line'. The 'edge line' is where the level of risk and level of difficulty of the challenge exceeds an individual's resilience capacity and levels of competency to overcome the stressor. If individuals pass through their edge line, the outcome may not only be a negative experience but also may result in 'misadventure' or 'devastation and disaster'. Most of the time the boys experienced psychological states represented in the top two quadrants of the diagram, where the challenges were either within a comfortable level, or just above manageable and within their 'Goldilocks Zone' (see Figure *5.27* and Figure *5.30*).



Figure 5.30. Inset 3: Possible outcomes filter and the breakthrough diagram.

Participants who experienced moving through the breakthrough line in the top two quadrants experience positive learning experiences and personal growth (see Figure 5.30 for examples). The process of passing through the breakthrough line resulted in success in adapting to the challenge. This typically occurred when;

- individuals experienced a challenge that was at the appropriate level of difficulty;
- they felt supported by their team and leader,
- the boys received coaching and mentoring from leaders; and
- they processed the experience often with assistance from the group leader (e.g., reflective experience or debrief).

In these cases, they were more likely to pass through the breakthrough line instead of hitting their limit line and turn back to their initial state of being (comfort zone).

This outcome indicates that the process of the boys moving through the breakthrough line may be accelerated by the support and coaching from the leader. For example, when group leaders were observed facilitating challenges and helping participants to reframe their thoughts, such as presenting a 'positive thinking model', the boys were more likely to have positive results and breakthrough their limit lines at each of the levels of the layers (refer to the 'positive thinking model' in Figure *5.3*). In order for the boys to experience psychological states in the top two quadrants of this model, there must be challenges presented that are engaging, are progressively more complex throughout the journey and suited to the individual's needs, aiming for optimal

levels of arousal and a peak experience (see Figure 5.30). Participants who experience an optimal level of challenge were stretched to learn more, increasing their personal development and become more competent at applying and practising copings skills and developmental assets (see Figure 5.29). A prime example of such a program challenge was when the boys were presented with challenge which enabled them to 'choose' their level of engagement was the jump rock challenge. Here the boys utilised a range of coping skills (e.g., positive selftalk, chunking and goal setting), external assets (e.g., support from peers) and internal assets (e.g., mental strength) to help them in successfully adapting to the challenge (Observation Diary: Day 6). The boys commented on this challenge not only being one of the most memorable moments of their experience but also it was a fun type of challenging experience.

Many of the boys hit their 'limit line' as they perceived the level of difficulty and risk of the challenge too much to manage, therefore retreating back to their comfort zone and choosing not to engage in the challenge. Interestingly, regardless of whether the boys chose to engage in the most difficult level of jumping off the top of the 9m rock, or whether they changed the challenge height to suit their individual needs, the outcome for all boys who engaged with the challenge was the same. All boys indicated that they were proud of themselves for either completing the challenge or overcoming their fears, resulting in them feeling empowered and having a sense of adventure and achievement. This outcome is supported by the results of Shellman's (2009) study showing that empowerment was highly correlated with factors such, the opportunity to develop and practice skills, the responsibilities participants were given and the help the participants received from the rest of the group, as well as help from group leaders.

The combination of both this study's findings and the results of Shellman's (2009) study indicate that outdoor education programs should aim for program challenges to be within the 'Goldilocks Zone' to ensure the appropriate level of challenge for individuals, as it is the perceived level of difficulty that is a crucial element to achieve the outcomes of personal growth and the enhancement of psychosocial attributes and coping skills. As the freedom to choose has been found to directly relate to one's ability to experience personal growth (Rohnke, 1989; Wallia, 2008), critical to adaptive dissonance is the need for programming to ensure challenges and choices are individualised to suit the participants' needs and include the freedom for them to choose their level of engagement, thus increasing their personal motivation and empowerment to engage with the challenge. In summary, the 'breakthrough diagram' highlights the significance of intentional program design, where levels of challenge are appropriate for individuals, that are aimed towards achieving optimal levels of arousal, to foster the development and transfer of resilience and coping skills through outdoor education programs.

#### **Transfer of Learning into Different Contexts**

The foundation and theories of resilience development relies on the concept of the transference of learning, where the understanding is that the skills, knowledge or behaviours that are developed in one context can then be transferred, drawn upon and applied in various situations and environments

(Beightol et al., 2009; Ewert & Yoshino, 2011; Gillespie & Allen-Craig, 2009). The topic of learning transfer from outdoor education programs into other contexts continues to be one of controversy (Brookes, 2003a; Brown, 2010; Sibthorp, 2003). As Brown (2010) argues, the concept of the transference of learning is highly problematic, and it is difficult to empirically validate beyond controlled experimental settings. It remains a consistent challenge of outdoor education programs to measure the extent of which the learning experiences of participants affect change beyond the immediate context of the outdoor environment (A. J. Martin & Leberman, 2005). The results of this study also indicate that the transference of skills is highly problematic. For instance, it could not be determined if the boys maintained their initial self-reported levels of increased overall resilience, as none of the boys commented on being exposed to specifically challenging opportunities to be able to practice the skills they had developed. As a group, the biggest challenge that they had post-program was either their exams or sporting activities.

Although, after the program, the boys were able to provide some examples of how the psychosocial attributes and coping skills that were developed during the program were maintained and applied in the six months post-program. These included examples of the application of internal assets, such as self-reliance, independence and tolerance that they applied in their everyday lives. Specifically, the boys identified the use of the same five problem-focused coping skills that they used during the program, to help them manage difficulties in their home lives after the program: (1) 'putting things into perspective, (2) 'removing oneself from the stressor', (3) 'ability to accept (and seek) social support', (4) 'coming to terms with difficulties', and (5) 'chunking'. These results indicate that outdoor education programs can have positive influences on the development and transference of resilience attributes.

Immediately after completion of the program, the boys had a 'sense of achievement' and feelings of relief that that had survived 'City to Summit'. Their initial comments indicated that the program needed to be more challenging in order for them to develop more resilience. However, comments in their follow-up interviews that inferred that the level of difficulty of the program challenges should not be changed and that the program challenges should be specifically catered for individuals. These results demonstrate that their initial feelings may have been attributed to groups experiencing 'post-trip' euphoria. This is the time directly after the end of the outdoor education program that can cause participants to feel a sense of euphoria, which has been known to cloud their sense of positive growth and learning that can potentially occur as a result of participation in the program (Ewert & Yoshino, 2008).

In the new school year, the boys tended to forget what they had learnt during the program and slipped back habits of their everyday lives. Without intentional follow-up, bringing attention to what they had learnt or opportunities to practice their resilience and coping skills, it appeared that their learnings dissipated. Opportunities for follow up programs or interventions post-trip which recall the learning experience may assist in the enhancement the learning transfer (D'Amato & Krasny, 2011). Griffiths (2011) also noted that it is important for the student's regular teaching staff to be present on the program to assist in the transferal of learning.

One critical finding of the qualitative data was the participants' inability to think abstractly and draw connections between their learnings within the outdoor education program context, and the context of their school lives. The boys specifically commented that their learnings felt 'unconscious'. This finding emphasises the importance of not just using self-reflective practices, such as the first generation of facilitation, known as 'letting the mountains speak for themselves' (Priest & Gass, 2005). These types of facilitation methods do not appear to support individuals when it comes to conceptualising coping skills and resilience. Outdoor education companies and practitioners can assist in the development of coping skills by explicitly teaching participants a range of problem- and emotion-focused coping skills, and to promote self-awareness of these skills during all three phases of the program design.

When participants are not assisted in the reflection and integration process post-program, there is a high risk that the experience will be viewed as a standalone experience, as most participants did in this study. If the intention of the program is to transfer learning from the context of the outdoor education program into other contexts, then the results of this study indicate that intentional program design during the Integration Phase can foster the students' long-term development when:

- Students are supported in debriefing and facilitating activities enhance the process of abstract thinking and making explicit connections between contexts of learning.
- Follow-up experiences which recall their learning experiences during the program.

- Opportunities to be exposed to appropriate challenges in other contexts.
- Opportunities for mastery and to practice the skills developed during the program.
- Goal setting and reflection activities for future personal growth.

### **Chapter Summary**

'City to Summit' used the process of stress inoculation and the concept of 'development-by-challenge' to gradually expose the boys to challenges and stressors throughout their journey. Program challenges provided physical, emotional and social challenges, which required the application of problemfocused (e.g., 'putting things into perspective') and emotion-focused coping skills (e.g., distraction). Along with coping skills, the boys developed internal assets (e.g., self-reliance, mental strength, perseverance) and external assets (e.g., positive relationships with peers, leaders and the natural environment) to help them manage their responses to difficulties during 'City to Summit'. The findings highlight the importance of participants positively experiencing challenging situations, whilst in a supportive learning environment, with both peers and leaders. The length of the program and program design of the Immersion Phase helped provide opportunities for the young people to practice (develop mastery) of these skills and help them to understand their personal responses to stressors, by creating challenges where the participants experienced eustress, rather than distress.

In particular, the boys made a clear link between the concept of the relationship between challenge and fun. Emphasising the importance of specifically tailored challenges for individual needs, whereby challenges for individuals are aimed at a level for optimal arousal. Based on conceptual understanding of primary appraisals in Lazarus's transactional model of stress and coping Lazarus (1984), outdoor education programs can aim to support psychological growth by creating situations during program implementation which create relevant challenges that generate a perceived level of difficulty, risk or threat; that also support participants to feel they are in a safe, supportive environment, to trial, practice and master coping skills in response to the program challenges. Outdoor education programs should therefore aim to be flexible in design and tailored to meet the specific needs of groups and individuals, with challenges aiming to be within the 'Goldilocks Zone', where optimal arousal and peak adventure is the goal of the participants' experiences.

Post-program, interviewees were able to identify resilience attributes that they applied when confronted with program challenges, such as independence and determination. However, apart from 'putting things into perspective', they were not able to explicitly identify the coping skills they used when confronted with program challenges. The boys were able to talk about what they did when they were challenged, but they were not able to identify the specific types of coping skills they used.

The boys found it difficult to draw connections between the environments of the context in which they developed the skills and the various contexts of other potential applications for the skills (e.g., outdoor education context of learning vs. the application of skills in the context of their school lives). The boys processed experiences using concrete thinking and struggled to make links between the different contexts of learning which demonstrated their inability to apply abstract thinking. The group leader's role in facilitation, debriefing and reflection activities became crucial in helping the boys to make connections between the different contexts of learning, therefore, highlighting the need for leaders to be explicit in teaching effective coping skills and developmental assets.

Emotional responses to challenges can be controlled and managed through a deeper understanding of an individual's patterns and responses when under stress. These personal understandings can be supported by the reflection and debriefing process facilitated by the group leader. This emphasised the importance of the role of the leader in creating supportive, optimal learning environments that help to foster the transference of learning.

Initially, the participants felt that they had increased their levels of resilience immediately after the program; however, that may have been a result of 'post-trip' euphoria. The boys found it difficult to maintain their learnings six months after the program. These findings challenge the assumption that learning is automatically transferred to other contexts.

Six months post-program, the boys felt their learnings had dissipated. This may have been influenced by their ability or lack of supported opportunities to recall the skills and apply them in other settings. Lastly, the results showing that the boys' learnings were difficult to maintain post-program, highlights the importance of the Integration Phase of the program design, whereby follow-up sessions with education staff, as well as opportunities to be exposed to challenging situations where they can apply the skills developed during the Immersion Phase of the program, may assist in the transference process postprogram.

In summary, the results of the qualitative data indicate that outdoor education programs can provide empowering, supportive environments that can help young people to experience their personal strengths and limitations, develop and practice coping skills and increase their resilience capacity. Ultimately, the program design, with critical input of group leaders, can assist young people with optimal growth, personal well-being, and social development by creating a supportive learning environment that encourages feelings of competence and the ability to relate to peers, through exposure to stress inoculation processes where the challenges are specifically tailored to an individual's needs. These elements help to create feelings of autonomy, sense of achievement and self-reliance over their learning journey. For the transfer of learning to occur in other contexts after the outdoor education program, the program design needs to be targeted, intentional, specific to individual needs and include Frontloading, Immersion and Integration Phases. The next chapter, therefore, moves on to discuss and explain in detail the recommendations for outdoor education program design to foster not only the development of resilience and coping skills throughout the program but also to foster the transference of learning into other contexts after the program.

# CHAPTER 6: DISCUSSION AND IMPLICATIONS FOR RESEARCH

# AND PRACTICE

Great things are done when men and mountains meet; This is not done by jostling in the street

– William Blake, Notebooks (1793)



Figure 6.1. Celebrating summiting Mt Kosciusko.

# **Final Discussion**

The previous chapter presented the results of the qualitative data and provided discussions about the interrelationships between the themes. Overall, this chapter summarises the findings and provides an analysis of the quantitative and qualitative data to formulate mixed method interpretations, cross-validate the data and confirm the findings. The first part of the chapter overviews the key findings based on the relationships between the quantitative and qualitative results in relation to the development and transference of resilience and coping skills. The chapter discusses the: (1) importance of understanding the concept of resilience; (2) the importance of development-by-challenge; (3) the importance of making connections between contexts; and (4) the importance of outdoor education programs in schools.

Secondly, this discussion directly addresses each of the research questions and presents recommendations for practice. Thirdly, the chapter presents the strengths, and limitations of the study and discusses the implications for future research and practice. Lastly, the chapter concludes by presenting implications and recommendations for future research, followed by a detailed summary of the chapter.

#### Importance of Understanding the Concept of Resilience

'Resilience' appears to have become a 'buzzword' in society. This may be influenced by the awareness that one in four young Australian's have reported having a serious mental illness and that suicide is reported as the leading cause of death for young people (Bullot et al.,2017; Mission Australia, 2016)". The awareness of these facts have heightened society's understanding that young people require on-going support to develop and strengthen their resilience capacity (American Psychological Association, 2011; World Health Organization, 2018). However, there is a common misconception in society where people view resilience as a personality trait or simply conceptualising resilience as 'just' the 'ability to bounce back' (American Psychological Association, 2011; Bounce Back, 2017; Netuveli et al., 2008; Spangler et al., 2012; Tugade & Fredrickson, 2004; West, 2012). It was clear in this research that this common understanding of the concept influenced the boys' understandings of resilience. This became evident from the boy's definitions of resilience (see Table 5.9). While the boy's comments varied, their explanations indicated some common understandings about the concept. Most of the boys felt resilience was their ability to 'bounce back' and cope with challenges through keeping a positive mindset and persisting until they had overcome the challenge.

It appears that the boys had a naive understanding of resilience. This is understandable considering the concepts of resilience and coping are complex and multifaceted. As it stands, researchers still find resilience a complex, multidimensional construct that is difficult to define (Allan et al., 2012; Masten, 2012; Miller, 2015; Stokes, 2009). While it is accurate that 'bouncing back' is the outcome of the process of applying coping skills and adapting in response to difficulties, this definition and understanding of resilience is very generalised and simplistic. Consistent with various definitions in the literature (e.g. Bounce Back, 2017; Smith, 2010; West, 2012), the boys' descriptions do not incorporate the understanding that both internal assets and external assets, such as relationships, are critical elements of demonstrating resilience.

Figure 6.2 presents an overview of both the qualitative and quantitative results and demonstrates the interrelationships between the internal and external assets that encompass an individual's resilience capacity, assisting them to flourish in challenging circumstances. As Figure 6.2 demonstrates, both emotion-focused and problem-focused coping skills acted as internal and external assets.



Figure 6.2. Resilience capacity flourishing flower.

Ungar (2008, 2011) highlights that definitions of resilience should be conceptualised within a social-ecological framework that acknowledges the complex interplay of the relationships, processes and protective mechanisms of resilience (see Figure 2.4). This understanding of resilience is confirmed by the results of this study that demonstrated that an individual's overall resilience capacity is determined by the participants ability to draw upon both their internal assets (e.g., self-reliance and perseverance) and external assets (e.g., supportive relationships with peers and leaders) to assist them in managing their responses to the program challenges (see Figure 6.2). This reinforces the literature, that resilience is not 'just a personality trait', nor 'just the ability to bounce back', but rather that resilience is the capacity of an individual to utilise both their internal and external supportive systems to positively adapt to difficulties (see Figure 6.2). In addition, the results provide evidence of the multifaceted interactions between an individual and their relationships with their environments and their ecological systems, such as the micro- and mesosystems as highlighted in Bronfenbrenner's (1977, 1981) social-ecology model of human development systems (refer to Figure 2.4) (Bronfenbrenner, 1981; Garbarino, 1982; Ungar, 2012).

The concept of resilience and how young people respond to challenging situations is not 'just bouncing back', it is about being able to identify what skills and assets they can apply to help them manage stressors. Resilience development is about individuals becoming more aware of how they respond to stress when confronted with challenges and understanding how they can apply their assets in different contexts when faced with stress. Understanding the concept of resilience is essential to the way that the boys responded to the presentation of program challenges, how they went about confronting the challenges, as well as the learning they gained through the process of managing their responses to the challenges. The results indicate that when the boys were assisted by the leaders to have a deeper understanding of; (a) the concept of resilience, (b) how it works, (c) what helped them to become more resilient individuals, and (d) understand how resilience functions within themselves; they were more likely to be able to conceptualise the association between contexts of learning.

By being explicit in teaching and breaking down the coping skills and assets required to increase an individual's resilience capacity (refer to Figure 6.2), practitioners may assist the participants in understanding the complexity of resilience, which may be more effective than focusing on developing resilience as an overall construct. Consequently, within the space of working with young people, the 'concept of resilience' may not be as useful to measure, but rather a more useful way to analyse and measure resilience may be to measure the internal and external assets and specific coping skills of individuals.

Helping the boys to develop a sense of belongingness through strong, positive, supportive relationships with peers and leaders during 'City to Summit' was crucial to develop resilience. It assisted young people to flourish and also acted as a buffer against developmental disruptions (Norrish et al., 2011; Shonkoff, 2015; Van Ryzin, Gravely, & Roseth, 2009). In addition, the positive social relationships developed during the program appeared to support and strengthen the development of other assets, such as self-esteem and persistence (Shaffer, 2005). This is consistent with the findings from Jain et al. (2011) who showed that supportive relationships are strong predictors of emotional resilience. Helping the boys to develop a sense of belongingness through strong, positive, supportive relationships with peers and leaders during 'City to Summit' was crucial for developing the boy's resilience capacity. These assets assisted young people to flourish and also to act as a buffer against developmental disruptions (Norrish et al., 2011; Shonkoff, 2015; Van Ryzin et al., 2009).

The results revealed that it was neither 'just the internal assets' or 'just external assets' alone, but rather the individual's ability to draw on the different assets and apply them in various environments and contexts according to the challenge presented. For example, for boys who found the 24-hour solo experience to be highly emotionally challenging it was effective for the boys to draw on their internal assets, such as personal determination, motivation, selfreliance and perseverance to manage their responses to the solo. Whereas, for boys who found being away from home to be highly emotionally challenging, it was effective for the boys to gain support from their peers and leaders to help them manage their feelings of being 'homesick'.

Fostering the development of external assets in young people, such as positive relationships with leaders, peers and family, is more likely to influence developmental pathways of resilience, rather than only focussing on the development of an individual's internal assets (Ungar, 2011). This highlights the importance of programs that focus on developing both internal and external assets, as this will support and strengthen their ability to cope with difficulties in their home lives (Barrett et al., 2014).

The boys showed that when internal and external assets are strong, there is an increased likelihood that when faced with a difficult situation, they will have more resources to draw upon, resulting in an increased resilience capacity, and therefore demonstrating their psychological flexibility (Fredrickson & Losada, 2005) (see Figure 6.2). For example, when Jake acknowledged that his emotional responses had improved, from being angry and frustrated at the start of the program, to showing his resilience capacity by moderating his responses to the same stressors later in the program (Group C: Interview 2). The more young people can understand their own responses to stressors under challenging circumstances, the more likely they will experience positive effects in other areas of their lives (Constantine & Derald Wing, 2006).

The results also indicate that the boys required opportunities for personal growth and assistance in developing positive human characteristics and internal assets, such as self-esteem, self-efficacy, purpose, perseverance and existential aloneness, to help them enhance their resilience capacity. For instance, the resilience attribute, perseverance, was drawn upon by the boys most commonly during emotionally challenging situations, such as the solo experience, and during physically challenging situations, such as the mountain biking and hiking steep uphill sections. Despite being discouraged, it was these sorts of challenging situations that required the boys to apply perseverance, and the determination to persist, which led to strengthening many of the boy's resilience capacity during

the course of the program (Wagnild, 2009, 2010; Wagnild & Young, 1993). The quantitative data supports these qualitative results, showing that 40.6% of the boys in the program group increased their scores of Perseverance, while 42% had no change in levels of Perseverance immediately after the program.

Although the qualitative data showed the importance of shared experiences and building supportive relationships with peers and leaders; developing a positive relationship with oneself was also an important element for personal development. During the 24-hour solo experience, the boys came to understand that some challenges in life must be faced alone and it is this understanding, which is an essential component for developing Existential Aloneness (Wagnild, 2009, 2010; Wagnild & Young, 1993). Wagnild (2009, 2010) confirms that Existential Aloneness is about knowing oneself and developing an understanding that as an individual, a person has the ultimate choice in making his or her own decisions and life choices (refer to the 'choices layer' in the 'breakthrough diagram' in Figure *5.30*).

Developmental assets are essential life skills and are not only important for managing mental health, but they also assist young people to cope with adversity, enabling them to actively participate in society at an optimal level of human functioning (Constantine & Derald Wing, 2006). A person's resilience capacity is dependent upon their ability to draw on both internal and external assets, therefore, it may be more constructive to focus on developing and strengthening a young person's repertoire of both internal and external assets, which will expand their resilience capacity and increase their likelihood to

maintain or regain personal well-being in response to stress. By helping young people to build and strengthen developmental assets, we can minimise the risk, likelihood and severity of adverse mental health conditions that may develop during adolescence and early adulthood (Friedli, 2009; Wagnild, 2010).

# **Importance of Development-by-Challenge**

City to Summit's program design used a stress inoculation approach by gradually exposing the boys to challenges over the period of the program (see Appendix W). Supported by other recent studies, the findings of this thesis highlight the effectiveness of outdoor education programs as a catalyst for personal growth and the development of resilience and coping skills through the process of supported, stress inoculation and experiential learning processes (Booth, 2015; Shellman, 2009). The value of stress inoculation is also maintained by Lazarus and Folkman (1984) who acknowledges that stress inoculation is gaining the coping skills, personal understandings and assets to enable and moderate the best ways of coping with stress.

A prime example of the effective application of stress inoculation that was used to build the boys' self-reliance and independence was the gradual exposure to 'solo time' (individual reflection time). The boys were exposed to small amounts of 'solo time' (time alone in the wilderness) each day that increased in time increments throughout the program starting with five to ten minutes and building to hours. This enabled the boys to practice and develop their skills before culminating in a 24-hour challenging solo experience. The results show that the solo time was instrumental in fostering the development of Self-Reliance, Existential Aloneness, mental toughness and a range of coping skills, such as 'putting things into perspective' and 'coming to terms with difficulties'. For instance, Elliot's comment below demonstrates an understanding of how the solo influenced the development of existential aloneness. His comment reinforces that the solo activity influenced participants to get to know 'oneself' and develop skills to enable them to feel comfortable spending time alone.

The solo day was definitely a good test of your independence. Your ability to be a friend with yourself, sort of just be by yourself and be happy with your own company I guess (Group A: Interview 1).

Developing these life skills is essential for young people to start taking ownership over their decision making and choices, to become independent, develop selfreliance, and advance in developmental tasks that will support them as they transition into adulthood (Kroger, 2004).

Outdoor Education is about students engaging in stressful activities with manageable risk (Priest & Gass, 2018a) and feeling a sense of adventure through experiencing unusual, exciting and stimulating emotions (Brendtro & Strother, 2007). The process of the boys feeling the emotions of a sense of adventure and sense of achievement was caused through the exposure to risk and challenge. These elements were essential to the boys experiencing personal growth. When the boys were pushed beyond their comfort zones and supported to successfully adapt to the program challenges they had positive learning experiences (see the 'breakthrough diagram' in Figure *5.30*).

The results of this study indicated that participants who were not pushed beyond their comfort zones felt 'cheated' and 'bored'. It was important for the level of challenge to extend participants beyond their comfort zone, as Brodie (Group A: Interview 1) explains *"it's definitely more rewarding when you finish it [a challenge]* ". Similarly, Shellman's (2009) study also revealed that a 'hardearned' sense of achievement was more satisfying to participants. Both these findings emphasise how important it is to have challenges within the 'Goldilocks Zone' that are perceived to be challenging, yet just manageable (see Figure 5.27) (Csikszentmihalyi, 1990; Yerkes & Dodson, 1908). If program challenges are not made difficult enough, participants are not going to be exposed to adversity and will not be provided with the opportunities to practice initiating the coping skills and psychosocial attributes required to positively adapt to the situation. Without opportunities for adaption development of resilience and coping skills is unlikely to occur.

An individual's resilience capacity is influenced by the individual's core state of being, including their knowledge, beliefs, skills and competencies; and their conditioning layers, through meso-systems, such as school, family, and peers (refer to Figure 6.2 and the 'breakthrough diagram' in Figure 5.30). Because of their individual differences, some of the boys may have appeared to have a greater resilience capacity compared to other individuals, as they demonstrated successful adaption to certain program challenges. However, their demonstration of adaption

was not necessarily reflective of their resilience capacity, as resilience is demonstrated when a person perceives a challenge to be difficult. For instance, some of the boys who were more physically fit found some of the program challenges easier than other students with lower levels of physical ability, hence, they did not have to draw on the skills and assets that make up their resilience. This example illustrates that the boys 'core' and 'conditioning' helped them to adapt to the situation presented (refer to Figure *5.30*). This highlights how each person interprets the level of challenge differently, depending on their core and conditioning layers. If new coping skills and assets are to be developed, it is important for the level of difficulty to push them beyond levels of comfort, so they are required to draw on their assets.

As discussed in Chapter 5, the results indicate the need for program challenges to be specifically tailored to meet the needs of the group, as well as the individuals within the group. Interestingly, the boys showed a level of understanding that it was difficult to design the program challenges in 'City to Summit' to cater for the specific needs of all individuals, as each person's needs and 'core' are different (refer to the 'core' in the 'breakthrough diagram' in Figure *5.30*). For example, Chris suggests that:

If the challenges got too hard, I think it would've broken a lot of kids down (Group B: Interview 2).

Program constraints make it difficult to cater for individual differences, learning styles and competence levels amongst individuals. However, by aiming to identify individual differences amongst learners during the planning phase of the program, practitioners can better understand the individual's needs, help to guide their learning process and plan the level of difficulty of program challenges accordingly. The process of 'group streaming' individuals into physical ability and challenge level groups may be one way of tailoring programs to suit individuals specific physical needs. In addition, by allowing students to actively 'choose' their level of challenge and engagement, this may also increase levels of empowerment, willingness to participate and personal growth (Rohnke, 1989; Shellman, 2009; Wallia, 2008).

Facilitated by the leaders, the supportive, trusting environment that developed within each group of boys fostered the development of positive psychosocial attributes and coping skills. For instance, Chris indicates that when the level of difficulty of challenges was hard, he felt the groups worked more as a team, highlighting that individuals were progressing by drawing on external assets, such as the positive, supportive relationships with peers and leaders:

Chris: When the challenges were really hard, we worked more as a team then. If it [the challenges and the program] were harder, it would've been a lot harder for us mentally to just cope with the whole thing and to wanna actually be there (Group B: Interview 2).

# Importance of Making Connections Between Contexts to Support the Transfer of Learning

The transfer of learning occurred both during the program and after the program. Consistent with the transfer literature (Barkley, 2010), the program data revealed four different factors affected the quality of an individual's ability for learning transfer during and after the program:

- 1. the similarity and differences between the skills learnt;
- 2. the individual's skills sets and prior knowledge;
- the ability to associate and make connections with what they had learnt; and
- 4. understanding the context of learning in the outdoor education setting.

The group leaders played an important role in assisting the participants to learn and transfer skills during the program through the delivery of the curriculum and the facilitation of experiential learning processes. As outlined by Priest and Gass (2018b), leaders assume various key roles throughout outdoor education programs. For example, the group leaders were observed demonstrating many key roles, including:

• The role of the 'translator': Such as when Group 9's leader facilitated the 'it makes me mad' debriefing activity to assist the boys in making connections between their challenges, translating what happened and suggesting how they could improve their behaviours and actions;

- The role of the 'initiator': When the leader from Group 9 introduced the use of the 'positive thinking model' to help manage the student's needs (see Figure 5.23);
- The role of the 'trainer': For example, when the boys received coaching for practical skills when they were self-guiding rafts during the whitewater rafting section;
- The role of the 'maintainer': Such as when Reece (Group B: Interview 1) remarked that the leaders kept their group motivated when they were struggling with the program challenges;
- Assuming an 'authority' position: For instance, when the boys in Group 9's behaviour needed to be addressed at Benambra when they re-entered into an urban resetting from the wilderness setting (Observation Diary: Day 9);
- Taking on the role as a 'guardian': Such as when Ian (Group 9) was feeling homesick, and the leader managed his emotional well-being (Descriptive Diary: Day 11);
- The role of being the 'exemplar': Leaders played an integral role as being perceived as a positive role model, mentor and coach to support the transfer of learning (Sibthorp et al., 2011). For example, leaders were observed modelling group expectations of leadership styles, behaviours and practices, such as appropriate ways of how to get the groups attention when a group meeting needed to be held.

The boy's comments indicate that the leader was the most influential factor that contributed to their ability to conceptualise and understand the skills learnt during the program. Sibthorp et al. (2011, p. 120) supports the belief that the leaders assist participants in the reflection and conceptualising process that leads to the transference of skills, suggesting that "it is possible that the intimate relationship between instructors and students in adventure education may act as a catalyst for transfer in ways that other forms of training may not be able to replicate".

Healthy decision-making skills are crucial abilities to learn during adolescence (Piaget, 1964, 1972). Piaget (1972) proposed that during this stage in the cognitive development cycle, young people develop formal operational thinking, which helps the ability for self-reflection and the ability to predict possible outcomes of different types of their own behaviour in applications in various contexts of their lives. However, when the reflection process was left to the boys to initiate, they struggled to make links between contexts of their lives or understand how they could use their leanings to benefit themselves in future challenging circumstances. The boys commented that the learning was more 'unconscious', showing the need for practitioners to initiate the reflection process. For example, Jason's comment below provides an indication of how most of the boys struggled to verbalise how the program impacted them, implying that they were consciously unaware of the learning outcomes. However, they were able to indicate that going through the experience had a significant impact on them: Jason: Yeah. I think if you do [transfer the skills], it's more subconscious as well cause you're not really thinking, "oh, this is linking back to what I learned from 'City to Summit'. It's just something that's become ingrained into you... And so you might not realise it at the time, but it still has an effect on you (Group D: Interview 1).

Jason's comment also highlights the importance of conscious reflection to support young people in making connections for the transference of skills and conceptual understandings between contexts. These findings emphasise the importance of engaging young people in deep, meaningful, reflective activities that use enquiring conversations that support individuals to develop abstract thinking and allow them to draw connections, conclusions and understandings. This is supported by Priest and Gass (2018b), who suggests that when participants emotions are involved in the experiential learning experience, they are more likely to retain their learning.

Ewert and Yoshino (2011) also supports the findings that group leaders can use debriefing and reflection processes as a supportive tool for resilience development, stating that "providing students with opportunities to experience a sense of perseverance, or a responsibility to others, and then using those experiences during the processing and debriefing sessions could be effective in segueing into the broader concept of personal resilience" (p.46). Therefore, the more group leaders can support learners to make connections and provide useful feedback in response to a learner's efforts in managing their thoughts, behaviours and actions in response to the stressor, the greater the chance of perceptual and conceptual development and learning to occur (Hammond et al., 2001).

Immediately after the program, interviewees 'predicted and assumed' that the psychosocial attributes, coping skills and learnings would automatically transfer into other contexts of their lives. This assumption that skills will automatically transfer is consistent with the results of Sibthorp's (2003) study on the transference of skills from an outdoor education program into participants home lives. Sibthorp's (2003) findings were also based on students 'perceptions and assumptions' of what skills 'might transfer' rather than collecting follow-up data to see if the life skills did actually transfer into their lives. However, in the interviews six months post-Immersion Phase, only some of the boys were able to indicate skills that did transfer (see Table 6.3); whereas the majority of the boys were unable to provide specific examples of how they applied resilience after the program. This confirms that even though skills may have been developed, they are not necessarily transferred automatically into other contexts.

The lack of transference or the inability to demonstrate the application of skills post-Immersion Phase may have been attributed to a lack of understanding of the concept of resilience or the lack of opportunities to practice and apply their resilience capacity in responses to stressors. For instance, Elliot's comment below emphasises how the boys were unable to understand how the psychosocial attributes and coping skills learnt in the outdoor education setting could also be applied in their 'metropolitan lifestyle':

I'm sure the skills we learnt definitely affect your resilience especially in a kind of outdoor environment... And the degree to which it affects you definitely depends on how much you needed to improve your resilience or how much you did struggle [during the program], but...transferring it back into a metropolitan lifestyle, like the centre of Melbourne, there's probably not a whole lot of areas that you can apply those skills in (Group C: Interview 2).

When the boys were able to comprehend less complex concepts and understand the similarities between applications of the assets, tasks and skills they were applying, it appeared easier for them to make the connection with how they may apply the attributes and skills in another context (e.g., perseverance, selfreliance and determination). For example, Conrad from Group 9 makes the comparison, "if I persevere and climb Mount Misery, surely I can persevere and sit the exams for my Year 12 studies" (Descriptive Diary: Day 18). In this example, he shows an understanding of the concept of perseverance and how it can be used in different challenging situations. In the literature, this type of transfer is known as positive transfer, where the individual makes positive connections and understands how they can integrate their learnings (Svinicki, 2004). There were countless examples of how the boys experienced 'specific transfer' of skills during the program from day to day, such as near-transfer tasks that involved tasks that are highly similar (e.g., learning how to put up a group tarp and transferring the same skills for setting up the personal shelter for the 24hour solo) (Priest & Gass, 2018a).

In relation to positive transfer, the leader played a key role in facilitating positive transfer through the use of experiential learning, whereby the leader supported the learner through metacognition, and construction of knowledge (Svinicki, 2004). An example of this type of transfer was seen when group leaders used metaphors to assist the boys to make the link or connection between different situations (Priest & Gass, 2018a). This occurred when Lachlan's (Group C: interview 1) leader used the metaphor that *"it only takes one drop of poison to poison a well"*, to help the boys make the connection that it only takes one person in a group to have a negative attitude or mindset for the entire group to be affected.

When the boys were able to establish an understanding of how applying their assets can help them manage difficulties, they were able to articulate how they could apply these skills to different contexts. This process of connection was greatly assisted by the group leader, through the debriefing and reflective processes. However, when the boys didn't understand the concepts, they often struggled to make links between the application and transfer of skills into different contexts. In the literature, this type of transfer known as non-specific transfer (far transfer) where there is a potential for the learnings and principles to be interchangeable between contexts (Priest & Gass, 2018a). Specifically, for concepts and skills that involved 'non-specific transfer' (e.g., the concept of resilience) to be understood and transferred between different contexts, the learning and connections needed to be explicitly pointed out to participants.

In addition, the participant's perceptions reflect the view that building resilience is related to the skills specific to the context in which they were developed. Without the assistance of the leaders to help them to make connections between contexts, they were unable to understand or conceptualise how the strengthening and development of assets could be used to help them to manage other challenges in their home life. This emphasises that awareness and conceptualisation of learning and personal growth are essential for young people (Grümme, 2011, p. 114) and are critical factors in assisting young people to make the connections, develop their ability to think abstractly and improve their selfawareness and reflection skills.

#### **Importance of Outdoor Education Programs in Schools**

The interviewees commonly stated that they felt they have been 'spoon fed' at home, which may have diminished their opportunities to develop psychosocial attributes, developmental tasks and coping skills that are essential life skills required for transitioning into adulthood (e.g., self- reliance, independence, planning). Instead of 'wrapping kids in cotton wool' by not exposing them to challenges and opportunities to develop independence or allowing for the development of healthy decision making around risk, young people need the freedom to confront difficulties (Coulson, 2017). The findings of this study show that outdoor education interventions that are compulsory within the school curriculum can foster the development of resilience, developmental assets and coping skills through participation in tailored programs that provide opportunities for:

- 1. Exposure to challenge, risk and adventure, which invites participants to practice internal assets, such as perseverance and independence.
- 2. Participants to push themselves and test their personal capabilities, strengths and weaknesses.
- Practising the application of developmental assets, including effective coping skills.
- 4. Reflection and refinement of coping skills and assets.

While previous research reports that approximately 40% of young people in Australia have good to high levels of resilience and emotional well-being (Resilient Youth Australia cited in Coulson, 2017), contrasting research outlines that a fifth of young people in Australian are experiencing very high levels of psychological distress (Lawrence et al., 2015), and that one in four young people have a serious mental illness and require the support the develop the coping skills required to manage difficulties (Bullot et al., 2017; Mission Australia, 2016). As young people spend most of their time at an educational institution, schools and education providers have the opportunity to create a supportive school network for young people, which focuses on the implementation of outdoor education programs that allow students to strengthen assets, create meaningful peer relationships and take ownership of their own social involvement (Department of Education Victoria, 1999).

Schools need to be aware of the current youth needs and support them in learning coping skills with the aim of young people flourishing, thriving and contribute positively to society. For instance, as Dalton, Fawcett, and WestBurnham (2001) discuss, innovation is necessary for best practice in schools in the 21<sup>st</sup> century. This means that schools should be focusing on implementing programs that foster challenge, courage and creativity throughout the curriculum (Dalton et al., 2001). On the whole, young people need more exposure to life experiences that provide the opportunity to develop the relevant problem-focused and emotion-focused coping skills required to help them respond to the challenges of the information age. Developing these skills and assets increases the likelihood of effectively managing stressors in other contexts of their lives as them transition into adulthood (Coulson, 2017; Marano, 2008; Schiffrin et al., 2013).

Programs, such as 'City to Summit, can provide a platform for young people develop their resilience capacity by; (a) explicitly being taught a range of coping skills; (b) providing opportunities to apply and practice a range of problem and emotion-focused coping skills; (c) exposing the participant to challenges aimed at a level of difficulty to induce optimal arousal; (d) helping them to learn how to manage their responses to stress; and (e) providing guidance and support for self-reflection and growth.

#### **Addressing the Research Questions**

The overarching aim of this research was to investigate if resilience is context specific. The four specific research questions have been discussed and addressed throughout this thesis. The following sections provide a summary response to directly answer the research questions relative to the relationships between the quantitative and qualitative results.
### **Research Question 1:**

Which attributes of resilience and coping are enhanced through participation in an extended secondary school outdoor education program?

The Principal of Yarra College (2012) stated that primary program aims and outcomes for 'City to Summit' was to; (a) present the boys with challenges; (b) build leadership skills; (c) build perseverance, resilience and teamwork skills; and (d) promote an appreciation of the importance of the natural world. As demonstrated by both the quantitative and qualitative results, the aims were clearly achieved, with participants reporting increases in their resilience and developmental assets, including a range of coping skills immediately after the Immersion Phase of the program (see Figure *6.2*). These results are consistent with other studies that also found significant changes in resilience scores (e.g., Neill & Dias, 2001; Shellman, 2009; Davidson, 2016) and coping skills (e.g., Booth, 2015; Yoshino, 2008) post-participation in an outdoor education program.

Both the qualitative and quantitative data identified increases and positive change in overall resilience for the program group immediately after the program. In all cases, the program group reported a higher percentage increase in resilience scores, with an increase in resilience attributes after participation in 'City to Summit' compared to the control group. The reported increases in the program groups overall resilience is supported by Booth (2015) who also found that over half of the 14 interviewees from the qualitative results, revealed a positive change in resilience and that on average, the quantitative data showed a small positive

increase in resilience. However, while Booth's (2015) study found that resilience levels of participants increased at Time 2 (M = 3.56), it is important to note that participants were considered to be moderately resilient at Time 1 (M = 3.42). This is similar to the results of the current study which also found that the program group (M = 136) and the control group (M = 135.29) were moderately resilient at T1.

The resilience attributes that were found to be statistically significant were also identified within the qualitative data. Specifically, Table 6.1 presents the connections between the data sets demonstrating that both the quantitative data and qualitative findings identified positive changes in resilience attributes. The researcher observations of the program also supported these comparisons.

Table 6.1

Quantitative Results	Qualitative Results				
Attributes that strengthened with statistical significance	Attributes that were identified as strengthened through the interviews and observation data				
Overall resilience	Overall resilience				
Perseverance	Determination (perseverance & persistence)				
Existential aloneness	Self-reliance and independence				
Purposeful life	Development of relationships with peers and leaders				
	□ Appreciation of relationships with family members				
	□ Appreciation of and connection with technology				

Quantitative and qualitative results comparisons of resilience attributes

The coping skills identified by the boys, involved the effort they took, both behaviourally and cognitively to help them to master, tolerate, reduce, or minimise stressful circumstances during 'City to Summit'. Table 6.2 provides a comparison of the two significant problem-focused coping skills that were identified during Phase I and the seven problem-focused and two emotion-focused coping skills that were identified in Phase II and Phase III. As Table 6.2 highlights, the results of the quantitative results are supported by the qualitative findings. While the interviewees emphasised the use of two emotion-focused coping strategies during the program, the qualitative results six months postprogram saw no transfer of emotion-focused coping skills. Interestingly, this is consistent with the quantitative data that revealed no significant increases recorded in emotion-focused coping skills during the Immersion Phase.

Apart from 'chunking', all the coping skills identified by the boys during the program could be classified to be a form of 'active coping', which is problemfocused coping skill. Chunking is a type of planning, which involved the boys breaking the challenge down into smaller, more manageable chunks. The use of chunking was observed on a regular basis throughout the program and is supported by the quantitative data in which the program group showed a clear increase in planning skills (42.4%). These findings are consistent with Yoshino (2008) who determined that social support and problem-focused coping were the most effective type of coping utilised by participants when faced with stress in an outdoor education context.

## Table 6.2

Quantitative and qualitative results comparisons of coping skills

Quantitative Results	Qualitative Results		
Coping skills that strengthened with statistical significance	Coping skills that were identified as strengthened through the interviews and observation data		
Problem-focused Coping Skills			
Active Coping	Putting Things into Perspective		
	Removing Oneself from the Stressor		
	Ability to Accept Social Support		
	Addressing the Issue		
	Coming to Terms with Difficulties		
	Cognitive Reframing and Applying Positive Thinking		
Planning	Chunking		
Emotion-focused Coping			
	Distraction		
	Avoidance		

# **Research Question 2:**

If attributes were enhanced, what supported and fostered the development of these resilience attributes and coping skills?

The program design during the Immersion Phase followed a similar process of learning to Walsh and Golins' (1976) widely accepted Outward Bound Process model. This process involves the learner being placed into a unique physical and social environment. The learner is given a set of problem-solving tasks, that can create a state of cognitive dissonance, which leads the learner to explore ways to manage the situation. The goal is to increase the students' mastery or competence and support the learner to reorganise and conceptualise the new understanding that has come from the experience (Walsh & Golins, 1976). This programming process had an overarching influence on the boy's personal development by providing a platform for participants to be exposed to eustress in a supportive environment, thus presenting opportunities to develop their psychosocial attributes and coping skills.

The program stressors that challenged the participants emotionally, physically and socially were critical components of the Immersion Phase (Ewert, 1989). The stressors and program challenges played a vital role in enhancing resilience. Stress is an essential element for an individual to develop and demonstrate their adaptive capacity, as their strengths and limitations only become apparent in the face of difficulty (Booth, 2015; Meichenbaum, 2017b; Rutter, 1993).

The application of a program design that was grounded in stress inoculation and experiential learning approaches to foster development-bychallenge, produced evidence that the program assisted the participants to:

- gain knowledge about themselves and how they respond to challenging situations;
- become familiar to stressful circumstances;
- learn psychological skills that can aid in the process of adaption;
- practice decision-making skills when under stress; and

• build confidence in their personal capabilities to manage themselves when in challenging situations.

Using 'challenge by choice' was essential in motivating, empowering and engaging the boys in the development-by-challenge process. When the level of challenge was within the 'Goldilocks Zone', that point where challenges were perceived to be 'just manageable', the students were seen to apply effective and relevant coping strategies, which helped them manage their responses to the stressors. Their effectiveness in adaption to the stressors resulted in the boys feeling a sense of achievement, which influenced gaining a positive sense of agency, self-esteem, self-confidence and self-efficacy of participants.

The results emphasise the significance of exposure to personally challenging experiences (e.g., 24-hour solo), as well as the importance of the reflection and debriefing process facilitated by the leader. Support for this finding is provided by Shellman (2009) who also found that debriefs, the solo experience and participant engagement, during an outdoor education program, were most highly correlated with resilience. In addition, Shellman's (2009) study identified that lessons from instructors, the debriefs and a personal, challenging event were most highly corrected with increases in empowerment of participants.

The data shows that the group leaders were instrumental in creating a supportive, collaborative environment for young people to thrive and feel safe during the program. Through the debriefing sessions facilitated by the leaders, participants developed their metacognitive skills by focussing the process of selfawareness, where they became more attuned to of their thoughts and behavioural

responses to challenging situations. This increase in awareness can affect the participant's personal growth by enabling them to manage their own learning. The group leaders who combined both experiential learning practices, along with information assimilation information appeared to strengthen the learning for participants (Piaget, 1964; Priest & Gass, 2018b).

Immersion in the natural environment not only provided the platform for participants to experience adventure, challenge and risk in a unique physical and social environment, but also it provided many other benefits, such as participants feeling the restorative healing effects of nature and experiencing less distractions that have become the norm of urban life in western cultures (e.g., technology, noises from the hustle and bustle). The boys emphasised that removing technology and not having other distractions, they were forced to engage with one another, helping them to connect with their peers and forge deeper relationships.

The length of the 'City to Summit', obligated the boys to spend a consistent and extended period with the same small group of people. Being in this situation provided countless opportunities to undertake shared experiences, support one another through difficulties, experience the challenges of group development processes and develop trusting relationships.

This extended time also fostered the development of peer to peer and peer to leader relationships. The results indicated that it took time for groups to develop trust over the course of the program. The more the group shared new experiences together, the more levels of trust increased between peers and leaders, which increased the likelihood for people to share feelings and be vulnerable in group debriefs. Program challenges that required a team effort, communication and trust also helped to build stronger peer to peer relationships.

### **Research Question 3:**

If attributes were enhanced, what attributes of resilience and coping were transferred and applied in contexts other than an outdoor education setting?

It could not be determined if the boys maintained their initial self-reported levels of increased overall resilience, as the as none of the boys reported having opportunities to demonstrate their resilience capacity in the six months after the Immersion Phase of the program. This observation highlights the need for the Integration Phase to include post-program follow-up activities and opportunities to continue to practice and apply psychosocial attributes and coping skills in supportive, challenging environments.

The results of the post-program interviews emphasised a range of resilience attributes and coping skills that had developed immediately after the program. Assets and coping skills appeared to transfer in some cases, but this was not consistent across all participants. Table 6.3 provides a summary of the psychosocial attributes that some of the boys reported as being transferred and applied in other contexts post- Immersion Phase.

# Table 6.3

## Psychosocial attributes that were transferred post-Immersion Phase

Problem-focused	<b>Emotion-focused</b>	Internal assets	External assets
coping skills	coping skills		
putting things into		self-reliance	positive
perspective			relationships with
			their peers
removing oneself		independence	positive
from the stressor			relationship with
			the natural
			environment
ability to accept		tolerance	
(and seek) social			
support			
coming to terms			
with difficulties			
chunking			

As Table 6.3 highlights, there were no reports of emotion-focused coping skills being transferred post-program. In addition to these skills and assets, onethird of the boys reported feeling an appreciation of their family members in the six months post-Immersion Phase. However, the other two-thirds of the boys commented on how difficult it was to maintain these initial feelings of appreciation of their family members six months after the program.

# **Research Question 4:**

How can resilience and coping skills be best developed and transferred into other contexts of learning?

The following provides a summary of the essential elements that best supported the development of resilience and coping skills which are discussed in detail in response to Research Question 2:

- Intentional program design that incorporates a Frontloading Phase, Immersion Phase and Integration Phase.
- Stress inoculation approach that uses gradual and increasing exposure to stressors.
- Program challenges set within the 'Goldilocks Zone' where the risk does not exceed the competence of participants.
- Setting up a supportive, trusting and collaborative learning environment in a small group setting.
- Engaging young people in deep reflective practice through experiential learning practices that involve meaningful, enquiring conversations.
- Group leaders supporting the participants to make connections and conceptualise key concepts.
- Individually challenging experiences, such as the 24-hour solo.
- Providing participants with opportunities to choose their level of challenge engagement to suit their individual needs.
- An extended program length that allows for practice and refinement of skills.
- Removal or limited use of technology.

The most critical factor in the transference of skills and assets postprogram was the boys' ability to conceptualise and understand the complexities and concepts of resilience and coping. The leader was the most influential factor in assisting participants to transfer learning between contexts, through helping the participants to draw connections between what they already knew or understood, and new information. Through the experiential learning and reflection processes, the leaders provided guidance on how to either consolidate learning through further practice and action and provided suggestions of how the boys could manage their responses to the stressors.

To increase the likelihood of the transference of the skills and assets developed during the program, it was essential for the students to be supported by staff to engage in abstract thinking and conceptual understandings of their learnings to understand how the concepts and skills they learn during 'City to Summit' can be useful other contexts. This is supported by researchers who suggest that for transference and application of resilience attributes to occur; the learning needs to be reflected on in the broader context of their lives (e.g., Anderson, P., 2008; Brookes, 2003a, 2003b).

The Integration Phase, also known as post-activity outdoor education was a fundamental element in supporting the transference of skills and assets across settings. This element is critical for supporting participants to (a) make sense of what they have learnt on the program; and (b) integrate their learning and skills from the program and transfer these skills to be able to apply in other contexts of their everyday lives. The phase also provides opportunities for self-reflection, to practice skills, receive feedback and allow for post-program follow-up.

### **Is Resilience Context Specific?**

The results of this study are consistent with the literature which identifies three environments can affect an individual's ability to develop resilience (Bronfenbrenner, 1977; Coulson, 2017; Garmezy, 1985; Werner, 1995, 2000; Windle, 2011). The first environment was the internal environment. This is the personal understanding about ourselves and involves an individual's ability to utilise their internal assets. The boys' internal environments were affected by selfawareness, thinking, cognitive processes, behaviours, feelings, emotions and actions (see the layers in the 'breakthrough diagram' in Figure 5.30). The second environment was each boy's relationship with the people that surrounded him. The quality and strength of the external assets relied on the strength of relationships with their peers, leaders and family members (Benard, 2004; Hanson & Kim, 2007; Olsson et al., 2003). The third environment influencing the development of resilience was the systems surrounding an individual in their broader environments, such as, fellow students in the same Year level, the wider school community and the natural environment (see Figure 2.4) (Bronfenbrenner, 1977, 1981; Garmezy, 1985; Werner, 1995, 2000; Windle, 2011).

The combination of the qualitative and quantitative that demonstrates young people can expand their resilience capacity through participation in outdoor education programs (refer to Figure 6.2), reinforces that by developing these psychosocial attributes, they increase their likelihood to be able to manage and self-regulate their response to stressors. However, it is important to note that positive adaptation may not occur across all domains of a young person's life (Masten, 2015). Different challenging situations, in different contexts, involve exposure to different stressors. The results indicate that an individual's resilience capacity is dynamic and can change over time. Consequently, individuals may show the ability to adapt in one context at a particular time and struggle to adapt in other contexts at different times of their lives. For instance, an individual may demonstrate a higher resilience capacity with respect to the stressors involved in the program challenges during an outdoor education context, but not be able to demonstrate adaption to other stressors in different environments, highlighting that resilience may be content- and context specific. For instance, challenges in the outdoor education context may require different skill sets, assets and competencies, compared to challenges in other contexts of the boys' lives (e.g., school, home, sporting groups, community groups) (refer to Figure 6.2). A young person may show adaption and success in overcoming one type of challenge, such as managing the stressors related to hiking up a steep hill, but the same young person may show a failure to adapt to the demands of studying for their final school exams. Therefore, depending on an individual's core and conditioning layers, their skills and assets may be better suited to manage stress in certain environments that require specific skill sets to be able to manage the stress. This highlights the importance of focussing on developing an individual's core and conditioning layers, to expand their resilience capacity and increase their likelihood to be able to cope across contexts (refer to 'core' and conditioning layers' presented in Figure 5.30).

Through direct personal experiences and by using a stress inoculation approach, participants can be explicitly taught coping skills and can expand and strengthen their psychosocial attributes through exposure to challenges (Passarelli et al., 2010; Scales & Leffert, 2004; Sesma Jr et al., 2003; Vera & Shin, 2006). The more people are inoculated to specific challenges, through gradual repeated exposure to the stressors, the more likely they are to achieve a higher level of mastery. However, it is important to note that a participant can experience repeated exposure to a stressor and not change their response.

In considering the findings of this study, it appears that an individual's resilience capacity is content and context specific to the environment; however, the psychosocial attributes that are strengthened and developed during outdoor education programs can be transferred and applied in other contexts, if the participants are able to conceptualise their learning. When an individual starts to understand concepts of resilience and how these work, they are more likely to be able to replicate the coping skills and assets in other contexts (Schenck & Cruickshank, 2015, p. 83). Therefore, the transference of learning can be supported by helping young people to conceptualise and understand the skills and assets they are developing and by helping them to comprehend how the skills and assets can be applied in different contexts. The results of this study indicate the importance of group leaders to be highly skilled, trained and educated not only in hard skills of adventure activities but, also be a good communicator, active listeners, effective problem-solvers, be knowledgeable about concepts of resilience and understand how to explicitly teach a range of coping skills.

Based on the overall results of this study, it is evident that the program design was the most significant factor that impacted the development and transference of the participant's resilience capacity. Therefore, the following section of this chapter provides recommendations about the program design. These recommendations support the development of resilience attributes and coping skills with the intention of these being transferred to the various contexts of a participants life.

### **Recommendations for Practice**

Adolescents have an increased risk of mental health concerns (Australian Bureau of Statistics, 2008a, 2016). The statistics that one in four young people in Australian's met the criteria for having 'serious mental illnesses' is a grave concern (Bullot et al., 2017; Mission Australia, 2016). These statistics should serve as encouragement for schools to take responsibility and be proactive in explicitly teaching an array of effective coping skills that are relevant to young people living in the digital and information age. When the transference of resilience is the intended learning outcomes for an outdoor education program, philosophically, the program design should be grounded on personal development, as this will promote more effective outcomes for participants (Marsh, 1999a, 1999b).

# Program Design to Foster the Development and Transference of Resilience and Coping Skills

If outdoor education programs intend to develop resilience attributes and coping skills to promote mental health and well-being; practitioners can combine preventative, proactive, strength-based approaches that focus on developing positive emotions, explicitly teach effective coping strategies and developmental assets (Seligman & Csikszentmihalyi, 2000). Booth & Neill (2017) agrees that in order for outdoor education programs to be more effective and consistent in achieving positive results in the development of psychological resilience, outdoor education program design and practitioners need to be more diligent in including psychoeducational coping curriculum into program design.

Fundamental, educational and psychological theory about resilience, coping and stress can be 'explicitly' taught to individuals during outdoor education programs that aim to have resilience development and coping skills as a program outcome. Therefore, based on the results of this study discussed in this chapter, it is recommended that the elements of the Resilience Development Program Design (RDPD) that is presented in this section be considered as it provides an effective approach when designing a resilience development intervention in an outdoor education context (see Table 6.4). Comparable to the design of 'City to Summit', the RDPD is presented as a single program design; however, it is comprised of three phases that include various key elements and actions that could be used to design a new program or integrated into a current outdoor education program. Currently, there appears to be a stigma attached to the terms used around interventions and programs associated with 'mental health' or 'therapy' (Ballenger-Browning & Johnson, 2010). For this reason, the terms 'Resilience Development Program Design' has been applied to help avoid the stigma attached to the traditional approach to mental health treatment and to view outdoor education interventions in positive, proactive terms (Ballenger-Browning & Johnson, 2010).

Consistent with wilderness and adventure therapy frameworks (Kimball & Bacon, 1993; Nadler, 1993; Newes & Bandoroff, 2004), stress inoculation training approaches (Meichenbaum, 2017b) and key elements of journey style outdoor education programs (Hattie et al., 1997; Neill, 2008; Priest & Gass, 2005); the RDPD proposes prerequisites during each phase of program design in an aim to transfer learning to and from the program. The RDPD presented in takes a proactive, preventative stance to mental health and well-being by providing a platform to (a) build developmental assets, (b) increase individuals repertoire of coping strategies, (c) develop an individual's resilience capacity, and (d) foster the transference of these skills and assets into other contexts.

In line with the phases of 'City to Summit' and the phases of stress inoculation training (SIT) outlined in the literature review, the Frontloading Phase can be utilised as the 'education phase and rehearsal phase', the Immersion Phase can be utilised as the 'rehearsal phase', 'application and follow-through' phase; and the Integration Phase can also be utilised as the 'application and followthrough' phase of SIT (Meichenbaum, 2017b, pp. 131-132). During all phases,

practitioners are recommended to use questions that promote personal awareness, cognitive processing and discovery to support learning and conceptualisation during the stress inoculation process (Meichenbaum, 2017b; Novaco, 1977).

In accordance with the design of 'City to Summit', the RDPD incorporates the three phases of program design (Frontloading Phase, Immersion Phase and Integration Phase). The purpose of the RDPD is to provide guidelines to support practitioners in assisting participants to expand their resilience capacity by developing and strengthening their psychosocial attributes and coping skills (refer to the RDPD in Table 6.4). In addition to the guidelines as outlined in Table 6.4 the RDPD can assist participants to develop and transfer their resilience capacity by:

- Teaching specific effective coping skills to help regulate negative emotions and enhance one's control over their physiological responses.
- Teaching cognitive approaches to managing stress and developing adaptive coping strategies, including changing maladaptive beliefs.
- Explicitly educating students the difference between problem-focused coping skills and emotion-focused coping skills, and when to best use them.
- Simplifying key concepts and supporting young people to draw conclusions and make connections based on their prior experiences and what they have come to know.

• Integrating learning back into school and home environments during the Integration Phase through follow-up activities and guided reflection.

These recommendations are not intended to be used for the specific treatment of mental health issues, but rather as a guide for educators and outdoor education companies to use practically, to support program design that fosters the development of resilience, effective coping skills and the transference of these skills into various contexts, therefore, helping young people to flourish and thrive in their everyday lives.

Given the nature of the risks associated with outdoor education programs, prior planning is a vital element of program design. In addition to risk management practices, standard operating procedures and guidelines for the designing and planning of outdoor education programs, the RDPD recommends considering prerequisites during each phase of the program design (see Table 6.4). The following sections expand on the prerequisites provided for each of the program design phases.

# Table 6.4

# Resilience Development Program Design

Program Design Planning:	Phase 1: Frontloading Phase		Phase 2: Immersion Phase		Phase 3: Integration Phase	
<b>Key Elements</b>	Pre-activity outdoor education		The outdoor education program		Post-activity outdoor education	
	Prerequisites		Prerequisites		Prerequisites	
Phase length	1 day or more		21 days or longer		1 day or more	
Environment	Normal learning environment (e.g., school, Tafe, University)		Intervention occurs in a semi-wilderness, unfamiliar environment		Normal learning environment (e.g., school, Tafe, University)	
			Participants are isolated from many or all for of technology and human impact or development	ms		
Group focus with small to	Interaction with the same group		Interaction with the same group	Rehearsal & Appli	Interaction with the same group	
medium size groups		Ed			Interaction with larger cohort of participants during celebration	Application & Follow
Stress inoculation and program challenges	Explicit teaching of problem-focused and emotion-focused coping skills	ducation & Rehe	Gradual exposure to a range of mentally, socially and/or physically challenging adventure activities that increase in level of difficulty relevant to the individuals 'core' and 'conditioning layers'		Nurture gradual mastery of coping skills	
	Set expectations	arsa	High expectations of participants	catio	High expectations of participants	thre
	Education and supported self-awareness activities to understand emotional and behavioural responses		Coaching and mentoring on problem-focused and emotion focused coping skills	on	Ongoing coaching and mentoring on problem-focused and emotion-focused coping skills	ough
	Provide opportunities for meaningful participation		Provide opportunities for meaningful participation		Provide opportunities for meaningful participation	
	Set expectations		Perceived risk, appropriate levels and forms of challenge for participants (physical, emotional, spiritual)		Perceived risk, appropriate levels and forms of challenge for participants (physical, emotional, spiritual)	
	Challenge by choice		Challenge by choice		Challenge by choice	
	Opportunities to apply and practice coping st taught during this phase	Opportunities to apply and practice coping skills when exposed to program challenges within Goldilocks Zone		Opportunities to apply and practice coping skills when exposed to new challenges specific to the school context		
		Immersion Phase includes challenges within Goldilocks Zone that provide both challengir and fun peak experiences	ıg	Integration Phase includes challenges within Goldilocks Zone that provide both challenging and fun peak experiences		

# Table 6.4 (continued)

<b>Program Design Planning:</b>	Phase 1: Frontloading Phase		Phase 2: Immersion Phase		Phase 3: Integration Phase	
Key Elements	Pre-activity outdoor education		The outdoor education program		Post-activity outdoor education	
	Prerequisites		Prerequisites	Prerequisites		
Facilitated reflection and debriefing	Personal reflection and awareness activities of current coping skills		Self-evaluation of behaviours, thoughts, actions, feelings and performance		Self-evaluation of behaviours, thoughts, actions, feelings and performance	
	Personal awareness of existing coping strategies and behavioural observations		Reflect and refine behavioural, practical and cognitive coping skills		Reflect and refine behavioural, practical and cognitive coping skills	
	Self-reflection activities		Self-reflection activities		Follow-up experiences which recall their learning experiences during the program	
	Leaders to provide coaching and mentoring on the application of coping skills and developmental assets		Leaders to provide coaching and mentoring of the application of coping skills and developmental assets	on	Ongoing coaching and mentoring opportunities	
			Daily reflection and debrief sessions		Debriefing and facilitating activities that enhance the process of abstract thinking and making explicit connections between contexts	
Goal setting	al setting Participants establish short-term, intermediate, and long-term personal program and life goals that are behaviourally specifiable	Education &	Monitor goals and re-assess if required	Rehearsal &	Goal setting and reflection activities fo future personal growth fo	
		& Rehe	Facilitated re-assessment of program daily goals	Applic	Structured reflection and re-evaluation goals post-program	
Relationship development:	Values and beliefs contracts or	arsa	There is daily and intense interactions	atio	Opportunities to connect with group	
Trust and team building	activities		within a small group setting requiring problem-solving and decision-making	'n	fortnightly basis)	
	Create a supportive, caring, positive environment		Create a supportive, caring, positive environment		Create a supportive, caring, positive environment	
	Initial trust, teambuilding and initiative activities that require team problem-solving		Adventure activities that require teamwork at trust (e.g. whitewater rafting, hiking)	nd	Group follow-up activities	
Problem-solving tasks and activities	Provision of concrete consequences, real actions and consequences through experiential learning experiences		Provision of concrete consequences, real actions and consequences through experiential learning experiences		Provision of concrete consequences, real actions and consequences through experiential learning experiences	
Solo experiences	Set expectations early		cheduled daily solo time that builds in length roughput the program		Opportunities for solo time during normal learn- ing contexts	
			Culminating solo experience (ideally 24 hour longer)	rs or		

## Program design planning.

Prior preparation of an interventions program design is critical for planning (a) the structure of how the individuals learn, what they will learn and when they will learn it, (b) what types of stressors and program challenges they were exposed to and how they are exposed to them, and (c) the structure of how individuals processed and reflect upon their learnings and experiences. Interventions aiming to foster psychological resilience should incorporate intentional program design that offers "a sequence of graduated mastery experience[s] that enable a child to experience success and build self-efficacy and motivation" (Masten & Reed, 2002, p. 85).

Consistent with the factors that lead to the development of resilience outlined this chapter, the RDPD follows a stress inoculation approach by allowing for smaller challenges to be present at the start of the program and move up the scale to more challenging experiences throughout the program. A prime example of a type of stress inoculation and development-by-challenge is introducing participants to solo time gradually over a period of time. Instead of sending individuals out into the wilderness for 24 hours by themselves without any prior experience, group leaders can intentionally set small blocks of solo time each day, increasing in the amount of time (the stressor) as the program continued, culminating with the 24-hour solo experience. Solo experiences are specifically recommended as part of the program design as it was the most beneficial activity to support self-reliance, independence and provide opportunities to apply a range

of problem-focused and emotion-focused coping strategies without relying on others.

It is essential to provide program challenges where the level of difficulty is relevant to the participants 'core' (e.g., prior, experience, knowledge and skill sets, physical capacity, fitness levels) and 'conditioning layers' (e.g., participants education, previous exposure to outdoors and adventure activities, cultural background, whether they are a minority group, such as youth at risk or they are a majority group, such as a mainstream educational institution) (refer to Figure *5.30*). It is also important to consider at what point in the program the challenges are presented to participants.

Ideally, program design should create relevant challenges that create a perceived risk, threat, or challenge to create a sense of disequilibrium and eustress (see Figure 5.28) that are within individuals 'Goldilocks Zone' with an aim for participants to have a peak experience (refer to Figure 5.27). Activities that allow for both individual and shared experiences, group development opportunities, facilitation and reflection early on in the program are favoured.

Walsh & Golins' (1976) suggests that the 'ideal' group size to foster and support learning outcomes consists of 10 participants. However, mainstream outdoor education program group sizes generally range between 6-18 participants. Therefore, to meet the criteria for small or medium-sized groups, it is proposed that groups ideally consist of 10-14 participants (Neill, 2008; Walsh & Golins, 1976).

#### **Phase 1: Frontloading Phase.**

The Frontloading Phase is the 'pre-activity' outdoor education phase which is critical for setting expectations, developing group trust, fostering supportive relationships, and initiating self-awareness. This phase in the program design is about adequately frontloading the experience prior to the Immersion Phase.

To support the transference of learning during each phase, it is necessary that programs aims and objectives are fully understood by the participant, where they understand the program's relevance to them as an individual (Cooley et al., 2016; Cooley et al., 2015). During the Frontloading Phase, participants can be supported by helping them to understand and see the value or need for them to engage in the challenge. If young people know in advance what the big concepts of the program are and how these concepts relate to each other conceptually, they are more likely to make sense of the information, be able to remember it and use it more flexibly (Hammond, L., et al., 2001).

The Frontloading Phase can also be used to explicitly educate young people about problem-focused and emotion-focused coping skills that help to focus on the modification, avoidance, and minimisation of the impact of stressors. These recommendations are consistent with Booth (2015, p. 28) who suggests that "program facilitators can actively assist and teach participants about their own coping strategy choices and guide them in making beneficial and productive choices to provide participants the best opportunities to develop resilience". Through understanding the five psychological attributes of resilience measured in this study (Wagnild, 2009, 2010; Wagnild & Young, 1993), it is clear that there is a direct link to the importance of goal setting before, during and after the Immersion Phase of the program. Considering the results of this study and the definitions of the resilience attributes measured in this study, all attributes highlight the importance of goal setting and self-awareness. Goal setting is therefore an essential element to the growth and development of an individual's psychological resilience and ability to strengthen adaptive coping mechanisms. For example:

- 'Purposeful life' is about understanding your life goals and knowing the steps and actions to work towards achieving them.
- 'Perseverance' is the ability to work towards achieving these goals despite difficulty or challenges.
- 'Equanimity' is the ability to keep a balanced perspective and to moderate our responses to these adversities.
- 'Self-reliance is the ability to understand your personal strengths, weaknesses and capabilities.
- 'Existential aloneness' relates to knowing yourself and being comfortable and confident in your own skin.

In addition, goal setting in the Frontloading Phase needs to be relevant, facilitated. Their goals need to be specific and difficult enough to increase performance or their current state of self-awareness in relation to coping. In addition, it is recommended that practitioners allow students to have input and control over their goals, personal development and activity selection during outdoor education programs (Cooley et al., 2016; Cooley et al., 2015; R. M. Ryan & Deci, 2000).

Assisting participants to develop goals that are directly related to their personal development has been shown to enhance well-being, resilience, and personal performance (S. Green, Grant, & Rynsaardt, 2007). Evidence shows that pathways to building resilience can be achieved through setting challenging, yet achievable goals (American Psychological Association, 2011; Benard, 1991; Werner & Smith, 2001). Setting clear goals at the before the Immersion Phase can also assist in creating engagement with the program and program challenges, where participants feel a sense of ownership and are committed to their goals and wanting to succeed throughout the program.

### Phase 2: Immersion Phase.

Along with the recommended prerequisites of the RDPD presented in Table 6.4), there is an abundance of literature that supports the use of cognitive behaviour modification techniques and somatophysiological coping strategies to aid in managing stress, building resilience, and maintaining mental health and well-being (Antoni et al., 2007; Bennett et al., 2013; Lazarus, 1984; Lillis et al., 2009; Miller, 2015; O'Connell, 2005; Wilson & Baer, 2010). The results of this study indicated the use and benefits of some of these coping strategies (e.g., journaling and positive reframing of thoughts).

The Immersion Phase provides an ideal platform for young people to be taught and have the opportunities practice the following cognitive behaviour

change processes, modification techniques and somatophysiological coping strategies that can be used as tools to manage stress in peoples everyday lives. Therefore, it is recommended that the program design includes the intentional use of coping strategies, such as:

- positive reframing of thoughts,
- journaling,
- mindfulness activities,
- meditation,
- relaxed breathing techniques
- self-monitoring and awareness,
- positive visualisations, and
- affirmations (Lazarus & Folkman, 1984; Meichenbaum, 2017b, pp. 131-132).

Debriefing and reflective sessions guided by the group leader should aim to 'hold space' where individuals learn to give and receive supportive, positive, yet constructive feedback from peer to peer and group leader to participant. Opportunities for participants to give group leaders feedback may also increase levels of mutual respect and group trust. However, caution should be taken as feedback that is not given and received in a constructive, positive and supportive manner could undermine efforts and cause conflict amongst team members.

Suitable program design and facilitation will set the tone for students to be able to share experiences with peers and teachers, which will allow for the development of relationships. The more group leaders can support the learners by providing useful feedback in response to a learner's efforts to manage their thoughts, behaviours and actions with their response to the stressor, the greater the chance of perceptual development and learning to occur (L. D. Hammond et al., 2001). This research shows that by supporting a young person to develop their emotional intelligence and self-awareness, they are more likely to be able to recognise and manage their emotional responses to stressors, solve conflicting situations and apply other resilience attributes, such as perseverance and selfreliance.

### **Phase 3: Integration Phase.**

Generally speaking, it appears that the Integration Phase is what is missing from most 'mainstream' outdoor education program designs in Australia. The Integration Phase is essential for supporting the transference of psychosocial skills and assets into other contexts. This final phase of program design provides the opportunity for participants to:

- Reflect on their goals, learnings and achievements of the program.
- Reset future goals and intentions for their upcoming year based on their learnings.
- Share these goals and intentions with important people in their lives who can assist them by holding them accountable.
- Experience supported, structured follow-up sessions throughout the year.
- Practice skills and be exposed to challenges in different contexts.

In supporting students to build their resilience capacity and explicitly teach them coping skills to manage mental health difficulties and life challenges, it is critical to have support from a positive school community, including parents and families (MindMatters, 2018b). In the follow-up interviews, six months post program, boys in all interview groups agreed that there was not enough emphasis on the culmination and celebration of the program. The boys also emphasised that they would have liked to have connected with their group leaders after the program finished. By celebrating the participants success of achievement with the school community and their families, this type of experience could potentially be used as a 'rites of passage' acknowledgement where important people in the boys' lives (e.g. family members, teachers, mentors within the community) come together to acknowledge not only the experiences they have had but also how they have developed, grown and changed as a person (Bell, 2003; Dooley-Feldman, 2016; Norris, 2011).

A participant-mentor relationship is similar to the participant-group leader relationship, which appears to occur during outdoor education programs naturally. Although mentoring was not included in program design of 'City to Summit', the boys highlighted the desire to; (a) connect with their group leaders after the program, (b) connect with their groups after the program, and (c) practice the skills they had learnt during the program. Therefore, for participants to develop mastery and transfer skills and assets learnt during the program, it is recommended that the Integration Phase considers setting up a mentoring protocol. The inclusion of youth mentoring in program design can assist in;

- preventing the emergence or continuation of psychosocial difficulties or problem behaviours;
- promoting positive adjustment through the development of individual competencies; and
- fostering integration with the community through opportunities for involvement (Keller, 2007, p. 27).

To support the facilitation of mentoring relationships during the Integration Phase, rather than face to face interactions, other alternatives for mentoring and feedback may be considered, including peer to peer mentoring, accountability partners or online mentoring through platforms such as Skype, Facebook or GoogleChat Rooms, where participants can be involved in group or individual discussions online.

In summary, the recommendation in the RDPD provides support for schools and practitioners to incorporate resilience and coping building interventions into mainstream outdoor education program designs. Application of the RDPD is recommended when mainstream outdoor education programs want to bridge the gap between their current programming and programs that intentionally aim to not only build skills and assets but also transfer resilience and coping skills across settings. To support the transference of skills, the Integration Phase is vital in assisting participants to conceptualise their learning and support the integration of skills and assets across different contexts of their lives.

The recommendations of the RDPD use proactive mental health promotion and prevention strategies by incorporating psychoeducational curriculum and explicitly teaching participants an array of effective coping skills. The program design uses strength-based approaches to foster the development of psychosocial attributes and coping skills to assist young people in expanding their resilience capacity, therefore increasing their likelihood to be able to manage and selfregulate their response to stressors in other areas of their lives.

#### Strengths of the Study

This study is unique in that it used a mixed methods longitudinal design, to investigate the impacts of participation in an extended journey style outdoor education program and its perceived effectiveness on enhancing resilience and coping skills in Australian young people, which is rare in this field of study. The study used empirical testing in Phase I to facilitate critical inquiry, validation of the data and enabled detailed annotations from Phase II and Phase III to support the findings. The collection of the qualitative data provided personal accounts which offered a greater understanding of the quantitative results. Moreover, by combining both quantitative and qualitative analyses, the researcher was able to obtain rich data for analysis which supports the methodological integrity of the study (Onwuegbuzie & Teddlie, 2003).

Research in this field has previously explored and measured both resilience and coping as separate constructs. However, Booth's (2015) research appears to be the first investigation to examine the relationships between both resilience and coping in the outdoor education domain. This study builds upon Booth's (2015) findings and provides a longitudinal data set which primarily

focuses on personal development sustainability and transference of skills sixmonths after the intervention.

Outdoor education practitioners and field staff can benefit from knowing the findings of this study, that outdoor education programs do have positive significant influences on the development of resilience and coping skills. The realworld applications of the results and implications for future research, and practice provided in this chapter, can be applied in practice by outdoor education practitioners, educators, schools and companies to not only foster the development of resilience and coping skills among young people but also provide the best opportunities for participants to transfer these skills into other contexts of their lives.

#### Limitations of the Study

While all research is subject to limitations through scope and sampling techniques, there are also limitations relating to data collection. For example, in this study, there was a potential problem with collecting data immediately after completion of the 'City to Summit', due to 'post-group euphoria' which is a common occurrence when people complete challenging experiences. Post-group euphoria can be a perplexing problem in the measurement of outdoor education and adventure education outcomes as it may accentuate some responses and muffle others (Ewert, 1989; Hattie et al., 1997; Marsh, Richards, & Barnes, 1986a, 1986b; Sibthorp, 2006). Marsh et al. (1986a) suggests that examining outcomes over the longer-term may reduce this phenomenon as further data is collected as the post-group euphoria subsides. The data collected six months post-program was used to inform the influence of post-group euphoria.

Gender differences in resilience and coping strategies are not explored in this study. In the context of this research, male adolescents aged 14 to 17 years were the participants in both the program and control groups.

There are also issues associated with the social representativeness of both sample groups. The young people sampled in this study all attended a private, single-sex boy's schools in Melbourne. The nature of the schools being private places a socio-economic selection with the boys coming from families who are able to pay the school fees and have an affinity with the school ethos.

Although this research examined self-reported levels of resilience and coping before the program, it did not assess the boys' current mental health or

psychological states. It is important to reflect on the use of self-report measures in this research, including the possibility that some participants may have underreported or over-reported their levels of resilience and coping due to varied conceptualisations and understandings of the concepts, as well as taking into account the participants ability to accurately evaluate, consider and report their response to the questionnaires.

The measures used were applicable in this study, however, both measures had limited intensive psychometric analysis. These measures can continue to be improved through new phases of psychometric analysis undertaken in future research.

Questions in the Resilience Scale are positively worded which could be viewed as providing 'leading questions', which can lead to a more 'socially desirable' response from the participants. In addition, the interviewees of the qualitative data may have been influenced by peers within the small group interviews, causing them to answer questions in a socially desirable manner, rather than responding to the questions with accurate, truthful responses.

The small sample size warrants consideration under the current study design. Even though (Hinkle et al., 2003) suggests that a total of 42 participants is required to achieve a large effect where d = 1.0,  $\alpha = .05$ , and power = .75, which was the final number of the control group, larger sample sizes may produce more conclusive and justifiable results. It is imperative to note that this study did not measure the actual individual stressors that participants may have experienced during 'City to Summit'. It should also be noted that the relatively small sample sizes and gender specificity; selective memory or lack of memory; or the ability or inability to recall past experiences over time; may cause limits on the generalisability of the outcomes of the qualitative data.

The complex and dynamic variables that are specific to outdoor education programs, such as the environment, program location, weather, group leader influences and impacts of the social group, cannot always be controlled. Consequently, these variables may have impacted the results of the study.

The use of small group, semi-structured interviews may influence participants to be reluctant to share their view with others in the group. However, these types of qualitative data assessments are suitable for an adolescent sample group.

### **Implications and Recommendations for Future Research**

Several methodological concerns were identified after collection of the quantitative data sets in Phase I. For instance; it would have been beneficial to have a longitudinal study that supported the qualitative data by using the same measures to empirically examine the program group and the control group six months prior to starting the 'City to Summit' and six months post completion. This follow-up data set could have been collected at the same time as the semistructured interviews in the post-data collection set.

While this study used a qualitative longitudinal design and showed that resilience is enhanced by participation in extended journey style outdoor education programs immediately after participation, future research should aim to include longitudinal designs that also include empirical measures to examine participants resilience capacity after a period of time, to examine whether initial increases in levels of resilience and coping skills show lasting results in other contexts. It is also recommended that future research using the Resilience Scale should consider adapting the tool by potentially including some negatively worded questions to ameliorate the effects of having all questions being positively worded. In addition, a Confirmatory Factory Analysis should be the test used to further investigate the validity of the factor structure of the Brief COPE Scale in future research.

This study did not take into account or measure participants' prior experiences in the context of exposure to challenge and stressors during outdoor education programs. Therefore, it is recommended that future research in this field address participants' prior experiences, as well as examining what specific types of challenges and what the specific level of difficulty of the challenges positively influences the participants and leads to positive outcomes, such as expanding and strengthening an individual's resilience capacity.

Additional research is warranted to determine whether gender difference impacts the development of resilience and coping after participation in extended journey style outdoor education programs. To enable a well-rounded view, future studies should examine various genders and groups from various socio-economic backgrounds, as well as being sourced from different cultures and counties.

Generally speaking, most mainstream outdoor education programs in Victoria name the 'development of resilience' as a program outcome. This is
problematic as it infers that all outdoor education programs, even the ones without intentional program design, do enhance resilience. If outdoor education practitioners and companies continue to claim that their programs develop resilience and coping skills in their participants, then there is a need for new teaching models and program designs in outdoor education. These new teaching models need to be grounded in research, able to be tested so as to determine validity, be simple to understand for practitioners, straight forward to use while including new ways of thinking.

As learning takes place in the brain, future research is warranted to further investigate the neurobiological processes that impact our ability to learn, retain information and skills learnt in outdoor education contexts. New and current research in psychoeducation, neuroeducation (Schenck & Cruickshank, 2015, p. 83), brain resilience (Allan et al., 2012) and neuroplasticity may help to inform best practices and educational models that can be applied in outdoor education programs.

There is a need for future research to investigate how to implement and integrate psychoeducational curriculum about developmental assets, coping strategies and stress in outdoor education programs which aim to bolster psychological resilience. The recommended Resilience Development Program Design (RDPD) presented in this chapter is based on research, extensive theoretical underpinnings and findings of this study. The intent is that the RDPD will lead to new research, new questioning and start further conversions about how best to support young people in not only developing these essential life skills

441

but also support them in transferring the skills in all contexts of their lives. It is essential for the proposed RDPD to be trialled and tested in a range of environments, with varying age groups, genders and demographics. It is advocated that the RDPD will evolve with testing and research, and should help to inform 'best practices' in building resilience and coping skills during outdoor education programs.

### **Chapter Summary**

This thesis strived to strengthen the research that supports the belief that participation in outdoor education programs enhances resilience, resilience attributes and coping skills, and to understand what practices support the development and transference of these psychosocial attributes and skills into other contexts. Given the self-reported increases in resilience and internal and external assets which includes coping skills; combined with the findings of the data collected from the Resilience Scale and the Brief COPE questionnaire's, the findings support the notion that outdoor education is a viable tool to for the development of psychosocial attributes.

The complex, multidimensional constructs of resilience and coping made untangling the relationships of the data difficult. Subsequently, this research confirmed the importance of using a mixed method approach to explore the themes revealed across the data sets. For instance, the results of both the qualitative and quantitative data emphasise that an individual's overall resilience capacity can be strengthened and enhanced by the development of the internal and external assets and copings skills (refer to the 'resilience capacity flourishing flower' in Figure 6.2). As resilience can only be demonstrated in times of challenge or adversity, it appears the development of these skills also occurs when participants are exposed to risk, challenges and adventure in an unfamiliar environment. Providing experiences of this type in a supported environment acts as a type of stress inoculation, through exposure to program challenges. Stress inoculation and experiential learning approaches used in the three-phase program design also created opportunities for individuals and groups to be exposed to different types and levels of challenges and provided an ideal platform for relationship development, reflection and personal growth to occur.

Furthermore, the results of this study demonstrate that providing opportunities for participants to take meaningful ownership over their engagement in challenges that are within the 'Goldilocks Zone', create conditions for both optimal arousal and peak adventure experiences at an individual level. The extended program length allowed time for the boys to practice and develop their skills in challenging situations while being in a supportive environment with peers and leaders. The combination of these factors resulted in participants feeling a sense of achievement, empowerment and higher levels of motivation to participate in future challenges. These findings, along with findings from current research (e.g., Booth, 20015; Shellman, 2009) indicate that outdoor education programs that provide the opportunity to experience achievement through exposure to 'just manageable' program challenges is a critical element in improving participants resilience capacity and coping skills. Leaders assumed various critical roles throughout the program to support the participants' growth and development. For instance, the results of this study demonstrate the importance and impact the group leaders had in the refection and debriefing process. When the participants were guided through the experiential learning processes and supported to conceptualise and understand key concepts of resilience; they were more likely to transfer the skills and assets developed during the program into other contexts of their lives. However, while the results of this study provide supporting evidence that some resilience attributes and coping skills developed can be developed, transferred and applied in other contexts of their lives, there is little evidence to support the long-term effects of these results.

The reported transference of skills and assets was more likely to occur when the concept, skill or asset involved specific transfer; however, when concepts and knowledge involved skills and tasks that involved non-specific transfer (far transfer), the boys struggled to make connections between the contexts of learning. Therefore, assisting participants to reflect on their experiences, supporting them to think abstractly, and helping them to understand how they can apply the skills learnt during the program into other domains of their lives, were critical roles that the leaders assumed during the program (e.g., the leader engaging in the role of the 'translator').

The findings of this study indicated that if an individual's contextual and conceptual understanding of resilience is limited, resilience is perceived to be content- and context-specific to the setting in which it is developed, unless the participants were guided to make clear connections between the different contexts of learning and application through supported debriefing and reflection processes. Because the boys lacked an understanding of some of the more complex concepts (e.g., the overall concept of resilience) and because the various environments 'appeared' to be so different to the boys, it may be useful to simplify the concepts associated with understanding resilience, so that the young people can connect with, and understand how specific assets can help them to cope and how they can apply skills in other areas of their lives (e.g., being more self-reliant or applying problem-solving skills).

Overall, the present findings provide a number of recommendations for practitioners. In essence, program design was the most significant contributing factor in supporting the development of resilience, learning effective coping skills and transferring these skills into other contexts. The way that a program is designed is integral to developing a framework that; (a) is consistent with the program aims and outcomes, (b) caters for participants 'core' and 'conditioning' layers, (d) addresses risk management guidelines, (e) provides appropriate levels of challenges, and (f) considers and plans for the transference of learning outcomes into other contexts.

Consideration must be given to the research findings that emphasise the importance of post-activity outdoor education during the Integration Phase of program design, as it was critical in supporting the transference of skills. The results emphasise that more needs to be done to support the transference and longevity of these skills during the Integration Phase, by providing opportunities to practice, develop, reflect and refine the skills learnt during the program. Postactivity outdoor education during the Integration Phase should be considered as an equally important element of program design as the Immersion Phase of programming.

An individual's resilience capacity depends on their ability to draw upon their internal and external assets, and the extent of adaption displayed by an individual in specific contexts may be related to the extent in which the environment has supporting factors that nurture the development of resilience (Ungar, 2008). Consequently, it is suggested that by combining proactive, strength-based, stress inoculation approaches into outdoor education program design, interventions can adopt a preventative approach to mental health and wellbeing by educating young people on specific coping skills and assets that may help them to cope.

The conclusions of this study will have significant applications in supporting schools and educators to develop programs that use best practices to expand an individual's resilience capacity and foster the transference of skills and assets long-term. The recommended RDPD presented in this chapter provides practical applications for practitioners to apply intentional program design that endeavours to develop effective coping skills and resilience during the program and how to best support the transference of these skills into other contexts.

As the dual axis model of mental health indicates, (see Figure 2.2) schools can aim to improve the health and well-being of everyone regardless of their current mental health status. Overall, the research findings indicated that by providing adolescents, who are still establishing their own identities, values and

446

beliefs, with opportunities for personal growth; schools and educators can use outdoor education programs to strengthen and develop psychosocial attributes and coping skills. The three-phase outdoor education program design (i.e., RDPD) can foster the development and transference of developmental skills and assets, and support the maintenance of young people's physical, mental, social and emotional well-being by expanding their resilience capacity and helping them progress to become more resilient individuals (Friedli, 2009; Wagnild, 2009, 2010; World Health Organization, 2009).

## **CHAPTER 7: SUMMARY AND CONCLUSION**

It is good to have an end to journey towards,

but it is the journey that matters, in the end

– Ursula le Guin



Figure 7.1. Last day sunrise hike to Mt Kosciusko.

The purpose of this study was to investigate if resilience is context specific to the field of outdoor education. Specifically, the thesis examined the impact of participation in an extended journey style outdoor education program on enhancing levels of resilience and coping skills in adolescent boys immediately after participation in the 21-day program, and again six months after the program concluded. The study also investigated how resilience and coping skills can be best developed and transferred into other contexts of learning. Statistically significant associations were found between the study variables within the program group and the control group. As the final chapter of this thesis, this chapter summarises the conclusions drawn from the quantitative and qualitative data and encapsulates a synopsis of the implications for practitioners and recommendations for future research.

#### Summary

The literature review presented in Chapter 2 establishes the importance of strengthening developmental assets and coping skills to foster the enhancement of adolescents' psychological resilience capacity. Outdoor education interventions that use a stress inculcation approach in the program design provide a 'learning playground' for participants to develop and practice coping skills and resilience attributes (Booth, 2015; Meichenbaum, 2017b; Rutter, 1993). In accordance, the challenge of this study was to measure the impacts of the outdoor education program on building resilience and coping skills, to analyse the short-term and long-term effects on young people and use these findings to assist in improving outdoor education program design that fosters the development and transference of these essential developmental skills.

The post-positivist, constructivist approach to the mixed method design allowed for the quantitative data to be cross-examined and reinforced by the qualitative data. The findings of this thesis advocate that an individual's resilience capacity is increased by developing and strengthening their internal and external assets through exposure to gradual stressful and challenging experiences that require adaption during outdoor education programs. It was critical to support the participants to develop their individual understandings of how they interact with their environments, how they respond to stress and challenges, and how they interact with people. However, resilience was perceived to be content- and context specific to the setting in which it is developed, unless the participants were supported through the debrief and reflection processes and guided to make clear connections between the different contexts of learning and application. To assist in the transference of skills and assets post-program, it may be more beneficial to simplify the ideas associated with understanding the concept of resilience and developmental assets, and explicitly teach psychoeducational curriculum and coping skills so that the young people can connect with, and understand how specific assets can help them to cope and how they can use these assets in other domains of their lives. As a result of the findings, the following conclusions can be drawn:

- Significant, positive changes were evident in the program group for resilience, compared to the control group. In addition, significant, positive changes were evident in several resilience attributes, such as perseverance, existential aloneness and purposeful life, indicating that outdoor education programs can have a small positive effect on a participant's overall sense of resilience.
- 2. Levels of internal developmental assets, such as perseverance, selfreliance and determination, can be enhanced through participation in extended journey style outdoor education programs.

- 3. Levels of external developmental assets, such as positive relationships with peers, leaders and family members, can be enhanced through participation in extended journey style outdoor education programs.
- Coping skills developed and strengthened during the outdoor education program, acted as both internal assets and external assets to mediate individuals responses to stress.
- Outdoor education programs can reduce coping mechanisms that undermine well-being in youth, such as Substance Use and Behavioural Disengagement.
- Outdoor education program design that focuses on the three critical stages of program design (Frontloading, Immersion and Integration Phases) can assist and support participants to transfer of learning into other contexts.
- 7. Outdoor education practitioners can actively assist young people to develop their resilience capacity through facilitating ongoing reflection and personal awareness activities that focus on an individual's understandings of their goals, attitudes, thoughts, behaviours and actions.
- Outdoor education practitioners can actively assist young people to develop coping skills through explicitly teaching various coping mechanisms and guiding them to become self-aware.
- 9. Outdoor education practitioners can actively assist young people to transfer their learning of resilience attributes and coping skills by

helping them to conceptually understand the concepts and make links between the application of the skills in different contexts.

10. Young people are very adaptable; however, without follow-up postprogram, in most cases, the young people in this study adapted straight back to the life they were living previous to their learning experiences.

### Conclusions

In conclusion, this study provided valuable support for applied methodological and empirical contributions to knowledge and research about the impacts of outdoor education programs on the development and transference of psychological resilience and coping skills into other contexts. This thesis supports past research that recognises that problem-focused coping strategies are associated with adaptive outcomes, such as enhanced resilience. The findings of this study demonstrate an important step in validating that psychological resilience, resilience attributes, developmental assets and coping skills can be enhanced through participation in extended journey style outdoor education programs.

However, the results also indicate that additional support is required to integrate and foster the transference of learning, and to maintain, continuously strengthen and further develop these essential life skills after participation in the intervention. It appears that outdoor education programs are currently being designed and implemented based on facilitator intuition, with the idea that 'because programs may have a positive effect on the development of resilience and coping skills, that these skills will automatically be transferred and utilised by participants in non-outdoor education settings'. Except, the findings of this research reveal that this is not the case and more needs to be done to foster the transference of these skills. Therefore, the recommendation presented in the previous Chapter that outdoor education program designs be revised to include a follow-up and Integration Phase, for interventions that intend to develop and transfer psychological resilience and coping skills in young people provides an integral concluding perspective.

These findings provide important data to guide future research to further investigate how outdoor education programs can be designed to intentionally enhance the development of resilience attributes and coping skills in participants and assist them to transfer and apply these skills in other contexts. The implications and recommendations of this research aim to inform teachers, schools, outdoor educators and industry providers, with the knowledge to develop and implement, effective outdoor education programs that aim to enhance resilience attributes and coping strategies in young people.

Schools continue to play a vital role in helping young people to maintain mental health and well-being. This was further enforced by the data that showed that schools who employ outdoor education programs can assist their students to develop their resilience capacity, developmental assets and coping skills. In addition, the programs curriculum and design impacts and influences the transference of these skills learnt during the outdoor education program into other contexts. As such, instead of taking a reactive approach, schools can take a preventative approach to mental health promotion, by utilising outdoor education

453

programs as an intervention to develop coping skills and their resilience capacity. By providing opportunities for young people to be engaged in outdoor education programs, that explicitly aim to develop their resilience capacity and repertoire of effective coping skills, schools can continuously support their students to transfer their learnings and integrate these skills into various life contexts.

Similarly, as an outcome of the findings, it is recommended that mainstream outdoor education programs aim to take a preventative approach to promote mental health and well-being in young people, by implementing program designs based on proactive resilience development frameworks, such as the Resilience Development Program Design (RDPD) guidelines presented in this research. It is the responsibility of schools and educators to continuously recap learning's, assist young people to transfer the skills learnt during programs and support them to integrate these skills long-term into contexts other than outdoor education.

As the field of outdoor education continues to move forward in terms of evidenced-based practices and empirical research, there is a continued need for the development of models (e.g., stress inoculation training and the RDPD guidelines), as well as substantiated theories in the areas of neuroeducation and psychoeducation that will enable the field to move forward with new findings and current research. Future research is required to; (1) further understand how learnings and skills developed during programs can be best supported, transferred, maintained and integrated into other contexts of participants daily lives; (2) how psychoeducational curriculum and implement proactive mental health promotion and prevention strategies, such a positive psychology and 'prehabilitation philosophies'; and (3) how cognitive behavioural techniques, such as cognitive reframing, and problem-solving, solution-focused approaches can be implemented into program design and implementation.

Ultimately, demonstrating positive change in young people's resilience and coping skills through outdoor education as an intervention is vital. By building on this work, researchers will be able to provide a clearer direction by documenting and examining not only how participants have changed as a result of the intervention, but also record how these changes impact their future and their future responses to stress and life challenges in varied contexts.

# **Concluding Comments**

Outdoor education programs and practitioners usually exist because they want to help people. The researcher believes that outdoor education programs can achieve the key ingredients of successful learning, optimal learning environments and personal development of essential life skills during and after the program through intentional program design and implementation, such as the RDPD presented in this study.

Whatever criticisms of the RDPD that has been suggested, the researcher posits that the guidelines warrant further investigation to understand the benefits and outcomes of the recommended program design. There are so many different variables that affect learning outcomes and the transferability of skills postparticipation in outdoor education programs. However, fundamentally, it is important to take a lifelong learning approach, where educators support the integration of essential life skills developed during outdoor education programs into other contexts of young people's lives.

# REFERENCES

- Ahern, N. R. (2006). Adolescent resilience: An evolutionary concept analysis. Journal of pediatric nursing, 21(3), 175-185. doi:10.1016/j.pedn.2005.07.009
- Ahern, N. R., Ark, P., & Byers, J. (2008). Resilience and coping strategies in adolescents. *Paediatric nursing*, 20(10), 32-36. doi:10.7748/paed2008.12.20.10.32.c6903
- Allan, J. F., McKenna, J., & Hind, K. (2012). Brain resilience: Shedding light into the black box of adventure processes. *Australian Journal of Outdoor Education*, 16(1), 3-14.
- Allen-Craig, S., & Miller, J. (2007). An evaluation of a cross section of Australian schools outdoor education program outcomes. Paper presented at the 15th National Outdoor Education Conference, Ballarat, Victoria.
- Allen, J. A. (1994). The constructivist paradigm: Values and ethics. *Journal of Teaching in Social Work*, 8(1-2), 31-54. doi:10.1300/J067v08n01\_03
- Allin, L., & Humberstone, B. (2010). Introducing 'journey(s)' in adventure and outdoor learning research. *Journal of Adventure Education and Outdoor Learning*, 10(2), 71-75. doi:10.1080/14729679.2010.539065
- American Psychological Association. (2011). *The road to resilience*. Retrieved from http://www.apa.org/helpcenter/road-resilience.aspx.
- Anderson, V. A., Anderson, P., Northam, E., Jacobs, R., & Catroppa, C. (2001). Development of executive functions through late childhood and adolescence in an Australian sample. *Developmental Neuropsychology*, 20(1), 385-406.
- Anderson, A., Hattie, J., & Hamilton, R. J. (2005). Locus of control, self-efficacy, and motivation in different schools: Is moderation the key to success? *Educational Psychology*, 25(5), 517-535.
- Anderson, P. (2008). Challenging experiences: What do students learn? *International Schools Journal*, 28(1), 55-58.
- Annerstedt, M., & Wahrborg, P. (2011). Nature-assisted therapy: Systematic review of controlled and observational studies. *Scandinavian Journal of Public Health*, 39(4), 371-388. doi:10.1177/1403494810396400
- Antoni, M. H., Ironson, G. H., & Scheiderman, N. (2007). *Cognitive-behavioral stress management*. New York: Oxford University Press.
- Arnett, J. J. (1999). Adolescent storm and stress, reconsidered. *American Psychologist*, 54(5), 317-326. doi:10.1037/0003-066X.54.5.317
- Artinian, N. T., Abrams, J., Keteyian, S. J., Franks, M. M., Franklin, B., Pienta, A., . . . Schwartz, S. (2009). Correlates of depression at baseline among African Americans enrolled in cardiac rehabilitation. *Journal of*

*cardiopulmonary rehabilitation and prevention, 29*(1), 24-31. doi:10.1097/HCR.0b013e31819276dd

- Association for Experiential Education. (2018). What is experiential education? Retrieved from http://www.aee.org/index.php?option=com\_content&view=article&id=11 0:what-is-ee&catid=20:other&Itemid=260
- Australian Bureau of Statistics. (2007). *National survey of mental health and wellbeing: Summary of results* (No. 4326.0). Retrieved from http://www.abs.gov.au/ausstats/abs@.nsf/mf/4326.0.
- Australian Bureau of Statistics. (2008a). *National survey of mental health and well-being: Summary of results 2007.* (No. 4326.0). Retrieved from http://www.abs.gov.au/ausstats/abs@.nsf/Latestproducts/4326.0Media%20 Release12007?opendocument&tabname=Summary&prodno=4326.0&issu e=2007&num=&view=.
- Australian Bureau of Statistics. (2008b). *Population Projections, Australia, 2006 to 2101*. Retrieved from http://www.abs.gov.au/ausstats/abs@.nsf/mf/3222.0.
- Australian Bureau of Statistics. (2012). *Australian social trend: Using statistics to paint a picture of Australian society*. Retrieved from http://www.ausstats.abs.gov.au/ausstats/subscriber.nsf/0/701575637CC4B BE3CA2579CE000BB833/\$File/41020\_astmar2012.pdf.
- Australian Bureau of Statistics. (2016). *Causes of death in Australia 2015*. (No. 3303.0). Canberra, Australia: Australian Bureau of Statistics Retrieved from

http://www.abs.gov.au/ausstats/abs@.nsf/Lookup/by%20Subject/3303.0~2 015~Main%20Features~Intentional%20self-harm:%20key%20characteristics~8.

- Australian Curriculum and Assessment Authority. (2012). General capabilities in the Australian curriculum. Retrieved from http://www.australiancurriculum.edu.au/GeneralCapabilities/General%20c apabilities.pdf
- Australian Curriculum and Assessment Authority. (2015). Guide to understanding ICSEA (index of community socioeducational advantage) values from 2013 onwards. Retrieved from https://acaraweb.blob.core.windows.net/resources/Guide\_to\_understandin g\_icsea\_values.pdf
- Australian Institute of Health and Welfare. (2011). *Risk Factors*. Retrieved from http://www.aihw.gov.au/risk-factors/.
- Australian Institute of Health and Welfare. (2016). Figure 3.2: Leading underlying causes of death, by age group, 2014–2016. *National Mortality Database*.

- Badr, H. (2004). Coping in marital dyads: A contextual perspective on the role of gender and health. *Personal relationships*, 11(2), 197-211. doi:10.1111/j.1475-6811.2004.00078.x
- Baker, J. P., & Berenbaum, H. (2007). Emotional approach and problem-focused coping: A comparison of potentially adaptive strategies. *Cognition and Emotion*, 21(1), 95-118. doi:10.1080/02699930600562276
- Bakker, A. B. (2005). Flow among music teachers and their students: The crossover of peak experiences. *Journal of Vocational Behavior*, 66(1), 26-44. doi:10.1016/j.jvb.2003.11.001
- Baldwin, T. T., & Ford, J. K. (1988). Transfer of training: A review and directions for future research. *Personnel Psychology*, 41(1), 63-105. Retrieved from https://onlinelibrary.wiley.com/doi/pdf/10.1111/j.1744-6570.1988.tb00632.x
- Ballenger-Browning, K., & Johnson, D. C. (2010). Key facts on resilience. Naval Center for Combat and Operational Stress Control. doi:10.1177/070674371105600504
- Banatao, E. J. (2011). Educational resilience: The relationship between school protective factors and student achievement. (Doctoral dissertation), San Diego State University, San Diego, CA. Retrieved from https://eric.ed.gov/contentdelivery/servlet/ERICServlet?accno=ED521456
- Bandura, A. (1977). Self-efficacy: Toward a unifying theory of behavioral change. *Psychological Review*, *84*(2), 191-215. doi:10.1037/0033-295X.84.2.191
- Bandura, A. (1986). Social foundations of thought and action: A social cognitive theory. Englewood Cliffs, NJ: PrenticeHall.
- Bandura, A. (1994). Self-efficacy. In V. S. Ramachaudran (Ed.), *Encyclopedia of Human Behavior* (Vol. 4, pp. 71-81). New York, NY: Academic Press.
- Bandura, A. (2006). Adolescent development from an agentic perspective. In F.
   Pajares & T. Urdan (Eds.), *Self-efficacy beliefs of adolescents* (pp. 1-43).
   Hartford, CT: Information Age Publishing.
- Barkley, E. F. (2010). Student engagement techniques: A handbook for college faculty. In The Jossey-Bass igher and adult education series. Retrieved from https://ebookcentral.proquest.com
- Barrett, P. M., Cooper, M., & Guajardo, J. G. (2014). Using the friends programs to promote resilience in cross-cultural populations. In S. Prince-Embury & D. H. Saklofske (Eds.), *Resilience interventions for youth in diverse populations* (pp. 85-108). New York, NY: Springer.
- Bartels, M., & Hudziak, J. J. (2007). Genetically informative designs in the study of resilience in developmental psychopathology. *Child and Adolescent Psychiatric Clinics of North America*, 16(2), 323-340. doi:10.1016/j.chc.2006.12.008

- Bartone, P. T. (1995). *A short hardiness scale*. Paper presented at the Annual Convention of the American Psychological Society, New York, NY.
- Bartone, P. T. (2007). Test-retest reliability of the Dispositional Resilience Scale-15, a Brief Hardiness Scale. *Psychological Reports*, *101*(3), 943-944. doi:10.2466/PR0.101.7.943-944
- Basile, C. G., & Copley, J. V. (1997). The effect of an outdoor nature investigation program on young children's ability to transform knowledge.
  Paper presented at the the Annual Meeting of the American Educational Research Association, Chicago, IL.
- Bedard, R. M., Rosen, L. A., & Vacha-Haase, T. (2003). Wilderness therapy programs for juvenile delinquents: A meta-analysis. *Journal of Therapeutic Wilderness Camping*, 3(1), 7-13.
- Beightol, J., Jevertson, J., Gray, S., Carter, S., & Gass, M. A. (2009). The effect of an experiential, adventure-based "anti-bullying initiative" on levels of resilience: A mixed methods study. *Journal of Experiential Education*, 31(3), 420-424. Retrieved from https://journals.sagepub.com/doi/pdf/10.1177/105382590803100312
- Bell, B. (2003). The rites of passage and outdoor education: Critical concerns for effective programming. *The Journal of Experiential Education*, 26(1), 41-50.
- Benard, B. (1991). Fostering resiliency in kids: Protective factors in the family, school, and community. Portland, OR: Western Center for Drug-Free Schools and Communities.
- Benard, B. (1995). *Fostering resilience in children*. (No. EDO-PS-95-9). Champaign, IL: ERIC Clearinghouse on Elementary and Childhood Education Retrieved from https://eric.ed.gov/?id=ED386327.
- Benard, B. (2004). *Resiliency: What we have learned*. San Francisco, CA: WestEd.
- Bennett, K., Manassis, K., Walter, S. D., Cheung, A., Wilansky-Traynor, P., Diaz-Granados, N., . . . Barrett, P. (2013). Cognitive Behavioral Therapy age effects in child and adolescent anxiety: An individual patient data metaanalysis. *Depression and anxiety*, 30(9), 829-841.
- Benson, P. L. (1997). All kids are our kids: What communities must do to raise caring and responsible children and adolescents. San Francisco, CA: Jossey-Bass.
- Benson, P. L. (2007). Developmental assets: An overview of theory, research, and practice. In R. K. Silbereisen & R. M. Lerner (Eds.), Approaches to positive youth development (pp. 33-58). London, UK: Sage Publications. Retrieved from https://uk.sagepub.com/en-gb/.
- Benson, P. L., Leffert, N., Scales, P. C., & Blyth, D. A. (2012). Beyond the "village" rhetoric: Creating healthy communities for children and

adolescents. *Applied developmental science*, *16*(1), 3-23. doi:10.1080/10888691.2012.642771

- Benson, P. L., Scales, P. C., Leffert, N., & Roehlkepartain, E. C. (1999). A fragile foundation: The state of developmental assets among American youth. In. Retrieved from https://www.search-institute.org/
- Berman, D. S., & Davis-Berman, J. L. (1989). Wilderness Therapy: A therapeutic adventure for adolescents. *Journal of Independent Social Work*, *3*(3), 65-77. doi:10.1300/J283v03n03\_06
- Berman, M. G., Jonides, J., & Kaplan, S. (2008). The cognitive benefits of interacting with nature. *Psychological Science*, 19(12), 1207-1212. doi:10.1111/j.1467-9280.2008.02225.x
- Bettmann, J. E. (2012). *Therapeutic outcomes of wilderness therapy for adolescent and young adult populations*. Paper presented at the American Psychological Association Convention, Orlando, FL.
- Block, J., & Kremen, A. M. (1996). IQ and ego-resiliency: Conceptual and empirical connections and separateness. *Journal of Personality and Social Psychology*, 70(2), 349-361. doi:10.1037/0022-3514.70.2.349
- BoingBoing. (2012). *Resilience Framework (Children & Young People)*. In. Retrieved from http://www.boingboing.org.uk/index.php/resources/category/9-resilienceframeworks#
- Bonanno, G. A. (2004). Loss, trauma, and human resilience: Have we underestimated the human capacity to thrive after extremely aversive events? *American Psychologist*, *59*(1), 20-28. doi:doi:0.1027/0003-066X.59.1.20
- Booth, J. W. (2015). *Coping strategies and development of psychological resilience in outdoor education*. (Unpublished honours thesis), University Canberra, Canberra, Australia.
- Booth, J. W., & Neill, J. T. (2017). Coping strategies and the development of psychological resilience. *Journal of Outdoor & Environmental Education*, 20(1), 47-54. Retrieved from http://outdooreducationaustralia.org.au/joee
- Boros, E., Hammer, P. L., Ibaraki, T., Kogan, A., Mayoraz, E., & Muchnik, I. (2000). An implementation of logical analysis of data. *IEEE Transactions* on Knowledge and Data Engineering, 12(2), 292-306. doi:10.1109/69.842268
- Bounce Back. (2017). What is resilience? Retrieved from http://www.bouncebackproject.org/resilience/
- Bowen, D. J., & Neill, J. T. (2013). A meta-analysis of adventure therapy outcomes and moderators. *The Open Psychology Journal*, *6*, 28-53. doi:10.2174/1874350120130802001

- Bowen, D. J., & Neill, J. T. (2015). Preliminary findings from a meta-analysis of adventure therapy program effects. In C. L. Norton, C. Carpenter, & A. Pryor (Eds.), Adventure therapy around the globe: International perspectives and diverse approaches (pp. 219-241). Champaign, Illinois: Common Ground Publishing LLC.
- Bowen, D. J., Neill, J. T., Williams, I. R., Mak, A. S., Allen, N. B., & Olsson, C. A. (2016). A profile of outdoor adventure interventions for young people in Australia(8), 26-40. Retrieved from https://doi.org/10.18666/JOREL-2016-V8-I1-7281
- Bradley, S. J. (1990). Affect regulation and psychopathology: Bridging the mindbody gap. *Can J Psychiatry*, *35*(6), 540-547.
- Brantlinger, E., Klingner, J., Richardson, V., & Taylor, S. J. (2005). Importance of experimental as well as empirical qualitative studies in special education. *Mental Retardation*, 43(2), 92-119. doi:10.1352/0047-6765(2005)43<92:IOEAWA>2.0.CO;2
- Breinbauer, C., & Maddaleno, M. (2005). Youth: choices and change: promoting healthy behaviors in adolescents. Retrieved from https://www.paho.org/hq/
- Brendtro, L. K., & Strother, M. A. (2007). Back to Basics Through Challenge and Adventure. *Reclaiming Children & Youth*, *16*(1), 2-6.
- Bronfenbrenner, U. (1977). Toward an experimental ecology of human development. *American Psychologist*, *32*(7), 513-531. doi:10.1037/0003-066X.32.7.513
- Bronfenbrenner, U. (1979). *The ecology of human development*. Cambridge, MA: Harvard University Press.
- Bronfenbrenner, U. (1981). *The ecology of human development: Experiments by nature and design*. In. Retrieved from http://www.hup.harvard.edu/catalog.php?isbn=9780674224575&content=r eviews
- Bronzaft, A. L. (2003). Noise pollution: A hazard to physical and mental wellbeing. In R. B. Bechtel, A. Churchman, & A. Ts'erts'man (Eds.), *Handbook of environmental psychology* (1 ed., pp. 499-510). New York, NY: John Wiley & Sons.
- Brookes, A. (2003a). A critique of Neo-Hahnian outdoor education theory. Part one: Challenges to the concept of "character building". *Journal of Adventure Education and Outdoor Learning*, 3(1), 49-62. doi:10.1080/14729670385200241
- Brookes, A. (2003b). A critique of Neo-Hahnian outdoor education theory. Part two:"The fundamental attribution error" in contemporary outdoor education discourse. *Journal of Adventure Education and Outdoor Learning*, *3*(2), 119-132. doi:10.1080/14729670385200311

- Brookes, A. (2004). Outdoor education fatalities in Australia 1960-2002. Part 3: Environmental circumstances. *Australian Journal of Outdoor Education*, 8(1), 44-56.
- Brooks, N. (2003). Vulnerability, risk and adaptation: A conceptual framework. (Working Paper No. 38). Norwich, UK: Tyndall Centre for Climate Change Research Retrieved from https://www.researchgate.net/profile/Nick\_Brooks2/publication/20003274
  6\_Vulnerability\_Risk\_and\_Adaptation\_A\_Conceptual\_Framework/links/0 fcfd50ac169e15865000000.pdf.
- Brown, M. (2002). The facilitator as gatekeeper: A critical analysis of social order in facilitation sessions. *Journal of Adventure Education and Outdoor Learning*, 2(2), 101-112. Retrieved from https://doi.org/10.1080/14729670285200211
- Brown, M. (2008). Comfort zone: Model or metaphor. *Australian Journal of Outdoor Education, 12*(1), 3-12.
- Brown, M. (2010). Transfer: Outdoor adventure education's achilles heel? Changing participation as a viable option. *Australian Journal of Outdoor Education, 14*(1), 13-22.
- Bruckauf, Z. (2017). Adolescents' mental health: Out of the shadows. Evidence on psychological well-being of 11-15-year-olds from 31 industrialized countries. (No. 2017-12). Innocenti, Florence: Innocenti Research Retrieved from https://www.unicef-irc.org/publications/896/.
- Brymer, E. (2002). Exploring expedition research methodology: A personal reflection. *Journal of Outdoor and Environmental Education*, 6(2), 44-57. Retrieved from https://outdooreducationaustralia.org.au/joee/
- Buchanan, C. M., Eccles, J. S., Flanagan, C., Midgley, C., Feldlaufer, H., & Harold, R. D. (1990). Parents' and teachers' beliefs about adolescents: Effects of sex and experience. *Journal of Youth & Adolescence*, 19(4), 363-394. doi:10.1007/BF01537078
- Buckner, E. B., Hawkins, A. M., Stover, L., Brakefield, J., Simmons, S., Foster, C., . . . Dubois, G. (2005). Knowledge, resilience, and effectiveness of education in a young teen asthma camp. *Pediatric Nursing*, 31(3), 201-210. Retrieved from https://www.pediatricnursing.org/
- Bullot, A., Cave, L., Fildes, J., Hall, S., & Plummer, J. (2017). Mission Australia youth survey report 2017. *Mission Australia*.
- Burdette, H. L., & Whitaker, R. C. (2005). Resurrecting free play in young children: Looking beyond fitness and fatness to attention, affiliation, and affect. *Archives of Pediatrics and Adolescent Medicine*, *159*(1), 46-50.
- Burke, J. R. (1997). Examining the validity structure of qualitative research. *Education*, *118*(2), 282-293. doi:10.1016/j.dss.2003.08.004

- Burke, L. A., & Hutchins, H. M. (2007). Training transfer: An integrative literature review. *Human Resource Development Review*, 6(3), 263-296. Retrieved from https://journals.sagepub.com/doi/pdf/10.1177/1534484307303035
- Burke, V., & Collins, D. (2004). Optimising skills transfer via outdoor management development. Part I : The provider's perspective. *Journal of Management Development*, 23(7), 678-696. doi:10.1108/02621710410549576
- Burridge, P. (2003). A study of the influences on participant change during a *journey style outdoor education program*. (Unpublished master's dissertation), La Trobe University Bundoora, Melbourne, Australia.
- Carpenter, B. N. (1992). *Personal coping: Theory, research, and application*. Westport, Conn: Praeger.
- Carr, A. (2011). *Positive psychology: The science of happiness and human strengths* (2nd ed.). New York, NY: Routledge.
- Carver, C. S. (1997). You want to measure coping but your protocol' too long: Consider the brief cope. *International Journal of Behavioral Medicine*, 4(1), 92-100. doi:10.1207/s15327558ijbm0401\_6
- Carver, C. S. (2007). Brief COPE. Retrieved from http://www.psy.miami.edu/faculty/ccarver/sclBrCOPE.html
- Carver, C. S., Pozo, C., Harris, S. D., Noriega, V., Scheier, M. F., Robinson, D. S., . . . Clark, K. C. (1993). How coping mediates the effect of optimism on distress: A study of women with early stage breast cancer. *Journal of Personality and Social Psychology*, 65(2), 375-390. doi:10.1037/0022-3514.65.2.375
- Carver, C. S., & Scheier, M. (1990). Principles of self-regulation: Action and emotion. In E. T. Higgins & R. M. Sorrentino (Eds.), *Handbook of motivation and cognition: Foundations of social behaviour* (Vol. 2, pp. 3-52). New York, NY: Guilford Press.
- Carver, C. S., & Scheier, M. F. (1981). Attention and self-regulation: A control theory approach to human behavior. New York, NY: Springer-Verlag.
- Carver, C. S., Scheier, M. F., & Weintraub, J. K. (1989). Assessing coping strategies: A theoretically based approach. *Journal of Personality and Social Psychology*, 56(2), 267-283. doi:10.1037/0022-3514.56.2.267
- Cason, D., & Gillis, H. L. (1994). A meta-analysis of outdoor adventure programming with adolescents. *Journal of Experiential Education*, 17(1), 40-47. doi:10.1177/105382599401700109
- Caspi, A., Taylor, A., Moffitt, T. E., & Plomin, R. (2000). Neighborhood deprivation affects children's mental health: Environmental risks identified in a genetic design. *Psychology Science*, 11(4), 338-342. doi:10.1111/1467-9280.00267

- Child, D. (2006). *The essentials of factor analysis* (3rd ed.). New York, NY: A & C Black.
- Cole, P., Vindurampulle, O., & Vindurampulle, S. (2006). Understanding Year 9 students: A theoretical perspective'. Retrieved from http://www.det.vic.gov.au/edulibrary/public/publ/research/publ/Understan dingYear9\_PartA-rpt.pdf.
- Commissioner for Children and Young People Western Australia. (2015). Our children can't wait- review of the implementation of the recommendations of the 2011. Report of the inquiry into the mental health and well-being of children and young people in Western Australia. Perth, Australia: Commissioner for Children and Young People Western Australia Retrieved from http://www.parliament.wa.gov.au.
- Connor, K. M., & Davidson, J. R. T. (2003). Development of a new Resilience Scale: The Connor-Davidson Resilience Scale (CD-RISC). *Depression* and anxiety, 18(2), 76-82. doi:10.1002/da.10113
- Constantine, M. G., & Derald Wing, S. (2006). Factors contributing to optimal human functioning in people of color in the United States. *The Counseling Psychologist*, *34*(2), 228-244. doi:10.1177/0011000005281318
- Cooley, S. J., Burns, V. E., & Cumming, J. (2016). Using outdoor adventure education to develop students' groupwork skills: A quantitative exploration of reaction and learning. *Journal of Experiential Education*, 39(4), 329-354. doi:10.1177/1053825916668899
- Cooley, S. J., Cumming, J., Holland, M. J. G., & Burns, V. E. (2015). Developing the model for optimal learning and transfer (MOLT) following an evaluation of outdoor groupwork skills programmes. *European Journal of Training and Development*, 39(2), 104-121.
- Cooper, G. (1994). The role of outdoor education for the 21st century. *The Journal of Adventure Education and Outdoor Leadership*, 11(2), 8-12.
- Cooper, G. (2004). How school groups benefit from outdoor experiences. *Horizons*, 25 (Spring), 10-18.
- Cortina, J. M. (1993). What is coefficient alpha? An examination of theory and applications. *Journal of Applied Psychology*, 78(1), 98-104. doi:10.1037/0021-9010.78.1.98
- Coulson, J. (2017). 9 ways to a resilient child. Sydney, Australia: Harper Collins.
- Creswell, J. W. (2013). *Research design: Qualitative, quantitative, and mixed methods approaches*. Thousand Oaks, CA: Sage Publications.
- Creswell, J. W., Plano Clark, V. L., Gutmann, M. L., & Hanson, W. E. (2003). Advanced mixed methods research designs. In A. Tashakkori & C. Teddlie (Eds.), *Handbook of mixed methods in social and behavioral research* (pp. 209-240).

- Cross, R. (2002). The effects of an adventure education program on perceptions of alienation and personal control among at-risk adolescents. *Journal of Experiential Education*, 25(1), 247-254. doi:10.1177/105382590202500109
- Csikszentmihalyi, M. (1990). *Flow: The psychology of optimal experience*. New York, NY: Harper & Row.
- Csikszentmihalyi, M. (1996). Flow and the psychology of discovery and *invention*. New York, NY: Harper Collins.
- Csikszentmihalyi, M. (2008). *Flow: The Psychology of Optimal Experience*. Harper Perennial: New York.
- D'Amato, L. G., & Krasny, M. E. (2011). Outdoor adventure education: Applying transformative learning theory to understanding instrumental learning and personal growth in environmental education. *Journal of Environmental Education*, 42(4), 237-254. Retrieved from https://www.tandfonline.com/doi/full/10.1080/00958964.2011.581313
- Dalton, I., Fawcett, R., & West-Burnham, J. (2001). Schools for the 21st Century; Developing Best Practice. Great Brittan: Pearson Education.
- Darbor, K. E., Lench, H. C., Davis, W. E., & Hicks, J. A. (2016). Experiencing versus contemplating: Language use during descriptions of awe and wonder. *Cognition and Emotion*, 30(6), 1188-1196. Retrieved from https://doi.org/10.1080/02699931.2015.1042836
- Davidson, C. (2016). Building character through adventure education: A study of levels of grit and resilience in outward bound students. (Doctoral dissertation, Indiana University), Available from ProQuest Dissertations & Theses database. (UMI no. 10108188)
- Dehn, M. J. (2014). *Essentials of processing assessment* (2nd ed.). Hoboken, New Jersey: John Wiley & Sons, Inc.
- Denzin, N. K., & Lincoln, Y. S. (1994). Entering the field of qualitative research. In N. K. Denzin & Y. S. Lincoln (Eds.), *Handbook of qualitative research* (pp. 1-18). United States: Sage Publications.
- Denzin, N. K., & Lincoln, Y. S. (2008). Collecting and interpreting qualitative materials (3rd ed.). United States: Sage Publications.
- Denzin, N. K., & Lincoln, Y. S. (2011). *The handbook of qualitative research*. Thousand Oaks, CA: Sage Publications.
- Denzin, N. K., & Lincoln, Y. S. (Eds.). (2003). *Collecting and interpreting qualitative materials* (2 ed.). United States: Sage Publications.
- Department of Education Victoria. (1999). Framework for student support services in Victorian government schools. Retrieved from http://www.eduweb.vic.gov.au/edulibrary/public/stuman/wellbeing/fwksss.pdf.

- Dewey, J. (1916). *Democracy and education: An introduction to the philosophy of education:* The Macmillan Company.
- Dewey, J. (1929). My Pedagogic Creed. *Journal of the National Education* Association, 18(9), 291-295.
- Dewey, J. (1938). Experience and education. New York: McMillian.
- Diamond, D., M, Campbell, A., M, Park, C., R, Halonen, J., & Zoladz, P., R. (2007). The temporal dynamics model of emotional memory processing: A synthesis on the neurobiological basis of stress-induced amnesia, flashbulb and traumatic memories, and the Yerkes-Dodson Law. *Neural Plasticity*, 1-33. doi:10.1155/2007/60803
- Donaldson, G. W., & Donaldson, L. E. (1958). Outdoor education: A definition. Journal of Health, Physical Education and Recreation, 29(5), 17-63. doi:10.1080/00221473
- Donnon, T., & Hammond, W. (2007). Understanding the relationship between resiliency and bullying in adolescence: An assessment of youth resiliency from five urban junior high schools. *Child and Adolescent Psychiatric Clinics of North America, 16*(2), 449-471. doi:10.1016/j.chc.2006.11.007
- Donnon, T., Hammond, W., & Charles, G. (2003). Youth resiliency: Assessing students' capacity for success at school. *Teaching and Learning*, 1(2), 23-28. doi:10.26522/tl.v1i2.109
- Dooley-Feldman, E. A. (2016). Outdoor adolescent rites of passage: Theoretical foundations, contemporary shortcomings, and the emerging model. (Master of Arts in Organizational Systems Unpublished Masters Thesis), Saybrook University, Oakland, California.
- Doucette, P. A. (2004). Walk and talk: An intervention for behaviorally challenged youths. *Adolescence*, *39*(154), 373-388.
- Driscoll, D. L., Appiah-Yeboah, A., Salib, P., & Rupert, D. J. (2007). Merging qualitative and quantitative data in mixed methods research: How to and why not. *Ecological and Environmental Anthropology (University of Georgia)*, *3*(1), 19-28.
- Duckworth, A. L., & Eskreis-Winkler, L. (2013). True grit. *The observer*, 26(4), 1-3.
- Duffy, M. E. (1987). Methodological triangulation: A vehicle for merging quantitative and qualitative research methods. *The Journal of Nursing Scholarship*, *19*(3), 130-133. doi:10.1111/j.1547-5069.1987.tb00609.x
- Duhachek, A. (2005). Coping: A multidimensional, hierarchical framework of responses to stressful consumption episodes. *Journal of Consumer Research*, *32*(1), 41-53. doi:10.1086/426612
- Dumont, M., & Provost, M. A. (1999). Resilience in adolescents: Protective role of social support, coping strategies, self-esteem, and social activities on

experience of stress and depression. *Journal of Youth and Adolescence*, 28(3), 343-363. doi:10.1023/A:1021637011732

- Durlak, J. A., & Wells, A. M. (1997). Primary prevention mental health programs for children and adolescents: A meta-analytic review. *American Journal of Community Psychology*, 25(2), 115-152.
- Dweck, C. S. (2006). *Mindset: The new psychology of success*. New York, NY: Ballatine Books.
- Dye, J. (2007). Meet generation C: Creatively connecting through content. *EContent*, *30*(4), 38–43. Retrieved from http://www.econtentmag.com
- Edward, K. I., Welch, A., & Chater, K. (2009). The phenomenon of resilience as described by adults who have experienced mental illness. *Journal of Advanced Nursing*, 65(3), 587-595. doi:10.1111/j.1365-2648.2008.04912.x.
- Ellis, M. J. (1973). Why people play. Englewood Cliffs, NJ: Prentice-Hall.
- Endler, N. S., & Parker, J. D. (1990). Multidimensional assessment of coping: A critical evaluation. *Journal of Personality and Social Psychology*, 58(5), 844-854. Retrieved from https://www.researchgate.net/profile/James\_Parker3/publication/20808460 \_Multidimensional\_Assessment\_of\_Coping\_A\_Critical\_Evaluation/links/ 5534fd770cf2df9ea6a41449/Multidimensional-Assessment-of-Coping-A-Critical-Evaluation.pdf
- Evans, G. W., Li, D., & Whipple, S. S. (2013). Cumulative risk and child development. *Psychological Bulletin*, *139*(6), 1342-1396. doi:0.1037/a0031808
- Ewert, A. (1989). *Outdoor adventure pursuits: Foundations, models, and theories*. Scottssdale, AZ: Publishing Horizons.
- Ewert, A., & Davidson, C. (2017a). *Behavior and group management in outdoor adventure education: Theory, research and practice.* New York, NY: Routledge.
- Ewert, A., & Davidson, C. (2017b). Relevant theories and their application in outdoor adventure education and behavior and group management. In *Behavior and group management in outdoor adventure education: Theory, research and practice* (pp. 54-71). New York, NY: Routledge.
- Ewert, A., & Garvey, D. (2007). Philosophy and theory of adventure education. In D. Prouty, J. Panicucci, & R. Collinson (Eds.), *Adventure education: Theory and applications*. Champaign, IL: Human Kinetics.
- Ewert, A., & McAvoy, L. (2000). The effects of wilderness settings on organized groups: A state-of-knowledge paper. USDA Forest Service Proceedings, 3(15), 13-26. Retrieved from <Go to ISI>://WOS:000165332700003

- Ewert, A., & Yoshino, A. (2008). A preliminary exploration of the influence of short-term adventure-based expeditions on levels of resilience. *Journal of Experiential Education*, 30(3), 262–266. doi:10.1177/105382590703000308
- Ewert, A., & Yoshino, A. (2011). The influence of short-term adventure-based experiences on levels of resilience. *Journal of Adventure Education & Outdoor Learning*, 11(1), 35-50. doi:10.1080/14729679.2010.532986
- Exeter, D. J. (2001). *Outward bound: Learning in the outdoors*. Golden, CO: Outward Bound Trust.
- Fanshawe, J. P., & Burnett, P. C. (1991). Assessing school-related stressors and coping mechanisms in adolescents. *British Journal of Educational Psychology*(61), 92-98. doi:doi:10.1111/j.2044-8279.1991.tb00964.x
- Fenemor, A., Deans, N., Davie, T., Allen, W., Dymond, J., Kilvington, M., . . . Smith, R. (2008). Collaboration and modelling: Tools for integration in the Motueka catchment, New Zealand. *Water SA*, 34(4), 448-455. Retrieved from http://www.scopus.com
- Fergus, S., & Zimmerman, M. A. (2005). Adolescent resilience: a framework for understanding healthy development in the face of risk. *Annu Rev Public Health*, 26, 399-419. doi:10.1146/annurev.publhealth.26.021304.144357
- Ferguson, E., & Cox, T. (1993). Exploratory factor analysis: A users' guide. International Journal of Selection and Assessment(1), 84–94. doi:10.1111/j.1468-2389.1993.tb00092.x
- Fjørtoft, I. (2001). The natural environment as a playground for children: The impact of outdoor play activities in pre-primary school children. *Early Childhood Education Journal*, 29(2), 111-117. doi:10.1023/A:1012576913074
- Fletcher, T. B., & Hinkle, J. S. (2002). Adventure based counseling: An innovation in counseling. *Journal of Counseling and Development*, 80(3), 277-285.
- Florian, V., Mikulincer, M., & Taubman, O. (1995). Does hardiness contribute to mental health during a stressful real-life situation? The roles of appraisal and coping. *Journal of Personality and Social Psychology*, 68(4), 687.
- Fontana, A., & Frey, J. (1994). The art of science. *The handbook of qualitative research*, 361-376.
- Fontana, A., & Frey, J. H. (2003). The Interview: From structured questions to negotiated text. In N. K. Denzin & Y. S. Lincoln (Eds.), *Collecting and interpreting qualitative materials* (2 ed.). United States: Sage Publications.
- Ford, P. (1986). Outdoor education: Definition and philosophy. *ERIC Digest: Outdoor Education, 1*(1), 1-14.

- Fossey, E., Harvey, C., McDermott, F., & Davidson, L. (2002). Understanding and evaluating qualitative research. *Australian and New Zealand journal* of psychiatry, 36(6), 717-732. doi:10.1046/j.1440-1614.2002.01100.x
- Fredrickson, B. L., & Losada, M. F. (2005). Positive affect and the complex dynamics of human flourishing. *American Psychologist*, 60(7), 678-686. doi:10.1037/0003-066X.60.7.678
- Friedli, L. (2009). *Mental health, resilience and inequalities*. In. Retrieved from http://www.euro.who.int/\_\_data/assets/pdf\_file/0012/100821/E92227.pdf
- Frydenberg, E. (2008). Adolescent coping: Advances in theory, research and practice. New York, NY: Routledge.
- Frydenberg, E. (2010). *Think positively! A course for developing coping skills in adolescents*. London, UK: Continuum International Publishing Group.
- Frydenberg, E., & Lewis, R. (1993). Adolescent Coping Scale (ACS): Administrator's Manual: ACER.
- Furlong, A. (2009). *Handbook of youth and young adulthood: New perspectives and agendas.* New York, NY: Routlegde.
- Garbarino, J. (1982). *Children and families in the social environment*. New York, NY: Aldine Publishing.
- Garmezy, N. (1985). Resiliency and vulnerability to adverse developmental outcomes associated with poverty. *American behavioral scientist*, *34*(4), 416-430. doi:10.1177/0002764291034004003
- Gass, M. A. (1993a). Adventure therapy: Therapeutic applications of adventure programming. Boulder, CO: Kendall/Hunt Publishing Company.
- Gass, M. A. (1993b). Foundations of adventure therapy. In M. A. Gass (Ed.), Adventure therapy: Therapeutic applications of adventure programming (pp. 3-10). Dubuque, IA: Kendall/Hunt Publishing Company.
- Gass, M. A. (1995). Programming the transfer of learning in adventure education. In K. Warren, M. Sakofs, & J. S. Hunt Jr (Eds.), *The theory of experiential education*. Boulder, CO: Association for Experiential Education.
- Gass, M. A. (2003). Kurt Hahn address 2002 AEE International Conference. *Journal of Experiential Education*, 25(3), 363-371. Retrieved from http://journals.sagepub.com/home/jee
- Gass, M. A., & Buell, L. (1986). *The season of ingenuity: Ethics in experiential education*. Paper presented at the annual conference of the Association for Experiential Education, Moodus, Connecticut.
- Gilbertson, K., Bates, T., McLaughlin, T., & Ewert, A. (2006). *Outdoor* education: Methods and strategies. Champaign, Ill: Human Kinetics.
- Gillespie, E., & Allen-Craig, S. (2009). The enhancement of resilience via a wilderness therapy program: A preliminary investigation. *Australian Journal of Outdoor Education*, *13*(1), 39-49.

- Ginsburg, K. R. (2007). The importance of play in promoting healthy child development and maintaining strong parent-child bonds. *Pediatrics*, *119*(1), 182-191. Retrieved from http://pediatrics.aappublications.org/content/pediatrics/119/1/182.full.pdf
- Gjerde, P. F., Block, J., & Block, J. H. (1986). Egocentrism and ego resiliency: Personality characteristics associated with perspective-taking from early childhood to adolescence. *Journal of Personality and Social Psychology*, 51(2), 423-434. doi:10.1037/0022-3514.51.2.423
- Glass, K., Flory, K., Hankin, B. L., Kloos, B., & Turecki, G. (2009). Are coping strategies, social support, and hope associated with psychological distress among Hurricane Katrina survivors? *Journal of Social and Clinical Psychology*, 28(6), 779-795. doi:10.1521/jscp.2009.28.6.779
- Gliem, R. R., & Gliem, J. A. (2003). Calculating, interpreting, and reporting cronbach's alpha reliability coefficient for likert-type scales. Paper presented at the 2003 Midwest Research-to-Practice Conference in Adult, Continuing, and Community Education, Columbus, OH.
- Goldstein, S., & Brooks, R. B. (2005). *Handbook of resilience in children*. New York, NY: Springer.
- Goossens, L. (2006). Affect, emotion, and loneliness in adolescence. In S. Jackson & L. Goossens (Eds.), *Handbook of adolescent development* (pp. 51-70). New York, NY: Psychology Press.
- Gorsuch, R. L. (1997). Exploratory factor analysis: Its role in item analysis. *Journal of personality assessment*, 68(3), 532-560. doi:10.1207/s15327752jpa6803\_5
- Green, G. T., Kleiber, D. A., & Tarrant, M. A. (2000). The effect of an adventurebased recreation program on development of resiliency in low income minority youth. *Journal of Park and Recreation Administration*, 18(3), 76-97.
- Green, S., Grant, A., & Rynsaardt, J. (2007). Evidence-based life coaching for senior high school students: Building hardiness and hope. *International Coaching Psychology Review*, 2(1), 24-32.
- Greenberg, M. T., Weissberg, R. P., O'Brien, M. U., Zins, J. E., Fredericks, L., Resnik, H., & Elias, M. J. (2003). Enhancing school-based prevention and youth development through coordinated social, emotional, and academic learning. *American Psychologist*, 58(6-7), 466-474. Retrieved from Prevention That Works for Children and Youth website: doi:10.1037/0003-066X.58.6-7.466
- Griffiths, P. (2011). Curriculum based Outdoor Education programs and their benefit to students. Retrieved from

- Grolnick, W. S., Kurowski, C. O., Dunlap, K. G., & Hevey, C. (2000). Parental resources and the transition to junior high. *Journal of Research on Adolescence, 10*, 465-488. doi:10.1207/SJRA1004\_05.
- Grümme, B. (2011). Vom erlebnis zur erfahrung: Anmerkungen zur erlebnispädagogik aus der sicht eines alteritätstheoretischen erfahrungsbegriffs. In V. Pum, M. Pirner, & J. Lohrer (Eds.), *Erlebnispädagogik im christlichen Kontext. Religionspädagogische Perspektiven* (pp. 95-118). Bad Boll, Germany: Evangelische Akademie.
- Guba, E. G., & Lincoln, Y. S. (1988a). Advantages of naturalistic methods. In E.
  G. Guba & Y. S. Lincoln (Eds.), *Effective evaluation: Improving the usefulness of evaluation results through responsive naturalistic approaches* (pp. 53-83). United States of America: Jossey-Bass Publishers.
- Guba, E. G., & Lincoln, Y. S. (1988b). Interviewing, observations, and nonverbal cue interpretation. In E. G. Guba & Y. S. Lincoln (Eds.), *Effective* evaluation: Improving the usefulness of evaluation results through responsive naturalistic approaches (pp. 153-219). United States of America: Jossey-Bass Publishers.
- Guba, E. G., & Lincoln, Y. S. (1988c). Naturalistic solutions to methodological problems. In E. G. Guba & Y. S. Lincoln (Eds.), *Effective evaluation: Improving the usefulness of evaluation results through responsive naturalistic approaches* (pp. 85-103). United States of America: Jossey-Bass Publishers.
- Haggerty, R. J., Sherrod, L. R., Garmezy, N., & Rutter, M. (1997). *Stress, risk, and resilience in children and adolescents: Processes, mechanisms, and interventions.* Cambridge, New York: Cambridge University Press.
- Hall, G. S. (1904). Adolescence: Its psychology and its relation to physiology, anthropology, sociology, sex, crime, religion, and education (1st ed.). Englewood Cliffs, NJ: Prentice-Hall.
- Hammerman, D. R., Hammerman, W. M., & Hammerman, E. L. (2001). *Teaching* in the outdoors (5th ed.). Danville, IL: Interstate Publishers.
- Hammond, C. (2004). Impacts of lifelong learning upon emotional resilience, psychological and mental health: Fieldwork evidence. *Oxford Review of Education*, *30*(4), 551-568. doi:10.1080/0305498042000303008
- Hammond, L. D., Austin, K., Orcutt, S., & Rosso, J. (2001). How people learn: Introduction to learning theories. In L. D. Hammond, K. Austin, S. Orcutt, & J. Rosso (Eds.), The learning classroom: Theory into practice, a telecourse for teacher education and professional development (pp. 1-22): Stanford University School of Education. Retrieved from https://web.stanford.edu/class/ed269/hplintrochapter.pdf.

- Hanewald, R. (2011). Reviewing the literature on "at-risk" and resilient children and young people. *Australian Journal of Teacher Education*, *36*(2), 16-29. doi:10.14221/ajte.2011v36n2.2
- Hans, T. A. (2000). A meta-analysis of the effects of adventure programming on locus of control. *Journal of Contemporary Psychotherapy*, 30(1), 33-60. doi:10.1023/A:1003649031834
- Hansmann, R., Hug, S.-M., & Seeland, K. (2007). Restoration and stress relief through physical activities in forests and parks. *Urban Forestry & Urban Greening*, 6(1), 213-225. Retrieved from https://www.sciencedirect.com/science/article/pii/S1618866707000623
- Hanson, T. L., & Kim, J. O. (2007). Measuring resilience and youth development: The psychometric properties of the healthy kids survey (Issues & Answers Report, REL 2007-No. 034). Retrieved from http://www.wested.org
- Hastings, R. P., Kovshoff, H., Brown, T., Ward, N. J., Degli Espinosa, F., & Remington, B. (2005). Coping strategies in mothers and fathers of preschool and school-age children with autism. *Autism*, 9(4), 377-391. doi:10.1177/1362361305056078
- Hattie, J. (2003). *Teachers make a difference: What is the research evidence?* Paper presented at the Australian Council for Educational Research Annual Conference on building teacher quality, Auckland, New Zealand. https://cdn.auckland.ac.nz/assets/education/hattie/docs/teachers-make-adifference-ACER-(2003).pdf
- Hattie, J., Marsh, H. W., Neill, J. T., & Richards, G. E. (1997). Adventure education and outward bound: Out-of-class experiences that make a lasting difference. *Review of educational research*, 67(1), 43-87. Retrieved from http://journals.sagepub.com/doi/abs/10.3102/00346543067001043 doi:10.3102/00346543067001043
- Havighurst, R. J. (1948). *Developmental tasks and education*. Chicago, IL: University of Chicago Press.
- Hayhurst, J., Hunter, J. A., Kafka, S., & Boyes, M. (2015). Enhancing resilience in youth through a 10-day developmental voyage. *Journal of Adventure Education and Outdoor Learning*, 15(1), 40-52. doi:10.1080/14729679.2013.843143
- Heilig, G. K. (2012). World urbanization prospects: The 2011 revision. United Nations, Department of Economic and Social Affairs (DESA) Population Division, Population Estimates and Projections Section.
- Herrman, H., Stewart, D. E., Diaz-Granados, N., Berger, E. L., Jackson, B., & Yuen, T. (2011). What is resilience? *Canadian Journal Of Psychiatry*. *Revue Canadienne De Psychiatrie*, *56*(5), 258-265.

- Higgins, P. (1997). Why educate out of doors? In P. Higgins, C. Loynes, & N. Crowther (Eds.), A guide for outdoor educators in Scotland (pp. 9-14). Penrith, UK: Reeds.
- Higgins, P., & Loynes, C. (1997). On the nature of outdoor education. In P. Higgins, C. Loynes, & N. Crowther (Eds.), A guide for outdoor educators in Scotland (pp. 6-8). Penrith, UK: Reeds.
- Hinkle, D. E., Wiersma, W., & Jurs, S. G. (2003). *Applied statistics for the behavioral sciences*. New York, NY: Houghton Mifflin Company.
- Holahan, C. J., & Moos, R. H. (1987). Personal and contextual determinants of coping strategies. *Journal of Personal Social Psychology*, 52(5), 946-955.
- Holman, T., & McAvoy, L. H. (2005). Transferring benefits of participation in an integrated wilderness adventure program to daily life. *Journal of Experiential Education*, 27(3), 322-325.
- Hong, B. S. S., Shull, P. J., & Haefner, L. A. (2011). Impact of perceptions of faculty on student outcomes of self-efficacy, locus of control, persistence, and commitment. *Journal of College Student Retention: Research, Theory* and Practice, 13(3), 289-309. doi:10.2190/CS.13.3.b
- Hopkins, D. (1985). Self-concept and adventure. *Journal of Adventure Education* and Outdoor Leadership, 2(1), 7-15.
- Howell, A. J., & Passmore, H.-A. (2013). The nature of happiness: Nature affiliation and mental well-being. In C. L. M. Keyes (Ed.), *Mental wellbeing: International contributions to the study of positive mental health* (pp. 231-257). Dordrecht: Springer Netherlands.
- Hulbert-Williams, N. J., Morrison, V., Wilkinson, C., & Neal, R. D. (2013).
  Investigating the cognitive precursors of emotional response to cancer stress: Re-testing Lazarus's transactional model. *British Journal of Health Psychology*, 18(1), 97-121. doi:10.1111/j.2044-8287.2012.02082.x
- Humphrey, J. (2002). Teenagers will be teenagers. New York: Kroshka Books.
- Hunt Jr, J. S. (2008). Make room for daddy...and mommy: Helicopter parents are here! *The Journal of Academic Administration in Higher Education*, *4*, 9-11.
- Hunter, A. J., & Chandler, G. E. (1999). Adolescent resilience. *Journal of Nursing Scholarship*, *31*(3), 243-247. doi:10.1111/j.1547-5069.1999.tb00488.x
- Hussein, A. (2009). The use of triangulation in social sciences research: Can qualitative and quantitative methods be combined? *Journal of Comparative Social Work*, 4(1), 1-12. Retrieved from http://journal.uia.no/index.php/JCSW/index
- Ihuah, P. W., & Eaton, D. (2013). The pragmatic research approach: A framework for sustainable management of public housing estates in Nigeria. *Journal*

*of US-China Public Administration, 10*(10), 933-944. Retrieved from http://www.davidpublisher.org/Home/Journal/UCPA

- Jaffe, M. L. (1998). Adolescence. New York, NY: Wiley.
- Jain, S., Buka, S. L., Subramanian, S. V., & Molnar, B. E. (2011). Protective factors for youth exposed to violence: Role of developmental assets in building emotional resilience. *Youth Violence and Juvenile Justice*, 10(1), 107-129. doi:10.1177/1541204011424735
- James, T. (1990). Kurt Hahn and the aims of education. *Journal of Experiential Education, 13*(1), 6-13.
- James, W. (1900). Talks to teachers and students. New York, NY: Holt.
- Jelalian, E., Mehlenbeck, R., Lloyd-Richardson, E. E., Birmaher, V., & Wing, R. R. (2005). 'Adventure therapy' combined with cognitive-behavioral treatment for overweight adolescents. *International Journal Of Obesity*, 30, 31. doi:10.1038/sj.ijo.0803069
- Jew, C. L., Green, K. E., & Kroger, J. (1999). Development and validation of a measure of resiliency. *Measurement and Evaluation in Counseling and Development*, 32, 75-89.
- Johnson, J. W. (2012). The effect of high school outdoor-based adventure leadership programs in independent schools on personal effectiveness and locus of control. (Master's thesis, Prescott College), Available from ProQuest Dissertations & Theses database. (UMI no.1021033992)
- Johnson, S. B., Blum, R. W., & Giedd, J. N. (2009). Adolescent maturity and the brain: The promise and pitfalls of neuroscience research in adolescent health policy. *Journal of Adolescent Health*, 45(3), 216-221. Retrieved from http://dx.doi.org/10.1016/j.jadohealth.2009.05.016
- Jordan, D. J. (2007). *Leadership in leisure services: Making a difference* (3rd ed.). State College, PA: Venture Publishing.
- Jostad, J., Sibthorp, J., Pohja, M., & Gookin, J. (2015). The adolescent social group in outdoor adventure education: Social connections that matter. *Research in Outdoor Education*, *13*(1), 16-37.
- Juster, F. T., Ono, H., & Stafford, F. P. (2004). Changing times of American youth, 1981–2003. Institute for Social Research, University of Michigan.
- Kaplan, H. B. (1999). Toward an understanding of resilience: Critical review of definitions and models. In M. D. Glantz & J. L. Johnson (Eds.), *Resilience* and development: Postive life adaptions. New York, NY: Kluwer Academic /Plenum Publishers.
- Kaplan, H. B. (2005). Understanding the concept of resilience. In S. Goldstein & R. B. Brooks (Eds.), *Handbook of resilience in children*. United States of America: Springer Science Business Media Inc.

- Kaplan, R. (2001). The nature of the view from home: Psychological benefits. *Environment and Behavior*, 33(4), 507-542. doi:10.1177/00139160121973115
- Kaplan, S. (1995). The restorative benefits of nature: Toward an integrative framework. *Journal of Environmental Psychology*, *15*(3), 169-182. Retrieved from http://www.sciencedirect.com/science/article/pii/0272494495900012
- Kaplan, S., & Talbot, J. F. (1983). Psycological benefits of a wilderness experience. In I. Altman & J. F. Wohlwill (Eds.), *Behavior and the natural environment* (pp. 163-203). New York: Plenum Press.
- Kashdan, T. B., & Rottenberg, J. (2010). Psychological flexibility as a fundamental aspect of health. *Clinical Psychology Review*, 30(7), 865-878. doi:10.1016/j.cpr.2010.03.001
- Keller, T. E. (2007). Youth mentoring: Theoretical and methodological issues. In T. D. Allen & L. T. Eby (Eds.), The Blackwell handbook of mentoring: A multiple perspectives approach (pp. 23-48). Malden, MA: Blackwell Publishing. Retrieved from https://ebookcentral.proquest.com.
- Keyes, C. L. M. (2006). Mental health in adolescence: Is America's youth flourishing? *American Journal of Orthopsychiatry*, 76(3), 395-402. doi:10.1037/0002-9432.76.3.395
- Ki-moon, B. (2016). Sustainability: Engaging future generations now. *The Lancet*, *387*(10036), 2356-2358. Retrieved from https://doi.org/10.1016/S0140-6736(16)30271-9
- Kidd, S., & Shahar, G. (2008). Resilience in homeless youth: The key role of selfesteem. American Journal of Orthopsychiatry, 78(2), 163-172. doi:10.1037/0002-9432.78.2.163
- Kim, T. H. (2001). *The comparative study on types of coping with stress put on the common juveniles and delinquent juveniles*. (Unpublished master's thesis), Hanyang University, Seoul, Korea.
- Kim, T. H. (2004). *The effects of hope and meaning of life on adolescent's resilience.* (Doctoral dissertation), Hanyang University, Seoul, Korea.
- Kimball, R. O., & Bacon, S. B. (1993). The wilderness challenge model. In M. A. Gass (Ed.), Adventure therapy: Therapeutic applications of adventure programming (pp. 11-42). Dubuque, IA: Kendall/Hunt Publishing Company.
- Kimemia, M., Asner-Self, K. K., & Daire, A. P. (2011). An exploratory factor analysis of the Brief COPE with a sample of Kenyan caregivers. *International Journal for the Advancement of Counselling*, *33*(3), 149-160. Retrieved from https://link.springer.com/article/10.1007/s10447-011-9122-8
- Kolb, D. A. (1984). *Experiential learning: Experience as the source of learning and development*. Englewood Cliffs, NJ: Prentice Hall.
- Kolb, D. A. (2007). *The Kolb learning style inventory: LSI workbook* (3.1 ed.). Boston, Mass: Hay Resources Direct, HayGroup.
- Kraft, R. J., & Sakofs, M. (1988). *The theory of experiential education*. Boulder, CO: Association for Experiential Education.
- Kroger, J. (2004). *Identity in adolescence: The balance between self and other* (3rd ed.). New York, NY: Psychology Press.
- Kvale, S. (1995). The social construction of validity. *Qualitative inquiry*, 1(1), 19-40.
- Lawrence, D., Johnson, S., Hafekost, J., Boterhoven De Haan, K., Sawyer, M., Ainley, J., & Zubrick, S. R. (2015). The mental health of children and adolescents. Report on the second Australian child and adolescent survey of mental health and wellbeing. *Department of Health, Canberra*.
- Lazarus, R. S. (1966). *Psychological stress and the coping process*. New York, NY: McGraw-Hill.
- Lazarus, R. S. (1984). Stress, Appraisal, and Coping. New York: Springer.
- Lazarus, R. S. (2006). *Stress and emotion: A new synthesis*. New York, NY: Springer Publishing Company.
- Lazarus, R. S., & Folkman, S. (1984). *Stress, appraisal, and coping*. In. Retrieved from http://www.springerpub.com/
- Lazarus, R. S., & Folkman, S. (1987). Transactional theory and research on emotions and coping. *European Journal of Personality*, 1(3), 141-169. doi:10.1002/per.2410010304
- Lederbogen, F., Kirsch, P., Haddad, L., Streit, F., Tost, H., Schuch, P., . . . Meyer-Lindenberg, A. (2011). City living and urban upbringing affect neural social stress processing in humans. *Nature*, 474, 498–501. doi:10.1038/nature10190
- Lee, J. H., Seo, M., Lee, M., Park, S. Y., Lee, J. H., & Lee, S. M. (2017). Profiles of coping strategies in resilient adolescents. *Psychological Reports*, *120*(1), 49-69. doi:10.1177/0033294116677947
- Lee, K., Bull, R., & Ho, R. M. H. (2013). Developmental changes in executive functioning. *Child Development*, 84(6), 1933-1953. doi:10.1111/cdev.12096
- Leppert, K., Gunzelmann, T., Schumacher, J., Strauss, B., & Brähler, E. (2005). Resilience as a protective personality characteristic in the elderly. *Psychotherapie, Psychosomatik, medizinische Psychologie, 55*(8), 365-369. doi:10.1055/s-2005-866873
- Lerner, R. M., Napolitano, C. M., Boyd, M. J., Mueller, M. K., & Callina, K. S. (2014). Mentoring and positive youth development. In D. L. DuBois & M.

J. Karcher (Eds.), Handbook of youth mentoring (2nd ed., pp. 15-28). Los Angeles: SAGE Publications.

- Lerner, R. M., & Steinberg, L. (2009). The scientific study of adolescent development: Past, present, and future. In R. M. Lerner & L. Steinberg (Eds.), *Handbook of adolescent psychology*. Hoboken, N.J: John Wiley & Sons.
- Levine, M. (2006). The price of privilege. New York, NK: Harper Collins.
- Lewis, C., T. (1890). *An elementary latin dictionary*. New York, NY: American Book Company Retrieved from https://americanbookcompany.com/.
- Lillis, J., Hayes, S. C., Bunting, K., & Masuda, A. (2009). Teaching acceptance and mindfulness to improve the lives of the obese: A preliminary test of a theoretical model. *Annals of Behavioral Medicine*, *37*(1), 58-69.
- Lin, Y., Mutz, J., Clough, P., & Papageorgiou, K. (2017). Mental toughness and individual differences in learning, educational and work performance, psychological well-being, and personality: A systematic review (Vol. 8).
- Long, T. (2011). Cognitive-behavioral approaches to therapeutic recreation. In N. J. Stumbo & B. Wardlaw (Eds.), *Facilitation of therapeutic recreation services: An evidence-based and best practice approach to techniques and processes* (pp. 289-306). State College, PA: Venture.
- Louv, R. (2005). Do your kids have nature deficit disorder? *Orion Magazine*. Retrieved from http://www.earthandspiritcenter.org/Cosmology/Cosmology2/3ED-4%20-%20Nature%20Deficit%20in%20Children%20-%20Richard%20Louv.pdf
- Louv, R. (2007). Leave no child inside. *Orion Magazine*. Retrieved from http://www.doi.gov/archive/hrm/SES%20Conference/Richard%20Louv% 20Congressional%20Testimony%202\_07.pdf
- Louv, R. (2013). Last child in the woods: Saving our children from nature-deficit disorder. New York, NY: Atlantic Books Ltd.
- Luckner, J. L., & Nadler, R. S. (1997). *Processing the experience: Strategies to enhance and generalize learning* (2nd ed.). Dubuque, IA: Kendall/Hunt Publishing Company
- Lugg, A. (2004). Outdoor adventure in Australian outdoor education: Is it a case of roast for Christmas dinner. *Australian Journal of Outdoor Education*, 8(1), 4-11.
- Luthar, S. S., Cicchetti, D., & Becker, B. (2000). The construct of resilience: a critical evaluation and guidelines for future work. *Child Development*, 71(3), 543-562.
- Mack, N., Woodsong, C., MacQueen, K. M., Guest, G., & Namey, E. (2005). *Qualitative research methods: A data collectors field guide*. In. Retrieved from https://www.fhi360.org/

- Marano, H. E. (2008). *A nation of wimps: The high cost of invasive parenting*. New York, NY: The Crown Publishing Group.
- Markstrom, C. A., Marshall, S. K., & Tryon, R. J. (2000). Resiliency, social support, and coping in rural low-income: Appalachian adolescents from two racial groups. *Journal of adolescence*, 23(6), 693-703. doi:10.1006/jado.2000.0353
- Marsh, H. W., Richards, G. E., & Barnes, J. (1986a). Multidimensional selfconcepts: A long term follow-up of the effect of participation in an Outward Bound program. *Journal of personality and social Psychology Bulletin*, 12(4), 475-492. Retrieved from https://doi.org/10.1177/0146167286124011
- Marsh, H. W., Richards, G. E., & Barnes, J. (1986b). Multidimensional selfconcepts: The effect of participation in an Outward Bound program. *Journal of Personality and Social Psychology*, 50(1), 195-204. Etrieved from http://dx.doi.org/10.1037/0022-3514.50.1.195
- Marsh, P. E. (1999a). *Executive summary: What does camp do for kids? A metaanalysis of the influence of organized camping experience on the self constructs of youth.* Indiana University.
- Marsh, P. E. (1999b). What does camp do for kids? A meta-analysis of the influence of organized camping experience on the self constructs of youth. Indiana University. Unpublished master's thesis.
- Martin, A. J., & Leberman, S. I. (2005). Personal learning or prescribed educational outcomes: A case study of the Outward Bound experience. *Journal of Experiential Education*, 28(1), 44-59.
- Martin, B., Cashel, C., Wagstaff, M., & Breunig, M. (2006). *Outdoor leadership: Theory and practice*. Champaign, IL: Human Kinetics.
- Martin, P. (2004). Outdoor adventure in promoting relationships with nature. *Australian Journal of Outdoor Education*, 8(1), 20-28. Retrieved from https://outdooreducationaustralia.org.au/
- Martin, P. (2010). Outdoor education and the national curriculum in Australia. *Australian Journal of Outdoor Education*, *14*(2), 3-11. Retrieved from https://outdooreducationaustralia.org.au/
- Martin, P., & Hewison, T. (2010). *Outdoor education and the national curriculum*. Outdoor Education Australia position paper. Retrieved from http://www.voea.vic.edu.au/resources/curriculum/nat\_curric/docs/OEStud entOutcomes.pdf.
- Martin, P., & Priest, S. (1986). Understanding the adventure experience. *Journal* of Adventure Education, 3, 18-21.
- Masten, A. S. (2001). Ordinary magic: Resilience processes in development. *American Psychologist, 56*(3), 227-238. doi:10.1037/0003-066X.56.3.227

- Masten, A. S. (2009). Ordinary magic: Lessons from research on resilience in human development. *Education Canada*, 49(3), 28-32. Retrieved from http://www.cea-ace.ca/education-canada
- Masten, A. S. (2012). Risk and resilience in development. In P. D. Zelazo (Ed.), *The Oxford handbook of developmental psychology, Vol. 2. Self and other* (pp. 579-607). New York, NY: Oxford University Press.
- Masten, A. S. (2015). *Resilience in development: Early childhood as a window of opportunity*. Paper presented at the ECHD Symposium University of New Mexico, University of Minnesota.
- Masten, A. S., & Cicchetti, D. (2010). Developmental cascades. *Development and Psychopathology*, 22(3), 491-495. doi:10.1017/S0954579410000222
- Masten, A. S., & Coatsworth, J. D. (1998). The development of competence in favorable and unfavorable environments: Lessons from research on successful children. *American Psychologist*, 53(2), 205-220. doi:10.1037/0003-066x.53.2.205
- Masten, A. S., & O'Connor, M. J. (1989). Vulnerability, stress, and resilience in the early development of a high risk child. *Journal of the American Academy of Child and Adolescent Psychiatry*, 28(2), 274-278. doi:10.1097/00004583-198903000-00021
- Masten, A. S., & Obradović, J. (2006). Competence and resilience in development. Annals of the New York Academy of Sciences, 1094(1), 13-27.
- Masten, A. S., & Reed, M. J. (2002). Resilience in development. In C. R. Snyder & S. J. Lopez (Eds.), *Handbook of positive psychology* (pp. 74-87). New York, NY: Oxford University Press.
- Matthews, G., & Zeidner, M. (2003). Negative appraisals of positive psychology: A mixed-valence endorsement of Lazarus. *Psychological inquiry*, 14(2), 137-143. Retrieved from https://www.tandf.co.uk/
- McCormick, C. M., Kuo, S. I. C., & Masten, A. S. (2011). Developmental tasks across the lifespan. In K. F. Fingerman, J. Smith, & T. C. Antonucci (Eds.), Handbook of lifespan development (pp. 117-140). New York, NY: Guilford Press. Retrieved from https://www.guilford.com/.
- McCurdy, L. E., Winterbottom, K. E., Mehta, S. S., & Roberts, J. R. (2010). Using nature and outdoor activity to improve children's health. *Current Problems in Pediatric and Adolescent Health Care*, 40(5), 102-117. doi:10.1016/j.cppeds.2010.02.003
- McDonald, G., Jackson, D., Wilkes, L., & Vickers, M. H. (2012). A work-based educational intervention to support the development of personal resilience in nurses and midwives. *Nurse Education Today*, 32(4), 378-384. doi:10.1016/j.nedt.2011.04.012

- McKenzie, M. D. (2000). How are adventure education program outcomes achieved? A review of the literature. *Australian Journal of Outdoor Education*, 5(1), 19-28.
- McKenzie, M. D. (2003). Beyond "the outward bound process:" Rethinking student learning. *Journal of Experiential Education*, 26(1), 8-23.
- Meichenbaum, D. (1977). Cognitive behaviour modification. *Cognitive Behaviour Therapy*, *6*(4), 185-192.
- Meichenbaum, D. (1996). Stress inoculation training for coping with stressors. *The Clinical Psychologist*, 49, 4-7.
- Meichenbaum, D. (2017a). *The evolution of Cognitive Behavior Therapy: A personal and professional journey*. In. Retrieved from https://www.routledge.com/9781138812222
- Meichenbaum, D. (2017b). Stress inoculation training: A preventative and treatment approach. In D. Meichenbaum (Ed.), The evolution of Cognitive Behavior Therapy: A personal and professional journey (pp. 120-146). London Taylor & Francis Group. Retrieved from https://www.routledge.com/9781138812222.
- Meichenbaum, D., & Cameron, R. (1989). Stress inoculation training. In D. Meichenbaum & M. E. Jaremko (Eds.), *Stress reduction and prevention* (pp. 115-154). New York, NY: Springer.
- Meichenbaum, D., & Deffenbacher, J. (1988). Stress inoculation training. *The Counseling Psychologist*, 16(1), 69-90. doi:10.1177/0011000088161005
- Melbourne Grammar School. (2012). Beyond the Gates. Retrieved from http://cdn.oeg.org.au/wp-content/uploads/2012/05/Beyond-the-Gates\_e0212.pdf
- Miles, J. C., & Priest, S. (1999). *Adventure programming*. State College, PA: Venture Publishing.
- Miller, J., & Allen-Craig, S. (2005). An evaluation of a cross section of Australian schools outdoor education program outcomes. Australian Catholic University,
- Miller, R. (2015). *The iRest program for healing PTSD: A proven-effective approach to using yoga nidra meditation and deep relaxation techniques to overcome trauma*. In. Retrieved from https://www.irest.us/research
- Miller, R. (2016). *Experiencing resiliency and well-being in every moment*. In. Retrieved from https://www.irest.us/research
- MindMatters. (2018a). *Developing resilience: Module* 2.2. In. Retrieved from https://www.mindmatters.edu.au/docs/default-source/learning-module-documents/mm\_module2\_2-moduleoverview.pdf?sfvrsn=2
- MindMatters. (2018b). *MindMatters mental health framework*. In. Retrieved from https://www.mindmatters.edu.au/tools-resources/component-framework

- Misra, R., Crist, M., & Burant, C. J. (2003). Relationships among life stress, social support, academic stressors, and reactions to stressors of international students in the United States. *International Journal of Stress Management*, 10(2), 137-157. doi:10.1037/1072-5245.10.2.137
- Mission Australia. (2016). Youth mental health report: Youth survey 2012-2016. *Mission Australia & Black Dog*.
- Nadler, R. S. (1993). Therapeutic process of change. In M. A. Gass (Ed.), *Adventure therapy: Therapeutic applications of adventure programming* (pp. 57-72). Boulder, CO: Kendall/Hunt Publishing Company.
- Nadler, R. S. (1995). Edgework: Stretching boundaries and generalizing experiences. *Journal of Experiential Education*, *18*(1), 52-55. doi:10.1177/105382599501800110
- Nakamura, J., & Csikszentmihalyi, M. (2005). The concept of flow. In C. R. Snyder & S. J. Lopez (Eds.), *Handbook of positive psychology*. New York: Oxford University Press.
- Neill, J. T. (1997). *Outdoor Education in the Schools: What can it achieve?* Paper presented at the Proceedings of the 10th National Outdoor Education Conference, Sydney, Australia.
- Neill, J. T. (2001). Simple, sophisticated and committed: Ways ahead for outdoor education in Australia. *Australian Journal of Outdoor Education*, 2.
- Neill, J. T. (2002). *Meta-analytic research on the outcomes of outdoor education*. Paper presented at the 6th Biennial Coalition for Outdoor Education in the Outdoors Research Symposium, Bradford Woods, IN. http://wilderdom.com/research/researchoutcomesmeta-analytic.htm
- Neill, J. T. (2008). Enhancing life effectiveness: The impacts of outdoor education programs. (Doctoral dissertation), University of Western Sydney, Sydney, Australia. Retrieved from https://researchdirect.westernsydney.edu.au/islandora/object/uws%3A644 1/datastream/PDF/view
- Neill, J. T., & Dias, K. L. (2001). Adventure education and resilience: The double-edged sword. *Journal of Adventure Education and Outdoor Learning*, 1(2), 35-42.
- Neill, J. T., & Richards, G. E. (1998). Does outdoor education really work? A summary of recent meta-analyses. *Australian Journal of Outdoor Education*, 3(1), 2-9.
- Netuveli, G., Wiggins, R. D., Montgomery, S. M., Hildon, Z., & Blane, D. (2008). Mental health and resilience at older ages: Bouncing back after adversity in the British household panel survey. *Journal of Epidemiology and Community Health*, 62(11), 987-991. doi:10.1136/jech.2007.069138

- Newes, S. L., & Bandoroff, S. (2004). What is adventure therapy? In S. Bandoroff
  & S. Newes (Eds.), *Coming of age: The evolving field of adventure therapy*. Boulder, Colorado: Association for Experiential Education.
- Norris, J. (2011). Crossing the threshold mindfully: Exploring rites of passage models in adventure therapy. *Journal of Adventure Education and Outdoor Learning*, *11*(2), 109-126. doi:10.1080/14729679.2011.633380
- Norrish, J., Robinson, J., & Williams, P. (2011). A model for positive education. *Corio: Geelong Grammar School*. Retrieved from www.ggs.vic.edu.au/PosEd
- Norton, C. L. (2008). Understanding the impact of wilderness therapyon adolescent depression and psychosocial development. *Illinois Child Welfare*, *4*(1), 166–178.
- Norton, C. L., Tucker, A., Russell, K. C., Bettmann, J. E., Gass, M. A., Gillis, H. L., & Behrens, E. (2014). Adventure therapy with youth. *Journal of Experiential Education*, 37(1), 46-59. doi:10.1177/1053825913518895
- Novaco, R. W. (1977). Stress inoculation: A cognitive therapy for anger and its application to a case of depression. *Journal of Consulting and Clinical Psychology*, *45*(4), 600-608. doi:http://dx.doi.org/10.1037/0022-006X.45.4.600
- Nunnaly, J. (1978). Psychometric theory. New York, NY: McGraw-Hill.
- Nussbaum, A. D., & Dweck, C. S. (2008). Defensiveness versus remediation: Self-theories and modes of self-esteem maintenance. *Personality and Social Psychology Bulletin*, 34(5), 599-612. doi:10.1177/0146167207312960
- O'Connell, E. (2005). Mood, energy, cognition, and physical complaints: A mind/body approach to symptom management during the climacteric. *Journal of Obstetric, Gynecologic, and Neonatal Nursing, 34*(2), 274-279.
- O'Dougherty Wright, M., & Masten, A. S. (2015). Pathways to resilience in context. In L. C. Theron, L. Liebenberg, & M. Ungar (Eds.), *Youth resilience and culture: commonalities and complexities*. Dordrecht: Springer.
- O'Dougherty Wright, M., Masten, A. S., & Narayan, A. J. (2013). Resilience processes in development: Four waves of research on positive adaptation in the context of adversity. In S. Goldstein & R. B. Brooks (Eds.), *Handbook of Resilience in Children*. New York: Springer Science+Business Media.
- Obradovic, J., Shaffer, A., & Masten, A. S. (2012). Risk and adversity in developmental psycho- pathology: Progress and future directions. In L. C. Mayes & M. Lewis (Eds.), *The Cambridge handbook of environment in human development environment of human development: A handbook of*

*theory and measurement* (pp. 35-37). New York, NY: Cambridge University Press.

- Olsson, C. A., Bond, L., Burns, J. M., Vella-Brodrick, D. A., & Sawyer, S. M. (2003). Adolescent resilience: A concept analysis. *Journal of adolescence*, 26(1), 1-11.
- Onwuegbuzie, A. J., & Johnson, R. B. (2006). The validity issue in mixed research. *Research in the Schools*, *13*(1), 48-63. doi:10.1016/j.jsis.2005.08.002
- Onwuegbuzie, A. J., Johnson, R. B., & Collins, K. M. T. (2009). Call for mixed analysis: A philosophical framework for combining qualitative and quantitative approaches. *International journal of multiple research* approaches, 3(2), 114-139. doi:10.5172/mra.3.2.114
- Onwuegbuzie, A. J., & Leech, N. L. (2007). Validity and qualitative research: An oxymoron? *Quality & Quantity*, 41(2), 233-249. doi:10.1007/s11135-006-9000-3
- Onwuegbuzie, A. J., & Teddlie, C. (2003). A framework for analyzing data in mixed methods research. In A. Tashakkori & C. Teddlie (Eds.), Handbook of mixed methods in social and behavioral research (pp. 351-383). Thousand Oaks, CA: Sage Publications. Retrieved from https://uk.sagepub.com/.
- Oshio, A., Kaneko, H., Nagamine, S., & Nakaya, M. (2003). Construct validity of the adolescent resilience scale. *Psychological Reports*, 93(3), 1217-1222. doi:10.2466/PR0.93.7.1217
- Panicucci, J. (2007). Cornerstones of adventure education. In D. Prouty, J. Panicucci, & R. Collinson (Eds.), *Adventure education: Theory and applications* (pp. 33-48). Champaign, IL: Human Kinetics.
- Park, C. L., & Blumberg, C. J. (2002). Disclosing trauma through writing: Testing the meaning-making hypothesis. *Cognitive Therapy & Research*, 26, 597-616.
- Park, N., Peterson, C., & Seligman, M. E. P. (2004). Strengths of character and well-being. *Journal of Social and Clinical Psychology*, 23(5), 603-619. doi:10.1521/jscp.23.5.603.50748
- Passarelli, A., Hall, E., & Anderson, M. (2010). A strengths-based approach to outdoor and adventure education: Possibilities for personal growth. *Journal of Experiential Education*, 33(2), 120-135.
- Pathways Health and Research Centre. (2018). *My FRIENDS youth program*. In. Retrieved from https://www.pathwayshrc.com.au/my-friends-youth/
- Patterson, J. M., & McCubbin, H. I. (1987). Adolescent coping style and behaviors: Conceptualization and measurement. *Journal of adolescence*, *10*(2), 163-186. doi:10.1016/S0140-1971(87)80086-6

- Patton, G. C., Sawyer, S. M., Santelli, J. S., Ross, D. A., Afifi, R., Allen, N. B., . . . Viner, R. M. (2016). Our future: A Lancet commission on adolescent health and wellbeing. *The Lancet*, 387(10036), 2423-2478. doi:10.1016/S0140-6736(16)00579-1
- Payne, S. C., Youngcourt, S. S., & Beaubien, J. M. (2007). A meta-analytic examination of the goal orientation nomological net. *Journal of Applied Psychology*, 92(1), 128-150. doi:10.1037/0021-9010.92.1.128
- Peacock, E. J., & Wong, P. T. P. (1990). The stress appraisal measure (SAM): A multidimensional approach to cognitive appraisal. *Stress Medicine*, 6, 277-236.
- Peterson, C., Park, N., & Seligman, M. E. P. (2005). Orientations to happiness and life satisfaction: The full life versus the empty life. *Journal of Happiness Studies*, *6*(1), 25-41. doi:10.1007/s10902-004-1278-z
- Peterson, C., Ruch, W., Beermann, U., Park, N., & Seligman, M. E. P. (2007). Strengths of character, orientations to happiness, and life satisfaction. *The Journal of Positive Psychology*, 2(3), 149-156.
- Piaget, J. (1964). Part I: Cognitive development in children: Piaget development and learning. *Journal of Research in Science Teaching*, 2(3), 176-186. doi:10.1002/tea.3660020306
- Piaget, J. (1972). Intellectual evolution from adolescence to adulthood. *Human Development*, 15(1), 1-12.
- Poggenpoel, M., & Myburgh, C. (2003). The researcher as research instrument in educational research: A possible threat to trustworthiness? *Education*, 124(2), 320-418. Retrieved from http://www.projectinnovation.com/education.html
- Priest, S. (1986). Redefining outdoor education: A matter of many relationships. *The Journal of Environmental Education*, *17*(3), 13-15. doi:10.1080/00958964.1986.9941413
- Priest, S. (1998). Physical challenge and the development of trust through corporate adventure training. *Journal of Experiential Education*, 21(1), 31-34. doi:10.1177/105382599802100107
- Priest, S. (1999). The adventure experience paradigm. In J. C. Miles & S. Priest (Eds.), Adventure programming (pp. 159-162). State College, PA: Venture Publishing.
- Priest, S., & Baillie, R. (1987). Justifying the risk to others: The real razor's edge. *Journal of Experiential Education*, 10(1), 16-22.
- Priest, S., & Gass, M. A. (2005). *Effective leadership in adventure programming* (2nd ed.). United States: Human Kinetics.
- Priest, S., & Gass, M. A. (2018a). *Effective leadership in adventure programming* (3rd ed.). Champaign, IL: Human Kinetics.

- Priest, S., & Gass, M. A. (2018b). Philosophy. In S. Priest & M. A. Gass (Eds.), *Effective leadership in adventure programming* (3rd ed., pp. 26-53). Champaign, IL: Human Kinetics.
- Priest, S., & Gass, M. A. (2018c). Social psychology. In S. Priest & M. A. Gass (Eds.), *Effective leadership in adventure programming* (3rd ed., pp. 196-215). Champaign, IL: Human Kinetics.
- Pryor, A., Townsend, M., Maller, C., & Field, K. (2006). Health and well-being naturally: 'Contact with nature' in health promotion for targeted individuals, communities and populations. *Health Promotion Journal of Australia*, 17(2), 114-123. doi:10.1071/HE06114
- Punch, M. (1994). Politics and ethics in qualitative research. In N. K. Denzin & Y. S. Lincoln (Eds.), *Handbook of qualitative research* (pp. 83-98). United States: Sage Publications.
- Rice-Oxley, M. (2019). Mental illness: is there really a global epidemic? Retrieved from https://www.theguardian.com/society/2019/jun/03/mentalillness-is-there-really-a-global-epidemic?CMP=share\_btn\_tw
- Richardson, G. E. (2002). The metatheory of resilience and resiliency. *Journal of Clinical Psychology*, 58(3), 307-321. doi:10.1002/jclp.10032
- Robinson, D., & Stevens, T. (1990). Stress in adventure recreation: types of stressors and their influences during an extended adventure-based expedition. *Journal of Applied Recreation Research*, 15(4), 218-238.
- Roehlkepartain, E. C., Pekel, K., Syvertsen, A. K., Sethi, J., Sullivan, T. K., & Scales, P. C. (2017). *Relationships first: Creating connections that help young people thrive.* In. Retrieved from www.search-institute.org
- Rohnke, K. (1989). *Cowstails and cobras II: A guide to games, initiatives, ropes courses & adventure curriculum*. In. Retrieved from https://www.kendallhunt.com/
- Rutter, M. (1985). Resilience in the face of adversity: Protective factors and resistance to psychiatric disorder. *British Journal of Psychiatry*, 147(DEC.), 598-611.
- Rutter, M. (1987). Psychosocial resilience and protective mechanisms. *American Journal of Orthopsychiatry*, 57(3), 316-331.
- Rutter, M. (1993). Resilience: Some conceptual considerations. *Journal of Adolescent Health, 14*, 626-631. doi:10.1016/1054-139X(93)90196-V
- Rutter, M. (2008). Developing concepts in developmental psychopathology. In J. J. Hudziak (Ed.), *Developmental psychopathology and wellness: Genetic and environmental influences* (pp. 3-22). Washington, DC: American Psychiatric Association.
- Ryan, A. B. (2006). Post-positivist approaches to research. In M. Antonesa, H. Fallon, A. B. Ryan, A. Ryan, T. Walsh, & L. Borys (Eds.), *Researching*

*and Writing your Thesis: a guide for postgraduate students* (pp. 12-26). Maynooth, Ireland: National University of Ireland.

- Ryan, R. M., & Deci, E. L. (2000). Self-determination theory and the facilitation of intrinsic motivation, social development, and well-being. *American Psychologist*, *55*(1), 68-78.
- Salvia, J., Ysseldyke, J. E., & Bolt, S. (2007). Assessment in special and inclusive education (10th ed.). Boston, MA: Wadsworth.
- Samuels, G. M., & Pryce, J. M. (2008). What doesn't kill you makes you stronger": Survivalist self-reliance as resilience and risk among young adults aging out of foster care. *Children and Youth Services Review*, 30(10), 1198-1210. doi:10.1016/j.childyouth.2008.03.005
- Santos, J. R. A. (1999). Cronbach's alpha: A tool for assessing the reliability of scales. *Journal of extension*, *37*(2), 1-5.
- Savery, J. R., & Duffy, T. M. (1995). Problem based learning: An instructional model and its constructivist framework. In B. G. Wilson (Ed.), *Designing constructivist learning environments* (pp. 135-138). Englewood Cliffs Educational Technology Publications.
- Savery, J. R., & Duffy, T. M. (2001). Problem based learning: An instructional model and its constructivist framework. *Center for Research on Learning* and Technology, Indiana University, 1-17.
- Sawyer, S. M., Afifi, R. A., Bearinger, L. H., Blakemore, S. J., Dick, B., Ezeh, A. C., & Patton, G. C. (2012). Adolescence: A foundation for future health. *The Lancet*, 379(9826), 1630-1640. doi:10.1016/S0140-6736(12)60072-5
- Scales, P. C., & Leffert, N. (2004). Developmental assets: A synthesis of the scientific research on adolescent development (2nd ed.). Minneapolis, MN: Search Institute.
- Schenck, J., & Cruickshank, J. (2015). Evolving Kolb: Experiential education in the age of neuroscience. *Journal of Experiential Education*, 38(1), 73-95. doi:10.1177/1053825914547153
- Schiffrin, H., Liss, M., Miles-McLean, H., Geary, K., Erchull, M., & Tashner, T. (2013). Helping or hovering? The effects of helicopter parenting on college students' well-being. *Journal of Child & Family Studies*, 23(3), 548-557. doi:10.1007/s10826-013-9716-3
- Schoeppe, A., Haggard, E. A., & Havighurst, R. J. (1953). Some factors affecting sixteen-year-olds' success in five developmental tasks. *The Journal of Abnormal and Social Psychology*, 48(1), 42. doi:10.1037/h0054913
- Schoon, I. (2013). Risk and resilience: Definitions. In I. Schoon (Ed.), Risk and resilience: Adaptations in changing times. New York, NY: Cambridge University Press. doi:10.1017/CBO9780511490132.004

- Schunk, D. H., & Meece, J. L. (2006). Self-efficacy development in adolescence. In F. Pajares & T. Urdan (Eds.), Self-efficacy beliefs of adolescents. Connecticut: Information Age Publishing.
- Schwandt, T. A. (2007). *The sage dictionary of qualitative inquiry*. Thousand Oaks, California: Sage Publications.
- Schwarzer, R., & Knoll, N. (2003). Positive coping: Mastering demands and searching for meaning. In S. J. Lopez & C. R. Snyder (Eds.), Positive psychological assessment: A handbook of models and measures (pp. 393-409). Washington, DC: American Psychological Association. doi:10.1037/10612-025
- Schwarzer, R., & Schwarzer, C. (1996). A critical survey of coping instruments. In M. Zeidner & N. S. Endler (Eds.), *Handbook of coping: Theory, research, applications* (pp. 107-132). Oxford, England: John Wiley & Sons.
- Seligman, M. E. P. (2011). Flourish. London: Nicholas Brealey Publishing.
- Seligman, M. E. P., & Csikszentmihalyi, M. (2000). Positive psychology: An introduction. *American Psychologist*, 55(1), 5-14.
- Selye, H. (1974). Stress without distress. In G. Serban (Ed.), *Psychopathology of human adaptation* (pp. 137-146). Boston, MA: Springer.
- Selye, H. (1975). Confusion and controversy in the stress field. *Journal of Human Stress, 1*(2), 37-44.
- Selye, H. (1976). Stress in health and disease. Boston, MA: Butterworth-Heinemann. Sheard, M., & Golby, J. (2006). The efficacy of an outdoor adventure education curriculum on selected aspects of positive psychological development. Journal of Experiential Education, 29, 187-209. doi:10.1177/105382590602900208
- Senninger, T. (2000). Abenteuer leiten. Münster/Germany: Ökotopia.
- Sesma Jr, A., Roehlkepartain, E. C., Benson, P. L., & Van Dulmen, M. (2003). Unique strengths, shared strengths: Developmental assets among youth of color. Search Institute Insights & Evidence, 1(2), 1-13. Retrieved from https://www.search-institute.org/search-institute-insights-evidence/
- Shaffer, D. R. (2005). *Social and personality development, 5th ed.* Belmont, CA, US: Thomson Wadsworth.
- Sheard, M., & Golby, J. (2006). The efficacy of an outdoor adventure education curriculum on selected aspects of positive psychological development. *Journal of Experiential Education*, 29(2), 187-209. doi:10.1177/105382590602900208
- Sheehan, P., Sweeny, K., Rasmussen, B., Wils, A., Friedman, H. S., Mahon, J., . . . Laski, L. (2017). Building the foundations for sustainable development:

A case for global investment in the capabilities of adolescents. *The Lancet, 390*(10104), 1792-1806. doi:10.1016/S0140-6736(17)30872-3

- Shellman, A. (2009). Empowerment and resilience: A multi-method approach to understanding processes and outcomes of adventure education program experiences. Unpublished doctoral dissertation. Indiana University. Indiana, MI. Retrieved from https://library.vu.edu.au/articles/4341894.38089/1.pdf.
- Shenton, A. K. (2004). Strategies for ensuring trustworthiness in qualitative research projects. *Education for information*, 22(2), 63-75. doi:10.3233/EFI-2004-22201
- Shernoff, D. J., Csikszentmihalyi, M., Shneider, B., & Shernoff, E. S. (2003). Student engagement in high school classrooms from the perspective of flow theory. *School Psychology Quarterly*, 18(2), 158-176. doi:10.1521/scpq.18.2.158.21860
- Shonkoff, J. P. (2015). Supportive relationships and active skill-building: Strengthen the foundations of resilience (No. 13). Retrieved from Cambridge, MA: http://developingchild.harvard.edu/resources/supportiverelationships-and-active-skill-building-strengthen-the-foundations-ofresilience/
- Shonkoff, J. P., Boyce, W. T., Cameron, J., Duncan, G. J., Fox, N. A., Gunnar, M. R., & Thompson, R. A. (2009). Excessive stress disrupts the architecture of the developing brain. Retrieved from http://developingchild.harvard.edu/
- Sibthorp, J. (2003). Learning transferable skills through adventure education: The role of an authentic process. *Journal of Adventure Education and Outdoor Learning*, *3*(2), 145-157.
- Sibthorp, J. (2006). An empirical look at Walsh and Golins' adventure education process model: Relationships between antecedent factors, perceptions of characteristics of an adventure education experience, and changes in self-efficacy. *Journal of Leisure Research*, *35*(1), 80-107. doi:10.18666/jlr-2003-v35-i1-611
- Sibthorp, J., Furman, N., Paisley, K., Gookin, J., & Schumann, S. (2011).
  Mechanisms of learning transfer in adventure education: Qualitative results from the NOLS transfer survey. *Journal of Experiential Education*, 34(2), 109-126. Retrieved from http://www.sagepub.com.wallaby.vu.edu.au:2048/journals/Journal202222
- Sibthorp, J., Paisley, K., & Gookin, J. (2007). Exploring participant development through adventure-based programming: A model from the National Outdoor Leadership School. *Leisure Sciences*, 29(1), 1-18. doi:10.1080/01490400600851346
- Skehill, C. M. (2001). *Resilience, coping with an extended stay outdoor education program, and adolescent mental health.* (Honours thesis Unpublished

honours thesis), University of Canberra, ACT, Australia. Retrieved from https://www.researchgate.net/publication/238100926\_Resilience\_Coping\_ with\_an\_Extended\_Stay\_Outdoor\_Education\_Program\_and\_Adolescent\_ Mental\_Health

- Smith, B., Dalen, J., Wiggins, K., Tooley, E., Christopher, P., & Bernard, J. (2008). The brief resilience scale: Assessing the ability to bounce back. *International Journal of Behavioral Medicine*, 15(3), 194-200. doi:10.1080/10705500802222972
- Smith, B. W., Tooley, E. M., Christopher, P. J., & Kay, V. S. (2010). Resilience as the ability to bounce back from stress: A neglected personal resource? *The Journal of Positive Psychology*, 5(3), 166-176. doi:10.1080/17439760.2010.482186
- Snyder, C. R., & Lopez, S. J. (2002). *Handbook of positive psychology*. New York, NY: Oxford University Press.
- Sofaer, S. (1999). Qualitative methods: what are they and why use them? *Health services research*, *34*(5 Pt 2), 1101.
- Somervell, J., & Lambie, I. (2009). Wilderness therapy within an adolescent sexual offender treatment programme: A qualitative study. *Journal of Sexual Aggression*, *15*(2), 161-177. doi:10.1080/13552600902823055
- Spangler, N. W., Koesten, J., Fox, M. H., & Radel, J. (2012). Employer perceptions of stress and resilience intervention. *Journal of Occupational* and Environmental Medicine, 54(11), 1421-1429. doi:10.1097/JOM.0b013e3182619038
- Stagman-Tyrer, D. (2014). Resiliency and the nurse leader: The importance of equanimity, optimism, and perseverance. *Nursing management*, 45(6), 46-50. doi:10.1097/01.NUMA.0000449763.99370.7f
- Statista. (2018a). Number of Facebook users by age in the U.S. as of January 2017 (in millions). Retrieved from https://www.statista.com/statistics/398136/us-facebook-user-age-groups/
- Statista. (2018b). Number of monthly active Facebook users worldwide as of 4th quarter 2017 (in millions). Retrieved from https://www.statista.com/statistics/264810/number-of-monthly-active-facebook-users-worldwide/
- Staunton, N. (2003). A meta-analysis of adventure therapy program outcomes. University of New Hampshire. Unpublished manuscript.
- Stokes, T. B. (2009). What freud didn't know: A three-step practice for emotional well-being through neuroscience and psychology. New Brunswick, NJ: Rutgers University Press.
- Sun, J., & Stewart, D. (2007). Development of population-based resilience measures in the primary school setting. *Health Education*, 107(6), 575-599. doi:10.1108/09654280710827957

- Svinicki, M. D. (2004). Authentic assessment: Testing in reality. In M. V. Achacoso & M. D. Svinicki (Eds.), *Alternative Strategies for Evaluating Student Learning*. (pp. 1-122). San Francisco: Jossey-Bass.
- Swanson, K. M. (2000). Predicting depressive symptoms after miscarriage: A path analysis based on the Lazarus paradigm. *Journal of Women's Health & Gender-Based Medicine*, 9(2), 191-206. doi:10.1089/152460900318696
- Swarbrick, N., Eastwood, G., & Tutton, K. (2004). Self-esteem and successful interaction as part of the forest school project. *Support for Learning*, *19*(3), 142-146. doi:10.1111/j.0268-2141.2004.00337.x
- Taylor, R. D., & Dymnicki, A. B. (2007). Empirical evidence of social and emotional learning's influence on school success: A commentary on "building academic success on social and emotional learning: What does the research say?," A book edited by Joseph E. Zins, Roger P. Weissberg, Margaret C. Wang, and Herbert J. Walberg. *Journal of Educational & Psychological Consultation*, *17*(2/3), 225-231. doi:10.1080/10474410701346725
- Taylor, S. J., & Bodgan, R. (1984). *Introduction to qualitative research methods: The search for meanings*. New York, NY: Wiley.
- Tedeschi, R. G., & Calhoun, L. G. (1996). The Posttraumatic Growth Inventory: Measuring the positive legacy of trauma. *Journal of Traumatic Stress*, 9(3), 455-471.
- The Department of Health. (2014). *The Mental health of children and adolescents: Report on the second Australian Child and Adolescent Survey of Mental Health and Wellbeing*. Retrieved from http://www.health.gov.au/internet/main/publishing.nsf/Content/mental-pubs-m-child2.
- The Search Institute. (2014). A research update from Search Institute: Developmental relationships. Retrieved from http://www.searchinstitute.org/downloadable/Dev-Relationships-Framework-Sept2014.pdf
- The Search Institute. (2016). 40 developmental assets for adolescents (ages 12-18). Retrieved from http://www.search-institute.org/content/40developmental-assets-adolescents-ages-12-18
- Toland, J., & Carrigan, D. (2011). Educational psychology and resilience: New concept, new opportunities. *School Psychology International*, *32*(1), 95-106. doi:10.1177/0143034310397284
- Tucker, A. R., Combs, K. M., Bettmann, J. E., Chang, T.-H., Graham, S., Hoag, M., & Tatum, C. (2018). Longitudinal outcomes for youth transported to wilderness therapy programs. *Research on social work practice*, 28(4), 438-451. doi:10.1177/1049731516647486
- Tudor, K. (1996). *Mental health promotion: Paradigms and practice*. London: Routledge.

- Tugade, M. M., & Fredrickson, B. L. (2004). Resilient individuals use positive emotions to bounce back from negative emotional experiences. *Journal of Personality and Social Psychology*, 86(2), 320-333. doi:10.1037/0022-3514.86.2.320
- Ungar, M. (2003). Qualitative contributions to resilience research. *Qualitative Social Work*, 2(1), 85-102. doi:10.1177/1473325003002001123
- Ungar, M. (2004). A constructionist discourse on resilience: Multiple contexts, multiple realities among at-risk children and youth. *Youth Society*, *35*(3), 341-365. doi:10.1177/0044118X03257030
- Ungar, M. (2008). Resilience across cultures. *The British Journal of Social Work*, 38(2), 218-235. doi:10.1093/bjsw/bcl343
- Ungar, M. (2011). The social ecology of resilience: Addressing contextual and cultural ambiguity of a nascent construct. *American Journal of Orthopsychiatry*, *81*(1), 1-17. doi:10.1111/j.1939-0025.2010.01067.x
- Ungar, M. (2012). *The social ecology of resilience: A handbook of theory and practice*. New York Springer.
- Ungar, M., Dumond, C., & McDonald, W. (2005). Risk, resilience and outdoor programmes for at-risk children. *Journal of Social Work*, *5*(3), 319-338. doi:10.1177/1468017305058938
- Ungar, M., Liebenberg, L., Boothroyd, R., Kwong, W. M., Lee, T. Y., Leblanc, J.,
  ... Makhnach, A. (2008). The study of youth resilience across cultures:
  Lessons from a pilot study of measurement development. *Research in Human Development*, 5(3), 166-180. doi:10.1080/15427600802274019
- UNICEF. (2011). The state of the world's children 2011: Adolescence an age of opportunity. In. Retrieved from www.unicef.org/sowc2011
- United Nations Department of Economic and Social Affairs: Population Division. (2017). *World population prospects: The 2017 revision*. In. Retrieved from http://www.un.org/en/development/desa/population/publications/pdf/agein g/WPA2017\_Highlights.pdf
- Van Ryzin, M. J., Gravely, A. A., & Roseth, C. J. (2009). Autonomy, belongingness, and engagement in school as contributors to adolescent psychological well-being. *Journal of Youth and Adolescence*, 38(1), 1-12. doi:https://doi.org/10.1007/s10964-007-9257-4
- Vera, E. M., & Shin, R. Q. (2006). Promoting strengths in a socially toxic world: Supporting resiliency with systemic interventions. *The Counseling Psychologist*, 34(1), 80-89. doi:10.1177/0011000005282365
- Victorian Curriculum and Assessment Authority. (2009). *Taking the VELS Outdoors: Meeting the VELS Through Outdoor Education*. Retrieved from http://vels.vcaa.vic.edu.au/hpe/outdoor\_education.html.

- Vissell, R. (2005). Effects of wilderness therapy on youth at risk's concept of self and other: A deeper understanding of the journey. (Unpublished doctoral dissertation, Palo Alto, California: Institute of Transpersonal Psychology), Available from ProQuest Dissertations & Theses database. (UMI no. 0419-4217)
- Von Culin, K. R., Tsukayama, E., & Duckworth, A. L. (2014). Unpacking grit: Motivational correlates of perseverance and passion for long-term goals. *The Journal of Positive Psychology*, 9(4), 306-312. doi:10.1080/17439760.2014.898320
- Von Glasersfeld, E. (1989). Cognition, construction of knowledge, and teaching. *Synthese*, *80*(1), 121-140.
- Wagnild, G. M. (2009). A review of the Resilience Scale. Journal of Nursing Measurement, 17(2), 105-113.
- Wagnild, G. M. (2010). Discovering your resilience core. Retrieved from The official home of The Resilience Scale website: http://www.resiliencescale.com/papers/pdfs/Discovering\_Your\_Resilience\_\_Core.pdf
- Wagnild, G. M. (2011). Will resilience help you be more successful? Retrieved from The official home of The Resilience Scale website: http://www.resiliencescale.com/papers/pdfs/Success.pdf
- Wagnild, G. M., & Guinn, P. E. (2011). The resilience scale user's guide: For the US English version of the resilience scale and the 14-item resilience scale. Worden, MT: The Resilience center.
- Wagnild, G. M., & Young, H. M. (1993). Development and psychometric evaluation of the Resilience Scale. *Journal of Nursing Measurement*, 1(2), 165-178.
- Wallia, S. S. (2008). Challenge by choice: A sojourn at the intersection of challenge and choice. *Australian Journal of Outdoor Education*, 12(2), 39-46. Retrieved from https://outdooreducationaustralia.org.au/
- Walsh, V., & Golins, G. (1976). *The exploration of the Outward Bound process*. Unpublished manuscript. Outward Bound School. Denver, CO. Retrieved from https://eric.ed.gov/?id=ED144754
- Warren, K., Sakofs, M., & Hunt Jr, J. S. (Eds.). (1995). *The theory of experiential education*. Boulder, CO: Association for Experiential Education.
- Watson, J. B. (1913). Psychology as the behaviorist views it. *Psychological Review*, 20, 158-177.
- Werner, E. E. (1995). Resilience in development. *Current directions in psychological science*, 4(3), 81-85. doi:10.1111/1467-8721.ep10772327
- Werner, E. E. (2000). Protective factors and individual resilience. In J. P. Shonkoff & S. J. Meisels (Eds.), Handbook of early childhood intervention

(2nd ed., pp. 115-132). New York, NY: Cambridge University Press. doi:10.1017/CBO9780511529320.008

- Werner, E. E., & Smith, R. S. (1982). Vulnerable but invincible: A longitudinal study of resilient children and youth. New York, NY: McGraw-Hill.
- Werner, E. E., & Smith, R. S. (1992). Overcoming the odds: High risk children from birth to adulthood. New York, NY: Cornell University Press.
- Werner, E. E., & Smith, R. S. (2001). Journeys from childhood to midlife: Risk, resilience, and recovery. Ithaca,NY: Cornell University Press.
- West, C. P. (2012). A mixed methods sequential explanatory study of the impact of chronic pain on family resilience. (Doctoral dissertion), James Cook University, Townsville, Australia. Retrieved from https://www.jcu.edu.au/
- Wildemuth, B. M. (1993). Post-positivist research: Two examples of methodological pluralism. *The Library Quarterly*, 63(4), 450-468. doi:10.1086/602621
- Wilson, K. G., & Baer, R. A. (2010). Assessing mindfulness and acceptance processes in clients: Illuminating the theory and practice of change. In The Context Press Mindfulness and Acceptance Practica Series.
- Windle, G. (2011). What is resilience? A review and concept analysis. *Reviews in Clinical Gerontology*, 21(2), 152-169. doi:10.1017/S0959259810000420
- Windle, G., Bennett, K. M., & Noyes, J. (2011). A methodological review of resilience measurement scales. *Health and Quality of Life Outcomes*, 9(1), 8-25. doi:10.1186/1477-7525-9-8
- Wolfe, B. D., & Samdahl, D. M. (2005). Challenging assumptions: Examining fundamental beliefs that shape challenge course programming and research. *Journal of Experiential Education*, 28(1), 25-43. Retrieved from http://www.sagepub.com.wallaby.vu.edu.au:2048/journals/Journal202222
- Wong, L. C. J., Reker, G. T., & Peacock, E. J. (2006). A resource-congruence model of coping and the development of the Coping Schemas Inventory. In P. T. P. Wong & L. C. J. Wong (Eds.), *Handbook of multicultural perspectives on stress and coping* (pp. 223-283). New York, NY: Springer.
- Wood, A. M., Linley, P. A., Maltby, J., Kashdan, T. B., & Hurling, R. (2011). Using personal and psychological strengths leads to increases in wellbeing over time: A longitudinal study and the development of the strengths use questionnaire. *Personality and Individual Differences*, 50(1), 15-19. doi:10.1016/j.paid.2010.08.004
- World Health Organization. (2009). *Milestones in health promotion: Statements from global conferences*. World Health Organization Retrieved from http://www.who.int/healthpromotion/Milestones\_Health\_Promotion\_0502 2010.pdf.

- World Health Organization. (2012a). *Mental health fact sheet*. World Health Organization Retrieved from http://www.wpro.who.int/mediacentre/factsheets/fs\_201203\_mental\_healt h/en/index.html#.
- World Health Organization. (2012b). Risk Factors. Retrieved from World Health Organisation website: http://www.who.int/topics/risk\_factors/en/
- World Health Organization. (2018). Adolescents: Health risks and solutions. Retrieved from http://www.who.int/news-room/factsheets/detail/adolescents-health-risks-and-solutions
- Yardley, L. (2008). Demonstrating validity in qualitative psychology. In J. A. Smith (Ed.), *Qualitative psychology: A practical guide to research methods* (2nd ed., pp. 235-251). Thousand Oaks, CA: Sage Publications.
- Yerkes, R. M., & Dodson, J. D. (1908). The relation of strength of stimulus to rapidity of habit-formation. *Journal of Comparative Neurology and Psychology*, 18(5), 459-482.
- Yong, A. G., & Pearce, S. (2013). A beginner's guide to factor analysis: Focusing on exploratory factor analysis. *Tutorials in Quantitative Methods for Psychology*, 9(2), 79-94. doi:10.20982/tqmp.09.2.p079
- Yoshino, A. (2008). Stress and coping in the context of adventure education: Testing a hypothetical model of perceived psychological growth (Doctoral dissertation). Available from ProQuest Dissertations and Theses database.
- Yusoff, N., Low, W. Y., & Yip, C. H. (2010). Reliability and validity of the Brief COPE Scale (English version) among women with breast cancer undergoing treatment of adjuvant chemotherapy: A Malaysian study. *Med J Malaysia*, 65(1), 41-44.
- Zimmerman, B. J., & Cleary, T. J. (2006). Adolescents' development of personal agency: The role of self-efficacy beliefs and self-regulatory skill. In F. Pajares & T. Urdan (Eds.), *Self-efficacy beliefs of adolescents*. Connecticut: Information Age Publishing.
- Zinc, R. (2004). The shifting sands of challenge-by-choice: Does it have a place in safety management in EOTC? *Journal of Physical Education New Zealand*, 37(1), 6-16. Retrieved from https://penz.org.nz/

# APPENDICES

Appendix A: Human Research Ethics Approval





Dr Peter Burridge	
School of Education	D
Department of Arts, Education a nd Human Development	
Victoria University	

ATE 10/082012

FROM Dr Debra Kerr Acting Chair Victoria University Human Research Ethics Committee

# SUBJECT Ethics Application - HRETH 12/198

Dear Dr Burridge,

TO

Thank you for submitting this a pplication for ethical a pproval of the project entitled:

HRETH 12/198 Resilience, adolescents and outdoor education: Is resilience context specific? (HREC 12/141)

The proposed research project has been accepted and deemed to meet the requirements of the National Health and Medical Research Council (NHMRC) 'National Statement on Ethical Conduct in Human Research (2007)' by the Victoria University Human Research Ethics Committe e. Approval has been granted from **10 August 2012 to 10 August 2014**.

Continued approval of this research project by the Victoria University Human Research Ethics Committee (VUHREC) is conditional upon the provision of a report within 12 months of the above approval date (**10 August 2013**) or upon the completion of the project (if earlier). A report proforma may be downloaded from the VUHREC web site at: http://research.vu.edu.au/hrec.php

Please note that the Human Research Ethics Committee must be informed of the following: any changes to the approved research protocol, project timelines, any serious events or adverse and/or unforeseen events that may affect continued ethical acceptability of the project. In these unlikely events, researchers must immediately cease all data collection until the Committee has approved the changes. Researchers are also reminded of the need to notify the approving HREC of changes to personnel in research projects via a request for a minor amendment. It should also be noted that it is the Chief Investigators' responsibility to ensure the research project is conducted in line with the recommendations outlined in the National Health and Medical Research Council (NHMRC) 'National Statement on Ethical Conduct in Human Research (2007).'

On behalf of the Committee, I wish you all the best for the conduct of the project.

Kind regards,

**Dr Debra Kerr** Acting Chair Victoria University Human Research Ethics Committee Appendix B: Invitation Letter to Program Group Principal



August 3, 2012

RE: Casie-Anne Chalman's, request for Yarra College to participate in research.

Dear Mr

My name is Casie-Anne Chalman, and I am writing to you to invite Yarra College to participate in a research project entitled Resilience, adolescents and outdoor education: Is resilience context specific? I am currently working as a Lecturer in the Outdoor Education Department at Victoria University and completing my Masters in Education (Outdoor Education) at Victoria University.

Yarra College would be a perfect candidate for involvement in this research project as I know you deliver outdoor education programs from years 7-9, as well as providing the City to Summit experience, a compulsory extended journey style outdoor education program for your Year 10 cohort. As developing resilience and educating for the whole person are core values of Yarra College, this research aligns directly with your school culture.

Increasing our understanding of the enhancement and application of resilience attributes in secondary school students, will enable school teachers, educators and outdoor education providers to develop improved and effective programs that assist the development and transference of resilience in young people.

Please refer to the information sheet attached which outlines the schools involvement and the requirements of participants.

Thank you for your time and assistance in this research project.

Sincerely,

Student Researcher:

Casie Chalman	
Phone:	
Email:	

Appendix C: Invitation Letter to Control Group Principal



3. August. 2012

**School of Education** 



RE: Casie Chalman's, request for Beachside College to participate in research.

# Dear Mr

My name is Casie-Anne Chalman, and I am writing to you to invite Beachside College to participate in a research project entitled Resilience, adolescents and outdoor education: Is resilience context specific? I am currently working as a Lecturer in the Outdoor Education Department at Victoria University and completing my Masters in Education (Outdoor Education) at Victoria University.

Beachside College would be a perfect candidate for involvement in this research project as I know you deliver outdoor education programs from years 7-9 and VCE Outdoor and Environmental Studies. As developing resilience and educating for the whole person are core values of Beachside College, this research aligns directly with your school culture.

Increasing our understanding of the enhancement and application of resilience attributes in secondary school students, will enable school teachers, educators and outdoor education providers to develop improved and effective programs that assist the development and transference of resilience in young people.

Please refer to the information sheet attached which outlines the schools involvement and the requirements of participants.

Thank you for your time and assistance in this research project.

Sincerely,

Student Researcher:

Casie Chalman	
Phone:	
Email:	

Appendix D: Information to Program Group (School)



# INFORMATION TO SCHOOLS INVOLVED IN RESEARCH

#### You are invited to participate

Yarra College is invited to participate in a research project entitled Resilience, Adolescents and Outdoor Education: Is resilience context specific?

This project is being conducted by a student researcher, Casie Chalman as part of a Masters by Research study at Victoria University under the supervision of Dr Peter Burridge from the Department of Arts, Education and Human Development and Dr Lauren Banting from the Institute of Sport, Exercise and Active Living (ISEAL).

#### **Project explanation**

This research will explore the experiences of students who have participated in the Year 10 *City to Summit*. The aim of this research is to investigate which attributes of resilience and coping are developed and transferred into the lives of young people through participation in a journey style outdoor education program. The first aspect will look at attributes of resilience and coping through an online questionnaire which will occur before and after the program.

The second phase of the research will invite twenty students in groups of four to be a part of a small group discussion to talk about their experience on the program. The information gathered from the research will then be used to inform future practice and development of outdoor education programs.

#### What will the school be asked to do?

The school will be asked to support the research by allowing access for Year 10 participants involved in the *City to Summit* outdoor education program, to complete two online questionnaires and provide a room to conduct the small group discussions. School email addresses of the students participating on the program will also be required to send the questionnaire links. Specifically, arrangements will be made with teachers from Yarra College to access a room and schedule time to administer the questionnaires on two occasions. These questionnaires will take approximately 10 minutes to complete.

#### What will school gain from participating?

The school will be provided with an end of project report that will help to inform future practice and improved programs which develop, enhance and transfer attributes of resilience and coping skills in young people.

#### How will the information be used?

All the information collected during the study will be used to identify which attributes of resilience and coping are developed and transferred into the lives of young people through participation in a journey style outdoor education program. These findings will be used to inform school teachers, educators and outdoor education providers to develop improved and effective outdoor education programs which facilitate the development and transference of resilience attributes in young people. The findings of the research will be presented in a Thesis, at conferences and in journal articles.

## What are the potential risks of participating in this project?

The risk to school participants is very small. There is the small possibility that during participation in the small group interviews or partaking in the questionnaires, students may reminded of an unpleasant past experience and students

may not feel comfortable talking about the experience. Should this occur, students do not have to complete the questionnaire or provide a response during the interview. If students are particularly distressed or uncomfortable, the interview will be stopped immediately by the researcher. During the interviews, students may also feel uncomfortable talking freely in front of other students in the small group. The researcher will be conducting the discussions to minimise any coercive behaviour and students will never be forced to respond to any questions. The researcher will stop the interview if she feels that any participant is particularly uncomfortable. Anyone who appears to be participating under duress will also be withdrawn from the study discretely. If during an interview a student becomes uncomfortable with or distressed by the material, that line of questioning will be stopped. If a participant becomes distressed they will be offered the option of additional assistance from the school student welfare counsellor, as the person most appropriate to speak to students from the school. If required, students will also be told of the option to speak to an independent clinical psychologist, Dr Carolyn Deans, who is a staff member at Victoria University but who is not involved in this research program. Please remember your participation is voluntary and you may withdraw from the research at anytime. Be assured that the small group interviews will be conducted with care and the researcher has extensive experience working with young people. All data collected will be kept confidential and will not be connected with the schools name in any way. This research is focused on the development of resilience and outdoor education programs and is not linked to the learning tasks or assessment tasks you may complete as part of your studies.

#### How will this project be conducted?

Information will be collected through two phases. In the first phase, students will receive a link via email to the questionnaires online. They will be given time to complete the questionnaire at school during two different times. The first questionnaire will be held on September 21<sup>st</sup> and 22<sup>nd</sup> 2012. The final questionnaire will be conducted throughout the month of November 2012 and will be completed the day after they complete the *City to Summit* program. The questionnaire should only take about ten minutes to complete.

The second phase involves group discussions, where twenty randomly selected students will be asked to participate in small group discussions. The first will occur straight after the *City to Summit* program and should last for approximately one hour.

#### Who is conducting the study?

Victoria University



Any queries about your participation in this project may be directed to the Chief Investigator listed above. If you have any queries or complaints about the way you have been treated, you may contact the Research Ethics and Biosafety Manager, Victoria University Human Research Ethics Committee, Victoria University, PO Box 14428, Melbourne, VIC, 8001 or phone (03) 9919 4148. Appendix E: Information to Control Group (School)



# INFORMATION TO SCHOOLS INVOLVED IN RESEARCH

#### You are invited to participate

Beachside College is invited to participate in a research project entitled Resilience, Adolescents and Outdoor Education: Is resilience context specific?

This project is being conducted by a student researcher, Casie Chalman as part of a Masters by Research study at Victoria University under the supervision of Dr Peter Burridge from the Department of Arts, Education and Human Development and Dr Lauren Banting from the Institute of Sport, Exercise and Active Living (ISEAL).

### **Project explanation**

This research will examine the student experience during their Year 10 studies. The aim of this research is to investigate which attributes of resilience and coping are developed and drawn upon by young people during their Year 10 experience. The research will look at attributes of resilience and coping through an online questionnaire which will occur over two different time periods to see if there is any change in their levels of resilience or coping over time.

#### What will the school be asked to do?

The school will be asked to support the research by allowing access for Year 10 participants who are not studying outdoor education to be involved in the study by completing surveys. School email addresses of the students participating on the program will also be required to send the questionnaire links. Specifically, arrangements will be made with teachers from Beachside College to access a room and schedule time to administer the questionnaires on two occasions. These questionnaires will take approximately 10 minutes to complete.

# What will school gain from participating?

The school will be provided with an end of project report that will help to inform future practice and improved programs which develop, enhance and transfer attributes of resilience and coping skills in young people.

# How will the information be used?

All the information collected during the study will be used to identify which attributes of resilience and coping are developed and transferred into the lives of young people during their Year 10 studies. These findings will be used to inform school teachers and educators to develop improved and effective programs which facilitate the development and transference of resilience attributes and coping skills in young people. The findings of the research will be presented in a Thesis, at conferences and in journal articles.

#### What are the potential risks of participating in this project?

The risk to school participants is very small. There is the small possibility that completing the questionnaires, may remind students of an unpleasant past experience. Should this occur, students do not have to complete the questionnaire. If students are particularly distressed or uncomfortable, completing the questionnaire will be stopped. If required, students will also be told of the option to speak to an independent clinical psychologist, Dr Carolyn Deans, who is a staff member at Victoria University and is not involved in this research program. Student participation is voluntary and you may withdraw from the research at anytime. All data collected will be kept confidential and will not be connected with the students' name in any way. This research is focused on the development of resilience and outdoor

education programs and is not linked to the learning tasks or assessment tasks you may complete as part of students' studies.

# How will this project be conducted?

Students' will receive a link via email to the questionnaires online. They will be given time to complete the questionnaire at school during two different times. The first questionnaire will be held during September 2012 and the final questionnaire will be conducted throughout the month of November 2012. The questionnaire should only take about ten minutes to complete.

# Who is conducting the study?

Victoria University

Chief investigator:



Associate Investigator:



Student Researcher:

Casie Chalman Phone: Email:

Any queries about your participation in this project may be directed to the Chief Investigator listed above. If you have any queries or complaints about the way you have been treated, you may contact the Research Ethics and Biosafety Manager, Victoria University Human Research Ethics Committee, Victoria University, PO Box 14428, Melbourne, VIC, 8001 or phone (03) 9919 4148. Appendix F: Program Group Consent Form (School)



# CONSENT FORM FOR SCHOOL INVOLVEMENT IN THE RESEARCH

# **INFORMATION TO PARTICIPANTS:**

We would like to invite Yarra College to be a part of a study entitled Resilience, Adolescents and Outdoor Education: Is resilience context specific?

This research will explore the experiences of students who have participated on the *City to Summit* outdoor education program. The aim of this research is to investigate which attributes of resilience and coping are developed and transferred into the lives of young people through participation in a journey style outdoor education program. The first aspect will look at attributes of resilience and coping through an online questionnaire which will occur before and after the program.

The second phase of the research will invite twenty students in groups of four to be a part of small group discussions to talk about their experience on the program. The information gathered from the research will then be used to inform future practice and development of outdoor education programs.

The research is voluntary and the information retrieved from data collection will be kept confidential and nobody will be named personally. Participants can withdraw from the research if they do not want to proceed.

# **CERTIFICATION BY SUBJECT**

I, .....(Position held)

I certify that the objectives of the study, together with any risks and safeguards associated with the procedures listed hereunder to be carried out in the research, have been fully explained to me by:

#### Casie Chalman

and that I freely consent to students being invited to voluntarily be involved in the study and that the below mentioned procedures can be conducted at the school:

- Students' are able to participate in completing two online questionnaires which will be conducted in September 2012 and November 2012.
- Students' are able to participate in small group discussions which will explore their experience after the program.

I certify that I have had the opportunity to have any questions answered and that I understand that the school can withdraw from this study at any time and that this withdrawal will not jeopardise the school in any way.

I have been informed that the information provided by staff and students will be kept confidential.

Signed:

(Principal or Authorised Representative of Yarra College)

Date:

Appendix G: Control Group Consent Form (School)



# CONSENT FORM FOR SCHOOL INVOLVEMENT IN THE RESEARCH

# **INFORMATION TO PARTICIPANTS:**

We would like to invite Beachside College to be a part of a study entitled Resilience, Adolescents and Outdoor Education: Is resilience context specific?

This research will examine the student experience of students who have chosen not to be involved in outdoor education during their Year 10 studies. The aim of this research is to investigate which attributes of resilience and coping are developed and drawn upon by young people during their Year 10 experience. The research will look at attributes of resilience and coping through an online questionnaire which will occur over two different time periods to see if there is any change in their levels of resilience or coping over time.

The research is voluntary and the information retrieved from data collection will be kept confidential and nobody will be named personally. Participants can withdraw from the research if they do not want to proceed.

# **CERTIFICATION BY SUBJECT**

I, .....(name)......(Position held)

On behalf of ......(name of school) give my consent for the school to participate in the study: Resilience, Adolescents and Outdoor Education: Is resilience context specific? This project is being conducted by a student researcher, Casie Chalman as part of a Masters by Research study at Victoria University under the supervision of Dr Peter Burridge from the Department of Arts, Education and Human Development and Dr Lauren Banting from the Institute of Sport, Exercise and Active Living (ISEAL).

I certify that the objectives of the study, together with any risks and safeguards associated with the procedures listed hereunder to be carried out in the research, have been fully explained to me by:

### Casie Chalman

and that I freely consent to students being invited to voluntarily be involved in the study and that the below mentioned procedures can be conducted at the school:

Students' are able to participate in completing two online questionnaires which will be conducted in September 2012 and November 2012.

I certify that I have had the opportunity to have any questions answered and that I understand that the school can withdraw from this study at any time and that this withdrawal will not jeopardise the school in any way.

I have been informed that the information provided by staff and students will be kept confidential.

Signed:

(Principal or Authorised Representative of Beachside College)

Date:

Appendix H: Information to Program Group (Parents and Participants)


# INFORMATION TO PARTICIPANTS INVOLVED IN RESEARCH

#### You are invited to participate

You are invited to participate in a research project entitled Resilience, Adolescents and Outdoor Education: Is resilience context specific?

This project is being conducted by a student researcher, Casie Chalman as part of a Masters by Research study at Victoria University under the supervision of Dr Peter Burridge from the Department of Arts, Education and Human Development and Dr Lauren Banting from the Institute of Sport, Exercise and Active Living (ISEAL).

#### **Project explanation**

This research will examine two aspects of the student experience during and after the *City to Summit* program. The aim of this research is to investigate which attributes of resilience and coping are developed and transferred into the lives of young people through participation in a journey style outdoor education program. The first aspect will look at attributes of resilience and coping through an online questionnaire which will occur before and after the program. The second aspect of the research will invite twenty students to be a part of small group discussions which will explore their experiences after the program. The information gathered from the research will then be used to inform future practice and development of outdoor education programs.

The research is voluntary and the information retrieved from data collection will be kept confidential and nobody will be named personally. Participants can withdraw from the research if they do not want to proceed.

#### What will I be asked to do?

You will be asked to participate in completing two online questionnaires at school. They will be the same questionnaires completed at two different times. The questionnaire should only take about ten minutes to complete. The first round of questionnaires will be completed before the program in September and the second round will be completed the next day after the program.

In addition, the twenty volunteers who will be randomly selected will be invited to participate in small group discussions which should last for approximately one hour. This will occur immediately after the *City to Summit* program. The small group interviews will be conducted during a reflective day for the camp which will involve various other small group activities.

#### What will I gain from participating?

You will have the opportunity to talk about your experiences or view your personal opinion about the *City to Summit* program. Your opinion is valued by the researcher and you will have the satisfaction that you have helped increase our understanding of the program which may impact the development of outdoor education programs for participants in the future.

#### How will the information I give be used?

All the information collected during the study will be used to identify which attributes of resilience and coping are developed and transferred into the lives of young people through participation in a journey style outdoor education program. These findings will be used to inform school teachers, educators and outdoor education providers to develop improved and effective outdoor education programs which facilitate the development and transference of resilience attributes in young people. This will be presented in the format of a Thesis, conference proceedings and/ or journal articles.

#### What are the potential risks of participating in this project?

There is the small possibility that during participation in the small group interviews or partaking in the questionnaires, that you may be reminded of an unpleasant past experience and you may not feel comfortable talking about the experience. Should this occur, you do not have to complete the questionnaire or provide a response during the interview. If you are particularly distressed or uncomfortable, the interview will be stopped immediately by the researcher. During the interviews, you may also feel uncomfortable talking freely in front of other students in the small group. The researcher will be conducting the discussions to minimise any coercive behaviour and you will never be forced to respond to any questions. The researcher will stop the interview if she feels that any participant is particularly uncomfortable. Anyone who appears to be participating under duress will also be withdrawn from the study discretely. If during an interview you become uncomfortable with or distressed by the material, that line of questioning will be stopped. If you become distressed you will be offered the option of additional assistance from the school student welfare counsellor, as the person most appropriate to speak to from the school. If required, you will also be told of the option to speak to an independent clinical psychologist, Dr Carolyn Deans, who is a staff member at Victoria University and is not involved in this research program. Please remember your participation is voluntary and you may withdraw from the research at anytime. Be assured that the small group interviews will be conducted with care and the researcher has extensive experience working with young people. All data collected will be kept confidential and will not be connected with your name in any way. This research is focused on the development of resilience and outdoor education programs and is not linked to the learning tasks or assessment tasks you may complete as part of your studies.

#### How will this project be conducted?

You will receive a link via email to the questionnaires online. You will be given time to complete the questionnaire at school during two different times. On the 14<sup>th</sup> of September 2012, you will complete the first round of online questionnaires. The questionnaires will be conducted during pre-trip preparation time at school when all the students attending the camp will also be completing a variety of different forms. On the last day of the *City to Summit* outdoor education program, you will have time allocated during your reflection time to complete the second round of questionnaires.

The twenty randomly selected students will be asked to participate in small group discussions. These students will be randomly selected from those that have consented to participate in the small group discussions as indicated on their consent form. Selected students will be notified that the discussions will occur on the reflection day immediately after completion of the *City to Summit* program. The small group discussions will be conducted by the Student Researcher. In groups of four, students will be asked about their experiences during the *City to Summit* program. The interviews will be conducted in a meeting room and all information will be kept confidential.

#### Who is conducting the study?

Victoria University

Chief investigator:

	Dr Peter Burridge Phone: Email:	
Associate Investi	gator: Dr Lauren Banting Phone: Email:	
Student Researc	her:	
	Casie Chalman	

Email:

Any queries about your participation in this project may be directed to the Chief Investigator listed above. If you have any queries or complaints about the way you have been treated, you may contact the Research Ethics and Biosafety Manager, Victoria University Human Research Ethics Committee, Victoria University, PO Box 14428, Melbourne, VIC, 8001 or phone (03) 9919 4148. Appendix I: Information to Control Group (Parents and Participants)



## INFORMATION TO PARTICIPANTS INVOLVED IN RESEARCH

#### You are invited to participate

You are invited to participate in a research project entitled Resilience, Adolescents and Outdoor Education: Is resilience context specific?

This project is being conducted by a student researcher, Casie Chalman as part of a Masters by Research study at Victoria University under the supervision of Dr Peter Burridge from the Department of Arts, Education and Human Development and Dr Lauren Banting from the Institute of Sport, Exercise and Active Living (ISEAL).

#### **Project explanation**

This research will examine the student experience during their Year 10 studies. The aim of this research is to investigate which attributes of resilience and coping are developed and drawn upon by young people during their Year 10 experience. The research will look at attributes of resilience and coping through an online questionnaire which will occur over two different time periods to see if there is any change in their levels of resilience or coping.

The research is voluntary and the information retrieved from data collection will be kept confidential and nobody will be named personally. Participants can withdraw from the research if they do not want to proceed.

#### What will I be asked to do?

You will be asked to participate in completing two online questionnaires. They will be the same questionnaires completed at different times. The questionnaire should only take about ten minutes to complete. The first round of questionnaires will be completed in September and the second round will be completed in November.

#### What will I gain from participating?

You will have the opportunity to voice your personal opinion and beliefs about your experience during Year 10. Your opinion is valued by the researcher and you will have the satisfaction that you have helped increase our understanding of resilience and coping in young people which may impact on the development of programs for participants in the future.

### How will the information I give be used?

All the information collected during the study will be used to identify which attributes of resilience and coping are developed and drawn upon by Year 10 students. These findings will be used to inform school teachers and educators to develop improved and effective programs which facilitate the development and transference of resilience attributes and coping skills in young people. This will be presented in the format of a Thesis, conference proceedings and/ or journal articles.

#### What are the potential risks of participating in this project?

The risk to school participants is very small. There is the small possibility that completing the questionnaires, may remind you of an unpleasant past experience. Should this occur, you do not have to complete the questionnaire. If you are particularly distressed or uncomfortable, completing the questionnaire will be stopped immediately by the researcher. If required, you will also be given the option to speak to an independent clinical psychologist, Dr Carolyn Deans, who is a staff member at Victoria University and is not involved in this research program. Student participation is voluntary and you may withdraw from the research at anytime. All data collected will be kept confidential and will not

be connected with the students' name in any way. This research is focused on the development of resilience and outdoor education programs and is not linked to the learning tasks or assessment tasks you may complete as part of students' studies.

#### How will this project be conducted?

During each stage, you will receive a link via email to the questionnaires online. You will have a week period to complete questionnaire at school or in your own time.

#### Who is conducting the study?

Victoria University

Chief investigator:

Dr Peter Burridge Phone: Email: Associate Investigator: Dr Lauren Banting Phone: Email: Student Researcher:

Casie Chalman Phone: \_\_\_\_\_ Email: \_\_\_\_\_

Any queries about your participation in this project may be directed to the Chief Investigator listed above. If you have any queries or complaints about the way you have been treated, you may contact the Research Ethics and Biosafety Manager, Victoria University Human Research Ethics Committee, Victoria University, PO Box 14428, Melbourne, VIC, 8001 or phone (03) 9919 4148. Appendix J: Program Group Consent Form for Questionnaires



# CONSENT FORM FOR STUDENTS INVOLVED IN RESEARCH QUESTIONNAIRES

#### **INFORMATION TO PARTICIPANTS:**

We would like to invite you to be a part of a study called Resilience, Adolescents and Outdoor Education: Is resilience context specific?

This research will examine two aspects of the student experience during and after the *City to Summit* program. The aim of this research is to investigate which attributes of resilience and coping are developed and transferred into the lives of young people through participation in a journey style outdoor education program. The first aspect will look at attributes of resilience and coping through an online questionnaire which will occur before and after the program. The second aspect of the research will invite twenty students to be a part of small group discussions which will explore their experiences after the program. The information gathered from the research will then be used to inform future practice and development of outdoor education programs.

The research is voluntary and the information retrieved from data collection will be kept confidential and nobody will be named personally. Participants can withdraw from the research if they do not want to proceed.

## **CERTIFICATION BY PARENT / GUARDIAN FOR STUDENT**

I, .....(name) of

certify that I am at least 18 years old\* and that I am voluntarily giving my consent for my child to participate in the study: Resilience, Adolescents and Outdoor Education: Is resilience context specific? This project is being conducted by a student researcher, Casie Chalman as part of a Masters by Research study at Victoria University under the supervision of Dr Peter Burridge from the Department of Arts, Education and Human Development and Dr Lauren Banting from the Institute of Sport, Exercise and Active Living (ISEAL).

I certify that the objectives of the study, together with any risks and safeguards associated with the procedures listed hereunder to be carried out in the research, have been fully explained to me by:

#### Casie Chalman

and that I freely consent to my child's participation involving the below mentioned procedures:

Participation in completing two online questionnaires which will be conducted in September 2012 and November 2012.

I certify that I have had the opportunity to have any questions answered and that I understand that my child can withdraw from this study at any time and that this withdrawal will not jeopardise him or her in any way.

I have been informed that the information I provide will be kept confidential.

Signed:

Date:

Appendix K: Program Group Consent Form for Small Group Interviews



## CONSENT FORM FOR STUDENTS INVOLVED IN RESEARCH DISCUSSIONS

## **INFORMATION TO PARTICIPANTS:**

We would like to invite you to be a part of a study called Resilience, Adolescents and Outdoor Education: Is resilience context specific?

This research will examine two aspects of the student experience during and after the *City to Summit* program. The aim of this research is to investigate which attributes of resilience and coping are developed and transferred into the lives of young people through participation in a journey style outdoor education program. The first aspect will look at attributes of resilience and coping through an online questionnaire which will occur before and after the program. The second aspect of the research will invite twenty students to be a part of small group discussions which will explore their experiences after the program. The information gathered from the research will then be used to inform future practice and development of outdoor education programs.

The research is voluntary and the information retrieved from data collection will be kept confidential and nobody will be named personally. Participants can withdraw from the research if they do not want to proceed.

## **CERTIFICATION BY PARENT / GUARDIAN FOR STUDENT**

I, .....(name) of

certify that I am at least 18 years old\* and that I am voluntarily giving my consent for my child to participate in the study: Resilience, Adolescents and Outdoor Education: Is resilience context specific? This project is being conducted by a student researcher, Casie Chalman as part of a Masters by Research study at Victoria University under the supervision of Dr Peter Burridge from the Department of Arts, Education and Human Development and Dr Lauren Banting from the Institute of Sport, Exercise and Active Living (ISEAL).

I certify that the objectives of the study, together with any risks and safeguards associated with the procedures listed hereunder to be carried out in the research, have been fully explained to me by:

Casie Chalman

and that I freely consent to my child's participation involving the below mentioned procedures:

□ Participation in small group discussions which will explore their experience after the program.

I certify that I have had the opportunity to have any questions answered and that I understand that my child can withdraw from this study at any time and that this withdrawal will not jeopardise him or her in any way.

I have been informed that the information I provide will be kept confidential.

Signed:

Date:

Appendix L: Control Group Consent Form for Questionnaires



## CONSENT FORM FOR STUDENTS INVOLVED IN RESEARCH QUESTIONNAIRES

#### **INFORMATION TO PARTICIPANTS:**

We would like to invite you to be a part of a study called Resilience, Adolescents and Outdoor Education: Is resilience context specific?

This research will examine the student experience during their Year 10 studies. The aim of this research is to investigate which attributes of resilience and coping are developed and drawn upon by young people during their Year 10 experience. The research will look at attributes of resilience and coping through an online questionnaire which will occur over two different time periods to see if there is any change in their levels of resilience or coping over time.

The research is voluntary and the information retrieved from data collection will be kept confidential and nobody will be named personally. Participants can withdraw from the research if they do not want to proceed.

## **CERTIFICATION BY PARENT / GUARDIAN FOR STUDENT**

.....(Suburb)

certify that I am at least 18 years old\* and that I am voluntarily giving my consent for my child to participate in the study: Resilience, Adolescents and Outdoor Education: Is resilience context specific? This project is being conducted by a student researcher, Casie Chalman as part of a Masters by Research study at Victoria University under the supervision of Dr Peter Burridge from the Department of Arts, Education and Human Development and Dr Lauren Banting from the Institute of Sport, Exercise and Active Living (ISEAL).

I certify that the objectives of the study, together with any risks and safeguards associated with the procedures listed hereunder to be carried out in the research, have been fully explained to me by:

#### Casie Chalman

and that I freely consent to my child's participation involving the below mentioned procedures:

Participation in completing two online questionnaires which will be conducted in September 2012 and November 2012.

I certify that I have had the opportunity to have any questions answered and that I understand that my child can withdraw from this study at any time and that this withdrawal will not jeopardise him or her in any way.

I have been informed that the information I provide will be kept confidential.

Signed:

Date:

## **Appendix M: Program Outline**

## City to Summit Outline 2012 (2 day split start) 17 groups of 12-14 students per group

		Group																
Date	Day	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17
20-Oct-12	1	Wangaratta Mt Beauty	Wangaratta Mt Beauty															
21-Oct-12	2	Start Windy Corner	Start Windy Corner															
		Pretty Valley	Heathy Spur															
22-Oct-12	3	Cope Hut	Johnson's Hut	Wangaratta Mt Beauty	Wangaratta Mt Beauty													
23-Oct-12	4	Kelly Hut	Fitgerald Hut	Start Windy Corner Pretty Valley	Start Windy Corner Heathy Spur													
24-Oct-12	5	Joker Flat	Big River Bridge	Cope Hut	Johnson's Hut	Wangaratta Mt Beauty	Wangaratta Mt Beauty											
25-Oct-12	6	Winter Take Out	Jokers Flat	Kelly Hut	Fitgerald Hut	Start Windy Corner	Start Windy Corner											
						Pretty Valley	Heathy Spur											
26-Oct-12	7	Hinnomunjie	Winter Take Out	Joker Flat	Big River Bridge	Cope Hut	Johnson's Hut	Wangaratta Mt Beauty	Wangaratta Mt Beauty									
27-Oct-12	8	Taylors Crossing	Hinnomunjie	Winter Take Out	Jokers Flat	Kelly Hut	Fitgerald Hut	Start Windy Corner	Start Windy Corner									
20.0.1.42		6	The last frequencies		WELL-THE OLD	to the Plan	01-01	Pretty Valley	Heathy Spur									
28-Oct-12	9	Cycle to Benambra	Taylors Crossing	Hinnomunjie	winter Take Out	Joker Hat	BIG RIVER Bridge	Cope Hut	Jonnson's Hut	wangaratta Mt Beauty	wangaratta Mt Beauty							
29-Oct-12	10	Cycle to Buenda Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	winter Take Out	Jokers Flat	Kelly Hut	Fitgeraid Hut	Pretty Valley	Heathy Spur							
30-Oct-12	11	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	Winter Take Out	Joker Flat	Big River Bridge	Cope Hut	Johnson's Hut	Wangaratta Mt Beauty	Wangaratta Mt Beauty					
31-Oct-12	12	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	Winter Take Out	Jokers Flat	Kelly Hut	Fitgerald Hut	Start Windy Corner Pretty Valley	Start Windy Corner Heathy Sour					
1-Nov-12	13	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	Winter Take Out	Joker Flat	Big River Bridge	Cope Hut	Johnson's Hut	Wangaratta Mt Beauty	Wangaratta Mt Beauty			
2-Nov-12	14	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	Winter Take Out	Jokers Flat	Kelly Hut	Fitgerald Hut	Start Windy Corner	Start Windy Corner			
3-Nov-12	15	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	501.0	Cycle to Buenha Hut	Cycle to Benambra	Taylors Crossing	Hinnomuniie	Winter Take Out	loker Flat	Big River Bridge	Cope Hut	Johnson's Hut	Wangaratta Mt Reputy	Wangaratta Mt Beauty	
5 100 12	15	TOT CLOBBIN	Durics Flam	chunics creek	buckwong nut	5020	Site	cycle to benambru	ruyiors crossing	, initial de la company de	Winter Fune Out	Joker Hat	Dig inver bridge	copende	John John J Hat	wangaratta int beauty	wangaratta int beauty	
4-Nov-12	16	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	Winter Take Out	Jokers Flat	Kelly Hut	Fitgerald Hut	Start Windy Corner Pretty Valley	Start Windy Corner Heathy Spur	
5-Nov-12	17	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	Winter Take Out	Joker Flat	Big River Bridge	Cope Hut	Johnson's Hut	Wangaratta Mt Beauty
6-Nov-12	18	Merritts Camp	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	Winter Take Out	Jokers Flat	Kelly Hut	Fitgerald Hut	Start Windy Corner Pretty Valley
7-Nov-12	19	SUMMIT DAY. Merrits	SUMMIT DAY.	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	Winter Take Out	Joker Flat	Big River Bridge	Cope Hut
8-Nov-12	20	Walk to Thredbo. Travel	Walk to Thredbo. Travel	Merritts Camp	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	Winter Take Out	Jokers Flat	Kelly Hut
9-Nov-12	21			SUMMIT DAY. Merrits Camp	SUMMIT DAY. Cootapatamba	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	Winter Take Out	Joker Flat
10-Nov-12	22			Walk to Thredbo. Trave	Walk to Thredbo. Travel	Merritts Camp	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie	Winter Take Out
11-Nov-12	23					SUMMIT DAY. Merrits	SUMMIT DAY. Cootapatamba	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing	Hinnomunjie
12-Nov-12	24					Walk to Thredbo. Travel	Walk to Thredbo. Travel	Merritts Camp	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra	Taylors Crossing
13-Nov-12	25							SUMMIT DAY. Merrits Camp	SUMMIT DAY. Cootapatamba	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site	Cycle to Benambra
14-Nov-12	26							Walk to Thredbo. Travel	Walk to Thredbo. Travel	Merritts Camp	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO	Cycle to Buenba Hut Site
15-Nov-12	27									SUMMIT DAY. Merrits Camp	SUMMIT DAY. Cootapatamba	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut	SOLO
16-Nov-12	28									Walk to Thredbo. Travel	Walk to Thredbo. Travel	Merritts Camp	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek	Buckwong Hut
17-Nov-12	29											SUMMIT DAY. Merrits	SUMMIT DAY.	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain	Charlies Creek
												Camp	Cootapatamba					
18-Nov-12	30											Walk to Thredbo. Travel	Walk to Thredbo. Travel	Merritts Camp	Dead Horse Spur	Leather Barrel Creek	Tom Groggin	Davies Plain
19-Nov-12	31				1									SUMIMIT DAY. Merrits	SUMMIT DAY.	Dead Horse Spur	Leather Barrel Creek	rom Groggin
20-Nov 12	32		1	1	+				1					Walk to Thredbo, Travel	Walk to Thredbo, Travol	Merritts Camp	Dead Horse Sour	Leather Barrel Creck
21-Nov-12	33	1	1	1	1		1	1	1		1	1		Tavel	to measo. Haver	SUMMIT DAY. Merrits	SUMMIT DAY.	Dead Horse Spur
									<u> </u>							Camp	Cootapatamba	
22-Nov-12	34															Walk to Thredbo. Travel	Walk to Thredbo. Travel	Merritts Camp
23-Nov-12	35																	SUMMIT DAY. Merrits Camp
24-Nov-12	36																	Walk to Thredbo. Travel

Appendix N: Phase I- Resilience Scale

Date: \_\_\_\_\_

Please read the following statements. To the right of each you will find seven numbers, ranging from "1" (Strongly Disagree) on the left to "7" (Strongly Agree) on the right. Circle the number which best indicates your feelings about that statement. For example, if you strongly disagree with a statement, circle "1". If you are neutral, circle "4", and if you strongly agree, circle "7", etc.

		Strongly Disagree				Strongly Agree		
1.	When I make plans, I follow through with them.	1	2	3	4	5	6	7
2.	I usually manage one way or another.	1	2	3	4	5	6	7
3.	I am able to depend on myself more than anyone else.	1	2	3	4	5	6	7
4.	Keeping interested in things is important to me.	1	2	3	4	5	6	7
5.	I can be on my own if I have to.	1	2	3	4	5	6	7
6.	I feel proud that I have accomplished things in life.	1	2	3	4	5	6	7
7.	I usually take things in stride.	1	2	3	4	5	6	7
8.	I am friends with myself.	1	2	3	4	5	6	7
9.	I feel that I can handle many things at a time.	1	2	3	4	5	6	7
10.	I am determined.	1	2	3	4	5	6	7
11.	I seldom wonder what the point of it all is.	1	2	3	4	5	6	7
12.	I take things one day at a time.	1	2	3	4	5	6	7
13.	I can get through difficult times because I've experienced difficulty before.	1	2	3	4	5	6	7
14.	I have self-discipline.	1	2	3	4	5	6	7
15.	I keep interested in things.	1	2	3	4	5	6	7
16.	I can usually find something to laugh about.	1	2	3	4	5	6	7
17.	My belief in myself gets me through hard times.	1	2	3	4	5	6	7
18.	In an emergency, I'm someone people can generally rely on.	1	2	3	4	5	6	7
19.	I can usually look at a situation in a number of ways.	1	2	3	4	5	6	7
20.	Sometimes I make myself do things whether I want to or not.	1	2	3	4	5	6	7
21.	My life has meaning.	1	2	3	4	5	6	7
22.	I do not dwell on things that I can't do anything about.	1	2	3	4	5	6	7
23.	When I'm in a difficult situation, I can usually find my way out of it.	1	2	3	4	5	6	7
24.	I have enough energy to do what I have to do.	1	2	3	4	5	6	7
25.	It's okay if there are people who don't like me.	1	2	3	4	5	6	7
26.	I am resilient.	1	2	3	4	5	6	7

© 1987 Gail M. Wagnild & Heather M. Young. Used by permission. All rights reserved. "The Resilience Scale" is an international trademark of Gail M. Wagnild & Heather M. Young.

Appendix O: Phase I- Brief COPE

These items deal with ways you cope with the stress in your life at school. There are many ways to try to deal with problems. These items ask what you do to cope with stressful events in your school life. Obviously, different people deal with things in different ways, but I'm interested in how you try to deal with stressful situations. Each item says something about a particular way of coping. I want to know to what extent you do what the item says. For example: How much or how frequently you do something. Don't answer on the basis of whether it seems to be working or not—just whether or not you do it. Use these response choices. Try to rate each item separately in your mind from the others. Make your answers as true FOR YOU as you can.

Please remember that no identifying information is being collected and we are unable to identify any of the responses. All your responses remain completely anonymous. Some of the questions (4 and 11) potentially require you to admit to illegal activities. You are under no obligation to answer these questions (or any other questions) if you are uncomfortable doing so. You can select option 5 *Prefer not to Answer* if you would prefer not to answer any specific questions.

Please **TICK** the following number that relates best to you.

- 1 = I don't do this at all
- 2 = I do this a little bit
- 3 = I do this a medium amount
- 4 = I do this a lot
- 5 = I prefer not to answer

#	Question	1	2	3	4	5
1	I turn to work or other activities to take my mind off things.					
2	I concentrate my efforts on doing something about the situation I'm in.					
3	I say to myself "this situation isn't real."					
4	I use alcohol or other drugs to make myself feel better.					
5	I get emotional support from others.					
6	I just give up trying to deal with the situation.					
7	I take action to try to make the situation better.					
8	I refuse to believe the situation has happened.					
9	I say things to let my unpleasant feelings escape.					
10	I get help and advice from other people.					
11	I use alcohol or other drugs to help me get through it.					

12	I try to see things in a different light, to make it seem more positive.			
13	I criticise myself.			
14	I try to come up with a strategy about what to do.			
15	I get comfort and understanding from someone.			
16	I give up the attempt to cope.			
17	I look for something good in what is happening.			
18	I make jokes about the situation.			
19	I do something to think about it less, such as going to movies, watching TV, reading, daydreaming, sleeping, or shopping.			
20	I accept the reality of the fact that the situation has happened.			
21	I express my negative feelings.			
22	I try to find comfort in my religion or spiritual beliefs.			
23	I try to get advice or help from other people about what to do.			
24	I learn to live with the situation.			
25	I think hard about what steps to take.			
26	I blame myself for things that happened.			
27	I pray or meditate.			
28	I make fun of the situation.			

Appendix P: Phase III- Interview Schedule

Data Collection		Program Group	Control Group	
Phase I:	Pre-test Questionnaires	21 & 22 September 2012	11 October 2012	
	Post-test Questionnaires	Throughout November 2012 (On day 22 of each activity groups program)	20 November 2012	
Phase II:	Program Observations	27 October to 18 November 2012	N/A	
Phase III:	Semi-structured Interviews: Data Set 1.	Interview Group 1: 20/11/2012 Interview Group 2: 21/11/2012 Interview Group 3: 22/11/2012 Interview Group 4: 28/11/2012	N/A	
	Semi-structured Interviews: Data Set 2.	Interview Group 1: 27/05/2013 Interview Group 2: 28/05/2013 Interview Group 3: 29/05/2013 Interview Group 4: 30/05/2013	N/A	

Appendix Q: Phase III- Semi Structured Interview Guide 1

#### Information statement to participants before starting:

I want to thank you for taking the time to meet with me today. My name is Casie Chalman and I would like to talk to you about your experiences participating in the *'City to Summit'* outdoor education program. I too have just come off the trip. I was one of the group leaders with Group 9. As one of the components of our overall program evaluation we are assessing resilience attributes and coping in order to capture lessons that can be used to develop effective outdoor education programs in the future.

I would like to remind you that the information retrieved from this interview will be kept confidential and nobody will be named personally. I will ensure all responses will be kept confidential. This means that your interview responses will only be shared with research team members and we will ensure that any information we include in our report does not identify you as the respondent. I will be taping the session because I don't want to miss any of your comments. Although I will be taking some notes during the session, I can't possibly write fast enough to get it all down. Because we're on tape, please be sure to speak up so that we don't miss your comments.

If at any time during the interview, you feel like you do not want to proceed or answer any questions, please let me know and we will stop. I'm estimating that the interview should last about the length of lunchtime, so if you would like a break, please let me know. During the interview I will ask a series of questions. Each of you will have time to answer these questions. So, we don't speak over each other, we will pass around a talking ball, so when you're ready to talk, you can ask for the ball by popping your hand up. Please take your time, have a think about the question and be honest. Remember, your name will not be attached to the comment if anyway.

Do you have any questions, or would you like me to explain anything before we begin? Great let's get started then...

#### Theme 1) Meaningfulness (Also known as purposeful life)

1. So I would like you to try and think back to the start of City to Summit. Can you tell me about any goals you set for yourself to achieve during City to Summit?

YES:

- What steps did you take to help you achieve these goals?
- > Why do you think you were successful in achieving these goals?

OR

Why do you think you didn't achieve these goals?

NO: Move to next question.

2. Can you tell me if participating in City to Summit influenced or had an impact on any of your personal values or beliefs?

YES:

- Why do you think that is?
- What made you feel that way?

NO: Move to next question.

## **Further Questions**

- Why do you value those things?
- Do you know your goals for the future? (Work, school, home, sport etc.)
- Do you have any strategies in place to help you achieve these goals?
- Can you explain what steps you took to overcome or conquer your fears or worries about...?
- Can you recall a situation or moment when...?
- Can you describe to me how you felt when...?

## Theme 2) Perseverance:

- 3. Can you tell me about what you found most challenging during the trip?
  - > What did you do to help you get through these challenges?
- 4. Can you recall any particular situation when you think you used determination or self-discipline to help you achieve something?
  - Please explain...
- 5. Can you describe a situation back at school where you think you might be faced with a similar challenge or difficulty?
  - > How do you think would respond if this was to happen now?

## Further Questions:

- What were the things that helped you through this?
- How did you end up achieving the desired outcome?
- Why is that?

## Theme 3) Self-reliance:

- 6. Can you describe a situation that springs to mind when you think you showed independence or relied only on yourself to achieve a desired outcome?
  - Is this a normal thing you would do in your school life, when you are trying to achieve something?
- 7. One of the focuses of City to Summit was looking at self, others and the natural world. Can you tell me something you learned about yourself?
  - > Do you think you will use this learning in your life back at school?
- 8. Did you face any difficulties when you were relying on yourself?
  - ➤ How did you deal with it?

## **Further Questions**

- Did you want or think you needed help from someone else?
- How did you overcome it?

### Theme 4) Existential Aloneness

- 9. At the start of the 24 hour solo experience, how did you feel about spending that much time alone?
- 10. Did your feelings change about spending time alone at any stage during or after the solo?
- 11. How did you feel about making decisions by yourself, without having to ask or rely on anyone else?
- 12. How do you feel about spending time alone now?

#### Theme 5) Equanimity:

- 13. Can you remember any situation where you became frustrated or angry with yourself another person or a situation during the trip?
- 14. Why? What happened?
- 15. How did you resolve the situation?
- NO: Move onto next question.
  - 16. Dis you encounter any problems between group members?
    - ➤ How was this resolved?
  - 17. During the trip, were you able to control your feeling and emotions when you were put in challenging situations?
    - Can you please tell me more?
    - Can you please explain

## Conclusion/Wrap Up:

- 18. Can you tell me anything you have learnt during the trip that you think you will take back and use in your school life?
- 19. Do you think you will actually use what you have learnt back in your life at school?

If YES:

> What would it be?

If NO:

- Why do you think that is?
- 20. Do you think that participating in City to Summit made your more resilient or gave you the tools to cope better with things in your life back at school?

## Whole Trip Questions

- If there was one thing that you learnt that you will take from the trip back to your school life, what would it be?
- Do you think you will actually use what you learnt during the outdoor education program back in your school life? Why/why not?
- Were there any new skills that you think you learnt over the course of the trip?
- What were they and why did you think you learnt them?

- Describe an experience in which...
- What did you learn from...?
- Do you think you will use your learning's in your everyday school life? When/where?
- When we get back to "reality" do think thing you will just forget the things you have learnt? Why/ why not?
- What things do you think would help you to actually apply what you have learnt in your school lives?
- Do you think you effectively communicated with others during the trip? Were there any times that you felt you didn't?

## Conclusion statement to participants:

- Before we wrap this up, is there anything you would like to add?
- Again I'd like to thank you all for volunteering for this interview. I hope that you have enjoyed reflecting on a fantastic trip. I will be hanging around for a while so if you would like to ask me any further questions, please feel free. I'll be analysing the information you and others gave me and submitting a draft report to the University in a month or so. I'll be happy to send you a copy to review at that time, if you are interested.

#### Thank you for your time.

Note: There will be no more than 15 open ended questions used during the interview.

#### **Further Questions**

- Could you talk to me a bit more about that?
- Why do you think that is?
- What do you think the reason was for that occurring?
- Would you give me an example?
- Can you elaborate on that idea?
- Would you explain that further?
- I'm not sure I understand what you're saying; can you please try explaining it in a different way?
- Is there anything else?
- What method did you use when...?
- What strategies did you use when...?
- During.... how did you feel about...
- What worked well? Please elaborate...
- What would you do differently next time? Please explain why...
- Were there any barriers that you felt you needed to overcome?

All volunteers and parent/guardians have completed consent forms to participant in the interviews.

#### Themes

## Theme 1) Meaningfulness (Also known as purposeful life):

Can be conceptualised as holding the belief that life has a purpose. This is about having an understanding of purpose and goals in life and knowing how you can achieve these goals, even in difficult or challenging situations.

## Theme 2) Perseverance:

Is known as persisting despite discouragement. Perseverance is really about applying determination and self-discipline to continue with a situation despite being discouraged by challenges.

### Theme 3) Self-reliance:

Is the ability to utilise one's own strengths and experiences to navigate difficult situations. This is about having a positive outlook about yourself, yet still knowing what your personal capacity and limitations are in a situation.

### Theme 4) Existential aloneness:

Is having a sense of uniqueness and independence. This is really about being comfortable in your own skin. This is also known as coming home to yourself. This is about being comfortable being alone and being able to make decisions by yourself, without the guidance of other people.

### Theme 5) Equanimity:

Has been defined as having a balanced perspective on life. It is also about having an optimistic outlook by being able to moderate extreme responses in difficult situations.

Appendix R: Phase III- Semi Structured Interview Guide 2

#### Information statement to participants before starting:

Hi guys. How are you all? I want to thank you for taking the time to meet with me again. It's great to see you all. As you know, from your previous discussion, one of the components we were evaluating was resilience attributes and coping in order to capture lessons that can be used to develop effective outdoor education programs in the future.

Just like last time, the information retrieved from this interview will be kept confidential and nobody will be named personally. I will ensure all responses will be kept confidential. This means that your interview responses will only be shared with research team members and we will ensure that any information we include in our report does not identify you as the respondent. I will be taping the session because I don't want to miss any of your comments. Although I will be taking some notes during the session, I can't possibly write fast enough to get it all down. Because we're on tape, please be sure to speak up so that we don't miss your comments.

If at any time during the interview, you feel like you do not want to proceed or answer any questions, please let me know and we will stop. I'm estimating that the interview should last about the length of lunchtime, so if you would like a break, please let me know. During the interview I will ask a series of questions. Each of you will have time to answer these questions. Please take your time, have a think about the question and be honest. Remember, your name will not be attached to the comment in anyway.

Do you have any questions or would you like me to explain anything before we begin? Before we get started could you please say your name one at a time so I can identify who is speaking on the tape? Thank you.

Great let's get started then...

I want you to all cast your minds back to six months ago when we were rafting the Mitta Mitta River and standing on the top of Australia.

1. What is your most vivid memory of City to Summit that first comes to mind for you?

## Experience (positive and negative) and Attitude (positive and negative)

- 1. So looking back now, how do you feel about CTS?
- 2. Overall, did you find it a positive or negative experience? Why is that?

#### Group development

- 3. How does the Year 11 cohort interact with each other after participating in CTS?
  - ➤ Have there been any changes to before?
  - > Can you think of any specific examples?
  - What about in your individual classes or during sporting activities?

#### Relationships (family, friends, teachers/staff)

- 4. You all mentioned that you made lots of new friends within your groups, how are your relationship now with your group members?
  - Why do you think that is?
- 5. Did you relationships initially change between your families or friends when you got back home?
  - ➤ How did they change?
  - What was different?

- 6. How are your relationships now? Did they go back to the way they were or did they stay the same?
  - > Why do you think that is?

## Appreciation

- 7. Now that you guys are back to "reality" as you called it, how have you gone at remembering the things you appreciated most?
  - Why do you think that is?

## Challenges (mentally, physically and perceived)

- 8. Have you had any big challenges you have had to deal with over the last few months at school?
  - > What did you do to help you get through these challenges?
- 9. Have you had to deal with any challenges between class mates?
  - ➢ How was this resolved?
  - > What strategies did you use?
  - Were you able to control your feelings and emotions when you were put in challenging situations?
- 10. Do you think that participating in CTS actually helped you deal with these challenges?

## **Further Questions:**

- What were the things that helped you through this?
- How did you end up achieving the desired outcome?
- Why is that?

## Tolerance

- 11. How are your tolerance levels now that you are back in the school environment?
  - ➤ Have they been tested at all?
  - Can you think of a specific situation?
  - ➢ How did you deal with this?

## Transference

- 12. Do you think you have put any skills that you learnt in CTS into practice at school?
- 13. Given that six months has past and you have had a chance to reflect, do you think that participating in City to Summit made your more resilient OR gave you the tools to cope better with things in your life back at school?
- 14. One of the things we were interested in was if outdoor education programs affect resilience and coping skills, and whether or not they transfer into your lives back at school. Do you think you have actually used any resilience or coping skills that you gained from participating in CTS in your lives back at school?
  - Can you think of any specific examples?

- 15. Can you tell me anything you have learnt during CTS that you have actually put into practice back at school?
  - Can you give me an example?
- 16. Was there any follow up lessons, debriefs or sessions in Year 11 to do with CTS?
  - What did you do in these sessions?

### **Conclusion Statement to Participants**

- Before we wrap this up, is there anything you would like to add?
- Again I'd like to thank you all for volunteering for this interview. I hope that you have enjoyed reflecting on a fantastic trip. I will be hanging around for a while so if you would like to ask me any further questions, please feel free. I'll be analysing the information you and others gave me and submitting a draft report to the University in a month or so. I'll be happy to send you a copy to review at that time, if you are interested. Thank you for your time.

#### **Further Questions**

- Do you remember the one thing you said you that you learnt that you will take from the trip and utilise it back to your school life? Do you think you have done that?
- Describe an experience in which...
- What did you learn from...?
- Do you think you will use your learning's in your everyday school life? When/where?
- Could you talk to me a bit more about that?
- Why do you think that is?
- > What do you think the reason was for that occurring?
- Would you give me an example?
- Can you elaborate on that idea?
- ➢ Would you explain that further?
- I'm not sure I understand what you're saying; can you please try explaining it in a different way?
- ➢ Is there anything else?
- What method did you use when...?
- ➤ What strategies did you use when...?
- During.... how did you feel about...
- What worked well? Please elaborate...
- What would you do differently next time? Please explain why...
- > Were there any barriers that you felt you needed to overcome?

Note: There will be no more than 15 open ended questions used during the interview

## Themes

Based on answers from participants in Phase 2, questions in the above Phase 3 interview will be guided around the following areas of interest:

- > Experience (positive and negative) and Attitude (positive and negative)
- Group development
- Relationships (family, friends, teachers/staff)
- > Appreciation
- Challenges (mentally, physically and perceived)
- ➤ Tolerance
- > Transference

(All volunteers and parent/guardians have completed consent forms to participant in the interviews)

Appendix S: Cycle 1- All Coded Themes

Name of the Theme	Number of Interview Sources	Number of Times Referenced
Coping (ALL Types)	8	87
Coping Dimensions	0	0
Active Avoidance	0	0
Avoidance	3	3
Negative Coping	4	5
Negativity	2	5
Not Coping (Avoidance)	4	12
Emotion-focused	0	0
Acceptance	3	4
Adaption	2	2
Attitude	7	49
Positive Thinking	7	26
Positive Coping	8	49
Putting things into Perspective	1	2
Problem Focused	0	0
Planning	4	7

Religious / Denial	0	0
Denial	2	2
Experience	8	49
Balance between Challenge and fun	5	11
Negative Experience	1	1
Positive Experience	8	70
Fun	3	14
Memories: most memorable moments	8	31
Previous Experience	7	17
Boarder	4	10
Increased Exposure to Outdoor Context	2	3
Shared Experience	5	18
Other	0	0
Camp Leaders Impact	7	23
Camp Leaders Negative Impact	2	2
Camp Leaders Positive Impact	2	6
Feelings	6	16

Leadership	4	13
Programming Issues	5	15
Suggestions	3	11
Support for staff	2	6
Thoughts on why we do the trip	4	6
What is resilience	3	5
Relationships	8	90
Conflict	2	7
Group Development	5	38
Trust Development	3	4
No Change	3	5
Peer pressure: Left out	4	9
Positive	6	21
Positive changes in family relationships	8	26
Stories from others	1	1
Resilience Attributes	0	0
Equanimity	0	0

Tolerance	7	30
Lack of tolerance	3	8
Existential Aloneness	0	0
Solo Existential Aloneness (2)	6	29
Perseverance	3	6
Determination and Self discipline	4	17
Purposeful Life_ Meaningfulness	0	0
Goal Setting (2)	5	12
Personal Learnings (Personal Reflection)	8	76
Confidence	2	2
Subconscious	3	8
Sense of Achievement	7	16
Self-Reliance	3	18
Independence	6	35
Resilience Development	0	0
Challenge	8	103
Character Building	3	13
Different sort of challenge	3	7
---	---	----
Mentally Challenging	7	16
Not Challenging Enough	7	23
Make more challenging	4	5
Perceived Challenge	8	38
Physically Challenging	7	22
Too challenging	2	2
Empathy	3	6
Lack of empathy	2	3
Sharing knowledge and experiences with year level below	2	3
Appreciation and Gratitude	8	57
Material Items	7	22
Nature	3	7
Others	5	14
Others opinions	2	5
Mindfulness / Awareness	0	0
Electronics- reduction in use (2)	5	8

Resilience	8	40
How much you were challenged	1	3
Mental resilience	2	3
Outdoors specific	1	1
Physically more resilient	1	1
Service to Others	0	0
Actions (Helping or Assisting) (2)	3	11
Transference	8	112
Lack of remembering_reflection	4	12
Learnings dissipated	5	30
Appreciation dissipated 6 months later	2	2
No changes in resilience or learnings	2	7
Lack of follow-up/ Reflection or Awareness	4	9
Possible transference of learnings	6	87
Struggles to make links between contexts	5	27
Transference of skills in same setting	3	3

Higher Order Themes Breakdown				
Original Theme Name (Parent Code)	Original Theme Name (Child Code)	Second-Order Theme Breakdown	First-Order Theme Breakdown	Number of times referenced
Acceptance		Coping Dimension	Emotion-focused Coping	4
Actions (Helping or Assisting)		Resilience Development	Service (doing things for others)	11
Adapt or adaption		Coping Dimension	Emotion-focused Coping	2
Appreciation		Resilience Development	Gratitude	57
Appreciation	Appreciation dissipated after 6 months later	Transference	Learning's dissipated	2
	Material Items	Resilience Development	Gratitude	22
	Nature	Relationship	Relationship with Natural Environment	7
	Others	Resilience Development	Gratitude	14
	Others opinions	Resilience Development	Gratitude	5
Attitude		Coping Dimension	Positive Coping (reframing)	49
Avoidance		Coping Dimension	Active Avoidance	3
Balance between Challenge and fun		Experience	Experience	11
Boarder		Experience	Previous Experience	10
Camp Leaders Impact		Other	Camp Leaders Impact	23
	Camp Leaders Negative Impact	Other	Camp Leaders Impact	2

	Camp Leaders Positive Impact	Other	Camp Leaders Impact	6
Challenge		Challenge	Challenge	103
	Different sort of challenge	Challenge	Challenge	7
	Mentally Challenging	Challenge	Challenge	16
	Not Challenging Enough	Challenge	Challenge	23
	Perceived Challenge	Challenge	Challenge	38
	Physically Challenging	Challenge	Challenge	22
	Too challenging	Challenge	Challenge	2
Character Building		Challenge	Challenge	13
Conflict		Relationships	Conflict	7
Coping		Coping Dimension	All Types Together	87
	Negative Coping	Coping Dimension	Active Avoidance	5
	Positive Coping	Coping Dimension	Emotion-focused Coping	49
	Putting things into Perspective (Reframing)	Coping Dimension	Emotion-focused Coping	2
Denial		Coping Dimension	Religious/ Denial	2
Determination and Self discipline		Resilience Attribute	Perseverance	17
Electronics- reduction in use		Resilience Development	Mindfulness/ Awareness	8
Empathy		Resilience Development	Empathy	6
Experience		Experience	Experience	49
	Negative Experience	Experience	Negative Experience	1
	Positive Experience	Experience	Positive	70

			Experience	
	Shared Experience	Experience	Shared Experience	18
Feelings		Other	Feelings	16
Fun		Experience	Positive Experience	14
Goal Setting		Resilience Attribute	Purposeful/ Meaningfulness	12
Group Development		Relationships	Group Development	38
Increased Exposure to Outdoor Context		Experience	Previous Experience	3
Independence		Resilience Attribute	Self-Reliance	35
Lack of empathy		Resilience Development	Empathy	3
Lack of remembering or reflection		Transference	Lack of remembering or reflection	12
Leadership		Other	Leadership	13
Make more challenging		Resilience Development	Challenge- Not Challenging Enough	5
Memories- most memorable moments		Experience	Positive Experience	31
Negativity		Coping Dimension	Active Avoidance	5
No changes in Resilience or learning's		Transference	No changes in Resilience or learning's	7
No Follow-up, Reflection or Awareness		Transference	No Follow-up, Reflection or Awareness	9
Not Coping (Avoidance)		Coping Dimension	Active Avoidance	12
Perseverance		Resilience Attribute	Perseverance	6
Personal Learning's (Personal Reflection)		Resilience Attribute	Purposeful Life/ Meaningfulness	76

	Confidence	Other	Personal Learning's (Personal Reflection)	2
Planning		Coping Dimension	Problem Focused	7
Positive Thinking		Coping Dimension	Emotion-focused Coping	26
Previous Experience		Experience	Previous Experience	17
Programming Issues		Other		15
Relationships		Relationships	Relationships	90
	No Change in Relationships	Relationships	Relationships	5
	Peer pressure -Left out	Relationships	Relationships	9
	Positive	Relationships	Relationships	21
	Positive changes in family relationships	Relationships	Positive	26
	Trust Development	Relationships	Group Development	4
Resilience		Resilience Development	Resilience	40
	How much you were challenged	Resilience Development	Resilience	3
	Mental resilience	Resilience Development	Resilience	3
	Outdoors specific	Resilience Development	Resilience	1
	Physically more resilient	Resilience Development	Resilience	1
Self Reliance		Resilience Attribute	Self Reliance	18
Sense of Achievement		Resilience Attribute	Purposeful Life/ Meaningfulness	16
Sharing knowledge and experiences with year level below		Resilience Development	Empathy	3

Solo- Existential Aloneness		Resilience Attribute	Existential Aloneness	29
Stories from others		Relationships	Stories from others	1
Struggles to make links between contexts		Transference	Struggles to make links between contexts	27
Subconscious		Resilience Attribute	Purposeful Life/ Meaningfulness- Personal Learning's	8
Suggestions		Other	Programming Issues	11
Support for staff		Other	Programming Issues	6
Thoughts on why we do the trip		Other	Thoughts on why we do the trip	6
Tolerance		Resilience Attribute	Equanimity	30
	Lack of tolerance	Resilience Attribute	Equanimity- Tolerance	8
Transference		Transference	Transference	112
	Learning's dissipated	Transference	Transference	30
	Possible transference of learning's	Transference	Transference	87
	Transference in same setting	Transference	Transference	3
What is resilience?		Other	What is resilience?	5

Appendix U: Cycle 3- Major Themes Breakdown

Theme	Major Theme	Organising Theme	First-Order Theme Clusters
Number			Clusters
1	Challenge	Challenge	Character building
		(Type of	Different sort of
		Challenge)	challenge
			Mentally
			challenging
			Not challenging
			enough
			Perceived
			challenge
			Physically
			challenging
			Too challenging
2	Perception of the experience	Experience	Balance between
		/Memorable Moment	challenge and fun
		(type of experience)	Negative
			Experience
			Positive
			Experience
			Previous
			Experience
			Shared Experience
			Feelings
3	Post-program understanding	Resilience	Empathy
		Development and	Gratitude
		Personal Development	Mindfulness
		(type of learning)	Resilience
			Service to others

			Leadership
			Thoughts on why
			we do the trip
			What is resilience?
4	Relationships	Relationships	Conflict
		(social and	Group
		environmental)	development
			No change
			Peer pressure
			Positive
			Positive changes in
			family relationships
			Camp leaders
			impact
			Relationship with
			natural environment
			Stories from others
5	Transference	Transference	Lack of
		(type of	remembering
		application)	Leanings
			dissipated
			No changes in
			resilience
			No follow-up:
			reflection
			Possible
			transference
			Struggles to make
			links between contexts

			Transference in the
			same setting
			Programming
			Issues
6	Response to challenge	Coping (all	
		types)	
		Coping	Active avoidance
		Dimension	Emotion-focused
			Problem-focused
			Religious/Denial
		Resilience	Equanimity
		Attributes	Existential
			Aloneness
			Perseverance
			Purposeful life
			Self-reliance
			Tolerance
			Determination
			Subconscious

Appendix V: Group 9 Consent for Use of Photos

Piroto Permission 16/11/2012 give permission for the researcher. Caste-Ane Chaiman to do the following -O Use photos containing me from the outdoor education program to use in her thesis. This includes print and digital form. (2) Contact me up to six months after the program if required for research reasons I have signed the following pages and provided my personal contact details to confirm permission



Appendix W: Progressive Levels of Difficulties of Adventure Activities

Day	Activity	Level of Difficulty
0	Initiative and team building activities	Low
1	Travel day	Low
2	Bushwalking	Low to medium
3	Bushwalking	Medium
4	Bushwalking	Medium
5	Bushwalking	Medium
6	Whitewater rafting	Medium to hard
7	Whitewater rafting	Medium to hard
8	Whitewater rafting	Medium to hard
9	Mountain biking	Hard
10	Mountain biking	Hard
11	Solo	Hard
12	Bushwalking	Medium to hard
13	Bushwalking	Medium to hard
14	Bushwalking	Medium to hard
15	Bushwalking	Medium to hard
16	Bushwalking	Medium to hard
17	Bushwalking	Medium to hard
18	Bushwalking	Medium to hard
19	Summit day: bushwalking	Medium
20	Walk and travel day	Low
21	Debrief day at school with staff	
22	Phase III interviews collected	

Note. The challenge levels are an overall average based on the activities presented in the program outline.